# SIMULATION-BASED LEARNING IN HEALTHCARE ETHICS EDUCATION

A Dissertation

Submitted to the Center for Healthcare Ethics

McAnulty College and Graduate School of Liberal Arts

Duquesne University

In partial fulfillment of the requirements for

the degree of Doctor of Philosophy

By

Kathryn E. Wilt, MSN, RN

December 2012

UMI Number: 3546094

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



UMI 3546094

Published by ProQuest LLC (2012). Copyright in the Dissertation held by the Author.

Microform Edition © ProQuest LLC. All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code



ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 - 1346

Copyright by

Kathryn E. Wilt, MSN, RN

#### SIMULATION-BASED LEARNING IN HEALTHCARE ETHICS EDUCATION

By

Kathryn E. Wilt, MSN, RN

Approved October 23, 2012

Henk ten Have, M.D., Ph.D. Professor and Director Center for Healthcare Ethics Duquesne University (Dissertation Director) Gerard Magill, Ph.D. Professor and Vernon F. Gallagher Chair for the Integration of Science, Theology, Philosophy and Law Center for Healthcare Ethics Duquesne University (Committee Member)

Sr. Rosemary Donley, S.C, Ph.D., ANP-BC, FAAN Professor of Nursing Jacques Laval Endowed Chair for Justice for Vulnerable Populations Duquesne University (Committee Member) James Swindal, Ph.D. Dean, McAnulty College and Graduate School of Liberal Arts Duquesne University (Dean)

# ABSTRACT

# SIMULATION-BASED LEARNING IN HEALTHCARE ETHICS EDUCATION

By

Kathryn E. Wilt, MSN, RN December 2012

Dissertation supervised by Dr. Henk ten Have.

The dissertation presents an argument for the use of a nontraditional teaching method – simulation – in ethics curricula in medical and undergraduate nursing programs. It reveals the soundness of simulation as an educational approach in light of its pedagogy of engagement and formation and its appropriateness in teaching problem solving in complex and uncertain situations that center on relationships. Simulation offers a holistic approach to teaching ethics that addresses the cognitive, practical and moral apprenticeships that comprise the signature pedagogy of medicine and nursing as characterized by Shulman. When used with the traditional methods of ethics instruction, simulation is well suited to assist in the development of the ethical practitioner by its inclusion of practice, feedback and reflection that encourages development and habituation of the professional virtues, including prudential reasoning.

iv

The dissertation includes the relevant literature on goals and methods of ethics education, virtue ethics, quality of care, and simulation. It responds to the virtue/skill dichotomy that is apparent in ethics curricula in medicine and nursing and argues for a balanced approach to ethics education. It engages the healthcare literature on virtue ethics to support the need to emphasize educational goals that prepare healthcare practitioners who exercise virtue as well as clinical ethics skills, such as ethical analysis and decision making skills.

In addition, the dissertation shows that while ethics education has many nonexclusive goals, an important and often overlooked goal of ethics education is improved quality of care for patients. The dissertation makes explicit the relationship between ethics education and the quality of care for patients and demonstrates the contribution of professional virtues and attributes to the patient perspective of quality care as measured by respect, empathy and satisfaction. It also shares an ethical imperative for the adoption of simulation in ethics education.

The dissertation concludes that simulation meets the goals of ethics education: the development of ethically-sensitive, reflective and ethically competent healthcare practitioners, with the potential, ultimately, to lead to improved quality of patient care. When used with traditional teaching methods, simulation allows the educator to better meet the ethics education goals by explicitly introducing specific virtues and values into learning experiences and providing essential practice opportunities for habituation and formation of professional virtues and skills. Suggestions for the inclusion of ethics simulations are provided. Although the focus of the scholarly research is limited to

v

medical students and baccalaureate nursing students, the results may be applicable to healthcare students and professionals in related disciplines.

# DEDICATION

This dissertation is dedicated to my husband, Peter Madsen, and children, Peter and Elizabeth, whose support, encouragement, and constant love sustain me.

#### ACKNOWLEDGEMENT

Successfully completing a dissertation might take a wee bit of luck, and in my case, it was the composition of my committee. I have been honored to have a committee of esteemed scholars and leaders. Together, they were dedicated to my success, and I am so grateful. To Dr. Henk ten Have, esteemed department chair and my advisor – your experience in ethics education and keen insight helped me shape this dissertation. I am very grateful for your patience and kindness. Dank u wel. To my committee members, Dr. Gerard Magill and Sr. Rosemary Donley – thank you for your steady support, wisdom and mentoring. To Glory Smith, HCE administrative assistant and "rock" – thank you for your friendship, assistance and unfailing encouragement over the years.

Thank you to the colleagues, friends and family who spurred me on and even offered assistance. Thank you, Dr. Marianne Burda, HCE alum and friend, who introduced me to Dr. Dennis Novack of Drexel's College of Medicine. As a result, I was able to observe the psych-ethics simulations that he facilitated with medical students. I especially want to thank Dr. Novack for sharing his perspective on ethics simulations and the permission to use his observations and comments in the dissertation. Thank you to so many others – to Diane Redington, for the gift of time as you double-checked my reference lists for completion. To Kathy McCauley, for critically listening to my arguments, even as you were writing your own book. To Sr. Mary Jean Flaherty, for her astute final reading of the document. To Dr. Michalene King for motivation and my "Git 'er done" mug. To Michele Kish and Charlotte Heide for helping me stay organized in my "regular" work while I did this research. To the dedicated staff of Gumberg Library and particularly, David Nolfi, for help and encouragement over the years. To the many faculty and staff at DUSON, too numerous to list, who offered encouragement. To Kathy Mc, Kathy M, Kathy D, Barbara, Diane, Margaret, Donna and Elainka, for not giving up on me and understanding that this "phase" would eventually end. A special note of gratitude to my college English teacher, Dr. Eleanor ("Ellie") Wymard, who I often had in mind as I crafted a sentence. Thank you for encouraging any abilities I have as a writer, and most of all, for mentoring me.

Finally, to my family, for whom I am most grateful. To Mom and Dad – thank you for showing me how to work hard. To my children, Peter and Elizabeth Madsen, who model discipline and grit in the face of challenges, I am so thankful for your love and support. Peter, thank you for rescuing me so many times with your technical know-how and spending your rare weekend at home helping me with computer "issues." Thank you also for patiently fielding countless questions about your experiences as a medical student. Elizabeth, thank you for editing side-by-side with me, burning the midnight oil as I raced to meet my final deadline (on the eve of my first all-nighter). Thank you for keeping my spirits up and making me laugh. To my daughter-in-law, Megan, thanks for doing a "happy dance" with me! You all had such faith in me that I couldn't fail.

Lastly, to my husband, Pete. You have inspired me and compelled me to be more than I ever thought I could be. You generously gave me the gift of time by managing so much of our lives in the last year and more. You patiently listened or stirred up creative dialogues, and I am so happy to have had that time with you. Moreover, you seemed to know even more than I that I could do this and encouraged me all the way. Thank you from the bottom of my heart.

# TABLE OF CONTENTS

	Page
Abstract	iv
Dedication	vii
Acknowledgment	viii
Introduction	1
A. Background of the Problem	1
B. Thesis Statement	5
C. Methodology	6
D. Summary	10
Chapter One – Ethics Education in Healthcare	15
A. Recent Influences in the Development of Ethics Education in Medical and	
NursingCurricula	15
B. History of Ethics Education in Health Care	21
C. The Importance of Ethics Education Today	29
D. Conclusion	41
Chapter Two – Goals and Methods of Ethics Teaching	44
A. The Goals of Ethics Education in Medicine and Nursing	44
B. The Methods Applied in Ethics Education	65
C. Evidence of the Effectiveness of Teaching-Learning Methods in Ethics	
Education	76
D. Conclusion	82

Chapter Three - Improving Quality of Care as a Fundamental Goal of Ethics

	Education	4
A.	Quality of Care Defined	4
B.	Empathy and Communication and Their Relationship to Quality of Care9	3
C.	The Contribution of Ethics Education to the Patient and Professional Perspectives	3
	of Quality of Care	2
D.	Conclusion107	7
Chapte	er Four – Virtue-based Ethics in Medicine and Nursing10	9
A.	The Relevance of Virtue Ethics in Health Care10	9
B.	What Virtues Are Important in the Nurse and Physician?11	7
C.	How Are Virtues Taught and Learned?	4
D.	The Relevance of Virtues to the Nurse and Physician14	2
E.	The Relationship of a Better Healthcare Professional to Quality of Care144	4
F.	Summary14	6
Chapte	er Five – Simulation in Healthcare Education14	8
A.	Simulation as a Teaching Method Defined14	8
B.	The History of Simulation as an Educational Method in Health Care158	3
C.	Factors Influencing the Development of Simulation as an Educational Method in	
	Healthcare Education17	1
D.	Types of Simulation Activities Employed in Healthcare Education	3
E.	Theories That Support the Use of Simulation in Healthcare Education	5
F.	The Structure and Components of Simulation in Healthcare Education195	5
G.	Empirical Studies That Illustrate the Use of Simulation in Healthcare	

Disciplines
H. Advantages and Challenges of Simulation in Healthcare Education233
Chapter Six – Simulation in Healthcare Ethics Education
A. The Relevance and Appropriateness of Simulation in Healthcare Ethics
Education
B. Ethical Reasons for the Application of Simulation to Ethics Education247
C. The Empirical Studies That Illustrate the Use of Simulation in Healthcare Ethics
Education
D. Suggestions for the Implementation of Simulation in Healthcare Ethics
Education
E. Final Considerations and Recommendations for Further Research
Reference List

# Introduction

#### A. Background of the problem

Many of the interactions that nurses and physicians experience on a daily basis with patients and their families have a moral dimension. There are "everyday ethics" that confront healthcare professionals routinely and are part of the day-to-day patient care or arise from the patient-provider relationship. Caring practices, information-sharing and patient teaching, pain management and treatment protocols, initiating fall precautions, and employing physical or chemical restraints, for instance, give rise to ethical concerns. For example, a patient's need for information about a recommended treatment places the patient and healthcare provider in a relationship for which the healthcare professional has inherent obligations. The healthcare professional must prudently explain the treatment plan and present information without bias, being aware of the possibility of inadvertently pressuring the patient and potentially coercing a decision. A routine interaction, such as sharing information, finds the healthcare professional balancing the obligation to provide benevolent care and at the same time protecting the patient's right to autonomy.

Ethical issues are part of the fabric of the patient-provider relationship based on the healthcare professionals commitment to care and occur with regularity. Ethical issues may arise in clinical settings for many reasons – insufficient resources to provide quality care, pressure to discharge patients earlier than desired and pressure to continue aggressive care for patients who have little chance of improvement are common examples. Physicians must frequently manage ethical issues or ethical dilemmas regarding confidentiality, informed consent, resource allocation inequities and end-of-life

care, such as advance directives, "do not resuscitate" orders, and withdrawing or withholding treatment. Nurses report frequent encounters with ethical issues related to protecting patients' rights, informed consent, advance directives, and surrogacy issues (Ulrich et al. 2010). In some instances more difficult ethical problems may occur that require the assistance of colleagues or ethics committees to resolve, such as establishing a surrogate decision maker or managing the unethical or incompetent conduct of a colleague. Few of the daily experiences of nurses and physicians require an ethics committee and even fewer healthcare practitioners face the sensational ethical cases that garner media attention. Fortunately, most ethical problems are managed "at the bedside" or within the patient-practitioner or family-practitioner relationship. By the intimate nature of the experiences and the dynamic nature of patient-provider relationships, however, nurses and physicians interact in relationships that call for healthcare providers to be ethically aware and responsive at all times, working from a strong ethical foundation. In the most basic sense of moral obligation, nurses and physicians are called to treat patients with dignity and respect, compassion and honesty, to practice with integrity and to provide the most competent care. No matter how specific ethical concerns are labeled or categorized – everyday ethics or ethical dilemmas – nurses and physicians must be adequately prepared to meet their many ethical responsibilities.

For ethics educators in healthcare disciplines, the focus of their work is to prepare healthcare professionals to be able to practice with the essential virtues and ethical skills that comprise ethical competency. Ethics educators strive to nurture the moral development of students and guide them to recognize and respond appropriately to the moral dimension of health care in every patient interaction. Through ethics curricula,

ethics educators assist students in the development of appropriate attitudes, behaviors and skills to effectively prevent, identify and manage ethical problems. The dissertation contends that the goals to develop virtues and skills in ethics are significant and important outcomes; however, ethics education should also be driven by an overarching, correlative goal to improve the quality of patient care. To attain this, students must reach short-term or interim goals directed toward enhancing ethical awareness and sensitivity, practicing with professional virtues and demonstrating competence in cognitive, behavioral and affective skills. Achieving these goals will potentially produce positive patient outcomes and improve the quality of care for patients.

Educators are challenged, however, to meet these goals in the present educational and healthcare environments. Classroom instruction in ethics, however, even when engaging and participatory, generally lacks the sense of immersion in the complex situation. Case study analysis and group discussions offer students the opportunity to engage in dialogue and test different viewpoints, but these methods offer little in the realm of role assimilation and formation. Students view case studies from the outside and without experiencing decision making in "real" time (Jones 1995; Smith et al. 2010). Well-scripted role play exercises are limited in their ability to allow students to develop ethical sensitivity and the ethical competencies needed as a practitioner. Role play can raise emotions and feelings, but it lacks the ability to situate the student in the functional role of the healthcare professional, with his or her inherent responsibilities. Clinical teaching, a valuable teaching strategy in the role formation of healthcare students, is also limited in its ability to aid in the development of specific professional virtues and ethics skills. Unfortunately, the clinical environment lacks predictable or consistent

opportunities for students to identify particular ethical issues and gain experience for future professional practice. Moreover, as many clinical faculty will attest, one learning moment moves to the next so rapidly in the clinical setting that the student has little time for reflection on learning, an important component of habituation that is essential for moral formation.

This dissertation reviews the traditional and nontraditional methods of ethics instruction and argues for the incorporation of an innovative model – simulation – to effectively meet the goals of ethics education. Simulation, a teaching-learning method is used presently in medicine and nursing to teach and assess critical thinking, promote confidence, develop communication skills, increase psychomotor skills and more. Simulation or simulation-based learning replicates clinical situations and provides opportunities for prepared practice, reflection and feedback, which can promote the development of ethically-sensitive, reflective and prudent nurses and physicians. It would seem to follow that a teaching-learning method such as simulation, which is used to promote the learning of complex skills, decision making and even teamwork, might also be a highly desirable component of ethics instruction. Simulation would offer students the opportunity to practice, receive feedback and reflect on experiences that have an ethical dimension and do so safely, without risk to patient or student. Moreover, it also seems reasonable to suggest that this strategy would be highly useful to inculcate the professional virtues, behaviors and skills necessary for the formation of an ethical practitioner. Simulation provides opportunities for students to practice and demonstrate virtues such as openness and compassion, virtues that are essential for awareness of and

sensitivity to possible ethical problems. Likewise, simulation assists students to develop and practice prudential reasoning, a virtue-based skill central to ethical decision making.

The dissertation also examines the increased attention to quality of care for patients that has arisen in the past two decades and expounds upon the relationship between ethics education and quality of care. Although improving quality of care has not been the primary goal of ethics curricula in medicine and nursing, it is an important potential outcome of ethics education that warrants analysis. The dissertation examines the relationship between developing virtuous, ethically competent practitioners and improving the quality of care as measured by the patient's perspective.

# B. Thesis Statement

The dissertation defends the following thesis statement that simulation, when used with traditional teaching methodologies, is a preferable method for teaching healthcare ethics to medical and baccalaureate nursing students when compared to traditional methods alone because it best meets the goals of ethics education: the development of ethically-sensitive, reflective and ethically competent healthcare practitioners who will have the capacity to positively influence the quality of care for patients. The results of the dissertation may be applicable to healthcare professionals in many disciplines, whether students or practitioners; however, the dissertation limits its focus to the ethics education of medical students and baccalaureate nursing students.

The above thesis statement is well-situated in the contemporary health care landscape and aligns well with current ethics education goals of medicine and nursing programs. While the literature shows that there is no consensus about what precise goals

should be set for ethics education in medicine and nursing, the dissertation examines the current discussion on these goals as well as the virtue/skill dichotomy found in ethics curricula (Eckles et al 2005). The dissertation contends that the goals to develop virtues and skills in ethics are significant and important outcomes; however, ethics education should also be driven by an overarching, correlative goal to improve the quality of patient care. The dissertation submits that simulation-based learning experiences allow the educator to *explicitly* introduce certain virtues and values that are essential to the professional behavior of nurses and physicians. The educator is able to manipulate circumstances to elucidate and reinforce particular virtues and values through simulated learning experiences and provide essential practice opportunities for habituation and formation of virtue, overcoming the lack of predictable learning experiences in clinical settings. Simulation provides an environment in which an educator can observe a student's behavior, present timely feedback related to virtues, values or ethical analysis, reinforce positive behaviors and encourage reflection on performance. Additionally, simulated learning experiences in ethics provide practice opportunities for the inexperienced or novice healthcare practitioner that lessen or remove the burden or risk of harm to patients, particularly moral harm in the form of disrespect or inequity. The dissertation argues that improving the virtues and ethics-related skills of the healthcare professional may benefit quality of care from the patient perspective by improving patient satisfaction as measured by respect, satisfaction and fulfillment of needs.

# C. Methodology

The methodology of the dissertation involves analysis of the literature on ethics education, virtue ethics, simulation and quality of care. Although the dissertation

extensively employs empirical data from the scholarly literature to provide theoretical arguments to justify the thesis, it does not undertake empirical studies. It analyzes the scholarly literature from the perspective of the thesis and its claims about the benefits of simulation to ethics education and quality of care. The dissertation research includes: a) identifying the goals of ethics education and the present traditional and nontraditional methodologies employed in ethics teaching, b) the benefits of simulation as a teaching method in ethics education, c) the benefit of emphasizing virtue ethics in medicine and nursing, and d) the positive impact of the virtuous practitioner on quality of care for patients. The literature engaged in the dissertation encompasses the following five areas of study: ethics education, essential virtues of the nurse and physician, the relationship between ethics education and quality of care, simulation as a teaching-learning method in healthcare in general, and more specifically, simulation as a teaching-learning method for health care ethics education. The literature that supports each chapter is outlined below.

The literature for Chapters One provides an examination of the historical, scientific, technological, sociocultural and economic factors that have influenced the rising prominence of ethics education since the 1960s. This chapter also surveys the literature on the problems and challenges that arise from the educational and professional practice settings that reflect an enhanced need for ethics education in the healthcare professions. The historical perspective on ethics education is provided by the early pioneers in healthcare ethics, among them, Edmund Pellegrino. The literature is also surveyed to identify the particular problems that challenge ethics educators today.

The literature consulted for Chapter Two provides support for an analysis of the goals and methods of ethics education for medical students and baccalaureate nursing

students. Reviews of ethics education in medicine are consulted (Eckles et al. 2005; Miles et al. 1989). The scholarship of Benner et al. (2010) and Fry (1989) are consulted regarding the state of ethics education in nursing. Curricular recommendations and accreditation agency standards for medical education from the Liaison Committee for Medical Education and the Accreditation Council for Graduate Medical Education are reviewed. Similarly, in nursing, the recommendations of the American Association of the Colleges of Nursing and the study results of the National Nursing Education Report of the Carnegie Foundation for the Advancement of Teaching (Benner et al. 2010) are examined. International perspectives on ethics education for healthcare professionals are also included, primarily the UNESCO core curriculum in bioethics (2008). The codes of ethics in medicine and nursing are also examined for guidance in developing ethics curricular goals.

Chapter Three includes the literature on quality of care. The writings of Brook, McGlynn and Shekelle (2000) and Grol (2001) are consulted for a basic understanding of quality of care. The literature review investigates the various stakeholder perspectives of quality of care, which include those of the patient, healthcare professional and policy maker. For this thesis, the patient and professional perspectives hold much relevance, particularly the patient perspective, which is therefore emphasized. Literature that demonstrates the relationship between the attributes of the healthcare professional – such as empathy, communication skills and affective behavior – and patient satisfaction is analyzed to establish the relationship between ethics education and quality of care.

The scholarship for Chapter Four consists of literature on the topic of virtue ethics as it relates to medicine and nursing. The literature draws from philosophy, medicine and

nursing and establishes the importance of virtue to achieving well-being or as understood by patients and healthcare professionals to mean good health. The writings of Pellegrino and Thomasma are significant to understanding the relevance of virtue education to medicine and by extension, to nursing. The literature reviewed for this chapter supports ethical sensitivity and prudential reasoning as essential virtues for nurses and physicians. The scholarship on the recommended ways in which educators can teach virtues is also examined.

The scholarship that is consulted for Chapter Five maintains that simulation is an effective pedagogical method in medicine and nursing. The definition of simulation is established by reviewing the perspectives of contributors from education, medicine and nursing. The scholarship of authors who are experienced with simulation in the disciplines of medicine and nursing are shared as well as the available empirical research on simulation in these fields. The literature for this chapter also includes educational theory that supports experiential learning. The early work of Dewey (1933) and later of Schön (1983) on reflective thinking, Dreyfus and Dreyfus' (1986, 2009) model of skill acquisition, Benner's work on the Dreyfuses' model as it applies to nursing (2001) and Shulman (2005a, 2005b) on the signature pedagogies of the professions are examined for their relevance to simulation.

Lastly, for Chapter Six, scholarly literature that is specific to the use of simulation in healthcare ethics education and which supports its application as an effective teachinglearning strategy for ethics education is reviewed. The chapter is supported by literature that provides empirical evidence for simulation as well as scholarly support for the use of simulation in the teaching of ethics.

It should also be noted that in the review of relevant dissertations unrestricted by year, there were no dissertations that focused on employing simulation to teach ethics to medical or nursing students. A search of related dissertations was conducted on the following databases: ProQuest Dissertation and Theses (PQD&T), DUCat, Networked Digital Library of Theses and Dissertations (NDLTD), and WorldCat. The search was conducted using variations of the following terms: ethics (ethics, virtues, moral), education (education, pedagogy, methodology, and teaching) and simulation (simulation, role play, standardized patients). It was unrestricted by year or discipline. The search yielded only one dissertation that focused on utilizing simulation to teach ethics in either medicine or nursing, yet this paper did not assess the simulation component. This dissertation by Garman (1994) compared the effectiveness of a lecture series that employed case studies with reflective practicum sessions that were conducted by an ethicist. Garman evaluated the effectiveness of the teaching-learning method 6-12 months later by assessing the performance of the medical students in a simulation with a standardized patient. The results of the study revealed that the medical students who participated in the reflective practicum scored higher than the control group in knowledge and in confidence. The dissertation provided support for the reflective practicum, an important component of simulation-based teaching strategies; however, it did not assess the effect of a simulation-based learning activity. There were no dissertations that assessed the effectiveness of using simulation to teach ethics to either medical students or baccalaureate nursing students.

#### D. Summary

As noted above, the dissertation proposes that simulation is a beneficial method to teach ethics to medical and nursing students and to promote attainment of the goals of ethics education. The chapters are organized to effectively present the supporting literature that contributes to this argument. The focus of each chapter is presented below.

Chapter One presents a brief history of the development of ethics education in medicine and nursing. It reviews the health care practices and cultural changes that have given rise to the increased attention to ethics. The chapter presents a brief history of ethics education in medicine and nursing and includes the work of early pioneers in healthcare ethics such as Pellegrino and the efforts of the Society for Health and Human Values in addressing ethics education in medicine. The chapter discusses the implicit and explicit approaches to ethics education and makes note of several phenomena – the "hidden curriculum," "ethical erosion," and "moral malaise" – as well as breaches in professional conduct that have been observed in medical students and physicians that strengthen the argument for enhanced ethics education. The nursing literature is also examined for the presence of these phenomena, as well as for lapses in professional conduct.

Chapter Two continues the discussion on ethics education in medicine and nursing, but with an emphasis on goals and methods of instruction. Traditional and nontraditional teaching-learning methods are reviewed, and simulation is introduced as a nontraditional method. The chapter includes a presentation of systematic reviews of curricula in medicine and nursing demonstrating the current goals, practices and pedagogical approaches in ethics instruction as well as a review of the effectiveness of these strategies. The chapter explicates the virtue/skill dichotomy in ethics education.

Furthermore, the chapter introduces the correlative goal of ethics education to improve quality of care. The chapter contains a discussion of the codes of ethics from both medicine and nursing and examines their significance and relevance to ethics education.

Chapter Three discusses quality of care in health care and presents its importance as correlative to the fundamental goals of ethics education. Various perspectives of defining quality of care are presented and the perspective of the patient and healthcare professional are examined. The chapter argues for the contributions of ethics education to quality of care, particularly when quality of care is defined from the patient's perspective. The chapter introduces the potential for simulation-based ethics education to reduce patient risk or harm and to better prepare an ethically-sensitive, reflective and prudent healthcare professional, consequently facilitating improvement in the quality of care for patients.

Chapter Four begins by establishing a definition of virtue ethics as it relates to medicine and nursing. The importance of virtue to the healthcare practitioner is established and the specific virtues that are essential to practice, such as compassion, integrity, honesty and prudence are explained. The chapter briefly addresses the age-old question of whether virtues can be taught, concluding affirmatively and reviews how virtues may be taught, supporting this with relevant literature. The chapter concludes with a discussion of the relationship of the virtuous healthcare professional to better quality of patient care.

Chapter Five develops the concept of simulation as a teaching-learning method in healthcare education. To begin, simulation is defined and a brief history of simulation in

medical and nursing education is shared. The types of simulation are described as well as the characteristic structure and components of simulation. The chapter includes theories that support the use of simulation as an effective pedagogy in medicine and nursing, many of these drawn from the discipline of education. Current research establishing the effectiveness of simulation is shared, along with a discussion of the advantages and disadvantages of simulation.

Chapter Six relates the use of simulation specifically to healthcare ethics education. The relevance and appropriateness of this method are presented as well as the advantages and disadvantages of simulation in ethics education. The chapter describes the ethical reasons for which simulation should be utilized in ethics education. Available research detailing the use of simulation to teach ethics to medical and nursing students is analyzed. The chapter concludes with suggestions for implementing ethics simulation with medical students and baccalaureate nursing students. These suggestions emphasize learning objectives that are consistent with the goals of ethics education.

In summary, the dissertation seeks to advance the literature in ethics education, virtue ethics, simulation, and quality of care for patients. It intends to establish the potential relationship between the development of morally sensitive and morally competent practitioners and improvements in achieving quality of care. It aims to show that while ethics education has many nonexclusive goals, an important and often overlooked goal of ethics education is improved quality of care for patients. Chiefly, however, this dissertation strives to demonstrate the strength of simulation as a method to teach healthcare ethics. Simulation meets the goals of ethics education: the development

of ethically-sensitive, reflective and ethically competent healthcare practitioners leading ultimately to improved quality of care for patients.

# **Chapter One – Ethics Education in Healthcare**

# A. Recent Influences in the Development of Ethics Education in Medical and Nursing Curricula

Ethics education in healthcare professions such as medicine and nursing has developed in response to advancements in the professions themselves as well as to critical social changes. It is important, therefore, to briefly identify those factors that have contributed more recently to the prominence of healthcare ethics in society. While an appreciation of the early roots of medicine and nursing and the beginnings of ethics in the professions would be interesting, this content is well documented elsewhere (Fowler and Tschudin 2006; Jonsen 2000) and does not hold specific bearing to the thesis. Therefore, the discussion will commence with more recent historical experiences that influenced the professions, beginning with the turbulent 1960s and 1970s, a period of scientific and social revolution that contributed significantly to the growth of healthcare ethics. The factors that contributed to this expansion include science and technology; patient rights and sociocultural value changes, economic market forces and health care delivery systems; and quality of care. These factors strongly directed the expansion of bioethics and, consequently, ethics teaching in medical and nursing programs. They are explored below.

To begin, the earliest forces that stimulated change in ethics teaching resulted from the discoveries and developments in science, medicine, and technology. The developments that began in the 1960s and which continue today at an explosive rate (Cassell 2007; Miles et al. 1989; Rosoff 2011), new areas of specialization (Miles et al.

1989) along with increasingly complex medical care and limited resources (Coutts 1991; Manson 2008; Miles et al. 1989) have heavily influenced health care ethics. Miles et al. observe that these changes present challenges to "medicine's and physicians" accountability to holistic, personal ends of patient care" (1989, 706). The explosion of scientific knowledge in health care and the increasing array of treatment options place an emphasis on the technical aspects of patient care and may present obstacles to the patienthealthcare provider relationship. Rosoff (2011) explains that the dominance of science over interactive skills in medicine may contribute to the loss of professionalism in medical students. New technologies also bring with them increasingly complex ethical decisions (Miles et al. 1989; Rosoff 2011). Gene therapy and stem cell therapies, for example, are two innovative techniques that have unknown outcomes and create unique ethical dilemmas for healthcare practitioners.

Second, shifts in values since the 1970s having to do with the rights of individuals (Cassell 2007), the conception of the patient as a "person" (Cassell 2007), and the deemphasizing of paternalism (Campbell, Chin, and Voo 2007; Cassell 2007) have also greatly contributed to the increased emphasis on healthcare ethics and ethics education. Patients are more aware of their rights and have better access to healthcare information, with many individuals having improved their health literacy (Campbell, Chin, and Voo 2007; Miles et al. 1989). Healthcare practitioners need to have a greater understanding and appreciation of patient agency and respect for patient autonomy as these have transformed the dynamics of the patient-provider relationship, leading to a softening of paternalism and moving toward a model of shared decision making (Campbell, Chin, and Voo 2007; Cassell 2007; Miles et al. 1989). Cassell (2007) challenges healthcare

professionals to move forward from the basic understanding of patient autonomy to a deeper recognition of the relationships that individuals hold, the influences of others upon decision making, and their relationships with healthcare practitioners. The past few decades have also been marked by an enhanced understanding and appreciation of cultural diversity and sociocultural differences. The Joint Commission (2010) has responded to the challenges that a diverse population brings and developed standards for healthcare professionals in cultural competence and patient communication to assist in overcoming linguistic and cultural barriers that may contribute to substandard care. Problems in obtaining informed consent are well recognized where there are language proficiency issues. Healthcare practitioners are obliged to understand the importance of such changes, shifts and differences and their impact on the interactions with patients, in general, as well as their obligations to protect patients' rights and respect differences.

Third, economic market forces have contributed to changes in healthcare practices, creating a new health care delivery model in the United States in the form of managed care. The marketplace disposition of managed care has brought about many unique ethical issues and stimulated changes in ethics teaching as a result. Present economic market forces reward efficiency and threaten "the primacy of patient welfare" (Pellegrino and Thomasma 1993, 170). Pellegrino and Thomasma observe that such an approach places the physician in the position of choosing between service to the patient or to the institution and acting in such a way as to lessen patient harm (171). Physicians have traditionally focused on the care of the individual, but the practices of the marketplace in many instances shifts the emphasis to the welfare of all members of the organization. Critiques of managed care point out that the imposed constraints on

physicians' autonomous decision making also have ethical ramifications. Costcontrolling measures such as restrictions on testing and consultations with specialists may contribute to negative patient outcomes (Agich and Forster 2000). Making clinical decisions that are based on less information than the physician considers necessary places both the patient and the physician in a difficult situation. The physician has less autonomy in such situations but the physician's obligation to the patient to practice beneficent, just and competent care remains. Additionally, physicians and other healthcare professionals who hold financial interests in healthcare corporations or who are encouraged to contain costs in the interests of the bottom-line, for instance, create ethical concerns such as unavoidable conflicts of interest (Pellegrino 1998). Pellegrino argues, "Whether for-profit or not, these arrangements expose the patient's vulnerability to exploitation, under the guise of efficiency or public service. It is illusory to think that physician-administered plans will be more respectful of patient welfare" (1998, 109). Benner (1998) concurs, critiquing the commodification of health care and its effects on patients as consumers. Pointing to profits as the driving principle in the healthcare industry, Benner observes that these changes increase vulnerability in patients. In the present market-driven healthcare industry, patients are expected to act as consumers and protect their own rights and interests, yet many individuals are unable to defend their needs or competently navigate the complex system because of illness, age, reduced language proficiency, frailty, and most notably, being uninsured.

Economic forces have a direct effect on the day-to-day operations of a healthcare organization itself, giving rise to considerable ethical dilemmas and leading to an increased need for ethically aware and sensitive healthcare professionals. In the United

States, for example, higher nurse-patient staff ratios, increasingly complex patients and the rapid transfer of patients from one setting to another encumber the typical healthcare institution. The resulting moral problems have been well documented in the literature and are briefly discussed below. The practices, which are designed to increase efficiency and reduce expenditures, increase the vulnerability of patients and threaten patient safety and the delivery of quality patient care (American Association of Colleges of Nursing 2002). Describing the effects of measures to reduce nurses' time with patients and families in an effort to reduce costs, Benner asserts that "adding profit incentives to "process" patients more quickly further undermines the *moral arts* of attentiveness and care" (1998, 124, italics added). Storch agrees, characterizing the present economic model in health care as one that "overwhelms a 'caring emphasis" (2009, 555). She reports on the stresses placed on nurses who fault the business model for the resulting problems, such as "dehumanizing practices, unnecessary suffering of patients, approaches that see some patients as more deserving than others, and failure of the system to deal with incompetence of health professionals" (556). White provides vivid accounts by nurses of moral conflicts that occur when nurse-patient communications are limited due to time constraints, resulting in poor caring practices, or when forced to choose between providing "insufficient care or no care at all" (1998, 19). She notes that nurses report frustration and demoralization as they struggle to provide more complex care to greater numbers of patients with insufficient resources. The changes brought about by the present economic models in healthcare and the restructuring of the healthcare delivery system have profoundly affected the development of ethics and ethics education in the past few decades, particularly in relation to justice principles such as equity, access to health care,

allocation of resources as well as professional ethics issues, such as conflicts of interests and moral distress.

Fourth, the prominence of quality of care initiatives in health care over the past two decades has also has influenced ethics education (Campbell, Roland, and Buetow 2000; Grol 2001). The ethical behavior of healthcare professionals bears a strong relationship to quality of care outcomes. Certain process measures of quality of care that are evaluated from the patient's perspective, such as the effectiveness of interpersonal care, can be enhanced by the actions of healthcare professionals (Campbell, Roland, and Buetow 2000). Van Mook et al. (2009) observe that studies reveal a positive relationship between the interpersonal skills of physicians and patient compliance, a quality of care indicator, concluding that efforts to develop professionalism may thereby improve patient outcomes. Other initiatives to enhance the quality of patient care are directly related to the view of the patient as an autonomous decision maker (Brook, McGlynn, and Shekelle 2000; Grol 2001). Grol (2001) notes that measures to support the patient as decision maker, such as increasing an individual's involvement and control of his or her care, are assumed to lead to improved adherence to the treatment regimen and better outcomes. In a review of research studies on physician-patient relationships and participatory decision making (an indicator of respect and autonomy), Beck, Daughtridge and Sloan (2002) report that the results revealed that patients who are more involved in the decision making process have more positive health outcomes. The relationship between quality of care and ethics education will be further developed in Chapter Three.

It is clear to see how medical and technological advances, shifts in values and increased recognition of patient autonomy, economic market forces and changes in

healthcare delivery and also quality of care concerns underscore the growing importance of healthcare ethics and ethics education in the present environment. The complex healthcare environment requires professionals who are better prepared to respond effectively to the maze of conflicting values and responsibilities. The professions of medicine and nursing have responded to the ethical challenges that this changing healthcare environment presents, recognizing the need for effective preparation in ethics. The accrediting agencies for schools of medicine and baccalaureate nursing programs require that ethics instruction be included in their respective curricula, and as a result, increased attention has been directed to the curricular goals and methods of instruction. Nonetheless, as this dissertation will review in more detail in Chapter Two, there is a lack of consistency in the goals or methods for the ethics curricula in medical or nursing programs. Furthermore, as the chapter will explore in a later section, despite the development of explicit measures to counteract the problems that the external environment presents, there is evidence that problems continue to exist and need to be satisfactorily addressed. The following section will provide a brief summary of the development of ethics education in medicine and nursing to set the stage for a discussion on the growing importance of ethics education in medical and nursing programs.

#### B. History of Ethics Education in Health Care

A brief description of the more recent history of medical ethics, nursing ethics, and ethics education in these professions is useful in preparing for the central discussion of an effective strategy for teaching ethics to medical and nursing students. In light of the history of medical ethics as a discipline preceding nursing ethics, the discussion will

begin with the history of medical ethics and ethics teaching in medicine. Jonsen's *A History of Bioethics as Discipline and Discourse* (2007, 3-16) provides the background for the following section.

Medical ethics has followed a steady development for most of the history of medicine. For centuries, up until the middle of the twentieth century, medical ethics conveyed fairly constant virtues and values that accompanied the traditional consideration of the physician-patient relationship (Jonsen 2007, 3-4). Medical ethics began a significant evolution around the mid-twentieth century, however, as a series of events took shape in the areas of human experimentation and science and technology. Early on, the Nuremberg Code in 1947 helped to define the guidelines for medical experimentation on humans, which provided the principles for medical research that are known today. Long-held philosophical perspectives in medicine were challenged by Joseph Fletcher in his 1954 text, Morals and Medicine, where he introduced a form of utilitarianism in the analysis of divisive medical topics, such as euthanasia and abortion (4). The 1950s were also marked by the introduction of the artificial ventilator, a technology that delayed or prevented cardiopulmonary death, but raised profound ethical concerns on consciousness. As Jonsen notes, anesthesiologists turned to Pope Pius XII for assistance with the unprecedented moral problems that accompanied the implementation of this technology. As a result, the traditional Catholic doctrine on extraordinary and ordinary means was raised in a new context and became influential in ethics dialogues regarding life-saving or life-extending measures (4-5).

Following the early seeds of change, the 1960s became a pivotal time in the development of medical ethics, and consequently, to the advancement of medical ethics

education. The discovery of hemodialysis in 1961 for patients with end-stage renal disease brought with it questions on the allocation of scarce resources and discussions on social utility (Jonsen 2007, 5). With insufficient numbers of hemodialysis machines to meet the demand, the ensuing debates centered on the means to select potential candidates. The moral arguments presented included utilitarian principles of social value; however, these discussions were abandoned and the selections were made by a lottery system. The successes in organ transplantation in the late 1960s, following the development of cyclosporine, raised ethical questions on excessive medical expenditures as well as an unprecedented ethical question in the case of heart transplantation on irreversible coma and the determination of death. The discussions led to a new determination of death – "brain death" – that opened up important dialogue in healthcare ethics (8).

The ramification of this rapid growth in science and technology was evident in medical curricula and witnessed by medical faculty. Dissatisfaction with medical curricula that emphasized the burgeoning medical knowledge and technical developments over the traditional humanistic stance in medicine led to a movement in this period to introduce humanities and social sciences to the curricula to counteract potential dehumanizing tendencies (McElhinney and Pellegrino 2001). Until this time of rapid change in medicine, ethics was taught informally and by implied methods, typically through professional enculturation and role modeling that stressed " 'bedside manner' and professional etiquette" (Gillam 2009, 585). One of the pioneers in medical ethics, Edmund Pellegrino, depicts the beginnings of bioethics and ethics education in *The Origins and Evolution of Bioethics: Some Personal Reflections* (1999) from which the

following historical perspective of medical ethics and ethics education is culled. He describes three stages of bioethics: educational, ethical, and global. The first phase, the education stage, is relevant to this discussion. It emerged at a time in which medical educators sought to counter the "perceived 'dehumanization' of medicine" that accompanied the scientific and technological advances and the trend toward specialization (Pellegrino 1999, 73). Concerns about the loss of the physician's humane, compassionate nature resulted in efforts to include courses on values, humanities and social sciences in medical school curricula. At the same time, ministers from medical school campuses formed similar undertakings to prevent the erosion of humanized medicine. Medical educators and campus ministers joined their efforts at national conferences and worked to enable the "ideal of the humanist physician" (77). One result of this collaboration was the formation in 1969 of the Society for Health and Human Values (SHHV) and its program arm, the Institute on Human Values in Medicine (the Institute), formed shortly after in 1971 (McElhinney and Pellegrino 2001). The SHHV along with The Society for Bioethics Consultation and The American Association of Bioethics eventually merged in 1997 to form the American Society for Bioethics and Humanities (ASBH).

Pellegrino (1999) attributes the work of the Institute as instrumental to the progress in medical ethics education that occurred in the 1970s. The Institute sponsored programs on faculty and program development and provided consultations to medical schools that were seeking to integrate values, ethics, and humanities into their curricula. Interdisciplinary programs for medical faculty and humanists were also offered. Medical and nursing faculty, humanists, and social scientists were selected for fellowships offered

through the Institute for advanced study in ethics, humanities or social sciences (McElhinney and Pellegrino 2001). In 1976, the Institute established The Journal of Medicine and Philosophy for the publication of scholarly writings on practical and theoretical ethics (Pellegrino 1999).

By 1976, forty-four percent of the medical schools (63 of 113) had required courses in ethics; by 1983 all but one medical school offered courses in medical ethics, although the courses were only required in approximately thirty percent (38 of 127) of the schools (Pellegrino et al. 1985, 52). During the 1970s, medical ethics education was in a period of tremendous change, with variation in its acceptance and the teaching methods employed (Pellegrino et al. 1985). Nonetheless, at the end of this decade marked by great scientific progress and social change, Pellegrino (1999) reports that human values, the humanities and medical ethics had converged. While many medical schools had established separate or integrated ethics programs by the late 1970s, Fox, Arnold and Brody (1995) report that by 1993, every medical school was required to teach ethics as part of the accreditation standards.

As previously noted, early ethics education in medicine relied more on implicit mechanisms, such as role modeling, for teaching ethics with little attention on explicit mechanisms such as formal classes or case studies (Gillam 2009). Efforts to enhance ethics education in medicine, however, were initially contested as faculty objected that first, it took time away from scientific knowledge that was rapidly expanding, and second, that ethics could not be taught (Bickel 1986; Veach and Sollitto 1976). Concerns regarding who would be best suited to teach ethics in medical programs, whether content should be integrated or taught as a stand-alone course and how it should be taught were

debated in the literature (Eckles et al. 2005; Fox, Arnold, and Brody 1995). A review of the literature from 1978-2004 by Eckles et al. (2005) revealed more support for ethicist-philosophers and physicians in a multidisciplinary team to teach ethics, while one article recommended that the faculty have additional experience in teaching methods. The review showed a consensus opinion on an integrated ethics curriculum, both horizontally and vertically. Importantly, the review demonstrated that ethics instruction should be considered "a process" encompassing medical students' four-year program (Eckles et al. 2005, 1148). Many studies have been conducted since the start of more formalized ethics instruction, which have evaluated the goals and methods of ethics education. These findings will be reviewed in Chapter Two. As ethics education efforts progressed, discussion on the inadequacies of implicit measures, such as role modeling, emerged. Problems inherent to the 'hidden curriculum' or the teaching that occurred outside of the formal curriculum were discussed and studied as well (Campbell, Chin, and Voo 2007; Hafferty and Franks 1994; Lakhan et al. 2009). This will be explored later in this chapter.

By comparison, nursing ethics and ethics education in nursing followed a less focused path than medical ethics education due in part to nursing's later development as a profession. Ethics in nursing was also greatly influenced by the developments in bioethics and medical ethics. The writings of Fowler and Tschudin (2006) as well as Storch (2009) are helpful in illuminating the development of nursing ethics and ethics education and are consulted in this section.

To begin, it is interesting to note, that similar to medicine, the development of nursing ethics and ethics education was affected by the sociocultural and sociopolitical changes and events occurring in the United States in the 1960s and 1970s albeit for

different reasons and with different results (Fowler and Tschudin 2006; Storch 2009). While nursing ethics and nursing ethics instruction were influenced by the advances and complexities in health care and medicine as well as by a new appreciation of patient rights, nursing as a female-dominant profession also responded to the changing social expectations and roles of women. Fowler and Tschudin observe, "The nursing moral literature, both indirectly and directly, reflected the emergence of women as more fully recognized participants in the professions and in society, as well as the emergence of nursing as a scientific profession struggling for public recognition as such" (2006, 16).

The focus of ethics in nursing initially centered on virtue ethics and service to others, with much of the emphasis on character and etiquette within those relationships that were central to the practice of nursing (Fowler and Tschudin 2006; Storch 2009). Early ethics education emphasized the feminine traits and the duties of the nurse to those to whom she was responsible - physicians, patients, colleagues, and herself. The good nurse was courteous, neat, and obedient and devoted to her duties (Storch 2009; Fry 2004). Some of the expected behavioral traits came from the profession's early religious influences on caring practices (Storch 2009). Florence Nightingale, known to be the founder of nursing, considered nursing to be a vocational calling and set Christian standards of service in nursing (Fry 2004). Nursing also carried over behaviors influenced by Nightingale's military service, such as orderliness, fidelity, the value of the hierarchy, and respect for duty (Storch 2009, 551). This now archaic stance was captured in nursing journal articles covering issues such as proper public attire and cooperation with physicians (Fowler and Tschudin 2006).

Ethics education in nursing programs, however, predated ethics education in medical schools. Ethics education has been valued from the profession's inception and well documented in many early texts and journal articles (Fowler and Tschudin 2006; Fry 2004). Fowler and Tschudin depict the history of ethics and ethics education in *The* Essentials of Teaching and Learning in Nursing Ethics (2006, 13-16), the resource for the section below, unless otherwise indicated. The first ethics text was written very early, in the late 1800s, and focused on virtue ethics in the context of relationships. As early as 1916, in California, the Board of Registration of Nurses had an ethics education requirement for all schools of nursing, requiring that ethics be taught in five out of the six half-years of a three-year program. The topics for coursework were broad reaching and included clinical ethics as well as sociocultural ethical concerns on housing and poverty. By the next year, the National League for Nursing Education required ten hours of ethics for all second-year nursing students, suggesting that courses include ethics theory, personal, professional and clinical ethics, and social ethics. From the period of 1900-1965, sixty-five text books on ethics in nursing were written by nurses, priests or social workers and covered such topics as the nurse's duty to the patient first, maintaining nonjudgmental attitudes and establishing relationships of trust and empathy (Fowler and Tschudin 2006; Storch 2009). These texts often served as the basis for ethics education coursework. Additionally, almost 450 articles were published on ethics in nursing from 1900-1980 (Fowler and Tschudin). The first empirical study on nursing ethics was conducted as a master's thesis by Vaughan in 1935 describing the moral problems experienced by nurses (in Fry 2004). The results showed that the most commonly occurring ethical problem for nurses at that time involved nurse-physician cooperation,

which Vaughan attributed to changes in the nursing role and responsibilities. Later research in nursing ethics studied the values and attitudes of nurses, moral development, ethical decision making, and ethical behavior (Fry 2004).

It was not until nursing instruction moved from diploma programs to baccalaureate nursing programs that there was a shift away from the predominance of virtue ethics in curricula and in nursing literature (Fowler and Tschudin 2006). With the expansion of the nurse's role and responsibilities to include independent work and increased accountability and responsibility for ethical decisions, there was a corresponding shift in nursing ethics from virtue ethics to duty-centered ethics (Fry 2004). By the 1970s, Fry reports that the relevant ethical problems now included protecting the rights of patients, balancing benefits and harm and allocating nursing resources. Frameworks for ethical decision making were critiqued, specific ethics content was identified, and analysis of ethical issues was introduced (Fry 2004). The ethics instruction of nurses changed to meet the new expectations of the profession as it continues to do so today.

# C. The Importance of Ethics Education Today

As the discussion of the influencing factors on healthcare ethics suggests, the present climate calls for healthcare professionals who can competently navigate an increasingly complex healthcare system and sociocultural environment that present more complex ethical dilemmas. Medical and nursing educators have responded positively to the substantial challenges identified above. A review of the ethics education literature reveals, however, that medical and nursing students are confronted by additional

difficulties that arise from the learning environment and the professional practice milieu itself which hinder the attainment of ethics educational goals. These problems include the 'hidden curriculum,' ethical erosion, moral distress and continued breaches in professional conduct- concerns that are more closely related to psychological influences, attitudes and communication patterns than to the knowledge and decision making skills that are typically emphasized in a traditional ethics curriculum. Although educators have been aware of these challenges to the moral development of medical and nursing students, their continued presence demonstrates a need for enhanced ethics educational programs today. A review of the specific challenges will establish the need for targeted curricular efforts that employ more effective instructional methods to defend against the negative influences inhabiting the learning and practice environment.

The first problem, the 'hidden curriculum,' is the informal teaching environment of an educational program and includes those lessons that fall outside of the planned formal instruction, which are unintentionally transmitted to students and often bear more weight than the formal curriculum (Hafferty and Franks 1994). It should be noted that the medical literature is quite extensive on this topic; nursing literature, while acknowledging the presence and effects of the informal curriculum, is limited here. The research in medicine by Hafferty and Franks (1994), though, is relevant to both professions. The hidden curriculum is comprised of implicit rather than explicit teaching methods and learning situations, such as those found in clinical learning experiences, which contribute to the socialization of medical and nursing students into their professions. While implied teaching methods, such as role modeling or clinical experiences are often positive, productive and essential, they are also an uncontrollable and insufficient means of ethics

education for healthcare professionals and can wield greater influence on the students' "moral enculturation" and ethical development than the formalized curriculum (Hafferty and Franks 1994, 861). Hafferty and Franks point out that only a portion of what is necessary for medical students to learn about ethics can be taught through formal instruction. Students internalize values and beliefs in experiences outside the classroom. Unfortunately, with poor role modeling, ethical norms, behaviors and attitudes may be incompletely or incorrectly communicated either overtly or indirectly.

Role modeling may be an effective method for socialization into the profession; however, as noted, this implicit method may be counterproductive and lead to harmful results when students are exposed to negative role models. Lempp and Seale summarize some of the negative outcomes in medicine associated with the hidden curriculum as the "loss of idealism, adoption of a 'ritualised' professional identity, emotional neutralisation, change of ethical integrity, acceptance of hierarchy, and the learning of less formal aspects of 'good doctoring'" (2004, 770). A qualitative study by Lempp and Seale on this phenomenon revealed that while ethics courses stress the concepts of professionalism, over half of the medical students in the study reported experiencing humiliation while in the clinical settings (771). Half of the students observed competition rather than cooperation among their peers in clinical settings and many described a prevailing hierarchical atmosphere (772).

Hospitals are complex institutions with set behaviors and norms for its members, which serve as the learning ground for the socialization of students (Storch and Kenny 2007). While these learning environments are necessary for the development of moral skills and behaviors, the unplanned learning in the clinical settings can challenge the

formal ethics curriculum. Gallagher notes that the practice environment "may challenge or contradict learning on the explicit curriculum" (2006, 225). Hafferty and Franks (1994) argue that medical school curricula must acknowledge the influence of the informal learning environment in planning ethics programs. They stress that the informal environment, such as interactions with peers, clinical experiences and role models are more powerful influences than formal lessons in ethics and at times "can be antithetical to the goals and content of those courses that are formally offered" (1994, 865). Nevertheless, the clinical environment is essential for internalization of values. Hafferty and Franks' well-known article on the hidden curriculum is almost two decades old, yet the effects of this tacit curriculum continue to be a concern for educators (Coulehan and Williams 2001; Gillam 2009; Mahood 2011). A lack of interest in ethical concerns, disrespect toward patients and staff, avoidance of patients' family members, cultural insensitivity and erosion of ethical principles are just some examples of the attitudes and behaviors that continue to permeate the learning environment (Mahood 2011), creating negative influences on the moral development of students. Ethics educators, therefore, have a responsibility to recognize this phenomenon and integrate teaching about the informal curriculum into the formal education plan (Gillam 2009; Hafferty and Franks 1994; Mahood 2011). They must support students and assist them with the skills to confront these negative influences on their moral and professional development.

What is more, educators need to consider that the hidden curriculum extends beyond the clinical setting and interactions with other healthcare professionals. Ethics educators themselves convey an informal hidden curriculum through the decisions they make related to the ethics content included and the approach and attitudes adopted in

teaching ethics (Gillam 2009; Hafferty and Franks 1994). For example, educators may stress ethical decision making skills over virtue ethics, attributing more worth to clinical analytic skills than to the attributes of the physician or nurse, sending the message that the character or attributes of the professional are less important than one's analytical skills. The time allotted o ethics instruction may also convey the value that ethics instruction holds to the educator. Ethics educators in the health professions must be aware of the influence that these implicit, informal measures have in the socialization of the student to the profession and the effects of these on the development of the moral agency of the physician and nurse. If the informal curriculum wields such influence, then ethics educators should attempt to mitigate its negative effects. Efforts that attend to this informal, 'hidden' learning environment are directly addressed in the accreditation standards for medical schools, which are presented in Chapter Two.

The second problem, ethical erosion, which is a loss of professionalism and dulling of moral virtues, has been observed in medical students and is closely related to the discussion on the hidden curriculum (Charon and Fox 1995; Crandall and Marion 2009; Hojat et al. 2009). While the phenomenon of ethical decline and its associated cynicism and skepticism may occur in nursing students, it does not seem to occur with notable frequency in nursing. In a summary of the Carnegie Foundation National Study of Nursing Education, researchers note that "very little cynicism" was observed in nursing students and nursing faculty (Benner et al. 2008, 476). Ethical erosion in medicine has been attributed in part to the emphasis of science and technology over the humanities in medical school curricula. As a result, students feel excessive pressures to acquire more knowledge in an ever-expanding discipline (Rosoff 2011). Ethical erosion

may be closely related to the informal curriculum described above, with negative influences in the learning environment breeding cynicism in medical students (Charon and Fox 1995). Significant to this discussion, Charon and Fox (1995) provided specific feedback from students, which illuminated the degree of ethical erosion experienced by students. In a report of medical students' responses to an essay assignment, "How can medical students best develop ethical thinking and behavior?," the students' essays revealed how they felt about their preparation in ethics. The authors noted that many students expressed concern "that their professional socialization abrades their ethical sensitivity and subverts their moral development" (Charon and Fox 1995, 767). Students worried that their education might erode their idealism and dehumanize them. Students shared concerns about witnessing or unwillingly participating in possible ethical wrongdoing and feeling powerless to avoid or stop their involvement in these situations. The essays identified many ethical challenges that medical students face, among these, withholding information, representing themselves falsely as physicians, and observing breaches in professionalism, such as misconduct or impairment of colleagues or medical staff. Students feared ethical erosion and wondered how to manage the stresses of being a physician without losing the virtues they value. Notably, the students identified some measures to thwart this phenomenon, suggesting that ethics education be included in all four years of a medical program, and in addition to principlism, instruction should include virtue-based ethics, casuistry and narrative ethics. Charon and Fox summarized the students' comments, noting, "ethics teaching should be rigorous, required, and examined" (1995, 767). Additional recommendations to thwart ethical erosion included methodologies that extend beyond lecture, involve students and include collaborative

teaching methods. Small group discussions were recommended to allow adequate time and provide a safe environment to examine views. Students requested dialogue that encouraged the "intricacy, ambiguity, and pluralistic differences intrinsic to ethical issues" (1995, 767).

In a review on identify transformation in medical students Cohen et al. (2009) outlined the stressors associated with the learning process and medical training environment that may contribute to a decline in professional role development, moral development and ethical erosion. While many role models inspire students with their professional attitudes and behaviors, regrettably, the presence of cynical, arrogant, and aloof role models has been well-documented and cited as a significant factor in identity diffusion among medical students (Cohen et al. 2009, 49). The authors reported that students are also exposed to humiliating situations in their clinical work, as well as in situations in which they must compromise their own values to "fit in" and avoid being chastised by health team members (49). In addition, the review found that students are negatively influenced by a system that ignores the indifference and disengagement of fellow students who cheat, make moral compromises and are driven by financial gain. Students require assistance in maintaining their ethical values system. Given this significant evidence, the authors contended that educators are obliged to assist students in identifying and forming relationships with positive role models. Cohen et al. stressed the integration of teaching methods that incorporate discussion and reflection to mitigate the negative effects of the learning environment and suggest faculty development programs centered on producing effective role models for students.

Other studies demonstrate a specific decline in empathy during medical school, declining in third year of the program (Hojat et al. 2009) and in the clinical years (Chen 2007). Empathy decline is relevant here as a specific trait or virtue important to the physician. Hojat et al. observed that the decline in empathy was twice as great in medical students in "technology-oriented" specialties compared to their peers in "peopleoriented" specialties, and a higher decline was observed in men versus women (2009, 1187-88). Similar to the research on ethical erosion and loss of professional identify, the authors attributed this trend to lack of positive role models, "affective distance," and environmental factors, among others (1188-89).

The third problem, moral distress, is the cognitive, affective or physical suffering that occurs when one is unable to pursue what he or she believes is the right course of action because of constraints (Jameton 1984; Pendry 2007; Wilkinson 1987). The definition of moral distress has been expanded from the basic definition first proposed by Jameton and later Wilkinson that recognized the external constraints or barriers, such as hierarchy and communication structures, legal obligations, and resource allocation, as hindrances to the ability to act as a moral agent (Pendry 2007). Current definitions of moral distress also recognize that internal barriers, such as lack of skills or confidence, may contribute to moral distress (American Association of Critical-Care Nurses 2005; Pendry 2007). Moral distress has been documented and studied extensively in nurses since the 1980s and more recently studied in nursing students, medical students and physicians as they encounter obstacles that prevent them from acting on their moral decisions.

Medical and nursing students may experience moral distress if they are unsuccessful in attempting to respond to ethical concerns that they witness or in which they are involved. Research on moral distress in nursing students reveals that students report discomfort as they witness breaches of patients' rights, including autonomy, a lack of professional integrity, unsafe conduct, breaches in confidentiality and disrespect toward patients (Callister et al. 2009; Cameron, Schaffer, and Park 2001). Woods (2005) examined evidence from several studies that illustrated the moral passivity and powerlessness of nursing students in such situations, noting that students feared rejection by staff nurses. Doane et al. (2004) described nursing students' moral passivity as resulting from their vulnerability and lack of influence in clinical situations. What may begin as a lack of moral confidence in nursing students may progress to moral passivity, moral compromise and eventually, moral distress (Drought 2006).

Moral distress has been acknowledged as a significant problem in nursing, and as a result, the North American Nursing Diagnosis Association- International (NANDA-I) created "Moral Distress" in 2006 as a new nursing diagnosis (Carpenito-Moyet 2009). Examples of ethical problems that result in moral distress for nurses include the inability to effectively advocate for the patient in situations where the nurse's authority is restricted, limitations in providing the best care due to lack of resources, and end-of-life issues in which the nurse is required to provide aggressive care that the nurse deems unwarranted. Nurses also acknowledge problems in competing obligations to the patient and to the institution that contribute to moral distress. In a survey of everyday ethical issues in nurses, Ulrich et al. reported that almost three-fourths of the nurse respondents experienced ethical problems about which they felt could do nothing and over a third

reported feeling "powerlessness" (2010, 2514). The survey revealed that younger, less experienced nurses in direct care positions reported more stress related to ethics than more experienced nurses.

Medical students also report experiencing moral distress in the clinical environment. Lomis, Carpenter and Miller (2009) assessed the causes and levels of stress present in third-year medical students by examining the students' reflective writing assignments, which recorded situations associated with either stress for the students or exemplary behavior that could be modeled. The findings overwhelming demonstrated that communication problems were cited most frequently, and those that occurred within the care team were associated with higher distress. The findings showed, however, that appropriately managed communication problems were associated with lower levels of distress. Other sources of distress were related to the structure of the hierarchy, the presence of negative role models, allocation of resources, and lack of access to care. One item deserves particular mention. The review of the writing assignments showed a strong correlation between a student's remorse over a lack of action in responding to ethical problems and higher levels of distress. Conversely, students who overcame their lack of confidence or fear of reprisal and took action expressed lower levels of distress. The authors acknowledged the value of preparing students for situations associated with moral distress to help lessen the degree of suffering. They suggested informing students of the lower stress experienced by students who took action with non-confrontational approaches and creating strategies to assist students to manage these situations. Suggestions included incorporating communication exercises directed toward encounters

with colleagues and creating discussion groups that are "safe environments" to share experiences (111).

Nurses and physicians share similar experiences in moral distress, reporting that end-of-life care situations are frequent triggers of moral distress. While many physicians and nurses are not sensitive to the other's moral distress, several commentators have addressed the similarities in moral distress between the groups. Oberle and Hughes (2001) found more similarities in the levels of distress experienced by nurses and physicians than differences. They note that the concerns and problems are similar, but the expression of associated responsibilities is different due to role. Each group experienced moral distress in end-of-life care situations that stemmed from value conflicts. communication problems, challenges inherent in the hierarchical processes and limited resources. Differences in their respective roles, however, influenced their perspective of ethical concerns. Another study by Hamric and Blackhall (2007) showed that while both physicians and nurses experience moral distress in end-of-life care situations, nurses report more moral distress and more intense levels of distress in these situations than physicians. Both groups, however, identify powerlessness and conflicts between institutional goals and patient care goals.

The final challenge considered here is the continued presence of professional ethics breaches in informed consent, confidentiality, conflicts of interest (Devettere 2010), and disciplinary actions (Papadakis et al. 2004), supporting the importance of ethics education today. Some of these problems in professional ethics may result from external factors, such as the market-driven healthcare environment or the rapidly expanding cultural diversity of the population that present new challenges to healthcare

professionals. Yet, there is some evidence of a correlation between the conduct of a student and his or her future professional behavior. One significant study revealed a correlation between the unprofessional behavior of medical students and future breaches in conduct. To identify predictors of ethical misconduct, Papadakis et al. (2004) reviewed the disciplinary records of physicians practicing in the State of California from 1990-2000 who were graduates of the University of California, San Francisco (UCSF). Disciplinary actions of the Medical Board of California that were included in the study were negligence, fraud, inappropriate prescribing, unlicensed activity, patient endangerment as the result of the physician's use of drugs and alcohol, conviction of a crime, sexual misconduct and unprofessional conduct (2004, 245). One percent of the UCSF graduates were disciplined by the State Medical Board during this period, the majority for negligence that resulted from unprofessional behavior rather than from incompetence. Interestingly, the study found that the students who displayed unprofessional behavior while in medical school (as reflected in Professional Evaluation Forms) were twice as likely to have disciplinary action taken against them when in practice (248). The authors stressed that professionalism is a critical competency for medical school graduates and argued that assessment and evaluation measures of professionalism hold significance. Educators need to consider the evidence of the impact of unprofessionalism in students on the quality of care for patients in planning ethics teaching and nurturing the professional growth of students.

A review of studies by Baxter and Boblin (2007) on the unethical behavior of nursing students reflects that the number of incidents in the classroom and clinical settings has been increasing. The behaviors involve lying, cheating, plagiarism, deception

and misrepresentation, falsifying medical records (for instance, claiming to have done something that was omitted), and failure to report data, errors or incidents (20). The unethical behaviors reflect a lack of integrity and hinder the development of trusting relationships in both the academic and clinical environments and pose potential harm to patients. The authors reported that there have been no studies with nursing students, however, that demonstrate the effect of unethical academic behavior on patient outcomes or any studies similar to Papadakis et al. that demonstrate a correlation between unethical behavior as a nursing student and behavior as a nurse. Baxter and Boblin strongly suggested that educators "create and sustain a positive learning environment that promotes honesty and allows students to make mistakes and solve problems so they may develop their moral thinking" (2007, 24).

### Conclusion

In spite of efforts to develop an explicit, formal ethics curriculum that meets the educational needs of students, the problems outlined above have been persistent and pervasive in medical and nursing programs. In addition, they tend to fall outside of the traditional ethics curriculum, which gives emphasis to ethical dilemmas and decision making. To better address the needs of students, explicit, formal ethics instruction must acknowledge and address the effects of the hidden curriculum, ethical erosion, moral distress and professional breaches in ethics that surface in prelicensure programs. The curriculum should acknowledge the presence of these influences and the associated negative effects on successful professional development and socialization in medicine and nursing. Ethics educators should identify measures, such as training, as an integral component of an ethics curriculum in the health professions to mitigate their effects. To

some degree, the problems are far-reaching, involving informal learning environments and individuals for whom ethics educators have no reach, and as a result, require institutional or system changes for better outcomes. A discussion of measures at the systems level is beyond the scope of this dissertation. This dissertation argues, however, that specific, targeted educational methods are necessary to better equip students with the skills and attributes needed to effectively cope with the potentially demoralizing effects of the hidden curriculum, to lessen the inevitable moral distress that arises in clinical situations and to resist ethical cynicism or erosion. Educational methods are also needed to assist educators in identifying unethical conduct in students that may have serious consequences.

Moreover, explicit measures in ethics teaching are needed to develop the professional attributes and virtues of the nurse and physician and sensitize the practitioner to the ethical issues inherent in the current healthcare system. As the review of the health care system changes and challenges reveals, it is fraught with complexity and places patients at risk for depersonalized care. Behaviors that focus on conveying respect through shared decision making, interpersonal care, and improved communications are important for positive patient outcomes and are driven by the nature of the practitionerpatient relationship. Development of professional virtues, such as compassion and integrity, sensitize the healthcare practitioner to ethical problems and support the creation of a meaningful practitioner-patient relationship. Ethics teaching can support these positive elements in the practitioner-patient relationship by explicitly teaching and reinforcing professional virtues and values as part of the formal curriculum.

A specific education method, simulation, provides students the opportunity to practice and develop the virtues and skills needed for ethically sensitive practice. Simulation experiences can also be used to enhance the awareness of ethical problems and develop ethical analytic skills. Additionally, simulation can be used to explicitly counter the negative influences in the learning environment. The benefits of this experiential learning method as an adjunct to traditional ethics teaching will be argued in Chapter Six.

# **Chapter Two – Goals and Methods of Ethics Teaching**

### A. The Goals of Ethics Education in Medicine and Nursing

It is important to begin with a discussion of the goals of an educational program because the goals or program objectives serve to direct the content and teaching methods that are used. This chapter identifies the goals of ethics education in medical and nursing programs and illustrates effective teaching methods for goal attainment. Section i begins with a review of the professional competencies and educational outcomes in ethics in medical and baccalaureate nursing programs. Section ii examines the codes of ethics for both disciplines, which through their identification of ethical practice provide the goals of ethics education. Section iii includes a review of curricular practices in ethics education in medical and baccalaureate nursing programs, identifying goals that are set as ideals, but perhaps not reached. Section iv identifies short-term and long-term goals in ethics education that serve to guide an ethics curriculum. Section v examines the fundamental goal of ethics education, that of improving the quality of care for patients.

 Review of the Professional Competency Standards and Educational Outcomes in Medicine and Nursing

This section is supported by the literature of the Liaison Committee for Medical Education (LCME) report *Function and Structure of a Medical School: Standards for Accreditation of Medical School Education Program Leading to the M.D. Degree* (2011); the American Association of Colleges of Nursing (AACN) publication, *The Essentials of Baccalaureate Education for Professional Nursing Practice* (2008); *The American Society of Bioethics and Humanities (ASBH) Task Force on Ethics and Humanities*  *Education in Undergraduate Medical Programs* (2009); and the Accreditation Council for Graduate Medical Education's (ACGME) "Six General Competencies" for resident education (Accreditation Council for Graduate Medical Education 2011). The global perspective on ethics education is captured with the UNESCO *Core Curriculum in Bioethics* (2008). In addition, the American Society for Bioethics and Humanities (ASBH) *Core Competencies for Health Care Ethics Consultation* (2006) is also consulted in light of its relevance to the ethics education of the developing physician and nurse.

Accrediting organizations in medicine and nursing have responded to both the burgeoning field of healthcare ethics and to the increase in clinical, professional and organizational ethical problems by requiring that the core curricula of the respective programs include instruction in ethics. The LCME report (2011) on Standards for Accreditation of Medical School Education Program Leading to the M.D. Degree includes criteria for ethics education in several of its educational objectives and reveals a strong commitment to ethics instruction. The language of the report carefully employs the terms "must" and "should" to indicate the degree of required compliance with the standards – "must" to reflect necessary compliance and "should" to reflect compliance except in "extraordinary and justifiable circumstances that preclude full compliance"(1). Educational objective ED-23 specifically addresses ethics instruction: "A medical education program must include instruction in medical ethics and human values and require its medical students to exhibit *scrupulous ethical principles* in caring for patients and in relating to patients' families and to others involved in patient care" (10, italics added). Furthermore, this objective stresses that education in ethics, values and

communication skills should occur before the student engages in patient care experiences. In addition to honesty, integrity and confidentiality, the most recent edition of the accreditation standards defines "scrupulous ethical principles" to include the virtue of respect and extends this responsibility to patients and their families, peers and other healthcare members (11). Of particular relevance to this thesis is the directive's call for the use of "formal instructional efforts" in the assessment and reinforcement of ethical principles (11).

This educational objective makes a strong statement for ethics education and aligns well with the AMA Code of Medical Ethics. The LCME report (2011) emphasizes the professional virtues and values that are included in the code of ethics such as honesty and integrity and calls for faculty to ensure that students adhere to ethical principles. It sets standards for educators to provide instruction on professional behaviors, attitudes and skills, communication skills, and ethical theory. Significantly, the LCME standards call upon educators to provide this instruction to students *prior* to engaging in the care of patients. Although the LCME does not specify the type or amount of instruction in ethics, the criterion emphasizes that the ethical behavior of medical students should be examined closely and evaluated. Objective ED-46 emphasizes the need for evaluation measures and specifically addresses the need for programs to collect outcome data on the "professional behavior of its graduates" (16).

Ethics education is supported in several other LCME standards, each of which addresses a specific area of concern. Objective ED-17-A focuses on the need for instruction in research ethics; objective ED-19 on communication skills with patients and families, colleagues and other professionals in health care; objective ED-20 on common

societal problems, such as violence and abuse; objective ED-21 on diverse cultures and belief systems, and objective ED-22 on gender and culture bias. Important to this thesis, the LCME standard MS-31-A directly stipulates the creation of a learning environment that "promotes the development of explicit and appropriate professional attributes in its medical students (i.e., attitudes, behaviors, and identity)" (2011, 21). This standard describes the need for careful management of both formal learning environments, as in planned activities, and informal learning environments, which include "the attitudes, values, and informal 'lessons'" (21), acknowledging the hidden curriculum. Educators are encouraged to enhance the learning environments and work to diminish the effects of negative influences on professional development. Objective IS-16 directs educators to develop an environment that engages students in the development of cultural competency and attributes that contribute to social responsibility, such as altruism and social accountability.

The ASBH task force on Ethics and Humanities Education in Undergraduate Medical Programs has also addressed medical ethics education. In 2005, the task force responded to the LCME requirements for the core content in medical school curricula. By 2009, the ASBH task force had developed a report to assist educators in the design and development of a curriculum that would enhance ethics education. The task force recommendations address specific LCME education objectives and provide a curriculum outline to assist educators in developing bioethics and humanities content for their programs. The learning objectives encompass both the skills and attitudes essential in ethics education. The task force also developed a "Template for Topic Development" to aid medical educators in curriculum development and evaluation (ASBH 2009). The

template acknowledges the domains of learning and addresses knowledge, skills and attitudes distinctly. Ethical skills and attitudes are assessed for knowledge ("know how to"), demonstration ("show how to") and competent performance ("do") (5).

Graduate medical education continues this emphasis on the importance of ethics education by establishing core competencies for residency programs. These competencies commonly referred to as the "Six General Competencies" were introduced in 1999 by the ACGME to clearly identify the expected knowledge and conduct of residents upon program completion (Doukas 2006). Many of the requirements set by the ACGME stress professionalism and ethical knowledge, skills and attitudes (ACGME 2011, IV.A.5). An examination of these competencies reveals, for example, the prominence of professional attributes and attitudes, such as compassion, respect, sensitivity and responsiveness, selfappraisal, ethical principles, communication skills, cultural competency, advocacy, and risk-benefit analysis, among others (Doukas 2006).

A review of The Essentials of Baccalaureate Education for Professional Nursing Practice (The Essentials) of the AACN (2008) shows a similar emphasis on ethics education in nursing. The Essentials provides competency standards for baccalaureate nursing education curricula that are required for accreditation. The recent revision of The Essentials in 2008 was written to address the Institute of Medicine's report on "core knowledge required of all healthcare professionals" (3). The document is comprised of nine Essentials, each of which represents a particular outcome for the baccalaureate nursing graduate. Essential VIII: Professionalism and Professional Values is pertinent to the teaching of ethics. This Essential addresses ethics as integral to nursing and cites the importance of respect, honesty, advocacy, accountability and civility to nursing practice.

Essential VIII also outlines the professional values of altruism, autonomy, human dignity, integrity and social justice. The document acknowledges nursing as a caring practice and emphasizes the significance of empathy, compassion and a caring stance to nursing while recognizing the relationship of professional values and "value-based behavior" to the nurse's ability to provide quality care (27). It is clear from The Essentials that there is a consensus view of nursing as a value-laden practice. Nursing programs are obligated to prepare graduates to practice with professional virtue and values. The report recommends that baccalaureate nursing programs prepare graduates who model professional behavior, are compassionate, accountable, self-reflective, respectful of confidentiality and privacy, employ ethical decision making frameworks and assist others in resolving ethical dilemmas (27-28). In addition, nursing students are to be prepared to identify risks that influence their professional and personal choices, prevent unethical and unsafe practices. and promote the image of nursing. The report offers suggestions for curricular content that include content on the code of ethics, ethical frameworks, communication, human rights, and moral agency.

The global perspective on ethics education as represented by the United Nations Educational, Scientific and Cultural Organization (UNESCO) reveals an emphasis on both ethical sensitivity and ethical reasoning in ethics curricula. The UNESCO publication, Bioethics Core Curriculum Section 1: Syllabus Ethics Education Programme (2008), presents minimum standards for bioethics instruction of medical students. It is important to note that although the core curriculum was developed for medical students, the organization suggests that this curriculum is appropriate for students in nursing, health sciences, dentistry, public health, law, philosophy and social sciences. The

curricular framework was based on the principles included in the Universal Declaration on Bioethics and Human Rights and moves from the individual perspective that is the typical focus of medical and nursing school ethics curricula to a social and community view. The framework stresses virtues and values, such as dignity, respect, social responsibility, and justice, among others, and the development of ethical reasoning skills, with the aim of developing practical wisdom. The bioethics curriculum clearly demonstrates a balance between the need for ethical sensitivity and reflection and the need for reasoning skills to make prudent decisions when faced with ethical conflict. Of note, the curriculum addresses the different types of harm that patients may face and presents an expanded view of harm – "harm to interests, harm as unfairness, harm as disrespect" – to include social-economic harm (role harm, stigmatization), psychological harm, physical harm, and moral harm (24). This expansive definition of harm holds relevance to later discussions of quality of care and ethics education that are explored in Chapter Three.

Another source for guidance in ethics education of nurses and physicians is the ASBH (2006) Core Competencies for Health Care Ethics Consultation. The ASBH Core Competencies include ethical assessment skills, process skills and interpersonal skills. Although the Core Competencies have been developed for clinical ethics consultation, the recommendations are not exclusively the province of ethics consultants, but are also relevant to the developing physician and nurse. Graduate physicians and nurses must promptly recognize an ethical problem and initiate the appropriate steps toward resolution. The attributes, attitudes and behaviors that are essential for ethics consultants are also fitting for nursing and medical students. The recommendations stress that

attributes and attitudes can be taught and supported through reflection, modeling, accountability, and evaluation. Ethics educators in medical and nursing programs can find guidance in the Core Competencies for Health Care Ethics Consultation and strive to nurture the knowledge, skills and attributes outlined in the recommendations.

By and large, the competency standards and professional outcomes identified by the above educational and professional organizations demonstrate consistency in the requirements for ethics instruction. To meet these requirements, ethics education must meet the following goals: to promote the development of the virtues and values inherent to the professions of medicine and nursing, to affect attitude formation, to increase understanding of ethical theory, to promote the prompt identification of ethical problems, and to improve ethical analysis and decision making. To achieve this, the standards reveal that ethics education for medical and nursing students should in general include a foundation of knowledge of ethical principles, codes of conduct, and application of theory; development of skills, such as communication and critical reasoning skills; and attitude formation, as seen in the development and nurturing of professional virtues and behaviors. The above standards suggest that a balance is needed in ethics education that supports professional values and virtues as well as the development of ethical skills. An examination of the codes of ethics in medicine and nursing that follows will also provide direction and assist in identifying the goals for ethics teaching in the respective professional programs.

### ii. Review of the Codes of Ethics in Medicine and Nursing

In addition to the above review of curricular guidelines and professional standards, it is also instructive to review the codes of ethics of nursing and medicine for

their utility in this discussion. It is not in the purview of this paper to do a thorough analysis of the professional codes in medicine and nursing; however, an overview of the codes illustrates their applicability to ethics education. A code of ethics describes the healthcare professional's expected ethical values and behavior. It is not a legally binding document; but rather, a code of ethics provides the model of professionalism to which its members aspire. The Code of Medical Ethics of the AMA (2001), the Code of Ethics of the American Nursing Association (ANA) (2001) and the International Code of Ethics of the International Council of Nurses (ICN) (2006) each affirm the expected virtues, values, responsibilities, scope of practice and standards of conduct of the respective professional. The codes reflect the goals of ethical practice; thus it can be inferred that the codes provide guidance for an ethics curriculum.

Physicians in the United States are guided by the AMA Code of Medical Ethics (2001), which includes the Preamble and Principles of Medical Ethics. The nine Principles are accompanied by Opinions and Reports that offer clarification of professional responsibilities in particular circumstances, for instance, on issues of informed consent and futile care. The Preamble of the code makes clear that the purpose of the code is for the "benefit of the patient," which is the physician's first and primary responsibility. The principles delineate the virtues and responsibilities of the physician and recognize the physician's duty "to patients first and foremost, as well as to society, to other health care professionals, and to self" (Preamble). Virtues and responsibilities described in the Principles of Medical Ethics include respect, compassion, honesty, and professionalism. Responsibilities identified in the code include competently providing medical care, protecting confidentiality and privacy, reporting incompetent physicians,

and advocating for changes that would benefit patient care. Although the code of Medical ethics was last updated in 2001, the inclusion of the Opinions and Reports reflects the AMA's considered response to the changing cultural and societal issues.

Nurses in the United States are guided in their professional practice by the ANA Code of Ethics with Interpretive Statements (2001). The Code of Ethics consists of nine Provisions, each of which contains statements that elucidate and further develop the themes of the Provisions. The first Provision is primary and establishes the nurse's responsibility to practice "with compassion and respect for the inherent dignity, worth, and uniqueness of every individual" (11). The code has three areas of focus: values and responsibilities; boundaries of duty and loyalties; and duties that extend beyond the patient. Revisions of the code are thoughtful and reflective of the profession's changing and expanding roles and the relational aspects of nursing. Fowler notes that the code's periodic reviews and revisions have been effective in addressing changing societal goals and standards while successfully maintaining the underlying values of the profession as a constant (2008, xiii). These constant virtues and values of the profession include respect, compassion, integrity, right to self-determination, advocacy, respect for privacy, confidentiality, accountability, among others (ANA 2001). Nurses are also expected to create environments that nurture virtues and values.

The ICN (2006) is another important source of guidance for ethics education. Similar to the ANA Code of Ethics, the ICN code stresses the values and virtues of rights, dignity, respect, privacy, competency, accountability, confidentiality, and advocacy. The ICN code places a heavy emphasis on social action and advocacy related to health care. The code differs from the ANA code in that it provides recommendations for its

application in various settings, such as practice, management, education, research and professional organizations. Of particular interest are the suggestions for nursing curricular development which recommend teaching the concepts of human rights, justice and equity issues related to health care, ethical decision making, and specific topics, such as informed consent, privacy and confidentiality.

The codes of ethics in medicine and nursing reveal the expectations for professional values and behavior, which are also echoed in the curricular recommendations for ethics education in the professions. The respective codes emphasize professional virtues and values, ethical sensitivity, the process for ethical decision making and the development of appropriate knowledge areas, skills and attitudes for ethical competency. Accreditation requirements for educational competencies in medicine and nursing, professional organizations' recommendations, and the codes of ethics provide guidance for ethics education in the respective programs. It is important to discern, however, whether these requirements (as in the case of accrediting organization guidelines) or recommendations (as in the ASBH task force suggestions) are actually implemented in medical programs and baccalaureate nursing programs.

iii. A Comparison of the Requisite Standards and Current Practices in Ethics Education

There is agreement that ethics education is integral to the instruction of medical students and baccalaureate nursing students – most significantly, it is a required component of the curricula of each profession, and its inclusion is essential for accreditation. Given this weight, however, it is surprising to find a lack of agreement in the literature on the primary goals of ethics education. A review of the literature reveals

two perspectives in ethics education that have emerged. One position stresses the development of skills in ethical analysis, and the other stresses the development of the virtues of the health professional. It is necessary to consider the major arguments in the literature related to the goals of ethics education for they have direct bearing on ethics curricula and, consequently, on the instructional methods employed and thus are particularly relevant to this thesis.

Discussion on the goals of ethics education has been a persistent topic in ethics literature for many decades. In a foundational work on ethics education, Daniel Callahan (1980) outlined the goals of ethics education. A review of the accrediting guidelines and professional codes of ethics shows that the general educational goals that he proposed continue to guide ethics curricula in undergraduate and graduate education as well as in professional programs. Thirty years hence, the dialogue on ethics education goals in medicine and nursing continues, but the initial goals that Callahan delineated continue to be relevant for educators today. Callahan put forward the following five goals of ethics education:

- Stimulating the moral imagination
- Recognizing ethical issues
- Eliciting a sense of moral obligation
- Developing analytical skills
- Tolerating –and reducing–disagreement and ambiguity (64-69)

In describing the first goal to stimulate the "moral imagination," Callahan noted the need to stir students' feelings and imaginations, which would serve as a necessary counterpoint to analysis. "Imagination without analysis is blind; analysis without imagination is sterile," he explained (1980, 65). Callahan's statement may serve to bridge

the differing positions toward ethics education that are reflected in the literature today – that of developing professional virtues, in particular, ethical compassion and sensitivity, and that of developing analytical skills. Moral imagination is needed to sensitize the student to ethical problems and to help "sort out those elements in emotional responses that represent appraisal and judgment" (65). Callahan reports that, in general, moral education includes a goal to instill virtues and improve behavior; yet, the classroom setting does not allow for this. He claims that there is no way to know if changes in attitude or behavior occur or if the desired behavior is consistently applied. Callahan contends, however, that it is an appropriate aim for educators to improve the way in which students think and feel about ethical issues. He argues that ethics educators should teach students how to articulate ethical positions, analyze ethical issues, and critique their positions. Callahan summarizes this behavioral goal as helping "students to know the importance of changing his or her behavior if that was what a moral judgment seemed to entail" (70). He expands these goals in a discussion of applied ethics and adds the importance for professionals to grasp the meaning of their disciplines' codes, mores, values, and ethical problems.

Pellegrino and Thomasma (1993) offer a different view of the goals of medical ethics education, providing support for the development of virtues as well as the ethical analysis skills that are emphasized by Callahan. They argue that medical schools have a responsibility to society to ensure that physicians practice from a moral stance, due to the moral nature of the healing relationship that physicians have with patients. They also contend that virtues are teachable "by practice, by example, and even by the study of ethics" (1993, 176). Pellegrino and Thomasma specify, however, that the virtues taught in

ethics education should fit the goals of medicine and be appropriate to the "physician *qua* physician," not to physician as person. The authors point to the "subtle" influence of personal virtues in the professional role and vice versa, but they contend that the goal of teaching virtues in medical programs is for the formation of the good physician as physician, not as person (176). This argument will be further developed in Chapter Four.

An early review of medical ethics curricula by Miles et al. (1989) presents ethics education goals that exclude the development of virtues. The authors contend that medical students should be taught to "recognize the humanistic and ethical aspects of medical careers... [and] examine and affirm their personal and professional moral commitments" (1989, 706). Even so, the authors reject a medical program's responsibility to teach or nurture virtue. Rather, they submit that students enter programs equipped with the moral characteristics required of physicians, and the school's responsibility is limited to providing students with the knowledge and skills to manage ethical issues.

Eckles et al. (2005) conducted an extensive review of the literature on formal undergraduate medical ethics education encompassing the period of 1978-2004. The authors reviewed sixty articles for features of ethics education, including goals, teaching methods, descriptive studies, empirical studies and so forth. The review revealed a lack of consensus related to the goals of ethics education and uncovered two perspectives in ethics education– a goal to produce virtuous physicians and a goal to produce physicians who skillfully analyze and manage ethical problems. The authors found three articles submitting that the primary goal is to develop the virtuous physician and ten articles that support the development of skills in ethical analysis and decision making as the primary

goal of medical ethics education. The authors refer to this as the "virtue/skill dichotomy" (2005, 1145). Eckles et al. note that they concur with the position of the majority of the research articles on ethics education curricula that support a primary goal to develop ethical decision making skills.

In nursing, Fry (1989) examined four educational models used to teach ethics in undergraduate and graduate nursing programs. She identified the goal of ethics education in nursing "to produce a morally accountable practitioner who is skilled in ethical decision making" (1989, 485). Two of the models include instruction on the moral agency of the nurse, nursing virtues and the value dimensions of nursing – the Moral Concepts and Moral Issues models. The Moral Concepts model emphasizes nursing virtues, values, ethical reasoning and ethical analysis, aligning well with proposed baccalaureate nursing program standards. The Moral Issues model includes a little discussion of virtues and values, but overall, stresses ethical theory, ethical dilemmas, and skills in ethical decision making.

Although no recent comprehensive reviews of nursing ethics curricula have been identified, one key study of the nursing profession illustrates the general practices in ethics teaching. The Carnegie Foundation National Nursing Education Study identified the signature pedagogies, competencies and methodologies of nursing education. The study revealed that nursing programs are effective in forming "professional identity and ethical comportment" (Benner et al. 2010, 11). The study gives support for two primary goals for ethics education that are consistent with accreditation standards and the code of ethics for nurses – developing the virtuous practitioner and developing the skillful, analytical practitioner. While the Carnegie Foundation Study has shown that nursing

holds both views as important, current educational practices reveal a different emphasis depending on the learning environment. The authors observed that formal curricula emphasize ethical issues, while the informal curricula of the clinical setting focus on the development of professional identity and attributes (Benner et al. 2010, 11; Benner et al. 2008, 475). The authors recommend that ethics instruction should include more than a focus on bioethics, principles and professional standards. They strongly encourage the development of ethical comportment, advocacy, social ethics and "good practice" (Benner et al. 2008, 475). Interestingly, the study includes nursing students' interviews that describe "good practice" as presence, meeting the patient as a person, preserving dignity, advocacy, improving competency and responding to substandard practice (476).

Despite the educational standards of accrediting organizations and the behaviors and skills identified in nursing codes of ethics, ethics education literature is skewed toward the development of critical thinking skills, ethical reasoning, and ethical problem solving (Gillam 2009). Gillam notes that this trend is also observed in ethics curricula in the health professions in general. Curricula are heavily weighted with content on ethical theories, principles, fundamental concepts (autonomy, patient rights), confidentiality, informed consent, problems in end-of-life care, and a laundry-list of special ethical topics related to reproduction, genetics, transplantation, euthanasia, research ethics and so on (Gillam 2009).

More recently, though, nursing literature reveals a renewed call for a virtue-ethics approach to ethics education. This is attributed in part to the growing recognition of an ethic of care in nursing based on the ethical relationship of the nurse and patient, which undergirds nursing practice (Gillam 2009; Johnstone 2009). Assessing ethics education,

Gastmans claims that what is needed is "an *attitude-* versus an *action-*focused ethics education" (2002, 500, emphasis added). He argues that actions are the result of a virtuous stance; that is, virtue is necessary to "do the right thing" (501). Woods (2005) summarized research on nurses and reports that although many are able to identify ethical problems, they are passive in ethical situations because they lack moral confidence and moral courage, are uncertain, and may experience moral distress. It is probable that preparing nursing students to be oriented toward virtuous behavior might strengthen the moral courage needed to face ethical challenges.

Internationally, there is also a trend to emphasize virtue over skill. The Global Ethics Observatory (GEObs), an initiative of UNESCO, provides a database of ethics programs and curricular materials from educational programs around the world (ten Have 2008, 58). Standardized information on ethics programs from member states is gathered and entered into the database, thus providing information on goals of ethics education in programs across the world. Ten Have, past director of UNESCO, observes that the database reveals that Central and Eastern European countries tend to emphasize virtues and values more than analytical skills (pers. comm.). This is in contrast to the review of formal medical and nursing ethics curricula in the United States that tend to put emphasis on ethical principles and analytical skills.

Additionally, it should be noted that there are fewer explicit means in place in both medical and baccalaureate nursing curricula for education on the development of professional virtues. Several explanations are offered to address this. Faculty in schools of medicine and nursing are in disagreement about the relevance and effectiveness of measures to nurture professional virtues and values (Eckles et al. 2005). Some ethics

educators believe that the virtues of students are set and cannot be further developed (Miles et al. 1989). Others observe that virtues and values have been difficult to control and to evaluate (Sellman 2007). Yet, the value of virtue-based instruction and its significance for medical and nursing students are sustained through the standards and guidelines of medical and nursing programs. Moreover, the professional organizations specifically address the virtues of the nurse and physician. The contribution of virtues to the ethics education of nurses and physicians will be argued in Chapter Three. It is important, however, to note at this point that ethics education can be effective in the development of virtues, for example, in developing prudential reasoning (Begley 2006), ethical sensitivity and a caring attitude (Callister et al. 2009; Vanlaere and Gastmans 2007).

#### iv. Short-term and Long-term Goals of Ethics Education

The assessment of ethics education in medicine and nursing, as outlined in program standards, codes of ethics, and as reflected in customary educational practices, yields short-term and long-term goals. Short term and/or intermediate goals aim to introduce and nurture professional virtues, values, attributes and behaviors in order to produce a beginning nurse or physician who is compassionate, reflective, honest, and more sensitive to ethical problems. Short-term goals include producing a practitioner who respects patient values, practices with integrity and is competent in communication and interaction skills. Furthermore, short-term goals are orientated towards creating a healthcare professional who is better able to apply ethical principles, identify and analyze problems, employ an ethical process and framework and assist patients and others in the decision making process. While medical students and baccalaureate nursing students are

generally exposed to wider community and societal issues and able to recognize ethical issues in social justice and distribution of resources (UNESCO 2008), program outcome goals do not require competency in these areas.

Long-term goals, on the other hand, focus on improving patient care (ASBH 2006; Pellegrino, Siegler, and Singer 1990; UNESCO 2008). As the professional codes of ethics for both nurses and physicians also indicate, the aim of developing excellence as a healthcare professional is ultimately for the benefit of the patient. Individually, medical and nursing students who practice with professional virtues, respect values and demonstrate sensitivity to ethical concerns have the capacity to produce a positive effect on the practitioner-patient relationship. In addition, individual students and practitioners who are better able to identify and analyze the ethical problems that arise may produce positive outcomes for patients, protect the dignity of the patient and prevent ethical harm (Pellegrino, Siegler, and Singer 1990). It is reasonable to contend that ethically practicing nurses and physicians also have the capacity to contribute to other long-term goals of ethics education. Ethics education can produce long-term goals aimed at improving work environments, enhancing organizational ethics and promoting the integrity of the medical and nursing professions. Ethically sensitive and skilled physicians and nurses can act as agents of change, creating an environment that supports the ethical behavior of others (AACN 2008), enhancing the image of the profession and the public's trust (UNESCO 2008), improving the ethics quality of the healthcare organization, enhancing the perception of the institution, and maintaining competency in the profession by reporting deficient practitioners (AMA 2001). In theory, ethically skilled practitioners also have the

capacity to reduce litigation by preventing ethical harm, thus improving their welfare and that of the institution.

Medical and nursing curricula focus chiefly on individual patients; likewise, the short-term goals of ethics education within these programs tend to focus on improving the learner or practitioner in his or her relationship with individual patients and their families. The reach of these goals predominately affects issues that arise in the "microethical" level (Parker 1995, 308) of the nurse-patient or physician-patient relationship rather than on issues larger in scope, such as social justice issues. These individualistic goals are critical, however, in advancing the overarching and ultimate goal of ethics education – to improve the quality of care for patients (ASBH 2006; Pellegrino, Siegler, and Singer 1990; UNESCO 2008).

#### v. Quality of Care

The review of accreditation standards and curricula in ethics education reveals largely practitioner-focused goals. This dissertation argues that while goals to improve the individual practitioner are essential and indispensable, the fundamental goal of ethics education remains to improve the quality of care for patients. The curricular goals to produce virtuous healthcare practitioners and to produce competency in ethical analysis are critical in advancing this ultimate goal. Furthermore, while skills in analysis are very important in achieving quality care, the attributes of the individual practitioner who applies the skills are ultimately more influential in obtaining better outcomes (Pellegrino, Siegler, and Singer 1990). Likewise, achievement of the long-term goals that are

identified above is dependent upon the attributes of the individual practitioner. This position will be fully developed in Chapter Four.

Quality of care can be viewed and evaluated from several perspectives. In brief, three perspectives are presented in the literature – the patient, professional, and policy perspectives. Recent quality measures related to the professional perspective include patient-centered care and shared decision making, which promote patient autonomy and are assumed to lead to better patient care (Grol 2001). It would seem that reaching success with practices that place more autonomy on patients requires healthcare professionals who have carefully come to know their patients, are open to dialogue, compassionate and empathetic. This requires development of a virtuous healthcare professional. The research of Grol is in medicine, but it is applicable to nursing as well as to other healthcare professions. The professional perspective of quality of care assesses appropriately applying standards or guidelines to individual patients, which involves knowledge and skill, but also requires sensitivity to the patient's wants and needs, an open and trusting relationship, effective interpersonal skills, and prudential reasoning to effectively weigh the benefits and burdens of standard of care guidelines in individual situations. The professional perspective is also responsible for the development of guidelines and standards, a process that is potentially influenced by values and biases (Shaneyfelt 2001). To successfully negotiate appropriate care, the physician and nurse require more than cognitive and behavioral skills – they must possess professional virtues and attributes.

Perhaps even more relevant to this argument is the patient perspective of quality of care. The patient perspective is generally defined, assessed and measured as patient

satisfaction, although it may be more specifically measured by fulfillment of expectations (or needs) and experiences (Sixma et al. 1998). Improvement in quality of care from the patient perspective necessitates a prudent and ethically sensitive healthcare professional who can successfully fulfill the patient's expectations and provide a satisfactory healthcare experience. Successfully meeting this quality standard requires a healthcare professional who is caring, sensitive and able to develop a trusting relationship in order to identify and respond to the patient's expectations or needs. The successful professional must be prudent in order to negotiate care with the patient, respecting and responding to the patient as person. Improvements in quality of care from the patient perspective, therefore, require prudence, sensitivity, compassion and respect, among other attributes. These issues are the province of virtue-based ethics. It is here that ethics education can make a significant contribution to quality of care. This argument will be further developed in Chapter Three.

## B. The Methods Applied in Ethics Education

## i. Relationship of Goals to Methods

Designing ethics education activities requires an understanding of expected outcomes as well as understanding the nature or category of goals in order to select the most effective pedagogical methods. The above review of educational and professional standards, codes of ethics, and curricular practices reveals three categories of goals or domains of learning that are required for ethical practice – cognitive, behavioral and affective. Cognitive goals include knowledge of ethical theory, principles, codes, moral reasoning skills, ethical decision making frameworks and analysis skills, values, virtues

and responsibilities, and other ethical issues or content common to medical ethics (see Miles et al. 1989 for specific content) and nursing ethics curricula (e.g., informed consent, confidentiality, withholding and withdrawing treatment) (AACN 2008). Behavioral goals include interactive or communication skills important in developing relationships with patients, families and other healthcare professionals and also process skills in ethical analysis, ethical decision making and management of ethical conflicts. Behavioral skills allow practitioners to "put their knowledge to work in everyday clinical encounters" (Pellegrino, Siegler, and Singer 1990, 177) such as explaining end-of-life considerations, advocating on behalf of the patient or breaking bad news. Affective goals represent attitudes and attributes (i.e., professional virtues, character traits) such as compassion, caring, interest, sensitivity, honesty, integrity, patience, altruism and prudence. Professional attitudes and attributes promote trust, increase sensitivity to ethical concerns, aid in preventing moral harm, and create ethical working environments.

Cognitive goals have received more emphasis in medical and nursing curricula, namely because they form the foundation from which skills arise and are easiest to teach and evaluate. Behavioral goals have received growing attention in curricula, with many programs including instruction on interactive and communication skills in the early years of their programs. Attitude formation in the development of virtues and values of the professional, while mentioned in all accreditation standards and a significant component of the codes of ethics in medicine and nursing, has received less attention in ethics curriculum programming as noted in the previous section (Eckles et al. 2005; Miles et al. 1989; Sellman 2007). Yet, as Pellegrino, Siegler and Singer observe, "ethics requires that the physician be a person of character, one who can be expected habitually to act in the

patient's interests when no one is watching" (1990, 177). This entails that adequate educational effort be directed to affective outcomes in order to produce virtuous physicians and nurses. This has been argued equally in nursing (Fry 1989) and will be further developed in Chapter Four on virtue-based ethics.

Review of the literature shows much discussion on the most effective means to teach ethics in medical and nursing programs. Students require both knowledge of ethical theory, values, professional virtues, decision making frameworks, and so forth, and also experience in developing professional virtues, interactive skills and management of ethical problems to promote competency in ethics. Educational strategies to attain the different types of goals require a variety of instructional methods (Campbell, Chin, and Voo 2007; Miles et al. 1989). In medical schools and baccalaureate nursing programs, formal teaching approaches range from traditional to nontraditional methods and didactic to interactive measures. Each teaching method has associated advantages and disadvantages and is better suited for the attainment of particular categories of goals. It follows then that it is important for the ethics educator to select an approach that is best suited to the desired outcome.

## ii. Traditional Methods

Traditional teaching methods primarily include lecture, small group discussion and case study analysis or a combination of these approaches. As Goldie (2000) notes, early ethics courses in medicine were taught by moral philosophers and theologians who focused on cognitive goals; therefore, lecture was the predominant pedagogical approach. Fox, Arnold and Brody observe that traditional approaches in ethics teaching tend to

stress the "process of moral deliberation more than its conclusions" (1995, 762). Lectures, however, remain an important instructional method (Fox, Arnold, and Brody 1995; Rosoff 2011; Woodring and Woodring 2011). Many medical programs have developed ethics curricula that provide formal instruction through lecture presentations in the first two years of the program, prior to clinical experiences, which sets a foundation of knowledge necessary for the development of behavioral and affective outcomes (Lehmann et al. 2004). Lectures can be an appropriate means to accommodate teaching larger groups of students and are efficient to communicate foundational concepts, such as ethical theory, ethical principles, communication theory and so forth. Didactic methods such as lectures or presentations may provide some opportunity for dialogue and reflection, yet these methods tend to be fact-based and driven by the teacher. They are useful in conveying complex content, however, such as ethical theory, and create a base for future experiences. Many medical schools employ lectures, as is evident in reviews of medical ethics curricula in the United States (DuBois and Burkemper 2002), the United States and United Kingdom (Lehmann et al. 2004) and in the United Kingdom (Mattick and Bligh 2006). Lehmann et al. (2004) surveyed the ethics curricula of the medical programs in the United States and Canada at the turn of the twenty-first century and found a variety of pedagogical methods for ethics instruction; however, the authors observed that the majority of instruction was in lecture and/or small group discussion. They reported that case discussion was used most often within ethics courses, with less use of reviews of empirical articles, moral philosophy and literature/humanities. Dubois and Burkemper's survey of medical schools in the United States revealed the frequency of inclusion of specific teaching methods (via rank-ordering) and also found a

predominant use of traditional methods with discussions/debates (84.5%), readings (82.8%), writing exercises (63.8%) and lectures (63.8%) used most often (2002, 434).

Many schools have adopted a model in which content is conveyed through large group presentations that are followed by small group discussions (Fox, Arnold, and Brody 1995; Lakhan et al. 2009; Lehmann et al. 2004). Small group discussions are used regularly to allow dialogue and reflection on the topics. Students appreciate discussion sessions as a secure space to "examine their own moral viewpoints and emotions in supportive environments" (Charon and Fox 1995, 767). Small group discussions are also integral to Problem-Based Learning (PBL) strategies, which have been instituted in many medical schools and other health-related programs (Cordingley et al. 2007; Parker 1995; Solomon 2011). Some schools employ PBL to teach medical ethics, finding that "meeting a variety of views in a friendly atmosphere will stimulate critical scrutiny of students' settled ideas" (Parker 1995, 308). Mattick and Bligh's (2006) report on medical ethics education in the United Kingdom revealed that educators preferred teaching small groups but felt that lectures were a sufficient method for particular topics.

Case study analysis has emerged as an important teaching method for medical and nursing students and is relied on heavily in ethics teaching. Lehmann et al. (2004) report that this method is used most often in medical ethics courses. Case analysis, like lecture and/or small group discussion, is also useful for reaching cognitive goals in ethics teaching and bringing out the complexity of moral dimensions in health care. Case discussion has the advantage of helping students understand the practitioner's moral agency and apply learned theory (Miles et al. 1989). Charon and Fox report on students' suggestions for ethics teaching and recommend exploration of the "intricacy, ambiguity,

and pluralistic differences" that present in ethics (1995, 767). Case analysis is useful here. Cases may be of sensational quality or everyday ethical problems, hypothetical or actual, teacher or student-generated (Southgate et al. 1987) but should preferably be relevant to students' current experiences (Charon and Fox 1995). Case study discussion is useful in advancing both cognitive and behavioral skills needed for critical reasoning and analyzing ethical problems. Fry (1989) recommends case study presentations with small groups of nursing students, finding this beneficial in helping students recognize the values of the patient and nurse, analyze moral problems and identify the nurse's role in ethical decision making. Moreover, Campbell, Chin and Voo (2007) observe that case study analysis can also assist students to reflect and develop empathy and sensitivity to ethical issues. The authors recommend case analysis as a means to teach the affective domain and also to assess the student's moral development.

#### iii. Nontraditional methods

Nontraditional methods have been introduced over the years, and along with group discussions and case analysis, they are favored because many of these approaches engage students in active learning, an approach favorable to adult learners (Woodring and Woodring 2011). Nontraditional methods may incorporate the use of literature, poetry and media, such as film and art to "stir the moral imagination," as Callahan recommends (1980, 65) and provide opportunity for reflection and discussion. Commercial films, such as *Wit* and *Miss Ever's Boys*, documentaries such as *The Suicide Tourist* (PBS Frontline), and interactive multimedia, for example, *A Right to Die? The Dax Cowart Case* (Anderson, Cavalier, and Covey 1996), expose students to challenging ethical situations

and raise theoretical and affective perspectives. Films, accompanied by discussion, may be useful for developing empathy and sensitizing students to challenging ethical problems (Fox, Arnold, and Brody 1995). Many documentaries also provide instructor guides to assist in developing group dialogue and critical reasoning skills.

Other nontraditional methods have advantages in developing knowledge, skills and attitudes. Student portfolios, case studies with reflection, development of case scenarios (Campbell, Chin, and Voo 2007) and reflective writing (Callister et al. 2009) have been used to nurture students' ethics competencies. One medical school reports enhancing ethics education with the integration of scholarly projects and elective coursework in ethics (Kanter, Wimmers, and Levine 2007). Games, panel discussions, debates, and quiz shows formats have-been used (Fox, Arnold, and Brody 1995) and can help to develop critical reasoning skills. Nontraditional teaching methods also include interactive computer programs (Fleetwood et al. 2000) and online instruction (Stoddard and Schonfeld 2011). Other schools incorporate ethics clinical conferences, post-clinical debriefing sessions and ethics grand rounds (Lakhan et al. 2009; Pellegrino, Siegler, and Singer 1990), but it should be noted that these are often not part of the formal curriculum.

Teaching for the behavioral and affective domains of learning particularly benefits from active, experiential methods. Presently, ethics literature supports teachinglearning activities that are better suited to these goals, such as role play (Fox, Arnold and Brody 1995; Garrett 2010; Lakhan et al. 2009), the use of standardized patients or actors (Litzelman and Cottingham 2007) and interviews. One program demonstrates an unusual way to provide students with practical experience by placing "unannounced" standardized patients in outpatient clinics to provide practice in managing ethically

challenging situations and to assess a student's performance as well (Litzelman and Cottingham 2007, 416). In addition to providing active learning, these methods present important practice opportunities for students to experience the emotions and feelings that accompany ethical situations as well as to develop relevant skills. Students are unaware that the interaction is an educational exercise, however, and one can rightly question the integrity of this approach. Nonetheless, in order for a medical student to compassionately and effectively "break bad news" or for a nursing student to successfully advocate for a vulnerable patient, students must develop a compassionate and sensitive orientation and possess an understanding of their role and responsibilities and the process and skills involved. Students also need a degree of proficiency with critical reasoning and communication skills to perform effectively. Practice opportunities with standardized patients or patient actors present students with needed opportunities to develop ethics competencies in safe, supported environments, avoiding risk for actual patients.

Many of the above mentioned active learning methods have been available for the past decades but have been used less frequently than group discussions and case study analysis. Dubois and Burkemper's (2002) survey reveals the following in a rank ordering of nontraditional teaching methods: multi-media presentations (29.3%), role playing/standardized patients (20.7%), clinical round/field visits (19.0%) and computer exercises (10.3%) (434).

The most recent method introduced to teach healthcare ethics is simulation. In brief, simulation involves the creation of a reality-based setting in which students engage with actors, standardized patients or high- or low fidelity mannequins to work through a teaching exercise, as a practice, assessment or evaluation session. Chapter Five focuses solely on simulation as a teaching method and explains it in more detail. It is important at this point, though, to clarify the differences between role play, mentioned above, and simulation. Some commentators use role play and simulation interchangeably, or see role play as falling on a continuum of simulated exercises, but there are significant and meaningful differences. Jones (1995) describes the subtle distinctions between these methods. In a role play activity, he submits that the participants take on their assigned roles for which, in many cases, personalities and emotions have been provided (for example, an angry patient or a bullying physician). Participants in role play have authorship of the role; that is, they can create and expand upon the role, in effect acting the part (1995, 10). Participants work to perform the role convincingly. In a simulation, there is a concerted effort to develop an environment that simulates reality, but the participants do not simulate their personalities. Rather, they take on functional roles and strive to carry out their professional roles most effectively. While many individuals would distinguish simulation from role play by the simulated environment, Jones notes that "the thoughts and attitudes, and the emotions and behaviour, are the evidence for distinguishing between simulations and other interactive techniques" (12). Participants are inwardly focused as they work to develop the competencies expected in the professional role. "Play acting" is discouraged in simulation, and the results of the participants' actions are considered real (135). In a simulation, the participants are provided extensive background information about the event or activity, perhaps a patient's medical history or physical exam results and an opportunity to incorporate this into their work in the simulation. The student is immersed in a professional "functional" role during the simulation and awareness of learning may not occur until the debriefing

session that follows. The learning that occurs with simulation often happens in the debriefing session or at a later point, after time for reflection.

Simulation is employed frequently in medical and nursing programs to teach cognitive and psychomotor skills, but it has been less frequently used to teach ethics. Some medical programs have incorporated simulation using standardized patients to teach ethics concepts (Fleetwood, Novack, and Templeton 2002). In these simulations, the standardized patient has learned the expected role, does not veer from his or her assigned role and displays the same functional role in repeated situations. The environment is simulated (an examination room, for instance). Simulation-based activities have also been used to evaluate the ethical behavior of medical students, using the Objective Structured Clinical Exam (Singer et al. 1996). Yet, a review of the literature shows that there are only a small number of articles that illustrate its use in teaching ethics to medical and nursing students (Arnold et al. 1988; Edinger et al. 1991; Fleetwood, Novack, and Templeton 2002; Gisondi et al. 2004; Gordon and Tolle 1991; Gropelli 2010; Perlman 2008; Vanlaere, Coucke, and Gastmans 2010). The few studies that have been published focus on using simulation to teach ethical analysis, clinical skills (communication and interpersonal skills), the role and function of ethics committees, and less commonly, to teach professional virtues. This methodology strongly supports practice and experiential learning and will be argued in more detail in Chapters Five and Six.

# iv. Other Pedagogical Considerations Related to Goals and Methods

The literature on the process and methods of ethics education reveals much discussion on the comprehensiveness of ethics education, including the relevant content

areas to include, the amount of time to dedicate to ethics education, whether it should be integrated into current courses or taught as a stand-alone course and who is best prepared to teach ethics content. In medicine, there is also significant discussion on the value of a standardized or uniform curriculum in ethics (Doukas, McCullough, and Wear 2010; Lakhan et al. 2009; UNESCO 2008).

These issues, while raising important process and methods concerns, are not expressly relevant to the thesis of this dissertation, and therefore, will only be briefly mentioned. It is important to the success of an ethics program that there is adequate time dedicated to ethics in the curriculum to allow for the development of ethics competencies. Another important consideration is having sufficient resources, such as adequate numbers of educators who are adequately prepared to teach ethics and sufficient financial support for labor-intensive educational approaches (Campbell, Chin, and Voo 2007). Ethics may be taught as a stand-alone course or integrated vertically and/or horizontally, and each approach has its advantages and disadvantages that are argued in the literature. Many ethics educators in medicine are working for the advancement of a uniform curriculum, which could assist in ensuring that all proposed goals of ethics education are more readily met. Lakhan et al. (2009) argue strongly for a standardized medical ethics curriculum and assert that it can benefit students and patients by increasing patients' trust in physicians, a measure of improved quality of care. Nevertheless, while these are valuable areas for research, the accrediting organizations have expressly declared the value of teaching professional virtues and attributes as well as ethical theory and skills and indicate that these should be taught explicitly as part of the curriculum. The guidelines and standards for ethics education in medicine and nursing provide clear directives that if effectively

instituted will help to ensure the development of ethically-sensitive, reflective and prudent practitioners.

C. Evidence of the Effectiveness of Teaching-Learning Methods in Ethics Education

It is important to review the teaching methods used in ethics education for their effectiveness in achieving the educational goals. Reviews of medical programs find no single favored method, and teaching methods vary according to the students' level in the program, moving from classroom to clinically-focused teaching (Dubois and Burkemper 2002; Eckles et al. 2005; Fox, Arnold, and Brody 1995; Lehman et al. 2004; Mattick and Bligh 2006). Many ethics teachers employ a variety of methods, pulling from the traditional and nontraditional categories, which is educationally appropriate since ethics curricula have many different objectives. Lectures are appropriate for factually-based theory, case studies and small group discussions are suitable to objectives that focus on analysis and developing sensitivity, and role-play and simulation have value for development of professional attributes, clinical reasoning, and interactive skills.

A summary of articles that review the effectiveness of specific teaching methods is included below. Many of the studies employ surveys that assess students' satisfaction with a teaching strategy or students' perceived learning (Kyle 2008). In theory, teaching methods can be evaluated for their effectiveness in meeting students' learning needs or their effectiveness in improving knowledge, skills (in moral reasoning, analysis, communication, for example) or attitudes and attributes (nurturing professional virtues and values), but few studies have attempted to measure the impact of ethics education on the student's behavior. Those that do are included below and reflect changes in moral

reasoning (Self, Wolinsky, and Baldwin 1989), ethical awareness (Goldie 2001), and improvement in ethical analysis (Callister et al. 2009; Smith et al. 2004), for instance. Improvements in moral reasoning have been identified following ethics education; however, the results measure skill sets in hypothetical situations, typically as students respond to case studies (Smith et al. 2004). Empirical studies can also be performed to measure the effectiveness of a particular teaching method in meeting students' learning needs or to gauge the effectiveness of the ethics curriculum overall in producing favorable outcomes (Sulmasy and Marx 1997, for example). There is a lack of consensus in the literature about which outcomes should be measured (Fox, Arnold, and Brody 1995). Theoretically, teaching methods could also be evaluated for their effectiveness in helping to achieve the long-term goals of the curriculum, for example, improving patient outcomes, but this task has proven difficult for ethics educators. No empirical studies were found that revealed the effect of ethics education on patient outcomes (Siegler 2001), although there is anecdotal evidence to suggest this potential (Gropelli 2009; Vanlaere, Coucke, and Gastmans 2010).

It is essential, though, for evaluation methods in ethics education to align well with the goals or objectives of teaching. For instance, to meet a goal to improve the understanding of ethical theories, a written test is an appropriate measure. To attain a goal to develop ethical analysis or decision making, many teachers would consider having students evaluate a case study and employ an ethical framework for analysis in writing or verbally discuss their approaches to the case. A more suitable evaluation method, however, might be a simulation-based exercise in which a student works through a simulated experience and is evaluated on demonstrated behaviors and skills in managing

the ethical situation. Additionally, to achieve a goal to develop professional attributes and virtues, simulation may be employed to allow faculty to observe and evaluate the student's performance in a simulated setting.

As noted previously, these evaluation methods measure students' learning at a particular moment, often in hypothetical situations, and are limited in their ability to predict future behavior of students. The scarcity of evidence for many of the approaches used in ethics education is an area that is ripe for study. Following are examples of some of the evidence on effectiveness of commonly used teaching methods in ethics education. Chapters Five and Six will examine the evidence that is available on simulation as a teaching method in health care and in health care ethics education.

Evidence of the effectiveness of small group discussions and use of case study analysis has been favorable and may account for the extensive use of these methods. In an influential study by Self, Wolinsky and Baldwin (1989), researchers sought to demonstrate the effectiveness of teaching medical ethics and the relative benefits of two methods, small-group case-study method and lecture method. A control group was used, which received no ethics instruction. The value of ethics education was assessed by its contribution to the moral reasoning of medical students as measured by the Sociomoral Reflection Measure (SRM) test. The study found that ethics instruction contributed to a statistically significant improvement of moral reasoning ( $p \le .0001$ ) in both the smallgroup case-study method and the lecture method groups when compared to the control group. Furthermore, when the two instructional methods were compared, students in the small-group case-study method had a statistically significant increase in scores than did

students in the lecture group (p  $\leq$  .03), demonstrating the advantages of the small-group case-study approach (757).

In a later study, Goldie et al (2001) compared the effectiveness of small group case discussion sessions to large group lecture format to achieve medical ethics education goals to increase first year medical students' awareness of ethical issues and to nurture self-awareness. The small groups also participated in seminars and sessions with experts in ethics and legal issues. The study used the Ethics and Health Care Survey Instrument (an adapted version) to measure the students' agreement with consensus views on ethical issues and observed that students in the small group discussion sessions had stronger agreement with consensus groups than the lecture format group. The authors concluded that the smaller group sessions promoted "normative identification with the profession of medicine" (301).

In a study of third-year medical students, Smith et al. (2004) compared the effectiveness of written case analysis alone to written case analysis with a group discussion and observed an increase in the ability of students to identify and assess ethical problems regardless of the teaching method used. Students who participated in the group discussion, however, experienced improvement in a final case analysis and also reported a greater increase in satisfaction with the ethics education sessions. The authors note the importance of demonstrating improvement in outcomes and positive feedback in light of the increased time and effort associated with facilitated group discussion sessions.

Southgate et al. (1987) illustrated the benefits of using an interactive, studentcentered approach to ethics instruction. Rather than analyze cases that were assigned by the faculty, medical students developed their own ethics cases for presentation in ethics session, using a provided framework for the development and analysis of the cases. The authors note that the cases reflected the students' experiences and interests and generally included common, recurring issues.

Teaching allied health students, Stoddard and Schonfeld (2011) compared online ethics instruction to traditional classroom teaching. The authors provided identical content and resources for students in an online course and a traditional face-to-face course. The courses only differed in that students in the online course posted responses to an asynchronous online discussion board to earn points, while students in the traditional classroom earned points through live discussions. The study showed no difference in the participation between the two groups. Although the study authors expected to find lower outcomes on writing assignments in the online student group, the results showed that online students performed better on the assignments. The authors attributed the finding, which they did not expect, to demographic characteristics of the students. The online students tended to be older, nontraditional students and perhaps more self-directed. Ethics educators interested in evaluating online ethics instruction for medical or nursing students might find it interesting to repeat this study.

In a qualitative descriptive study on reflective journaling, Callister et al. (2009) found this teaching method to be effective in capturing ethical reasoning in nursing students. Students were asked to identify, describe, analyze and reflect on ethical problems that they encountered in their clinical experiences. Students received

foundational ethical knowledge to aid in the assignment. The authors assessed the reflective journals for the students' level of reflection and analyzed the themes. The results showed a high level of commitment to nursing and a high level of critical thinking ability. Callister et al. found that teaching ethics by reflective journaling was an effective teaching method to meet goals of increasing ethical sensitivity, critical thinking and ethical reasoning.

A qualitative study by Kyle (2008) evaluated the effectiveness of anonymized reflection as a method to teach ethics to nursing students. This teaching method consisted of anonymously written summaries of actual clinical experiences, which students share within a small group. Each group participant randomly selected a case summary and anonymously reviewed it, employing an ethical framework in the process. The group then discussed the anonymous reflections and reached a consensus decision, mimicking actual clinical practice. The qualitative study measured the students' perceptions of the teaching method on achieving learning outcomes. All students rated the learning experience as relevant and felt that their knowledge had increased, but the anonymized reflections were not highly rated, leading the researchers to conclude that the students felt that learning took place in the discussion segments. The author acknowledged the study's limitation because of the subjective nature of evidence, but points out that the experience was an "exciting and relevant way of teaching and learning ethics" (2008, 14).

Peer-tutored problem-based learning was evaluated for its effectiveness in comparison to conventional teaching ethics to senior nursing students (Lin et al. 2010). Although the problem-based learning (PBL) method has been used in medicine for many years, it is fairly new to nursing education. It involves teaching from real problems,

includes group discussion and differs from conventional teaching in that it is studentgenerated. This study employed peer tutors as group discussion leaders rather than expert tutors. There was a statistically significant difference between the ethical discrimination scores of the PBL and conventional method groups, with higher scores in the PBL group. Both groups showed statistically significant improvements in the ethical discrimination scores from their pre-intervention scores; however, this study showed the PBL group to be a more effective teaching method (379-380). The PBL group also scored significantly higher than the conventional group in self-motivated learning and critical thinking.

Some empirical studies have focused on the overall curriculum rather than on particular teaching methods. For instance, an early study by Sulmasy and Marx (1997) conducted in 1992-1994 with medical house officers (residents) evaluated the long-term effects of a two-year ethics program that included monthly ethics reports and scheduled didactic conferences. The authors provided an integrated curriculum with opportunities for house officers to analyze ethics cases that were of concern to them. The study assessed for improvement in knowledge, confidence and attitude following the two-year curriculum. The study showed a statistically significant increase in knowledge scores, which improved 14% on average and an increase in confidence (89). Confidence levels increased in the ability to recognize ethical problems, form and justify ethical decisions, obtain informed consent and so on. Interestingly, the study showed that knowledge and confidence tended to go up together (91). With one exception, attitudes did not change significantly; the proportion of residents who felt that ethics should be required during residency increased, however, from 57% to 80% (90).

## D. Conclusion

In summary of this chapter, the review of relevant sources reveals that while there is no consensus on the goals of ethics education, there is agreement on the essential importance of ethics education for the developing nurse or physician. The review shows that there are two goals of ethics education that have emerged- that of developing the professional virtues and that of developing ethical analytical skills. Short-term goals tend to focus on developing the ethics competencies of the individual student in relationship to the patient. Long-term goals focus more broadly on communities and larger social issues. The overarching goal of ethics education that drives these important objectives is to improve the quality of care for patients. The review of current methods for instruction in ethics reveals while there is no teaching-learning methodology that emerges as exclusively the best method, it is important to chose methods that best meet specific goals. There is growing support for incorporation of nontraditional methods to teach ethics and methods that actively engage the learner in critical thinking and reflection. Simulation's benefits as an engaging, active and reflective practice will be demonstrated in Chapters Five and Six, showing it to be advantageous as a method for ethics instruction.

# Chapter Three – Improving Quality of Care as a Fundamental Goal of Ethics Education

A. Quality of Care Defined

As Chapter Two has established, the ultimate goal of ethics education is to improve the quality of care for the patient. Just as the goal of medical and nursing education, in general, is directed to improving patient care, the goal of ethics education is for the betterment of the patient. To reach this goal, there are short-term or interim goals in ethics education that must be met, which focus on the knowledge, skills and behaviors that are essential to ethical practice. For instance, physicians and nurses must develop and exhibit professional virtues, attain knowledge essential for ethical analysis, and develop skills in ethical decision making in order to best meet the patient's needs and to avoid engaging in ethical harm. The long-term goal of ethics teaching to enhance the quality of patient care is aided by the attainment of these short-term or intermediate goals. In addition, nurses and physicians have an ethical responsibility to improve the quality of care for patients, which is made clear in professional codes and accrediting organizations' standards as presented in the previous chapter. Nurses and physicians are obligated to provide benevolent, ethically-sensitive, competent care that appropriately balances risks and benefits to promote optimal health for patients.

The relationship of ethics and quality of care is reflected in the definition of quality of care. Quality of care can be defined by the nature of care and the characteristics of care delivery. It can also be described from the different perspectives of stakeholders. In addressing the first sense of quality of care, Brook, McGlynn and Shekelle identify

two central components. The first element is "providing care of high technical quality ....[such that] the patient receives only the procedures, test, or services for which the desired health outcomes exceed the health risks by a sufficiently wide margin; and that each of these procedures or services is performed in a technically excellent manner" (2000, 282). This component of quality care addresses the nature of the care received and stresses that quality care consists of more than highly competent care; it requires the prudent weighing of risks and benefits. The inclusion of the phrase "the patient receives only the procedures or services..." in the definition also recognizes the importance of resource allocation and justice issues in the provision of care.

The second element of quality of care acknowledges, "that all patients wish to be treated in a humane and culturally appropriate manner and be invited to participate fully in deciding about their therapy" (Brook, McGlynn, and Shekelle 2000, 282). This component emphasizes the characteristics of the provider, calling for respect, cultural competency, and patient-centered care that engages the patient in decision making. It underscores the relational aspect of health care and the significant role that professional virtues and values play in the delivery of quality of care. Patient-centered care is an important aim of ethics in that it respects the patient as a person, "placing the patient at the center of the provision of care" (Grol 2001, 2582). It necessitates empowering the patient and respecting the patient's autonomy, a fundamental principle of ethics.

Quality of care has also been defined from the different perspectives of the stakeholders who interface within medical settings. In a broad sense, quality of care can be defined from the perspectives of the patient, the healthcare professional and the policymaker (Grol and Wensing 1995). The perspectives of additional stakeholders such

as payer organizations, and aggregates such as communities, organizations or specific patient populations may also be included in quality of care evaluations. For the purposes of this dissertation, only the perspectives of the patient and healthcare professional will be reviewed. Undoubtedly, the contributions of individual experts, groups and organizations in medicine and nursing are essential in establishing policies that directly influence patient care outcomes, for example, setting state-mandated nurse-patient staffing ratios in California (Serratt et al. 2011) and advising and collaborating in health care reform (AMA 2012; ANA 2012). The emphasis of this dissertation, however, is on the ethics education of medical and nursing students as novice practitioners; therefore, while the dissertation may make mention of the importance of this stakeholder perspective, the policymaker perspective on quality of care involves issues more suited to a discussion of organizational ethics, which is beyond the purview of this research.

As discussed in Chapter Two, ethics education bears a strong relation to the patient perspective, a quality measure that is generally gauged by patient satisfaction regarding expectations and experiences (Sixma et al. 1998). Although the patient perspective has been studied extensively and defined by many experts, Campbell, Roland and Buetow provide a useful operational definition of the patient perspective on quality of care as "whether individuals can access the health structures and processes of care which they need and whether the care received is effective" (2000, 1614). This definition identifies two areas– access and effectiveness of care– that are essential components of quality of care from the patient perspective. Access includes items such as geographic and physical accessibility, the availability of facilities and services, and affordability of health care (Campbell, Roland, and Buetow 2000). While access is certainly an important

aspect of quality of care, it is not directly relevant to the concerns of this investigation. Effectiveness of care, however, is highly relevant when considering quality of care from the patient perspective, and its parameters will be discussed here.

Campbell, Roland and Buetow (2000) characterize the effectiveness of care in two domains- clinical care and interpersonal care. In the conceptual framework that the authors develop, they maintain that both domains are equally important in the quality process. Healthcare professionals are highly engaged, and therefore, highly influential in both the clinical and interpersonal care of the patient. The knowledge, behavior and attitudes of healthcare professionals, therefore, will have a significant effect upon patients' views of the care they receive. It would follow that improving the professional's competencies in these areas should significantly improve the quality of interpersonal care and clinical care as measured from the patient's perspective. This understanding reinforces Brook, McGlynn and Shekelle's definition of quality of care - to be treated humanely, with culturally appropriate, patient-centered care. Interpersonal care necessarily involves sensitivity and respect along with effective communications skills in order to successfully attend to the psychosocial needs of the patient and develop an effective, therapeutic patient-provider relationship. The practitioner must develop trust, convey empathy and obtain and relay information successfully (Campbell, Roland, and Buetow 2000). The patient perspective, therefore, underscores the central role of ethics in the quality of care. Achieving high levels of quality in health care means that professionals need to acquire certain virtues and attributes, such as sensitivity, integrity, empathy, respect and prudence. Consequently, enhancing the virtues and attributes of the professional through ethics education can in turn effect desirable improvements in quality

of care. In other words, ethics education is a crucial step in the direction of providing quality care to patients. This will be examined further in Section C.

It is also important to consider the professional's perspective as a stakeholder in quality of care assessment. The professional perspective includes variables such as setting standards of care and employing evidence-based guidelines (Grol 2001). While the goal of evidence-based guidelines is to ensure best practices, the guidelines are general approaches and not required treatment formulas. For that reason, selection and implementation of guidelines require the professional's experienced interpretation regarding the suitability for individual patients. Interestingly, Grol (2001) observes differences in the professional perspective across the globe, noting that despite the availability of similar research findings, the interpretation of research may vary widely in diverse countries, leading to different recommendations for the treatment and management of patients' problems.

To illustrate the significance of the professional perspective in defining quality of care and also the role of the professional in shaping quality care outcomes, it is helpful to use a recent example from medicine. The latest treatment guidelines for thrombotic therapy demonstrate the need for the practitioner's prudential judgment and expertise in applying the recommended standards (Guyatt et al. 2012). These recommendations have been developed to address a wide range of patient situations, and yet the practitioner's careful consideration remains critical in weighing the benefits and risks of treatment and prevention of thrombosis for individual patients in clinical settings. Various factors strengthen or weaken the need for the recommended protocols, while patient preferences

regarding medication therapy and associated risks also significantly influence the type of treatment, if any, that is initiated.

The process of formulating the standards was also conducted with care. In establishing a panel of experts to formulate the treatment standards, the Executive Committee of the American College of Chest Physicians (ACCP) took into consideration the likelihood of panelists' personal biases and conflicts of interest as potential influences in the formation of the guidelines. To manage this, the committee carefully screened the potential panelists for their financial relationships with external agencies (pharmaceutical companies, for example) and, for the most part, excluded participants who held significant financial conflicts of interest. The ACCP was also careful to prohibit participants with intellectual conflicts of interest from voting on the proposed guidelines (Guyatt et al. 2012). Intellectual conflicts are those viewpoints that arise from academic work, such as research or publications, which would unfairly influence and bias participants' opinions on the guideline recommendations (Guyatt et al. 2010). The ACCP developed a method to both engage the opinions and expertise of professionals with conflicts of interest but at the same time safeguard the final recommendations from undue bias. As a result, experts with conflicts of interest were able to present and review research findings, but they were prohibited from discussions that would influence the voting committee members (Guyatt et al. 2010). Additionally, the ACCP took into consideration sociocultural concerns when formulating the thrombotic therapy guidelines. The ACCP addressed this concern by including "health economists and experts on patient values and preferences" among the committee members in a concerted effort to address "issues related to resource allocation and variations in patient management related to

individual and societal norms" (Metersky and Nathanson 2012, 285). As a result of its approach to formulating recommendations for therapy, the ACCP constructed guidelines that were sensitive to particular groups and acknowledged the need for individualization in setting treatment plans for patients.

As the above example illustrates, evidence-based guidelines are subject to interpretation by health care practitioners and may be influenced by personal and sociocultural values (Grol 2001; Shaneyfelt 2001). In selecting and applying the recommendations, practitioners require an appreciation of the patient's understanding of his or her own needs, a practice that is consistent with patient-centered care, rather than a broad and general application of standards (Campbell, Roland, and Buetow 2000; Naylor 1995). This requires skillful communication that sensitively and effectively uncovers the patient's wishes and needs, respecting the patient's autonomy. As Naylor efficiently summarizes, "even good evidence can lead to bad practice if applied in an unthinking or unfeeling way" (1995, 841). The expert, prudent judgment of the professional is integral to successful utilization of any guideline.

The professional perspective of quality of care requires a delicate balancing of the practitioner's judgment of the best course of action, the patient's values and wishes, and cost considerations (McGlynn 1997). Despite efforts to take into consideration many patient variables, formulated guidelines can never fully address the complexity of patient situations; therefore, decisions are often challenging (Grol 2001). Practitioners need to be aware also that while many guidelines arise from high-quality research with a high confidence level, recommendations may also follow from lower-quality evidence or may result from bias and "vested interests of specific parties or industries" (Grol 2001, 2579).

Decision making in this context requires clinical competency, ethical awareness of the potential problems, excellent communication skills, and professional virtues, such as prudence and integrity, to achieve an optimal outcome. Practitioners also need to be sensitive to their own values and potential for bias as well as possible conflicts of interest in selecting the optimal treatment plan for patients. Ethics education, therefore, can positively shape the professional perspective of quality of care by preparing healthcare professionals with the knowledge, behavior and attributes essential to ethically develop, select and apply the most beneficial standards or guidelines for quality patient care and develop patient-centered approaches to care.

Surveys of patient-provider interactions and variables such as patient satisfaction, compliance and recall have been traditionally conducted to gauge the effectiveness of the patient-provider relationship (Hall, Roter, and Katz 1988). These surveys provide an indicator of quality of care from the patient perspective (Barnes et al. 1995; Davidson and Mills 2005; Hall, Roter, and Katz 1988). A meta-analysis by Hall, Roter and Katz (1988) evaluated medical encounters and patient outcome measurements and identified specific provider behaviors that are more consistently associated with measures of increased patient satisfaction. The meta-analysis found that factors such as the amount of information provided, positive nonverbal communication, positive talk, partnership building, and technical and interpersonal competence have been correlated with increased patient satisfaction (666). Among the variables included in the studies, patient satisfaction was more consistently related to the amount of information given, with more information contributing to increased satisfaction. Greater sharing of information was also associated with increased compliance (661). There was also a relatively large

positive association between patient satisfaction and interpersonal competence of the provider (663). A second meta-analysis by Hall and Dornan (1988) rank-ordered aspects of patient satisfaction, identifying how satisfied patients are with particular correlates of care. The meta-analysis of 107 articles revealed that higher patient satisfaction was associated with the overall quality of care, humaneness ("warmth, respect, kindness, willingness to listen, appropriate nonverbal behaviors, and interpersonal skill"), and competence of the care provider (936-937). Interestingly, the analysis revealed that certain aspects of care such as humaneness and "informativeness" (related to diagnosis and treatment) were more frequently assessed in patient satisfaction studies, while other aspects, such as attention to psychosocial problems and continuity of care were evaluated much less often.

In a review of the literature on patient satisfaction, Miaoulis, Gutman and Snow (2009) identified eight dominant themes. These include empathy/being cared for as a person, communication/information, satisfaction with medical service, improving patient satisfaction, perceptions of quality, importance of the physical environment, patient satisfaction and compliance, and organizational change (58-60). The authors found that the most important factor in patient satisfaction is 'empathy/being cared for as a person,' noting that ''a lack of empathy is interpreted by most patients as being disappointing and equivalent to a lack of quality'' (59). The review showed that second most important factor is communication/information, including both the amount and quality of provided information. The authors noted that the theme of 'satisfaction with medical service' included caring behaviors of the provider as well as the provider's competence, but they

concluded that "satisfaction is related more to the human dimensions of how the service was delivered, and less to the technical competence of the provider" (58).

The above analyses of patient satisfaction reveal that professional attributes, such as empathy, effective communications skills that build partnerships, and behaviors that convey 'humaneness,' are highly related to the patient perspective of quality of care. These predominant themes will be briefly examined below with an emphasis on their relationship to medical and nursing ethics.

B. Empathy and Communication and Their Relationship to Quality of Care

The definitions of quality of care presented above confirm the role of professional ethics in achieving positive outcomes for patients. While technical competence is vital to quality care, both the patient and professional perspectives of quality of care are also greatly influenced and determined by the professional's values, virtues, attributes, and decision making abilities. In the next chapter where virtue ethics is the focus, the significance of virtues to the healthcare professional will be underscored, demonstrating, for example, how integrity is critical to establishing trust, and prudence is vital to decision making. These virtues are outward expressions of the values of the healthcare professional and instrumental to developing and maintaining the patient-practitioner relationship. Properly engaged, they are influential in creating positive outcomes for patients.

Empathy, communication skills, and affective behavior, while not virtues, are essential to ethics, and correlatively, they are also fundamental to the quality of care for patients. Empathy, for example, has been defined as having a moral component that

enables the development and expression of altruism and ethical agency (Mercer and Reynolds 2002). Oxley observes, however, that "empathy is not intrinsically moral" (2011, 4). She asserts that "empathy *alone* is insufficient as a moral guide"; yet, when used with appropriate ethical principles, empathy supports moral deliberation, reflection and action (5). Empathy and affective behavior also assist in establishing effective communication, which is vital to agency in ethics. The meta-analyses and reviews of the patient perspective of quality of care presented in the above section stress the importance of specific professional behaviors that are fundamental to ethics - empathy and communication - which are explored below.

The definition of empathy has long been debated in medical literature, particularly as to whether empathy is a cognitive or affective construct or if it is actually comprised of both cognitive and affective elements (Hassenstab et al. 2007; Hojat 2009; Kim, Kaplowitz, and Johnston 2004). This point, although interesting, is not significant to the present discussion. A practical definition of empathy for the health care professional is "the patient's perception... of his or her feelings of being understood and accepted" (Kim, Kaplowitz, and Johnston 2004, 239). The Scottish Enlightenment scholar, Adam Smith (1790), developed a useful argument on empathy that will be shared here, although it should be noted that Smith used the term sympathy to describe this construct. Commentators observe that the word for empathy was not in use during Smith's time (Agosta 2011, Introduction); therefore, Smith's argument will be presented using the terminology of his era. Smith understands the fundamental position of sympathy (i.e. empathy) as a moral sentiment and argues that it is the foundation of moral judgment. He writes that a person's idea of right and wrong stems from an "immediate

sense and feeling" (Smith 1790, VII.III.14). For Smith, moral judgment flows from sympathy, which he describes as "fellow-feeling," rather than from reason (I.I.3 and VII.III.14). He writes that sympathy "is the source of our fellow-feeling for the misery of others, that it is by changing places in fancy with the sufferer, that we come either to conceive or to be affected by what he feels" (I.I.3). Changing places "in fancy" is to imaginatively enter the other's situation or as in the German translation of empathy, *Einfühlung*, "feeling one's way into" the other (Agosta 2011, Introduction).

For Smith, the moral sentiments encourage an appreciation of the experiences of the person and serve as the foundation for making moral judgments. This process of thought and feeling, which is based upon an affective understanding of the position of another, clearly highlights the importance of relationships. To use Brooks' parlance about human relationships, it fills the "spaces between people" (2010). It also brings to the forefront the significance of the professional's skillfulness in empathy as vital to the process of ethical decision making. Göçmen observes, "in order to understand and communicate with one another, human beings must, according to Smith, sympathise with one another and therefore always go beyond themselves. Furthermore, if they cannot understand one another, they also cannot understand themselves" (2007, 3-4). The significance of empathy to relationships, particularly in the context of healthcare delivery, cannot be understated (American College of Obstetricians and Gynecologists [ACOG] 2011; Platt and Platt 1998; Reynolds and Scott 1999). Empathy requires the person to imagine or "step into the other's shoes" in order to appreciate the experiences and feelings of the other. Additionally, empathy in a healthcare setting also necessitates communicating one's perceptions of the patient's feelings or experiences for validation

(Platt and Platt 1998; Reynolds and Scott 1999). Successfully appreciating and communicating the other's experiences or feelings requires skills formed with experience, practice and habituation.

Smith's eighteenth century view places emphasis on the healthcare professional's empathy as an essential skill and strengthens the fundamental role of the person in meeting the ethical needs of the other, establishing the constitutive position of empathy in ethics. Smith's views of the relational aspect of empathy also reinforce the essence of communication to empathy - effective communication is vital for empathy and, therefore, to ethics, in general. Empathic physicians and nurses require effective communication skills and education on these skills (ACOG 2011). Highly empathic nurses and physicians have developed effective communication skills and have honed their abilities to interpret verbal and nonverbal cues of their patients. Empathic practitioners, therefore, can gain a more accurate view of the patient's perspective.

The literature treating empathy in medicine and nursing, therefore, has relevance to ethics. Empathy may be seen as an outward expression of respect and valuing of the person, setting into motion beneficence, altruism, caring practices and justice (Agosta 2011; Gelhaus 2012; Oxley 2011). As an emotion, it reflects a dimension of our "interpenetrated" social existence, as Brooks (2010) would describe our human connection. Empathy also has considerable bearing in communication processes. Excellent communication skills are necessary for "recognizing and responding empathically to patients' verbal and nonverbal clues, thereby inviting patients to express their concerns" (ACOG 2011, 4). The empathic practitioner, therefore, characteristically displays openness, compassion and sensitivity.

Literature from medicine, nursing, psychotherapy and related health disciplines offers much evidence to demonstrate the therapeutic benefit of empathy within the practitioner-patient relationship and how it leads to positive patient outcomes. A review by Squier (1990) highlights the range of patient outcomes to include patient satisfaction, enhanced adherence to treatment, reduced anxiety and stress, increased trust, better exchange of information and increased self-efficacy. Squier maintains that the evidence shows that healthcare professionals not only have an ethical responsibility to attend to the quality of their relationships with patients, but quality relationships are "necessary to ensure that patients in fact benefit from the health care which is being given them" (326).

Kim, Kaplowitz and Johnston (2004) measured patients' perceptions of empathy in their physicians and the effect on patient outcomes. The study showed that empathic communication by physicians had a statistically significant influence on patient satisfaction and compliance (246). The results indicated that "the cognitive component of physician empathy led to better exchange of cognitive information, and the affective aspect of physician empathy led to partnership" (244). Furthermore, the study revealed that partnership was shown to increase interpersonal trust, and in addition, affective empathy and partnership contributed more significantly to increased satisfaction and compliance. The authors concluded, "The effective use of empathic communicative skills may be one of the best ways to improve patient satisfaction and patient compliance" (248). Additional favorable outcomes may follow from enhanced satisfaction and adherence, such as improved health outcomes, reduced health care costs and decreased malpractice claims.

Zachariae et al. (2003) investigated the communication behaviors of physicians in relation to outcomes of satisfaction, lowered emotional distress, self-efficacy and perceived control in oncology patients. The study authors found that cancer patients who rated their physicians as higher in attentiveness and empathy reported greater satisfaction, less distress and higher self-efficacy. Conversely, patients of physicians who received lower ratings for attentiveness and empathy had lower levels of patient satisfaction. In a factor analysis, empathy was a strong predictor in reducing distress. Of interest, the study findings also showed that patients in palliative care were less satisfied with the communication sessions than the other participants, which is unfortunate because patients in palliative care have a greater need for support and empathy. This finding calls attention to the need for practitioners working with patients in palliative care to form better quality relationships.

A study by Williams (1979) examined the relationship of empathic communication to the outcome of self-concept in patients in long-term care facilities. The study revealed that the presence of higher empathic communication by a nurse therapist enhanced the self-concept of the patients in long-term care facilities. Conversely, lower empathic communications resulted in a decline in self-concept. The results illustrate the power of empathy to support patients' psychological and social dimensions and presents opportunities for physicians and nurses to draw on this communication skill to enhance patient outcomes.

A review of the literature conducted by Beck, Daughtridge and Sloane (2002) on physicians' communication behaviors and their relationship to positive patient outcomes found a positive correlation between physician behavior and favorable patient outcomes.

The database for the review was comprised of research studies of primary care office interviews conducted from 1975-2000 that measured either verbal or nonverbal communication skills. Of the studies measuring verbal communication skills, the majority assessed patient outcome measurements of satisfaction (n= 23), while others evaluated compliance with a medical regimen (n= 13) and less often, comprehension. The review showed that twenty-two verbal behaviors, such as communicating empathy, reassurance, support, patient-centered behavior, courtesy, openness, listening behavior and more, were associated with favorable patient health outcomes (31). Conversely, the review revealed fourteen verbal behaviors that were negatively associated with patient outcomes. The negative behaviors included lack of respect and compassion, inattentiveness, one-way flow of information, directiveness, and dominance, among others. The review reinforces the value of empathic behavior and competency in communication and supports medical education that develops these abilities.

As the above studies of empathy and communication skills reveal, empathy is a powerful interpersonal skill of the healthcare professional. Although empathy has been traditionally linked to improved patient satisfaction in the emotional and social dimensions of care, studies have also shown a positive association between empathy and improvement in physical health outcomes. Hojat et al. (2011) examined the relationship between physician empathy and improvements in the physical health status of diabetic patients. Health status was assessed by blood glucose control as measured by hemoglobin A1c (A1c) and the levels of low-density lipoprotein cholesterol (LDL-C). Empathy was assessed with the Jefferson Empathy Scale; physicians were categorized as high-, moderate-, or low-scoring. Overall, the association between empathy and A1c (p<.001)

and empathy and LDL-C (p<.001) was highly significant, revealing that empathy had a statistically significant and substantial effect on the measurements (361). Specifically, the results demonstrated that patients of physicians who scored higher in empathy were more likely to have good control of A1c (p<.01). Patients of physicians who scored high in empathy also were likely to have good control of LDL-C (p <.01). The study authors hypothesized that empathy enhances trust and facilitates the patient-physician relationship, leading to improved diagnosis and adherence to therapy (362). The authors submit that the results add further support to efforts to develop the empathic skills of medical students and physicians.

Empathy of the healthcare provider also has the potential to enhance the placebo effect (Meissner et al. 2011; Turner et al. 1994; Wells-Federman et al. 1995). A study by Kelley et al. (2009) on placebo effect and patients with irritable bowel syndrome (IBS) revealed that empathic practitioner communication style has the potential to actually heighten the placebo effect. In this study, licensed acupuncturists provided placebo or "sham" acupuncture to two groups of patients with IBS. One group received placebo acupuncture with limited interactions that were "neutral and business-like" (790). The other group received placebo acupuncture but with augmented interactions in which the practitioner was intentionally "more empathic, more sensitive to the patient's feelings, more supportive of the patient, more validating of the patient's perceptions, more nonjudgmental and accepting of the patient, and more responsive and affectively involved" (794). A third group served as a control and received neither acupuncture nor an interaction with a practitioner. The study demonstrated a significant placebo effect with improved symptoms in both the limited and augmented interaction groups, but

greater results in the augmented group. More pertinent to this discussion, the results showed a statistically significant practitioner effect on the treatment outcomes, with greater effect in the augmented group, inferring that empathy, sensitivity and other characteristics of a therapeutic relationship produced more favorable patient outcomes in this population.

It is clear to see that empathy and communication cannot be easily separated. Communication is used to bridge the gap in relationships, allowing a person to discover and affirm the joys, needs, and suffering of the other. The effect of communication on patients' well-being has been established, but it is not yet well understood. Communication skills that effectively convey respect and empathy, affirm perceptions, and provide information in the physician-patient or nurse-patient relationship have the potential to lead to positive outcomes for patients. Street et al. theorize various "pathways" through which therapeutic communication leads to positive patient outcomes (2009, 295). The authors call for more direct studies that identify specific pathways linking communication and patient outcomes. For instance, in a study by Stewart et al. (2008) patients' perceptions of patient-centered communication were associated with self-reported positive outcomes in health, such as lower levels of discomfort and improved mental health. The results were obtained two months after the patient-physician encounter, inferring a longer duration of effect from patient-centered communication. The study authors controlled for confounding variables and concluded that the positive results infer a "pathway [that] suggests a process through which patient-physician communication influences patients' health, by first influencing the patients' perceptions of being a full participant in the discussions during the encounter" (799). For the most

part, researchers may infer the effect of a particular behavior or attribute on a patient outcome, but it is difficult to control for other influences, such as from additional care providers, that may contribute to the outcome. It is encouraging to observe that the effect of patient-centered practice is associated with prolonged positive outcomes. It is difficult, though, to distinguish the effect of this behavior from additional influences during the two-month period.

The healthcare literature provides many excellent examples that demonstrate the positive impact of good communication skills. As a result, it is critical that medical and nursing education programs continue to place emphasis on developing and nurturing excellent communication skills in students to enable empathic, therapeutic relationships with patients.

# C. The Contributions of Ethics Education to the Patient and Professional Perspectives of Quality of Care

The above elements of medical and nursing ethics have been examined over the past decades for their effect in improving patient outcomes, and the results confirm that there is a positive relation between provider behaviors - empathy, communication skills, openness, ability to consider the patient's perspective, competency in weighing benefits and risks - and quality of care. It follows then that ethics education that teaches, facilitates and reinforces these behaviors can ultimately lead to better outcomes for patients. This is particularly relevant at this point in the history of medical and nursing education, as some commentators and researchers have observed a decline in ethical behaviors as noted in Chapter One in the discussion of ethical erosion. Moreover, there

has been pressure to increase competencies in other areas of medicine and nursing to accommodate the growth of knowledge and new initiatives, such as quality and safety, which can potentially alter educational priorities and lessen programming in ethics education, professionalism and communication (Ward et al. 2012), factors which contribute to the patient and professional perspectives of quality of care.

The strength of the professional attributes to foster better quality care outcomes for patients demonstrates how vital and necessary continued educational efforts are to develop and nurture these characteristics and skills in healthcare students and professionals. For example, physicians who received an eight-hour training session in communication skills that focused on recognizing and managing psychosocial problems were more proficient in identifying patients' emotional distress and employing strategies to assist patients (Roter et al. 1995). Patients of physicians who received communication skills training also reported lower levels of emotional distress. This group of physicians also scored higher in empathy skills. Furthermore, the effects of training were evident over a six-month period; the patients of the physicians who had received communication training had reduced levels of emotional distress when assessed six months following the physician-patient encounters (1883). The results of this investigation suggest that education and training that enhances psychosocial skills and interpersonal competence, such as active listening, empathy, acknowledgment and reassurance, can create positive outcomes and enhance the quality of care.

The following study did not measure quality of care outcomes but provides evidence for the effectiveness of educational programming in empathy and communication training. In a study on communication skill training to improve patient-

centered discourse, Sheldon (2011) noted that communication training with oncology nurses has the potential to enhance empathic behavior, reassurance, allow for more uninterrupted dialogue, improve reciprocity and reduces medically-oriented dialogue.

In an analysis of the effectiveness of empathy education in nursing, Brunero, Lamont and Coates (2010) examined seventeen quantitative and qualitative studies in undergraduate and graduate programs. The majority of the studies used an active learning approach, such as case-based learning, role play or media-based learning. Interventions varied widely in length, from six to 105 hours, and a variety of measurement tools were used to assess for a change in empathy levels pre- and post-intervention. The authors found that a majority of the studies (approximately 65%) demonstrated a statistically significant improvement in empathy skill level in the participants following educational interventions. The authors noted that eight of the eleven studies that employed an experiential method observed an increase in empathy skills. Interestingly, shorter educational interventions were associated with favorable outcomes (70). These findings suggest that focused, active learning experiences can produce positive results in developing empathy, an essential component of ethics education.

An analysis of empathy research in medical programs shows similar findings. In a review of empathy education in medical school programs. Stepien and Baernstein (2006) identified thirteen research articles focusing on methods to enhance empathy. Similar to the findings of Brunero, Lamont and Coates above, the studies included both quantitative and qualitative design and used various methods for measuring empathy. The research articles also showed that a variety of educational methods were employed; however, the majority used active learning methods, such as communication workshops, and two used

experiential approaches. The most highly experiential approach placed medical students as patients with fake diagnoses in a teaching hospital for a 24-30 hour period (Wilkes, Milgrom, and Hoffman 2002). The students were cared for by residents who were unaware that the students were not actual patients. The experience was highly effective in producing empathic responses in the students. All of the research articles included in this analysis reported improvements in empathy among the medical students following the interventions.

As the review of studies on professional attributes such as empathy and patientcentered communication implies, activities that strengthen professional attributes contribute to the development of ethically sensitive and responsive practitioners. The ethically-responsive nurse or physician can better understand the patient as a person and develop an enriched patient-practitioner relationship. Approaching the patient-provider relationship in this manner is more likely to improve patient satisfaction and outcomes, thus advancing certain measures of quality of care. In addition, ethics education aims to produce a prudent healthcare professional who is better able to discern and weigh the benefits and harms of established standards of care and consider the patient's preferences in clinical decision making. Skills in prudential reasoning will facilitate the selective application of recommended standards as appropriate to individual cases, producing better patient outcomes, and thus improving quality of care as measured by the professional perspective.

A recent initiative by the National Center for Ethics in Health Care (NCEHC 2012) showcases a major educational and organizational program that aims to improve the quality of care in ethics. In an effort to proactively prevent, identify and/or manage

ethical problems more effectively and at an early stage, the NCEHC developed IntegratedEthics: Improving Ethics Quality in Health Care for the Veterans Health Administration (VHA) (2012). This is a comprehensive program that uses a quality approach to systematically address and manage actual and potential ethical issues in the healthcare system by narrowing "ethics quality gaps" (Fox et al. 2010, 15; Fox et al. 2012). The three-pronged program includes major modules dealing with such areas as "Ethics Consultation," "Preventive Ethics," and "Ethical Leadership." The goal of "Ethics Consultation" is to improve the resolution of ethical concerns, thereby improving quality of care. "Preventive Ethics" addresses recurring ethical issues using systems measures. "Ethical Leadership" seeks to provide an ethically supportive environment. Each arm of the program has an educational component, employing a variety of teaching methods for the healthcare team that include video courses, training exercises and written materials. The IntegratedEthics program has demonstrated that education, effective resources and a supportive administration can create an environment that has a positive effect on patient outcomes.

The following example from the IntegratedEthics program illustrates the effectiveness of ethics education on the quality of patient care. In an instructional video included in the "Preventive Ethics" educational materials (NCEHC 2012, Preventive Ethics), a healthcare team of nursing and social service employees address the ethical concerns posed by patients who are without any identified surrogate (or proxy) decision makers. The healthcare team recognizes the potential problems that might arise should such a patient lose his or her decision making capacity. The team considers the frequency with which patients were admitted without an identified surrogate decision maker.

Although the team noted that a hierarchy for decision making would be employed, they observe that this protocol is not always the most effective means to manage the situation. "Preventive Ethics" makes use of systems theory to aid in problem identification, goal setting, strategies, implementation and evaluation to address recurring issues, called the ISSUES approach (NCEHC 2010). The team's interventions have significantly reduced the numbers of patients who do not have identified surrogates, demonstrating that education, effective resources and a supportive administration can create an environment that has a positive effect on patient outcomes. It is reasonable to suggest that the efforts of nursing and social services to obtain surrogate information in advance to protect patients' rights are positive influences in achieving better interpersonal and clinical care and would be perceived satisfactorily by the patients. New guidelines to ensure and maintain standards of care, in this case, to protect the patient's right to self-determination, resulted from the health teams' efforts, thereby improving quality of care from the professional perspective.

# D. Conclusion

The chapter began by linking ethics and quality of care and concludes by making explicit the relationship between ethics education to quality of care. This is principally in its contribution to the development of the provider's attributes and behaviors, such as empathy, communication and prudence. Examination of quality of care from either the patient or professional perspective shows that success is dependent upon the professional. Importantly, ethics education can and must explicitly direct its goals to positively influencing the quality of care. Healthcare professionals must act with empathy, respect and care to contribute positively to both the interpersonal and clinical care measures of

quality of care. Furthermore, they must exercise prudence in selecting and applying standards of care and act with consideration of the patient's preferences. Standards and guidelines are not foolproof, nor mandated; they must be prudently considered and applied, not imposed upon patients. By including effective ethics education in medical and nursing school curricula, schools better meet their responsibilities to the profession and to society to prepare graduates who are most prepared to relate successfully to patients, communicating with the patient as a person. It is important for educators and students to keep in mind that the ultimate goal of ethics and ethics education is for the betterment of the patient.

Developing the capacity of students to relate to the patient as a person is a critical component of ethics education and entails the development and nurturing of professional virtues. Educational efforts at the institutional level, as the VHA program illustrates, continue the development of the professional and improve goal attainment in the quality initiatives. Responding with ethical-sensitivity and reflection will enable prevention, early identification, and satisfactory resolution of ethical problems. In the next chapter the relevance of virtues to health care and the specific virtues that are essential to the nurse and physician are examined.

# **Chapter Four – Virtue-based Ethics in Medicine and Nursing**

#### A. The Relevance of Virtue Ethics in Health Care

As demonstrated in Chapter One, the need for an increased emphasis on ethics education is due in part to the current state of affairs in health care delivery that presents obstacles to the humanistic aspects of the patient-provider relationship. In addition, the increased stress on students in educational programs to acquire vast amounts of knowledge has worked to the detriment of instruction in ethics in both medical and nursing programs. The persistent negative effects of the hidden curriculum and a decline in professionalism - decreased empathy, moral erosion, increased cynicism as well as the continued presence of moral distress- continue to create challenges to humanistic care. As established in the preceding chapters, ethics teaching that features virtue-based ethics has the potential to counteract these dehumanizing tendencies.

Presently, ethics education in healthcare is characterized by a reliance on traditional ethical theories, such as deontology and utilitarianism, and sets of principles (e.g., principlism) that stress the obligation or duty of the healthcare professional (Devettere 2010). The review of ethics education in medical and nursing programs presented in Chapter Two confirms this state of educational affairs, where the teaching of ethics predominantly emphasizes decision making skills over the virtues, character traits or attributes of the professional. Theory and principles are certainly important to ethics education, but an overreliance on these has resulted in an incomplete set of tools in ethics education for nurses and physicians. Theory and principles, in effect, are necessary but not sufficient in producing the best educational results (Devettere 2010). Duty or

obligation-based theories and principles are concerned with the actions a person ought to take, but these theories and principles do not focus on the qualities of the person who serves as the moral agent. Virtue-based ethics, alternatively, offers another perspective and means of applying ethics in healthcare, one that focuses on the disposition and character of the person. With roots in the ancient cultures and first formulated by Aristotle, virtue-ethics has been influenced by the Judeo-Christian tradition as well as by secular philosophy. It centers on both moral and intellectual elements along with highlighting the dispositions of the moral agent and ethical decision maker. Virtue-based ethics, while not sufficient by itself either, is essential in the ethics preparation of nurses and physicians. Ethics is complex, and reliance on one approach alone may not provide the student with adequate tools to navigate the ethical dimensions of the patient-provider relationship and any subsequent ethical quandaries. The role of virtue ethics has been minimized at times since the 1970's (Johnstone 2009), and yet it provides excellent guidance to the practitioner in terms of the dispositions necessary to be an effective moral agent.

The major categories of ethics – principles, duty, virtue and rights – are interrelated and essential to understanding ethics (Drane 1988). Drane explains their relatedness, observing that principles "serve as the ground of duties," rights provide claims for what one is due and claims for noninterference, while virtues "establish dispositions and habits of living according to objectively right standards" (1988, 155). He observes:

Virtues are not the whole of ethics, but they contribute to good behavior, good persons and even to good societies. Virtuous acts contribute both to the good (fulfillment) of others and the good of the acting person by creating a readiness

and orientation to right action. As virtue is developed, it provides practitioners with a practical moral wisdom, and with a way of achieving the good which is different from act analysis and norm creation (Drane 1988, 159).

Although Drane is writing about virtue ethics and the development of the "good doctor," his observations apply to nurses, as well. He rightly observes that not all of ethics in medicine (and nursing, by extension) involves ethical dilemmas for which act analysis and rule applications are required. Physicians and nurses encounter many situations that have an ethical dimension, but these are not always conflict-based and for which "what is needed is not a strategy or a rule but rather a style of life, or way of being that fits or is appropriate" (Drane 1988, xi). For Drane, virtue-based ethics fills this need. Virtue-based ethics is understood as "lived ethic," where ethics arises from the vital relationship of the patient and practitioner (Drane 1988, 16). As a lived ethic it comprises both the internal and external dimensions of the person; that is to say, what is important in virtue ethics is both the private character of the person and the public action that the person takes as a moral agent. Individuals should embody the virtues needed to do what is right (Athanassoulis 2004). As noted in the passage by Drane above, virtue provides the disposition or "readiness and orientation" to do the right thing. The fundamental intellectual virtue of prudence or practical wisdom enables the person to understand the right action (Drane, 1988). As this conception of virtues shows, virtue is closely aligned with character.

Virtue-based ethics has its origins in the philosophy of Aristotle and while always an important set of tenets, it enjoyed a resurgence starting in 1958 with the writing of Anscombe and again in the 1980s with the work of MacIntyre (Athanassoulis 2004; Pellegrino and Thomasma 1993). Presently, there is growing appreciation of virtue ethics

in medicine and nursing (Pellegrino and Thomasma 1993). The word virtue comes from the Greek, *arête*, meaning excellence, which is that toward which individuals should aim (MacIntyre 2007). Aristotle wrote of virtue or excellence as having three forms- moral virtue (excellence of character), intellectual virtue (excellence of intelligence) and physical or bodily excellence (Begley 2006). It can be noted that in contemporary society, physical excellence is no longer considered relevant to one's moral being (Begley 2006); commentators on virtue-based ethics in health care, however, are concerned with the moral and intellectual virtues, and this will be described in more detail in section B below. Moral virtues are attributes or dispositions of character and include temperance, justice and courage. The intellectual virtues include both theoretical knowledge and practical wisdom or prudence. Prudence or *phronesis* is knowledge applied to practical concerns. Aristotelian scholars note that moral virtues require practical wisdom. One needs excellence of character for prudential reasoning, and prudence is instrumental in achieving moral excellence (Devettere 2010; MacIntyre 2007). As Devettere explains, moral virtue and practical reasoning work in concert:

A degree of moral virtue is necessary for the relevant intellectual virtue, prudence, to function well, and a degree of prudence is necessary for morally virtuous decision making in each particular situation. Every moral virtue presupposes prudential reasoning, and sound prudential reasoning presupposes the person has already developed some level of moral virtue (2010, 28).

Writing on the historical perspective of virtues, MacIntyre (2007) notes that virtue is a social phenomenon and contextually-driven, and as such, the understanding of virtue has changed over time. He points to the different conceptions of virtues - from the Aristotelian view of virtues as aimed at attaining *eudaimonia* or well-being to the

medieval view that virtues are directed to overcoming hardship; of the historic shift from pagan virtues to Christian virtues; and the juxtaposition in cardinal virtues (justice, temperance, courage, prudence) with the theological virtues of faith, hope and charity. This conceptualization frames contemporary virtues in their social contexts and strengthens the importance of relationships in virtue-based ethics. It is from the relational aspects of the patient-nurse or patient-physician relationship that the lived experience of ethics arises.

Arguments in support of the relevance of virtue-based ethics to health care turn on the relational aspect of health care and the personhood of the patient and practitioner in navigating the complex issues that comprise the healthcare system, many of which have no ready answer (Drane 1988, 18). As Drane observes, virtue ethics has an important place in clinical ethics but not to the exclusion of norms and principles. Most nurses and physicians would agree. Virtue-based ethics supports the development of a caring practitioner who may be better disposed to nurture the patient-professional relationship. Cassell states simply, "Ethics is about relationships" (Cassell 2007, 21). He observes that practitioners may have no "answers except as fellow human beings" Cassell reminds the reader of the intricacies of relationships, submitting that "whatever is done for one person inevitably has implications for others, some of whom are so close to that person as to be almost one with him or her" (20). From this perspective it is argued that healthcare professionals need particular virtues, attributes and excellent interpersonal skills to build a trusting relationship and effectively engage the patient as a person. While it is important that a physician or nurse can competently articulate and defend a value or ethical position, solve a pressing dilemma or sort out an ethical issue, it is more important that

the same physician or nurse be virtuous, considering an issue from a position of caring, integrity and honesty. Furthermore, it is the humane and compassionate physician or nurse who is better able to recognize and respond to a potential or actual ethical problem. The intellectual virtue of prudence, in particular, appreciably helps the person to shape the action that is needed.

Gastmans provides support for the significance of ethics in nursing that is based on the foundation of the nurse-patient relationship. The nurse-patient relationship is a personal model, established on a model of caring that is "ethically laden" (2002, 9). In such an engagement, a nurse's "ethical practice becomes concrete through the personal relationship between the nurse and the patient" (2002, 496). As a practice, nurses strive to provide holistic care that comprises all dimensions of the person, including the physical, psychosocial, relational, spiritual and moral levels (Gastmans, Dierckx de Casterle, and Schotsmans 1998). The nature of the care provided is goal-oriented that seeks to achieve an ideal of the best care for patients. Nursing practice entails care decisions from many options, requiring careful consideration as they are applied to particular patients to achieve good care. Aristotle understood virtue (arête) or excellence as the means toward achieving the "good." It follows then that a nurse's practice is a moral practice, because it has as its aim producing a particular good for a patient, namely a healthy body and/or mind or a healthy patient-practitioner relationship (Gastmans, Dierckx de Casterle, and Schotsmans 1998). The relational nature and the goals of the relationship give rise to the ethical practice of the nurse. Caring practices require much more than technical competency to achieve excellence. Good care necessitates a caring attitude. Gastmans, Dierckx de Casterle and Schotsmans clarify, "It is only by integrating a virtuous attitude

of caring with the competent performance of care activities (*caring behavior*) that *good care* can be achieved" (1998, 45).

Writing prominently on the value that virtue has and should have to the healthcare professional, Pellegrino's stance supports virtue-based ethics from his conceptualization of the patient-physician relationship. Like Drane, while his arguments have been written mainly for medicine, his concepts are also highly relevant to nursing and other healthcare professionals as his position highlights the ways in which virtues nurture the relationship between the patient and the practitioner. This is particularly true in nursing, whereby the nurse-patient relationship encompasses a holistic patient perspective, as Gastmans, Dierckx de Casterle and Schotsmans (1998) underscore. The value of virtue stems from the human nature of health care and the relationships formed that have a "fundamental human grounding" (Pellegrino 2008b, 152). To navigate these human relationships, the healthcare professional requires a virtuous stance. Pellegrino submits that virtue ethics does not provide answers to ethical problems, but it provides a balance to the reliance on ethical norms and principles (Pellegrino and Thomasma 1993). The development of the virtuous professional consequently nurtures the relationship with the patient. The virtuous professional will also protect patients and guard against questionable practices that although legal, border on the unethical (Pellegrino and Thomasma 1993). Nurses and physicians may find themselves functioning at the "moral margin" - the place "where law, custom, or principle would allow certain acts (or fail to prohibit them) that virtue would prohibit" (Pellegrino and Thomasma 1993, 169).

Pellegrino and Thomasma provide an example of the moral margins that result from the current economic model of health care, where unavoidable conflicts of interests

arise when health care professionals become owners of medical equipment, laboratories or skilled care facilities. Virtue-based ethics, they argue, is incompatible with the contractual model of health care implied in these entrepreneurial efforts. They also caution against the "moral malaise" that develops when self-interests trump professional virtues (1993, 145). The authors describe situations in which healthcare professionals circumvent care for those who are difficult to treat- notably, poor patients with complicated cases that pose a malpractice risk. Moral malaise also presents in nurses and physicians who fail to challenge a healthcare institution's practices that threaten quality of care or pose harm to patients. Identifying and responding to moral complicity in a situation such as this requires a virtuous professional who not only perceives the potential harm but is also disposed to act to protect patients. Educating for and nurturing virtue in nurses and physicians will help to guard against these moral gray zones where acquiescing to hierarchical pressures creates inequities or conflicts of interest (Pellegrino and Thomasma 1993). In light of the subtle, insidious nature of moral margins, moral complicity and moral malaise, one can argue that while virtues, ethical theory and principles are all necessary to the ethics education of the healthcare practitioner, virtuebased ethics that focuses on the dispositions or characteristics of the moral agent is in effect more fundamental and necessary in affecting the well-being of patients than theory or rule-based ethics. In essence, the virtuous physician or nurse perceives potential or actual problems and holds the necessary virtues to affect the good. Furthermore, many of the ethical problems that nurses and physicians encounter are not conflict-based but stem from the ethical dimensions of the patient-practitioner relationship, a metaphorical space that appropriately applied virtues can bridge.

# B. What Virtues Are Important In The Nurse And Physician?

The preceding section and chapters have argued that virtues are essential to the nurse-patient and nurse-physician relationship. The question remains, though, as to which *particular* virtues should be developed and nurtured. One can make a determination of which virtues or dispositions the professional should hold and educationally develop by examining several pertinent sources, such as classic virtue-based ethics, the expectations of accrediting organizations, the provisions of professional codes of ethics and the demands of quality of care perspectives.

Traditional accounts of virtue-based ethics provide an instructive place to begin. While numerous lists of virtues have been posited from the classical period to the present, Aristotle maintained that the dispositions that one needs are those that will lead to *eudaimonia*, often translated as well-being, flourishing, or good (Devettere 2010; MacIntyre 2007, 148). One can debate about what Aristotle and subsequent writers of virtue-based ethics understand by "good"; however, in healthcare, producing good or what is right is generally understood as providing excellent care for the patient. From the classical perspective, the moral virtues or dispositions that are required to bring about good include love, justice, dignity, temperance, courage and gentleness, among others (Devettere 2010). Significantly, the Aristotelian perspective contends that it is of equal importance to possess both moral virtues (dispositions of character) and intellectual virtues (theoretical and practical wisdom) to achieve what is good or right (Begley 2006; MacIntyre 2007). MacIntyre explains that for Aristotle "excellence of character and intelligence cannot be separated" (2007, 154), an important consideration in the ethics teaching of the nurse and physician. Thus, both are needed to create what is right or good.

Some have maintained that while both moral virtues and intellectual virtues are essential, a certain virtue is more important than the others. MacIntyre makes a strong case for the centrality of prudence or *phronesis*. He claims that the intellectual virtue of prudence is necessary to apply the moral virtues, submitting, "it is that intellectual virtue without which none of the virtues of character can be exercised" (2007, 154). To show wisdom is to "exercise judgment in particular cases" (MacIntyre 2007, 154), knowing what is good or right, and in addition, to know which virtue to exercise to achieve this aim. According to MacIntyre, Aristotle's views on the relationship of the virtues underscore the inseparability of moral and intellectual virtues:

The exercise of intelligence is what makes the crucial difference between a natural disposition of a certain kind and the corresponding virtue. Conversely the exercise of practical intelligence requires the presence of the virtues of character; otherwise it degenerates into or remains from the outset merely a certain cunning capacity for linking means to any end rather than to those ends which are genuine goods for man. (2007, 154).

Prior concurs, noting that "it is the virtuous person who sees what is morally salient in most situations, and the virtuous person sees the morally salient fact as a reason for action because he or she already is predisposed by virtue to act appropriately" (2007, 59). The perception that Prior describes is a feature of prudence. One needs to be predisposed to act appropriately on these perceptions, revealing the relatedness of the moral and intellectual virtues. Prudence or practical reasoning compels professionals to reflect upon and recognize the effect of their dispositions toward patients. The nurse or physician also requires practical wisdom or prudence in everyday practice to deliberate among choices when weighing projected outcomes of care.

Writing specifically on virtues in nursing, Johnstone explains that the end or *telos* of nursing is a "moral end" in which the nurse works toward the promotion of health and wellbeing and toward the alleviation of suffering (2009, 60). This goal is understood as "good" care and establishes the moral behavior or "moral practice" of the nurse (60). She observes that good care as an end encompasses both competent care and "virtuous caring" (see Gastmans, Dierckx de Casterle, and Schotsmans 1998, above). Johnstone clarifies that virtuous caring comprises both the virtuous dispositions of the nurse as well as the motivation to act morally, that is, "caring about" the patient, family or community (60). Therefore, the virtues that are required for nursing consist of those that support caring practices as well as "right attitudes" and include "compassion, empathy, concern, genuineness, warmth, trust, kindness, gentleness, nurturence, enablement, respect, mutuality, 'giving presence' (being there), attentive responsiveness, mindfulness, providing comfort, providing a sense of safety and security and others" (60).

Pellegrino and Thomasma offer a slightly different account of the position of virtue-based ethics to the physician. The authors write on the virtues of the physician, but their position can be applied to the healthcare practitioner, in general. Generally, for Pellegrino and Thomasma, virtue is defined by the ends or goals of the profession, not by the definitions of good character and is evident in the "ends to which those professions are dedicated" (1993, 146). They describe the virtues of the "physician *qua* physician, not the virtues of the physician *qua* person, which is a broader and more private enterprise" (1993, 178). Thus, they assert that only the virtues of the physician as he or she acts in the professional role are of concern. The excellent professional knows what virtues are necessary to apply from examining his or her moral responsibilities as a professional.

Pellegrino and Thomasma's observations, therefore, would necessitate that the physician reflect on the values and responsibilities of the professional ends or goals and act with the appropriate virtue. For example, if the physician has a responsibility to provide informed consent, the correlative virtues needed to meet this end are respect, truthfulness, sensitivity and prudence. If the physician's duty is to protect patients from medical harm and the physician observes an impaired colleague, moral courage and advocacy are required. If the nurse has a responsibility to provide comforting practices for the dying patient, the correlative virtues that would enable the attainment of this end are care, compassion and empathy. If the nurse's goal is to preserve the dignity of the drugdependent patient, virtues of respect, presence, and care are required. In Pellegrino's terms, the virtues appropriate to a situation are those "essential to achieving the ends of medicine optimally and without which those ends would be frustrated or attained in less than optimal fashion" (1995, 268). This approach to virtue ethics stops short of referring to codes of ethics and laundry lists of virtues. Rather, the virtuous practitioner acts in accordance with the virtue required for the best outcome, being guided by a welldeveloped sense of prudence and practical wisdom.

Yet, well-considered lists of virtues are helpful in teaching ethics to students. A medical or nursing student cannot always perceive the virtues that should be applied to reach certain ends or goals. It is essential that medical and nursing students learn which virtues and attributes are fundamental to the profession and be given opportunities to practice and internalize these (Kopelman 1999). Some commentators assist here in defining the virtues that are essential to nurses and physicians. Kopelman (1999) presents respect, fidelity, confidentiality, relieving suffering and promoting well-being as

important virtues. One can argue that confidentiality is actually a professional responsibility and not a virtue. The virtue of respect ensures that the patient's right to privacy is protected. Tsai and Harasym (2010) identify a broad range of virtues to include curiosity, sensitivity, respect, trustworthiness, honesty, caring, compassion and humanity (871). Curiosity in the context of health care could well be understood as a willingness and openness to learn about the patient.

Drane (1988) provides five fundamental virtues for physicians- benevolence, respect, friendliness, truthfulness and justice. Many of these have been mentioned as attributes of professionals that are valued by patients and lead to improved patient satisfaction in quality of care studies. Although the virtues that Drane provides are selfexplanatory, benevolence merits further discussion here, as it shares similarities with caring, a virtue central to nursing practice. Benevolence is the virtue that disposes the practitioner to carry out acts of good or beneficence "according to highest ethical standards" (33). Furthermore, Drane notes that benevolent practice "disposes the self of the doctor to the needs of the patient who is a person and not just a body. It opens the doctor to the lived experience, personal intimacies, personal concerns, personal needs, personal fears, and to a recognition that they are all connected with the patient's illness" (37). Benevolence, therefore, supports the relational aspect of care, allowing the practitioner to respond to the patient as a person. Drane's description of benevolence evokes the virtue of caring that is fundamental to the nurse-patient relationship. In her review of caring, Brody (1988) shares how the concept of care has multiple meanings in nursing: as a virtue (an ideal), as an act (a virtuous act) and as a responsibility. She informs that the nursing literature primarily recognizes caring as a virtue, that is, the work

of nursing is understood as both technically and compassionately delivered, where the feelings and dispositions of the nurse are central to care (92). In addition, Brody explains that the act of caring can be understood from the relational aspects of the act alone, apart from the moral disposition of the nurse, where the merit of caring is in the virtuous, caring act. As she notes, the writings of Noddings (see *Caring: A Feminine Approach to Ethics and Moral Education* 1984) and Gilligan (see *In a Different Voice* 1982), who have written extensively on the ethics of care, also underscore this perspective. The third perspective that Brody offers places care in a central role as a responsibility that defines nursing as a profession. The characterization of care from these three perspectives makes clear its place as a principal virtue for nurses.

As demonstrated in Chapter Two, ethical codes of conduct in medicine and nursing and the standards of accrediting institutions in medicine and baccalaureate nursing programs reflect the professions' positions on the dispositions and behaviors required of nurses and physicians to produce what is right or good in the patientpractitioner encounter. The American Nursing Association *Code of Ethics for Nurses with Interpretive Statements* (2001) makes clear that the virtues of honesty, integrity, caring, compassion, patience, fairness, justice, respect, self-respect, courage and professional competence are essential (Lachman 2008). The American Medical Association *Code of Medical Ethics* (2001) emphasizes the virtues of respect, compassion, honesty, professionalism, advocacy and courage. The accreditation standards of the American Association of Colleges of Nursing (2008) for baccalaureate nursing education are consistent with the American Nurses Association *Code of Ethics for Nurses with Interpretive Statements* (2001) and underline the virtues of respect, honesty, altruism, integrity, empathy, compassion, caring stance, accountability, selfreflection, and moral agency. In medicine, the Liaison Committee for Medical Education (2011) highlights honesty, integrity, respect, altruism and professional behavior, attitudes and skills. The Accreditation Council for Graduate Medical Education (2011) emphasizes professional attributes and attitudes that include compassion, respect, sensitivity, responsiveness, self-appraisal, and advocacy. These various virtues identified in the codes and accreditation standards reflect a long tradition in the professions and clearly indicate the dispositions and behaviors that are considered essential for excellence in medicine and nursing.

Finally, in the discussion of quality of care developed in Chapter Three, it became evident that the virtues of the healthcare practitioner can enhance the patient-provider relationship. Quality of care has been shown to encompass care that is provided with technical excellence and whose benefits widely surpass any risks. Moreover, quality care consists of humane and culturally appropriate treatment that ensures respect and preserves the patient's autonomy (Brook, McGlynn, and Shekelle 2000). Quality patient outcomes may include partnership building, increasing a patient's satisfaction with care, enhancing empathy, reducing anxiety, increasing trust, enhancing self-efficacy and improving adherence to therapy (Hall, Roter, and Katz 1988; Squier 1990; Zachariae et al. 2003). Specific professional behaviors have been associated with quality patient outcomes, such as empathy, respect, reassurance, support, courtesy, openness, and effective listening (Beck, Daughtridge, and Sloane 2002). Humaneness and culturally appropriate care as noted in the definition of quality of care are associated with the professional skills, and

empathy/being cared for as a person (Hall and Dornan 1988). Other professional behaviors central to achieving quality care are integrity, trust and prudence. In achieving quality outcomes the practitioner needs to strive for excellence in moral disposition and also needs to know how best to exercise these virtues.

# C. How Are Virtues Taught and Learned?

The question of *how* virtues are taught and learned presupposes that virtues *can* in fact be taught. This age-old question has intrigued scholars and suggests that virtue education is highly problematic. Most contemporary philosophers and educators accept that virtues are formed and cultivated through experiences (Brody 1988; see also Gillam 2009; Pellegrino and Thomasma 1993; Prior 2007; Siegler 2001); therefore, with and through the right experiences, they may be taught and learned (Begley 2006; Brody 1988). However, such an understanding has not always held sway. The writings of the ancient Greek scholars, Plato and Aristotle, introduced this provocative question and provided a paradigmatic formulation of virtue theory in response to it (Pellegrino 1995). In *Meno*, a dialogue on moral knowledge, Plato attempts to answer the question of whether or not virtue can be taught. Speaking through the character of Socrates who answers Meno's questions on virtue, Plato concludes that virtue is a form of knowledge and for that reason, it can be taught; but because no teacher of virtue can be identified, it is also likely that virtue cannot be taught:

Socrates: And there are no teachers of virtue to be found anywhere? Meno: There are not. Socrates: And if there are no teachers, neither are there scholars? Meno: That, I think, is true. Socrates: Then virtue cannot be taught?

Meno: Not if we are right in our view. But I cannot believe, Socrates, that there are no good men; and if there are, how did they come into existence? (Plato 1949, 56-57)

The dialogue in *Meno* concludes with the possibility "that virtue is neither natural nor acquired, but an instinct given by God to the virtuous" (61). Some scholars have proposed that Plato may have been writing ironically about the lack of teachers of virtue, airing his disapproval of some of the renowned teachers of the period, such as the Sophists (Gooch 1987). Be that as it may, Aristotle, Plato's student, concluded otherwise. He did affirm that virtue is a special type of knowledge and that this knowledge can be acquired through teaching and experience:

Virtue, then being of two kinds, intellectual and moral, intellectual in the main owes both its birth and its growth to teaching (for which reason it requires experience and time), while moral virtue comes about as a result of habit, whence also its name *ethike* is one that is formed by a slight variation from the word *ethos* (habit). From this it is also plain that none of the moral virtues arises in us by nature; for nothing that exists by nature can form a habit contrary to its nature. (*Nicomachean Ethics*, Book II, sec 1, 1103a)

As noted in the earlier section on virtue, Aristotle identifies the forms of virtue to include intellectual virtue (excellence of theoretical and practical reasoning) and moral virtue (excellence of character). The intellectual virtues are of various forms of knowledge, some of which are acquired through theory - the form of wisdom as *sophia* or scientific knowledge, for instance - while other forms of knowledge - *phronesis* or practical wisdom and *techne* or technical skill - are acquired through "art" or practice (McKie et al. 2012). MacIntyre explains that Aristotle distinguishes between these virtues by how they are taken up:

"intellectual virtues are acquired through teaching, the virtues of character from habitual exercise. We become just or courageous by performing just or courageous acts; we become theoretically or practically wise as a result of systematic instruction. Nonetheless these two kinds of moral education are intimately related. As we transform our initial naturally given dispositions into virtues of character, we do so by gradually coming to exercise those dispositions *kata ton ortbon logon*" (i.e., according to right reason) (2007, 154).

For Aristotle, the moral virtues require practice that allows the individual to form virtuous habits. The individual also needs practical wisdom, gleaned from experience, to know how to apply the right virtue in a particular circumstance. For Aristotle, then, "exercising" one's dispositions requires experience, a point that is central to the thesis of this dissertation. More so, the intellectual virtue of practical wisdom is also developed through life lessons or lived experience. Practical learning develops habits that are formed from repeated experiences. Theoretical knowledge, on the other hand, is developed through theory.

Aristotle's insight on virtue has significantly influenced the present understanding of virtue education. Informally, virtues have been taught experientially, through oral traditions, such as story-telling, observations of the behaviors of others, or through personal experiences that reinforce the effects of "good" or "bad" behavior. Writing on character education, Noddings (2002) observes that, historically, people have been engaged in teaching virtues either individually or as communities through narratives, traditions, exemplars and role modeling. As part of a society, "we are products of as well as contributors to traditions of behavior" (Noddings 2002, 62). Noddings illustrates how virtue education, in the form of character education, was part of the early public school and religious education curricula. She points to the McGuffey reader series that were used in the early twentieth century as well as the Character Education Development League curriculum, both which were developed for the moral education of youth (see the

Project Gutenberg eBook, www.gutenberg.org, for examples of the McGuffey readers). The use of exemplars whether fictional (Atticus Finch, Luke Skywalker, Belle) or real (Jesus, Mother Theresa, Gandhi) points to paradigm persons who by their actions convey the traits or attributes that individuals are encouraged to develop (Pellegrino 1995). The paradigm person, consequently, is the person who possesses excellence and "sets the standards of noble conduct for a culture" (Pellegrino 1995, 255). Like Pellegrino, Begley observes that the norm for virtue ethics is based on the virtuous moral agent; therefore, the concept of the paradigm person is integral to understanding and incorporating virtue into the healthcare professions. Exemplars in medicine and nursing could include historical (Elizabeth Blackwell, Florence Nightingale) or contemporary (Paul Farmer, Henry Marsh) individuals that students learn of through literature, narrative and media but would more likely be found among those with whom they interact with in their student lives. Positive role models in everyday clinical experiences can serve as exemplars that have a beneficial influence on the hidden curriculum.

Noddings notes that, in general, the early efforts in character education in many disciplines were supplanted in the 1980s by the cognitive developmentalism of Kohlberg that created a shift in moral education from character education to an emphasis on moral reasoning (2002). Facility for moral reasoning continues to be highly valued at present, and for instance, testing instruments that measure moral reasoning are sometimes used as part of the admissions process in medical programs (Glick 1994). Nonetheless, while a deemphasis in virtue education was observed in the later part of the twentieth century, the educational value of virtues has maintained its importance in medical and nursing programs (Pellegrino 1995). Importantly, most philosophers and educators accept that

virtues are formed and cultivated through experiences, and therefore, with the right experiences, they may be taught and learned. Begley submits that "if we accept that *virtue is a type of knowledge then virtue can be taught*" (Begley 2006, 259).

Virtues as a form of knowledge, therefore, can and should be taught, developed and nurtured. This is no less true for the medical or nursing student than it is for a growing child. A person learns the value of virtues from life lessons – for example, to behave in a particular way in order to develop trusting relationships or through experiencing what it means to be treated with respect. People are also influenced by negative experiences and learn, for example, the downside of dishonesty as either victims or as the result of their own bad decisions. Begley reminds the educator that while students can be taught about virtue, "not all students will learn" (2006, 264). Methods presented in this section aim to illustrate that with creative teaching and careful feedback and reflection, medical and baccalaureate nursing students are better able to acquire those dispositions and habits that are necessary for excellence. Begley suggests realistically, though, that those students who cannot learn the virtues that are required of them as professionals (physician *qua* physician, nurse *qua* nurse) should be directed to other careers.

The principal question, then, about how to teach virtues is addressed by considering the type of knowledge that is required, how it is acquired and the educational goals or outcomes. Developing a virtuous disposition requires knowledge, habituation, and action (Campbell, Chin, and Voo 2007; see also Begley 2006; Brody 1988). The student needs to learn about virtues and their relation to other ethical theories, principles, and rules to have a base for habituation and action. Students require habituation or

practice activities to promote the development of empathy, sensitivity and critical reasoning. Opportunities for the student to put virtues into action also are essential. Determining which instructional method is most effective to teach a virtue is similar to determining how to teach any concept in ethics or for that matter in education in general - that is, teaching methods should be based on how particular knowledge is acquired as well as the aims of education (Issenberg et al. 2005). The educator must take into consideration whether the goal is a theoretical understanding of moral virtues and prudential reasoning, for which theory-based teaching methods are appropriate or whether the goal is developing and cultivating moral virtues and prudence or practical wisdom, for which teaching through practice via experiential learning is appropriate. It is very appropriate that educators set goals in virtue-based ethics education that seek competency in both theoretical knowledge and practical knowledge; consequently, teaching methods would engage the various domains of learning (cognitive, affective and psychomotor) to achieve competency in theory, attitude and behavioral skills. It is also necessary to consider the overarching goal of ethics education of providing quality care for patients; therefore, teaching needs to be focused on the specific virtues that contribute to excellence of character and excellence of intelligence that have been identified in the previous section. It is important to note that the language used in this dissertation thus far with regard to teaching and learning ethics has consistently applied "growth" terminology, such as developing, nurturing and cultivating, that takes into account this kind of learning as a life-long process (McKie et al. 2012). Whichever pedagogical approaches are adopted, educators in medical programs and baccalaureate nursing

programs should view ethics education as a long-term process and view themselves as nurturers of traits and abilities that will improve with additional exposure and maturity.

Virtue education, therefore, should be comprised of both theory and opportunities for practice and habit formation. There are many methods recommended to reach these goals, but it is essential that the way in which virtues are taught is appropriate to the type of knowledge to be acquired and the educational goals. Shumway and Harden (2003) summarize those methods that are well-suited in producing particular learning outcomes in medical students' education. The authors report that more complex learning outcomes that require a change in attitude or behavior, as virtue-based ethics does, are at the top level of the learning pyramid ("knows," "knows how," "shows," "does") that moves from comprehension to application and action-oriented practice (2003, 580). Outcomes related to attitudes, ethics, decision making, role of doctor, and personal development are classified at the topmost level - "does." Consequently, the methods that educators select to reach higher level outcomes must include opportunities to apply learning in practice situations and eventually in actual clinical situations. While lectures may be effective for conveying fundamental theoretical knowledge about moral and intellectual virtues essential to the nurse and physician, the hierarchy of learning outcomes supports that they are not sufficient for knowledge that is formed through practice, habituation and action. A review of the various methods proposed and used to teach virtue ethics - lecture, case studies, humanities and narrative, portfolio, role play/drama format and simulation - is presented below.

While lecture alone is an insufficient teaching method for virtue knowledge, this is not to say that lectures have a limited role in virtue ethics education. The lecture format

has traditionally been used in ethics education and is a convenient and useful way to convey complicated and thorny information. In a review of lecture as a pedagogical method, Benner (2012) shares the example of how an educator can effectively apply lecture to teach the intellectual virtue of practical reasoning or *phronesis*. Understanding that Aristotle's writings on virtues are complex, she points out that lecture can be used to help students work through the difficult concepts of the text (in this case, *Nicomachean Ethics*) and create a dialogue about the many interpretations that his writings on prudence and practical reasoning have raised. She reminds the educator that lectures need to be engaging and reciprocal where both teacher and students participate in the exchange of ideas to avoid a one-directional flow of information. Benner suggests that lectures that are interpretative and translational and used "in the context of a lively, interactive, engaged learning community ... can guide the emotional learning climate of the classroom" (2012, Conclusion). Nonetheless, she cautions against an over-reliance on lecture and argues for focused mini-lectures of perhaps 10 to 15 minutes or so that are used in conjunction with other instructional methods. Mini-lectures are valuable in aiding comprehension of complex topics (such as Aristotelian virtue ethics), to develop students' moral sensitivity and insight, and to develop clinical reasoning skills. Many of the suggestions that Benner presents to develop concepts in lecture-style classes draw upon narratives, either stories of students' clinical experiences or unfolding case studies. As an example, in a lecture that aims to bring about a deeper understanding of moral virtues, a mini-lecture that reinforces assigned readings can help to focus small group discussions, debates or case analysis, applying virtue ethics to solve dilemmas. Additionally, following an interactive lecture in which questions are clarified and

perspectives are shared, students can be asked to identify the appropriate virtues to use in certain situations or to solve a healthcare problem using prudential reasoning.

As indicated above, case-based discussion or casuistry is another teaching method that provides students the context in which to apply their learning in practice situations. Cases can be actual accounts (classic, current ethics cases, or students' clinical experiences) or conveyed through various media. Begley highlights the importance of case studies, noting, "Students... need to be introduced to the subject of a case before the dilemma arises. They need to see the person affected by the situation..." (2006, 262). She suggests that the humanities in the form of literature, poetry and film have a valuable role to play in providing vicarious experiences from which students can develop moral sensitivity, compassion, empathy and moral insight (262). Teaching methods that allow students to observe or read about virtues in contexts, such as film, literature, case analysis, interactive multimedia, and reflective writing, provide opportunities for students to reflect upon character traits and practice decision making. In Medical Ethics and Living a Life, Robert Coles (Coles 1995) makes a strong case for incorporating the humanities in healthcare education as a way for students to develop moral sensitivity and learn about the ethical problems and decisions they will face. Integrating literature into healthcare education, for example, can cultivate skills of observation, interpretation, expressiveness, self-awareness and curiosity, and in addition, enhance moral education and empathy (Hunter, Charon, and Coulehan 1995). Hunter, Charon and Coulehan (1995) outline three approaches to including literature into medical education that have relevance for the ethics educator considering this pedagogy. The three approaches – ethical, aesthetic and empathic – differ in their educational aims and in the selection of

literature. The ethical approach considers the lived experience (much like Drane's conceptualization) and focuses on moral reflection. The authors suggest that typical readings in this approach would focus on justice issues, views of illness, and the ethical dimension that is inherent to the patient-practitioner relationship. The aesthetic approach engages the student in the patient's narrative and "encourages tolerance for the ambiguity and turmoil of clinical situations" (789). Readings for this approach are used to demonstrate aesthetic distance, identification and perspective. The empathic approach centers on understanding the patient's feelings, emotions and perspective as a patient. Readings that support this approach often include persons who vary in age and culture. While there is overlap in these approaches, the authors explain that the educator should understand the goals of the approach. Teaching methods should include the use of small group discussion, writing exercises, along with interactive methods such as role play (Hunter, Charon, and Coulehan 1995).

Illingworth offers that the richness of the humanities comes from its capacity to draw out the narrative in a person's life, engage the learner in other perspectives, and recognize the emotional and psychological responses of the practitioner, "thereby reminding all that healthcare is a *human* science" (2004, 55). Many ethics educators successfully engage narratives as a method for teaching ethics (Charon 2001; Hunter, Charon, and Coulehan 1995). In her work in medical ethics, Charon calls for "narrative competence," which she describes as the ability of physicians to "listen to the narratives of the patient, grasp and honor their meanings, and be moved to act on the patient's behalf" (2001, 1897). The appropriateness in employing narratives to teach virtues lies in the ability of a story to elicit professional virtues that are exhibited or required, such as

moral sensitivity and awareness, compassion, empathy and courage, through dialogue and reflection. Illingworth cites the value in using humanities-based learning activities in ethics education:

Indeed if one accepts, that the aims of healthcare education are to produce practitioners who are reflective, empathetic and humane, professional, patient-centred, honourable and responsible, drawing on emotion and offering contrasting human perspectives on moral dilemmas in healthcare may be an essential part of ethics teaching and learning. (2004, 55)

Ashcroft finds narrative to be useful as an educational tool given its ability to contextualize an individual's life and problems, showing life as "fuzzy" and "rarely, if ever linear," a contrast to the linear thought processes and decision making frameworks that are often part of ethics education (2000, 289). He offers that narratives which place the patient at the center have the advantage of reducing the emphasis on the problem or dilemma (quandary ethics) and instead contribute to patient-centered care, explaining:

The role of narrative ethics as a supplement to virtue ethics (and the ethics of care popular in nursing circles) is to handle decision-making in the context of a patient's life-story; something which is rarely, if ever, linear. In this understanding, dilemmas are to some extent artefacts of the narration, and it is the shape of the narrative as a whole, not just its turning points, which is important-with the consequence that dilemmas are less important to ethical reflection. (2000, 289)

Ashcroft's comment highlights the importance of placing the patient's story at the center in the student's deliberate processes. Narrative, an instructional method abundant in the humanities as well as in classic ethics case studies, helps the medical and nursing student develop an awareness of the wholeness of the patient. Being patient-centered is foundational to virtue ethics in that the end or telos of virtue ethics is that which is good and right – interpreted as best care outcomes for the patient. In addition to ethics case studies, which are plentiful in medical and nursing ethics literature, there are online resources that provide patient stories. One of these, Healthtalkonline.org, produced by Health Experience Research Group at the University of Oxford, houses many video interviews of patients and caregivers. As ethics is embedded in the patient experience, any of these interviews would provide excellent patient stories to be used as starting points for a class dialogue.

When possible, using cases developed by students from their actual clinical experiences can provide rich discussions for virtue ethics education. Caldicott and Faber-Langendoen (2005) found that while most student-generated cases involved treatment decisions (for example, conflicts between the physician and patient concerning an intervention, improperly obtained informed consent or futility concerns), ten percent of the cases involved issues of deception and deliberate lying. Nine percent of the cases dealt with discriminatory treatment of patients, unfair resource allocation and inadequate care. Remarkably, twelve percent of the cases revealed students' reluctance to speak up or challenge moral misconduct that they witnessed for fear of retaliation. The study's authors noted that the students included fear of reprisal as justification for their response or lack thereof to the ethical missteps they observed, indicating a need for discussion and practice related to moral courage. This study also underlined the pervasiveness of ethical problems in clinical settings that pivoted on professional virtues and illustrated the benefit of using student-developed cases to guide inquiries. Caldicott and Faber-Langendoen found "The pedagogic process of working through students' own cases transforms ethics from a theoretical ideal into an actual code of professional conduct capable of cultivating moral habits of action" (2005, 872). While the study had

limitations – students were from one medical school and written cases did not represent all clinical experiences – the experiences are telling of students' perceptions.

The use of portfolios as a teaching method has been recommended for its value in encouraging analysis and reflection (Campbell, Chin, and Voo 2007). Although portfolios are often considered a medium to collect evidence of student work, they can also be an effective means to develop professional competencies and can include journals and diaries. A Best Evidence Medical Education (BEME) systematic review of portfolio use in medicine and biological sciences sheds light on their appropriateness as a teaching method in general, rather than as a means for teaching ethics, but the results are applicable here as well (Buckley et al. 2009). The review notes that portfolio uses have become more widespread in medicine, nursing and allied health, likely due to the emphasis on reflection as part of health profession education. The review also suggests that portfolio use that requires active engagement and reflective practice by students may be an effective way to help students translate theory into practice and increase selfawareness and reflection (Buckley et al 2009, 351). While the systematic review did not define reflection, reflection is generally understood to mean critical assessment on experiences that creates new knowledge (Mann, Gordon, and MacLeod 2009). This may lead to an understanding about one's self, skills, or the meaning in a particular situation. In the systematic review, portfolio use was associated with improvements in critical thinking, decision making, communication skills and ability to learn independently. Some students found portfolios helpful in identifying their own learning needs. In regard to attitude and behavioral changes, some articles reported that portfolios were associated with improved self-confidence, professionalism and empathy. Most of the articles in the

BEME review measured self-reported or teacher-reported appraisals of the effectiveness of the portfolio use on learning. Fourteen percent of the articles, however, included the results of direct observation of changes in knowledge, skills, or attitude/ behavior (Buckley et al. 2009, 346). Interestingly, instructors reported that reviewing portfolios enhanced their understanding of the students' needs and prompted changes in teaching as a result. The review found that the major disadvantage of portfolio use is that it is timeconsuming as reported by students. It also reported that while portfolios encourage reflection, there are concerns with the quality of the reflections. Some students reflect at a basic level, offering only descriptions of feelings and experiences versus higher level reflection that engages critical analysis, indicating a need for enhanced preparation and guidance on reflection.

The advantages of portfolio use to develop self-awareness make it highly suitable as an educational tool to teach virtues. As a teaching method, portfolios can be employed to enhance the students' understanding of virtue theory as applied in practice. Portfolios are also appropriate in teaching virtues through the engagement of reflective questioning and journaling techniques that encourage moral sensitivity and insight. Jaeger defines moral sensitivity as the ability "to recognize when an act, situation or certain aspects of a situation have moral implications" and coming to understand what one's moral responsibilities are in the situation (2001, 132). The reflective component of portfolio writing has the potential to develop the student's moral perception. When used over a length of time, the portfolio can provide evidence of cultivating professional virtues. In a similar vein, students can participate in regularly scheduled writing assignments (while not a part of a formalized portfolio) that relate to professional competencies and assess

these for their inherent virtue content. By reflecting on clinical experiences, students learn to identify the professional virtues that are required of them in their role and record and measure their own progress in developing these. Ethics educators can also encourage students to write about paradigm cases culled from their professional experiences, thereby engaging students in reflective activities and helping students to track their growth in professional virtues. The authors of the BEME systematic review stress that it is important that the process of portfolios be efficient and integrated into course work (Buckley et al. 2009). The results support that students must also be mentored appropriately on the purpose of reflective writing in order to best achieve the desired educational goals.

Role play is another method that has been recommended to teach ethics, particularly given that this format encourages active, experiential learning and helps students to translate theory into practice (Fox, Arnold, and Brody 1995; Garrett 2010; Miles et al. 1989; Seiler et al. 2011). Role play allows students to take on a given role, often with assigned personalities and emotions. In most role play scenarios, the players have "authorship" of the role and participate by acting a part (Jones 1995, 10). Role play has a long history of use in medicine and nursing and has been demonstrated to successfully develop communication skills and empathy (Brunero, Lamont, and Coates 2010; Chan et al. 2003; Fisher, Taylor, and High 2012; Roter et al. 1995). It has been used to teach "breaking bad news," the responsible conduct of research, and many other activities that require ethical and affective competencies (Chauhan and Long 2000; Seiler et al. 2011). Role play is suited to teaching in the affective domain and can provide students with essential practice in dealing with multiple perspectives, attitudes,

interpersonal conflicts, and their own personal feelings and emotions as well as those of the role they assume (Schoenly 1994). Likewise, role play is an appropriate method for teaching virtues and has advantages over static or non-experiential methods to develop professional attributes, enhance moral reasoning and moral courage, identify values, beliefs and biases, and apply the correct attitudes in particular situations (Brown and Gillespie 1997; Chauhan and Long 2000; Doron 2007; Seiler et al. 2011; Tsai and Harasym 2010). As previously demonstrated, developing and nurturing moral and intellectual virtues requires experience. A learning activity such as role play provides opportunities for practice and rehearsal and aids in developing self-perception, helping the student gain a sense of how he or she might act when confronting actual clinical situations. Tsai and Harasym (2010) acknowledge the difficulty in changing students' attitudes and yet make a strong case for ethics education that incorporates opportunities for practice, feedback and reflection, asserting, "The development of proper attitudes is not instigated by lecturing (i.e. affirming, declaring, claiming, telling or stating), but by educating (i.e. questioning, challenging, inspiring, showing, motivating, problem solving and reflecting)" (871).

Illingworth (2004) suggests that role play or alternatively, live drama, is useful in enhancing students' awareness of ethical issues and developing their ability to consider the views of others. Live drama employs the use of actors who present a scene or interaction, perhaps of a patient or family member and healthcare practitioner. Students observe the drama unfold, much like watching a scene from a film; however, the live action format is appealing and moving. Following the scene, students may question the actors, who respond in character. The use of live drama is similar to Medical Readers'

Theater (Savitt 2002) in which medical and/or nursing students read from scripted plays about healthcare situations to an audience, followed by a discussion on the issues raised by the reading. Savitt (2010), a pioneer of Medical Readers' Theater, shares that this method brings students closer to ethical and sociocultural issues where they can experience the emotive side of healthcare issues and patients' perspectives.

Any of these methods have the benefits of increasing ethical sensitivity, insight, and empathy and enhancing perspectives as well as giving students the chance to apply appropriate moral and intellectual virtues. Stoker (Garrett 2010) uses role play or drama format regularly in a healthcare ethics course with predominantly nursing and allied health students. She notes that even students who have a good grasp of the theoretical concepts of ethics have difficulty applying them in action, pointing out, "Competence in theory doesn't translate to effective moral agency" (Garrett 2010, 10). She finds that this active teaching method helps students move from theory to practice (see Smith 2010 for video interview with Stoker). Participatory and dynamic methods such as role play provide students the opportunity to practice professional virtues and practical reasoning skills, experiment with their approaches to ethical situations and prepare as moral agents. Similarly, Brown and Gillespie (1997) have utilized improvisational theatre in a bioethics class with graduate students in occupational therapy. They highlight the advantage of this format to develop ethics in "complex interactions" that are much like those that students will encounter in clinical learning environments and in their work as practitioners. They view the advantages of practicing ethical competencies over and above a fundamental understanding of ethical principles and problem-solving as a means to support students'

moral courage and moral agency in sessions that act as "rehearsals for 'real' life" (1997, 108).

These various teaching methods, while not forming an exhaustive list, offer the ethics educator many options in reaching the educational goals related to virtue knowledge. Each has advantages and disadvantages that the educator must consider when selecting pedagogical approaches to the teaching of virtue ethics. The use of experiential learning, though, whether through integration of case studies, humanities, or active learning methods such as role play or drama formats provides necessary opportunities for students to explore and express their emotions and feelings that are part of moral reasoning and the exercise of virtue prior to working with patients. "Ethics is about relationships," as Cassell (2007, 21) states, and these active learning methods put ethical problems into context, highlighting the behaviors and skills required of students to anticipate and confront awkward or challenging clinical situations.

The final method that is offered is simulation, which will be shown to have many advantages in providing realistic opportunities for the essential practice and application of virtue knowledge. Chapters Five and Six exclusively address simulation; hence, the description here will be very brief. Simulation is an interactive learning method, which at times has been compared to role play and theater-inspired teaching methods. While some definitions of simulation include role play as the lower end of a continuum of simulation activities, there are meaningful distinctions between the teaching methods and a few bear repeating. First, simulation differs from role play in that it provides a high level of realism in the learning environment. Second, while the participants in a role play attempt to act or "simulate" their roles convincingly, in simulation, the participants do not

simulate their personalities; rather, they take on functional roles and work toward effectively carrying out the competencies of the professional roles. Students in simulation activities attempt to convey the thoughts, attitudes, and behaviors that will best achieve the professional role competencies.

Third, simulation allows learners to apply knowledge in a controlled practice setting and receive feedback from the instructor. Reflection upon the activity, as recommended by many educators (Campbell, Chin, and Voo 2007; Vanlaere and Gastmans 2007) is also an essential component of simulation. This method, therefore, is well-suited to teaching virtues in that it provides the opportunity to work through clinical situations, applying the appropriate virtues and experimenting with new approaches. Teaching methods need to be focused on specific virtues, active to allow practice that is directed toward eliciting the virtuous behavior, freely encourage experimenting with new approaches, reiterative to allow for habituation, and reflective to allow students to raise concerns, gain clarification and gauge the effectiveness of their approaches in patientpractitioner interactions. A better understanding of the ethics dimension inherent to the patient-practitioner or patient-family-practitioner relationship may be developed in students through practice opportunities in "human engagement" (Caldicott and Danis 2009, 285) and carefully assessed to allow for feedback and growth. Simulation-based learning activities can fill the need for the practice of virtue in an interactive manner and accomplish these objectives in a creative, engaging way.

## D. The Relevance of Virtues to the Nurse and Physician

Thus far, this chapter has established the relevance of virtues to health care in general and has demonstrated those virtues that are most important to the nurse and physician. Additionally, it has offered various effective methods for teaching virtues in the healthcare professions. As a final point of justification for virtue education, it is important to clarify the relevance of virtues to the nurse and physician. The value of virtue, as Pellegrino and Thomasma (1993) stipulated, is in its contribution to the development of a healthcare professional who successfully meets the goals of the profession. Furthermore, it is important to teach the virtues that will develop the moral agency of the healthcare professional (Caldicott and Danis 2009). The virtues are relevant to the professions as a means of achieving what is right and good - a state of health or well-being for the patient. Moreover, prudence or practical reasoning is a central virtue and inseparable from the moral virtues. Employing the moral and intellectual virtues to develop and nurture the patient-practitioner relationship and to better understand those relationships that are important to the patient helps to counteract the tendency seen in ethics education and in professional practice to reduce ethical decisions to a dichotomous choice between competing values, virtues, or principles (Caldicott and Danis 2009), and in addition, helps the nurse or physician work with the patient in reaching a decision that best achieves well-being, flourishing or good.

Virtue in action promotes good patient outcomes. One must have certain virtues to be sensitive to the patient's feelings and needs, to make prudent decisions (for example, balancing benefits and harms), and to put the needs of the other before oneself. Nurses and physicians also need to develop and display virtues that are effective in relationships. As noted earlier, relationships are integral to ethics, and healthcare

professionals are obliged to develop particular attributes and skills so that they may develop ethical sensitivity, a reflective disposition and successfully engage with patients. Virtue in action, such as respecting the dignity of patients and respecting and appreciating their relationships with others, honoring confidentiality, treating patients with compassion, adopting a caring attitude, being open and sensitive to the patient's feelings, respecting the autonomy and vulnerability of patients, prudently considering the appropriate care options, skillfully negotiating patient-centered care and more, is expected of the physician and nurse. Virtue education can cultivate the dispositions required of nurses and physicians to act as moral agents.

## E. The Relationship of a Better Healthcare Professional to Quality of Care

Quality of care is a noble aim of medicine and nursing and has been presented as the fundamental goal of ethics education. The relationship between the attributes and behaviors of the healthcare practitioners and the improvement in quality of care has been well-established in patient satisfaction studies included in Chapter Three. In a patient's terms, a good outcome is humanistic care, being treated with respect, being comfortable, having needs fulfilled, being listened to, and so on. The studies that measure quality of care are imperfect at an institutional level, but individual patients recognize quality care when it is offered.

Undoubtedly, the current healthcare environment presents many ethical challenges that complicate efforts to provide quality care, from the quickly changing technologies and resources to the maze of conflicting values and obligations that confront the practitioner. Furthermore, there is the understanding that the moral dimension of the

patient-practitioner relationship must be considered at all times in an effort to produce what is right and good. These professional goals require virtuous nurses and physicians who are ethically-sensitive, reflective and prudent in order to capably create better outcomes for patients. The virtues that best meet the patient's needs and contribute to quality care have been discussed at length in the previous sections. The effects of the healthcare professional on the patient perspective in terms of more effective interpersonal and clinical care and on the professional perspective in the form of appropriate consideration and application of standards and guidelines illustrate the benefits of exercising virtue to attain positive outcomes for patients.

Some commentators have observed an additional obligation of ethics education that should be noted – the duty to sustain the profession's values. Begley (2006) asserts that nursing programs have a responsibility to sustain the virtues and values of the profession for the protection of the public. Pellegrino and Thomasma concur, maintaining that professional schools enter into a "covenant" with society and have a responsibility to produce virtuous practitioners (1993, 178). Consequently, the professional schools need to assume responsibility for educating for a virtuous professional who is an effective moral agent. They observe that "the essential moral nature of healing compels attempt to instill virtue as well as technical knowledge" (1993, 178). Their arguments noted earlier on the need to mitigate moral malaise illustrate that it takes vigilance on the part of educators to point out morally ambiguous practices and educate for virtue. Begley notes the responsibility of ethics education to sustain the values of the profession. These poignant observations provide support not only for the contribution of virtue ethics education to quality of care but also for its contribution to protecting the public and

sustaining the professions. It is only through vigilance and courage that ethics educators can impress upon medical and nursing students the value of virtues, their obligations to society as a whole and to their professions.

#### F. Summary

The principal objectives of this review of virtue ethics have been to clarify the value of virtues and virtue education, to justify a renewed emphasis on virtue ethics in medical and baccalaureate nursing education, and to identify a more effective method of teaching virtue ethics that is better suited to the understanding of virtue ethics as practicebased knowledge. The virtues that are vital for nurses and physicians to acquire and exercise are culled from the professional codes of ethics, guidelines and standards of professional organizations and accreditation agencies and informed by expert opinion and an understanding of virtue theory. It has been demonstrated that teaching virtues requires attention to both the theoretical knowledge of virtues and practical knowledge. Keeping in mind that opportunities for practice are "rehearsals" for actual patient contact and that a goal to prevent harm to patients is ever-present, a teaching method that minimizes harm, whether moral, psychological or physical harm, thoughtfully prepares medical and nursing students to better meet the fundamental goal of ethics education – enhancing the quality of patient care. Simulation emerges, therefore, as a teaching method that can fill the need for the practice of virtues. It can be utilized to fulfill the requirements of knowledge, habituation and action essential to virtue education. This teaching method will be extensively analyzed in both Chapters Five and Six. Chapter Five reviews simulation as it is used in many areas of healthcare education, while Chapter Six

demonstrates its current position in teaching virtues and its utility in meeting the goals of ethics education for medical and nursing students.

# **Chapter Five – Simulation in Healthcare Education**

#### A. Simulation as a Teaching Method Defined

The relevance of simulation as a teaching method in ethics has been introduced in the previous chapters, for the most part found in Chapter Two on the discussion of nontraditional methods in ethics education, but is elaborated upon in this chapter. Simulation, in its broadest sense of teaching using practice experiences that imitate or replicate clinical learning with real patients, has been a part of medical and nursing education since the formalization of their educational programs. Role play, anatomical models and task trainers, for example, have long been used in healthcare education (Nehring and Lashley 2009). Early in medical and nursing education, students learned basic interviewing skills through peer-to-peer role play and developed rudimentary psychomotor skills, such as injection techniques and dressing changes, in practice sessions that simulated the care of actual patients. Since that time, however, there has been a slow and steady progression in the demand for and refinement of technology that would create a more realistic learning experience for clinical education. In the last decade or so, simulation has rapidly advanced to the present state of fully integrated and reactive patient simulators that respond in a convincing, true to reality, manner (Gaba 2004; Nehring 2010), the types of which are described in section D below.

Prior to beginning an analysis of simulation, however, it is necessary to define simulation and identify its primary characteristics. Simulation is understood as a form of experiential learning that holds particular applicability in the education of medical and nursing students whose professions center on complex human interactions. Gredler (2004), an educational psychologist, has defined simulation and identified the characteristics that set it apart from other experiential teaching methods, such as games and role play. Gredler notes that simulations are "open-ended evolving situations with many interacting variables," further defining simulation as an "evolving case study of a particular social or physical reality in which the participants take on bona fide roles with well-defined responsibilities and constraints" (2004, 571). She identifies four characteristics that are important to simulation experiences.

First, simulations mimic reality, a feature known as fidelity or validity, with the intent to "transport the student to another setting" (Gredler 2004, 572). A simulation with high-fidelity more accurately or more closely imitates reality than does a low-fidelity simulation and is therefore more authentic, allowing the student to suspend disbelief. Fidelity often refers to considerations of the physical environment and equipment, but simulation also requires validity of causal factors and relationships (Gredler 2004). The literature on fidelity of simulation and technology reveals various properties of fidelity that may be considered in a simulation, including physical fidelity, equipment fidelity, psychological or perceptual fidelity, environmental fidelity, functional or task fidelity and more (Allan, Buffardi, and Hays 1991). The common threads in fidelity in simulation, particularly those involving devices or simulators, typically concern physical fidelity (appearance) and function (Allan, Buffardi, and Hays 1991; Rehmann 1985). Gredler also asserts that the data set that is provided to students to work with through the experience as well as the relationships between the participants, and the responses to the participants' actions, must be valid and realistic. Second, the participants' roles in the simulation are to be clearly outlined, genuine and internalized. Gredler's understanding of the role

students adopt in a simulation is very similar to Jones' (1988) conception of the participant's "functional role" that differentiates simulation from role play, which involves improvisation or "play-acting." In a simulation, the student internalizes the role, is actively engaged in the activity and assumes control of the learning exercise (572), making this a very rich learning experience. Third, simulation relies on a set of data that allows the student to control the action. The simulated environment and preparatory materials provide the information or materials needed to navigate the learning exercise. Finally, simulation provides feedback that emerges from the student's action. Theoretically, a simulated learning exercise may progress in many different directions depending on the student's decisions and actions. The participants may receive feedback immediately (for example, successfully or unsuccessfully resuscitating a patient in a mock code) or during a reflection session that follows the simulated activity. Students often experience the consequences of their dispositions, attitudes and affective responses, technical and interpersonal skills, and reasoning abilities immediately, gaining important feedback from simulation activities. Gredler adds that a goal of simulation is to engage the students in solving "ill-defined" problems rather than to manage unambiguous, straightforward problems. Simulation allows the educator to create learning activities that advance the application of knowledge, skills and attributes to solve the more difficult problems that represent those found in the "real world" (2004, 573). Thus, the selection of the problem and case study materials that direct the simulation is critical and central to the learning experience.

Writing from the perspective of medicine, Gaba offers a definition of simulation that is frequently cited, describing it as "a technique, not a technology, to replace or

amplify real experiences with guided experiences, often immersive in nature, that evoke or replicate substantial aspects of the real world in a fully interactive fashion" (2004, i2). Gaba's definition underscores that the emphasis in simulation is not on the technology, equipment or individuals (patient actors, for example) that are utilized in the educational interaction, but rather on how one structures and works with the components of the simulated experience. As with any educational method, the stress is always on the educational goals and outcomes. Although educational technology provides support for the teacher, success in reaching goals and producing desired outcomes is determined by the educator's skillfulness in utilizing the educational materials and technology. Gaba's definition shares similarities with Gredler's characterization of simulation, highlighting the immersion of the student in the learning environment, the need for fidelity of the experience, and the value of active learning. An effective simulated learning environment with high fidelity and validity can create a realistic substitute for the actual clinical experience and allow students to more easily suspend disbelief and go about the issue of problem-solving as if they were acting as clinicians in the actual role.

In nursing education, the National Council of State Board of Nursing (NCSBN) provides a definition of simulation to guide educators, identifying this as "activities that mimic the reality of a clinical environment and are designed to demonstrate procedures, decision making and critical thinking through techniques such as role-playing and the use of devices such as interactive videos and mannequins" (2005, 2). This definition emphasizes the component of fidelity, but more so, it stresses the value of simulation to enhance reasoning skills and problem solving as well as a method to learn psychomotor skills. It should be noted that NCSBN includes role play as a type of simulation.

Following Gredler's defining views of simulation, role play would only be considered a simulation when certain criteria are met: the student takes on a functional role and internalizes the feelings of the role (versus "play acting" a role); the student is highly involved and actively contributes to the learning experience; the experience contains a data base; and the student receives feedback from the activity itself or from the facilitator. The difficulty in labeling experiential learning as simulation is apparent throughout the healthcare literature as well as in other disciplines.

The NCSBN definition also references the use of devices in simulation, which are generally referred to as simulators. Simulators are devices (actual or computerized) that assist in providing realism; however, the mere presence of a simulator or simulation technology does not constitute a genuine simulation. Simulation often is conducted without any technology or device. It is common within the healthcare professions, however, to define activities that involve the use of a simulator as a simulation. Depending on the definition of simulation that one accepts, the use of a simulator (for example, an arm model for practicing intravenous catheterization) could be considered simulation or it may be understood as simply a practice session – another and different type of experiential learning. By adding complexity to the learning problem, (for instance, by placing the arm model alongside an individual who is acting as the patient), the experience progresses from being a technical skill practice to a simulation experience as understood by Gredler. The student must now perform the task while solving a complex problem that reflects the issues encountered in the real world (Kneebone et al. 2002). In the intravenous catheterization example involving both the arm model simulator and an individual, the student is given the opportunity to develop appropriate

interpersonal skills, convey empathy, allay potential anxiety of the patient, and explain the purpose or goals of the intravenous therapy in conjunction with the practice of a technical skill.

The Simulation Innovation Resource Center (SIRC), a program of the National League for Nursing (NLN), defines simulation as "an attempt to mimic essential aspects of a clinical simulation with the goal of understanding and managing the situation better when it occurs in actual clinical practice. A technique that uses a situation or environment created to allow persons to experience a representation of a real event for the purpose of practice, learning, evaluation, testing, or to gain understanding of systems or human actions" (NLN SIRC; Glossary, Simulation). Similar to Gaba's views, this definition stresses the method or technique applied in replicating an experience for students, over and above an emphasis on the technology. Moreover, the SIRC definition is very similar to one proposed by Morton in 1995, as "an attempt to replicate some or nearly all of the essential aspects of a clinical situation so that the situation may be more readily understood and managed when it occurs for real in clinical practice" (66). All of these perspectives emphasize the value of reproducing clinical learning for the benefit of preparing students prior to actually encountering situations in the "real world." Furthermore, implied in all of these definitions is the desire to improve clinical performance in order to improve patient care.

For Gredler, simulation represents an open-ended, evolving case study, which generally involves addressing difficult or complex situations and often resolving illdefined problems. For Gaba and the NCSBN, simulation may present complex, illdefined problems for students to consider, yet their common definition implies that

simulation may also be task-oriented or focus on only one particular skill. To further illustrate this distinction, several examples can be offered. Medical or nursing students may prepare for administering injections to patients by first practicing on an injection pad. For some healthcare educators, this is considered a simulation exercise, albeit with low-fidelity and with no simulated response to the students' actions. The pad does not feel like a patient's tissues and does not respond like tissue to the presence of the injected fluid. On the other hand, students may practice injection administration on a mannequin that is fitted with an embedded injection pad in its arm. This pad also does not respond like tissue but the placement of the pad within the mannequin's arm allows students to practice how to position the patient's arm to receive the needle, obtaining the correct angle for insertion and overall providing more realism for students. If the learning exercise requires the student to interact with the mannequin during the activity, the experience becomes more complex and allows for enhanced fidelity and improved transference of knowledge to the actual clinical setting. When injections are practiced on a patient actor (a volunteer or a standardized patient) in a realistic setting, the experience becomes more open-ended and challenging in that the student now must attend to the patient's responses, placing learning in a realistic context (Kneebone et al. 2002).

The student also better internalizes the role and responses of the clinician in the above situation. This example of experiential learning, therefore, better meets the definition of a simulation according to Gredler's understanding. In another example, anesthesiology students may prepare for the intubation of an actual patient by practicing the activity on a simulated anesthesia model. This anatomical simulator, unlike the arm model in the above example, is more sophisticated and responds to the student's

manipulations, transmitting the "feel" of intubation and providing an immediate response to the student of successfully intubating the patient (or failing to do so) (Issenberg and Scalese 2008, 38). The realism of the experience adds to the student's ability to be submersed in the activity. This situation may not be open-ended or ill-defined. If factors are added to the situation that increase its realism and complexity, the experiential learning activity better represents Gredler's characteristics of simulation and provides the opportunity for teaching and nurturing professional behaviors and dispositions under stress, as well as critical thinking and technical skills.

It is important to note the ways in which Gredler's definition differs from that of many writers in medicine and nursing. Gredler's conceptualization of simulation is general and written so that it can readily be applied to any discipline, yet the defining characteristics move simulation from the simple practice of a skill to the application of knowledge, attributes and skills to solve complex, open-ended problems. The definitions commonly accepted in medicine and nursing do not always reference an underlying problem or case study that is related to a particular patient situation. Simulations in medicine and nursing are often competency- or skill- building exercises but not directed to solving clinical dilemmas or patient-practitioner problems. This is not to say that definitions of simulation used in medicine and nursing that focus on the practice of a behavior on a simulator are short-sighted or incomplete; these definitions are representative of only part of a broad categorization of simulation. Moreover, the distinction in these definitions may not carry weight for many medical and nursing educators. Appreciating the varying dimensions and characteristics of simulation that demarcate it from other experiential learning techniques aids the teacher in appropriately

meeting the goals of ethics education. When medical and nursing ethics education is the central concern as it is in this work, Gredler's definition of simulation particularly stands out as the most appropriate one and best aligned with the goals of this dissertation of those considered. This definition will be applied in Chapter Six to support the use of simulation in ethics education.

Simulation is also characterized by the specific type of activity in which students are engaged. These activities may range from simple to complex and from low fidelity to high fidelity. Decker et al. provide a typology of simulations that includes the following: partial-task trainers (a model or mannequin used to practice simple procedures or skills), peer-to-peer learning (such as practicing a health assessment on a peer), screen-based computer simulations, virtual reality (computer-based; includes tactile, visual and auditory experience), haptic systems (a simulator combining touch and virtual worlds), standardized patients (paid actors, volunteers or individuals that simulate the patient), and full-scale simulations (computerized mannequin that provides a physiologic response in medium to high fidelity) (2008, 75). Gaba (2004) also categorizes simulation by the technologies employed, including the absence of technology, which uses individuals only, but presents a different organizing structure than Decker et al. He places simulation on a continuum ranging from simple to complex in terms of technology: verbal (role play, "what if" situations), standardized patients, part-task trainers (ranging from simple substitutions for reality, such as using fruit for skin, anatomical models, and virtual reality), computer patient (interactive; software-based or virtual reality-based) and electronic patient (fully interactive mannequin or full virtual reality, often with a recreated clinical setting) (Gaba 2004, i4). More recent developments in simulation have

occurred in the use of high-fidelity patient simulators that respond physiologically to the student's action or inaction. Yet, simulations may involve no technology, instead using actors or other individuals to portray patients in reproducing a clinical experience for the student. This type of simulation may involve specially trained individuals, referred to as standardized patients or patient-actors, to portray patients in particular situations. Other times, simulations utilize peers or untrained volunteers to fill the patient role. Nehring (2010) identifies a similar continuum of activities that includes games among the list of simulated activities. While games can be seen as creative and interactive learning exercises to teach critical thinking, by definition they are competitive with the objective to "win" and thus are not open-ended (Gredler 2004). For some educators, by definition, games would not qualify as an educational simulation for medical or nursing students.

The nomenclature used by Gaba and others in healthcare (Decker et al. 2008; Jeffries 2007; Nehring 2010) reveals a continuum ranging in general from role play to high-fidelity human patient simulation. By including role play as simulation, these categorizations differ from Gredler's. However, a review of the literature shows inconsistency regarding role play as simply a form of experiential learning and as a simulation learning experience. Some articles reviewed in preparing this research identify learning exercises as "role play" when they meet the definition of a simulation – the students work with rich data in a simulated environment and adopt and internalize functional roles rather than simply "acting out" a role. Other articles describe an activity as a simulation, when in fact the participants are only "play-acting" and have little opportunity to truly adopt the functional role. Educators are encouraged to carefully

define and describe the learning activity, particularly in the way in which it promotes the internalization of a role.

B. The History of Simulation as an Educational Method in Health Care

The history of the development of simulation as an educational method is extensive and has been well-documented in the literature (Cooper and Taqueti 2004; Gaba 2000; Gredler 2004; Nehring and Lashley 2009; Page 2000; Rosen 2008). Some of the more relevant and interesting aspects in its progression to its current state in healthcare are shared here, most of which originated outside of healthcare in industries that are heavily laden with risk and complexity, such as the military, aviation and space exploration, to name a few. This history will include the development of mannequinbased and computer-based simulators as well as the incorporation of individuals, in particular, standardized patients, in healthcare education.

The use of simulation has a very long history, having emerged in military war strategic planning in the 1600s (Gredler 2004). The use of simulation in the military continued through the 1800s and was seen most notably during the Cold War of the 1950s when it was employed to test possible responses to bomb threats (Gredler 2004, 571). During that time, citizens also participated in simulation exercises. Some readers may recollect the drills that took place regularly in schools and communities during this period for the purpose of training citizens on how to respond in the event of a bomb threat. The cognitive, behavioral and affective responses to these simulated exercises have been imprinted on those who participated in the activities. These exercises foretold of the now common mock disaster training in healthcare and crisis management exercises that represent large-scale, evolving, case study-type simulations.

Developments in the flight industry at the start of the twentieth century and continuing through the space exploration program in the 1960s set in motion the technological advances required for the sophisticated simulators presently used in medicine and nursing education (Cooper and Taqueti 2004). In "A Brief History of Flight Simulation," Page (2000) summarizes the development of training devices created in aviation that were made to replicate the flight experience while reducing or eliminating risks to the pilot. He notes that the advances in aviation paralleled the advancing technology in computers. The earliest flight simulator, the Sanders Teacher, was a crude device consisting of a light aircraft body tethered to a post. The device was dependent on the wind to provide the experience of flying and thus was not efficient, yet in a rudimentary way, the simulator met the objective to train pilots more safely. Other devices were crafted that mimicked the movements of the cockpit, thus simulating the "feel" of the flight. The eventual advances in mock instrumentation, electronics and analogue computers during World War II led to the development of the modern flight simulator (Page 2000). Page reports that the success of the space program was dependent upon realistic, sophisticated simulation that could demonstrate the conditions of an environment never experienced by humans. The impressive developments in the aviation industry are too numerous to detail here, but these advancements paved the way for the sophisticated simulation technology in healthcare (Cooper and Taqueti 2004). Page asserts that the aviation industry contributed not only to simulation technology but, importantly, in establishing standards for flight simulation that ensured consistency in outcomes, an important contribution to simulation training. Moreover, the developments in simulation in the aviation industry focused not only on improving technical

performance, but emphasized the need for behavioral and organization skills (for example, Crew Resources Management) to improve outcomes (Gaba 2000).

The early roots of simulation in healthcare were seen in the mid-sixteenth century, emerging from the field of obstetrics as a response to high infant mortality and complications to infants and mothers in midwife-assisted deliveries (Buck 1991). Buck reports that midwifery licensing laws were enacted to ensure better preparation of midwives. Physicians developed courses that consisted of lectures, along with illustrations, for current or prospective midwives; however, they were unsuccessful in improving outcomes. In time, the Gregoire obstetric simulator was developed to replicate childbirth. This device, known as a phantom, consisted of a female pelvis with which educators could demonstrate childbirth and possible complications (Buck 1991). The simulator was constructed of a basket-type frame, with an oil-skin covering at the pelvis and likely at the genitalia and cloth over the remainder of it. Remarkably, preserved fetuses were used along with this simulator to demonstrate normal and complicated childbirth. Despite this, the device apparently had a low level of realism and concerns surfaced that knowledge learned on the simulator would not likely be transferred to an actual birth (Buck 1991). Evolution in the design of training models to practice obstetric maneuvers continued throughout the seventeenth and eighteenth centuries in an attempt to increase the fidelity of the learning experience, even creating bladders to hold and release amniotic fluid (Buck 1991). By the nineteenth century, medical students in the United States were taught about childbirth using a simulator, given that the prevailing customs made it difficult for them to attend an actual childbirth (Buck 1991; see Buck

1991 and Gardner and Raemer 2008 for a more detailed history of simulation in obstetrics).

Medical and nursing programs introduced simulation in their formal instruction in the early part of the twentieth century as a way to teach clinical skills and improve patient safety. Simulation in medicine and nursing as it is known today had humble beginnings. Nehring (2010) recounts one of the earliest recorded views of simulation-based teaching in nursing, citing Lees who wrote in 1874 that nursing schools should have "a mechanical dummy, models of legs and arms to learn bandaging, a jointed skeleton, a black drawing board, and drawings, books and models" (Lees, 34, in Nehring, 2010, 10). Unfortunately, a "mechanical dummy" for use in nursing education was not available for almost forty years. A precursor of the life-size simulator models used in healthcare today might have had its origin in the development of the mannequin known as "Mrs. Chase," whose development is chronicled by Herrmann (1981, 2008). The cloth, full-size, jointed mannequin was commissioned by A. Lauder Sutherland, principal of the Hartford Hospital Training School for Nurses, to replace "straw-filled dummies" that were used in the nursing school for practice experiences (2008, 53). The concern of the school director was not only obtaining practice experience for the students, but also "sparing patients" possible discomfort" (Herrmann 1981,1836). The mannequin was manufactured in 1911 by the M. J. Chase Company, a doll manufacturer, who later produced child and infant versions that were used to teach child care to nursing students and inexperienced mothers. Within three years, Mrs. Chase was updated with arm injection sites and internal reservoirs for students to practice urinary, vaginal and rectal procedures and treatments. It should be pointed out that, similar to today, expenses often precluded schools from

investing in the latest equipment, and many schools could not afford to purchase the mannequin. As a result, a pattern was made available through Columbia University Teacher's College for schools to build their own "proxy patient" (1981,1836). The U.S. Army commissioned a male version of the mannequin in the 1940s to use in the training of medical corpsmen. The mannequins were used extensively in nursing programs across the globe up until the 1970s (Herrmann 2008). Interestingly, Hartford Hospital has an original Mrs. Chase mannequin, now over 100 years old in its archives (Weir 2012).

In addition to the early simulation work in obstetric simulators in midwifery, some of the earliest simulation training in medicine reportedly began in the field of anesthesiology in the 1920s with the efforts of Dr. Lundy at Mayo Clinic, an anesthesiologist who used cadavers to teach anatomy and anesthesiology to surgical residents at Mayo Clinic (Burden 2011). Lundy observed that the residents who practiced on cadavers in the anatomy lab had a better understanding of surgery than those who did not have this experience. Consequently, Lundy produced a simulated operating room suite where surgical residents were able to study anatomy and perform procedures while receiving feedback (Burden 2011). This model of teaching provided a rich experiential learning environment for the residents. The early task trainer for cardiopulmonary resuscitation (CPR), Resusci® Anne, was also the idea of anesthesiologists and created by toy manufacturer, Asmund S. Laerdal, in the 1960s (Cooper and Taqueti 2004; Devita 2009) who had begun to create training kits for wound care (Laerdal Medical). This CPR trainer was designed to allow manipulation of the head and neck, and later, with the addition of springs in the thorax, permitted a more realistic simulated experience of maintaining an open airway, respirations and performing chest compressions. In the mid-

1960s, Sim One, a prototype of the computer-based mannequins used in simulation settings today, was developed to replicate an anesthesia simulator for training and evolved from a machine to a full-body mannequin. The prototype was constructed and used by the developers, but it was never mass-produced (Cooper and Taqueti 2004). Sim One was a digital/analog computer hybrid with life-like, sophisticated features (moving chest, blinking eyes, dilating pupils, moveable jaw), but the cost of production and challenges to its acceptance halted production. Cooper and Taqueti report that not only was Sim One very expensive, but poor marketing limited its perceived usefulness and demand (2004, i12)

By 1968, however, the part-task trainer mannequin, "Harvey," was introduced in medical schools to teach cardiovascular disease assessment. This part-task trainer was quickly adopted in all levels of medical programs and was later used in nursing programs and continuing education programs. Harvey was used for both teaching and assessing cardiovascular skills in students and practitioners. The use of this particular task trainer has been well studied, and its effectiveness has been demonstrated (Cooper and Taqueti 2004). Students who have been taught cardiovascular assessment skills using Harvey are better able to identify similar problems in patient encounters than students who do not use the task trainer (Cooper and Taqueti 2004, i13). Harvey is considered to be the first successful task trainer - its acceptance made way for more portable versions of cardiac simulators (Cooper and Taqueti 2004).

Developments in the 1980s that were directed toward teaching skills and improving patient safety related to anesthesia resulted in two types of simulators that have greatly influenced simulation training as it is known today. Gaba, along with other

Stanford Medical School colleagues, designed an anesthesia training simulator known as CASE (Comprehensive Anesthesia Simulation Environment) that utilized a mannequin with computer-controlled physiological responses, computer software to control data, and an operating room to produce a realistic environment for student training (Cooper and Taqueti 2004). Critical events in anesthesia were recreated and participant responses were analyzed, leading to a major productivity of research in performance assessment. Modifications of this model by the CAE-Link Corporation enabled the mannequin to be manually controlled by a facilitator in response to pharmacological and physiological variables. The airway could be adapted to create realistic intubation challenges, its pulses were palpable, the lungs and heart could be auscultated, and more (Cooper and Taqueti 2004). The ability to demonstrate mannequin responses enhanced the fidelity of the simulation. Interestingly, Cooper and Taqueti note that a similar anesthesia simulator was being developed at the University of Florida at the same time as the CASE system but independent of it. The Gainesville Anesthesia Simulator (GAS) was developed to diagnosis anesthesia-related errors and had a complex system that was capable of recognizing injected drugs as well as demonstrating pre-programmed responses to the drugs, a feature not available in the CASE model. Cooper and Taqueti report that an early study of the effectiveness of GAS training demonstrated that the training increased the rate at which residents learned (2004, i14). The simulator was eventually owned by Medical Education Technologies Inc. (METI), a major manufacturer who renamed it the Human Patient Simulator (HPS). Cooper and Taqueti observe that the early developers of the anesthesia simulators worked independently. It was not until the late 1980s that conferences on simulation were held that allowed for the exchange of work and

dissemination of research (Cooper and Taqueti 2004). Today's high-technology, highfidelity computerized patient simulators have the capacity to respond to the student's actions, displaying both physiologic and pharmacologic responses (Nehring 2010). A more detailed description of these simulators is included in section D.

Developments in task trainers continued during this period with models created for skill training, particularly in the areas of assessment, anesthesia, surgery, and pharmacology (Cooper and Taqueti 2004). Cooper and Taqueti list over 20 types of task trainers that have been used in medicine alone, many of these to train students and healthcare practitioners in endoscopic skills, vascular access and repair, and surgical techniques. More complex task trainers were designed to meet the needs of surgeons training in minimally invasive surgical techniques. Task trainers can range from the complex (as in the flexible endoscopic models) to simple (as in simulated wounds) and are the most common form of simulation (Nehring 2010). In many nursing programs, task trainers that simulate anatomical organs and body parts as well as life-sized mannequins are used to teach a large array of fundamental as well as complex psychomotor skills, such as urinary catheterization, intravenous catheterization, wound and ostomy care, and airway management, as well as for physical assessment skills, such as ausculating breath, heart, and bowel sounds (Jeffries 2007; Nehring and Lashley 2009). Rosen observes that a confluence of factors contributed to the progress in advanced task trainers, including three-dimensional images of the human body (from the Visual Human Project), simulators that provide haptic - or tactile - feedback, and the growing interest in minimally invasive surgery (2008, 160).

As noted earlier, simulation can be achieved without technology. One early form of simulation that involved the use of individuals to play the roles of the patient or family member – role play – was an interactive, experiential means for educators to teach communication skills and interviewing techniques (Nehring 2010). Often this involved peer-to-peer teaching exercises, while at other times, faculty members or volunteers participated in the learning exercises. In 1963, however, an original and revolutionary (by the standards of the era) form of interactive teaching emerged at the University of Southern California medical program under the leadership of Dr. Howard S. Barrows, a neurologist on the faculty at the university (Wallace 1997). Dr. Barrows sought a more accurate way to both teach and evaluate medical students who were in a neurology clerkship, having been dissatisfied by the subjective, inconsistent or nonexistent evaluation measures that were currently in place. Wallace (1997) provides a detailed portrayal of the beginnings of the era of standardized patients, Barrows' contribution to standardized patient research, and the revolution in medical education that resulted from his initial vision. Her research provides the history that is included here. Despite Barrow's success, Wallace reveals that his work with standardized patients was highly criticized by other medical school programs. The prevalent view at the time was that trained patient actors were deemed unnecessary, expensive and harmful to the image of medical education.

Barrows' referred to the actors as "programmed patients" and, along with Abrahamson, wrote the first article on the use of such "patients" to assess student performance (Barrows and Abrahamson 1964, 803; Wallace 1997). Not only was Barrows avant-garde in his design of a standardized patient scenario; the educational techniques that he adopted highly influenced medical education and are still in use today. Barrows underscored the need for active and participatory learning, consistency in experiential learning, and immediate feedback (Wallace 1997). Barrows developed a checklist that the standardized patient completed after a student's simulated clinical encounter that provided feedback on the student's performance.

With the success he experienced with students, he designed continuing education workshops in neurology for physicians, incorporating similar techniques using standardized patients. For the workshops, Barrows invited a small number of expert neurologists to serve as tutors for the program and trained these physician-tutors for the possible situations that they might encounter as teachers. He anticipated all possible circumstances. As expected, he prepared the standardized patients to act as patients with neurological problems. Remarkably though, he also trained the actors to simulate the characteristics of workshop participants who often present challenges in continuing education programs – those who disrupt the learning environment with inappropriate questioning, disinterest, or overbearing natures, for example. Not only were the standardized patients prepared to act as patients with neurological conditions, they were trained to act as physician participants. As a result, the tutors could practice and develop effective techniques to better communicate and manage these situations (Wallace 1997, 9). His tremendous attention to detail served him well. Barrows' work stimulated research in education, leading to discoveries in stimulated recall and clinical reasoning, as just one example, furthering work in this area by his colleagues, Lee Shulman and Art Elstein (10). Wallace explains Barrows' teaching philosophy as "the student should be given an opportunity to learn in the same manner as the student is going to practice" (10).

Simulation provides this opportunity for students, as Barrows discovered in his pioneering work with "programmed patients."

Others were influential in simulation, notably Patricia Stillman, director of a pediatric clerkship, whose work with Ray Helfer at Michigan State University involved "programmed mothers" or "simulated mothers," with whom medical students interacted (Wallace 1997). Stillman worked to standardize a rating scale for standardized patients to use when evaluating a student's performance, developing the Arizona Clinical Interview Rating Scale ("Arizona Scale"). Importantly, Stillman conceived the idea of "patient instructors" – standardized patients who not only assessed the student's performance, but who were trained to teach the student how to correctly perform a technique (Wallace 1997). Wallace reports that while the "patient instructors" were not experts in medicine or physical assessment, they were competent in using their own bodies to teach the process of physical assessment. Stillman considered standardized patients as "co-educators" (Wallace 1997, 14).

Further advancement in the use of patient instructors or patient educators by physicians such as Kretzschmar, who incorporated the use of the patient educators for teaching communication skills along with gynecological examinations, have assisted in the evolution of standardized patients as they are understood today (Wallace 1997). The evolution in the role of standardized patients changed to meet the educational needs of the curriculum, with the inclusion of standardized patients who have known chronic conditions in addition to healthy actors. The contributions of Barrows, Stillman, Kretzschmar, Abrahamson and others, in conjunction with the perceived need for curricular reforms established a permanent place for the incorporation of standardized

patients into medical school curricula. By 1984, requirements for medical students to pass directly observed clinical performance exams prior to graduation were being discussed (16). In effect, this set the foundation for the adoption of the now standard clinical competency examination of the National Licensing Examination Board, the Step 2 Clinical Skills exam (see USMLE -http://www.usmle.org/step-2-cs/), which relies on standardized patients and is required for all graduates of United States and foreign medical students (Wallace 1997, 25).

Standardized patients are used frequently today in medical and nursing programs for teaching and evaluation, in continuing education programs, in credentialing for professional groups and in remediation for healthcare students and professionals. While the integration of standardized patient simulation is not as extensive in nursing, their use in nursing programs has increased since the mid 1990s, particularly in graduate nursing programs and to a lesser extent in undergraduate nursing education (Nehring 2010). In a ten-year review of the literature from 1996-2005 on standardized patients in health care education, the majority of articles were from medicine, with only 15% of the articles from nursing (May, Park, and Lee 2009, 489). No detailed history of the inclusion of standardized patients in nursing was found in preparation of this dissertation. Nonetheless, hundreds of articles have been written on the use of standardized patients in nursing in the past decade, and its incorporation into nursing education is likely to increase.

The final forms of simulation in healthcare to emerge are computer-based and initially included computer-assisted instruction (CAI) that appeared in the late 1970s, which was used to teach physiology, pharmacology, health assessment, skills, and

decision making (Maran and Glavin 2003; Nehring 2010). What began as novel computer-aided instruction, often using CD-ROM, has progressed to three-dimensional computer imaging, virtual worlds, and ultimately to computer-based learning that is accompanied by touch, known as haptics. Some of the early CD-ROMS programs used in healthcare ethics were mentioned in Chapter Two. The evolution of this form of simulation in healthcare mirrors the developments in technology and education and has benefited simulation by enhancing the fidelity, level of student interaction and quality of feedback. Computer-aided instruction can be enhanced by combining it with a tasktrainer, allowing surgeons to practice techniques such as endoscopy and laparoscopy (Maran and Glavin 2003). When the system allows the operator to "feel" the activity through haptics (touch technology), the experience becomes very life-like (Maran and Glavin 2003; Rosen 2008). This allows the teaching of "complex procedures that are too dangerous to practise on live patients" (McGaghie et al. 2010, 56-57). One of the latest forms of computer-based instruction consists of two-dimensional images of life-size virtual patients that are projected onto a wall (Nehring 2010). The technology allows evaluators to track the students eye movements during the interactions with the virtual patient to view where the student is looking.

As this synopsis of the history of simulation suggests, medicine has generally taken the lead in establishing some of the forms of simulation, in particular, high-fidelity patient simulators, complex task trainers, and standardized patient simulations, with nursing tending to follow the progress set in medicine. This is likely a result of the difference in structure between the professional programs. While both professions emerge from the apprenticeship model, medical students have less direct supervision

when on the clinical units and spend shorter times with the physician supervisors during their clerkships than baccalaureate nursing students. Physicians find themselves supervising medical students while simultaneously managing their caseload. Undergraduate nursing students, on the other hand, typically have clinical experiences under the direct supervision of a nursing instructor and spend on average a half semester or more with one particular instructor. It might also be the case that the different forms of prelicensure education (baccalaureate, associate degree, hospital-based programs, accelerated BSN programs) with different curricula and program lengths ranging from one to four years make early adoption of new technologies more challenging. It is also likely that the expenses of the new technologies have played a large part in the nursing profession's restrained approach to adopting these. As will be discussed in the challenges of simulation, for some programs the costs have been prohibitive, particularly when first introduced, but the technologies often become more affordable with time and widerspread adoption. Although its application as a teaching method is "nontraditional," the incorporation of simulation in medical and nursing programs has increased significantly over the past decades. Medical and nursing programs are making significant investments in simulation, developing simulation learning centers and purchasing equipment. It is very likely that today's student will expect simulation activities to contribute significantly in their curricula.

# C. Factors Influencing the Development of Simulation as an Educational Method in Healthcare Education

The modern era of simulation arose from the confluence of several factors: the need to find a more effective means to evaluate medical students' clinical competencies

and the dynamic changes in the healthcare system that ultimately had a transformative effect on the education of medical and nursing students. As noted earlier, almost fifty years ago in his seminal work on standardized patients, Barrows (Barrows 1964; Wallace 1997) outlined the difficulties that faculty members had in assessing the clinical preparedness of medical students. Faculty members were challenged not only to consistently find appropriate patients, but also to find sufficient time to observe medical students during their clerkships and to note their clinical performances with adequate detail. When a faculty member had the opportunity to directly observe a student's interaction with a patient, however, the faculty member's presence affected the dynamics of the student-patient relationship (Barrows 1964, 802). In addition, the lack of a reliable clinical competency examination troubled educators. The examinations were written or oral, but they depended upon assigned patients, a process that was unpredictable. As well, there was no system in place to compare performances among students (Barrow 1964, 803). The need for a more reliable evaluation method prompted the development of an innovative form of simulation, the standardized patient, which has had lasting success.

The development of simulation has also been highly influenced by the changes in the health care climate over the last several decades, which have had a profound effect upon the education of medical and nursing students. Changes in healthcare delivery, the rapid expansion of knowledge, research and technology, the emphasis on safety and quality, and a greater emphasis on the ethical considerations in patient care contributed to transformations in the way in which medical and nursing students are educated, particularly in the way in which simulation has been developed, accepted and integrated into medical and nursing programs (Kneebone 2010; Nehring 2010; Satava 2009).

The implementation of managed care in the United States, for example, has reduced the average length of patient stay, subsequently decreasing the opportunities for sustained student-patient interactions (Kneebone 2010; Okuda et al. 2009). Patients who are admitted to acute care facilities tend to be more acutely ill and complex, limiting the opportunity for medical and nursing students to work independently (Kneebone 2010; Okuda et al. 2009). Advances in health care, research and technology have produced a considerable growth in information, creating pressure on medical and nursing programs to expand the curricula to accommodate this growth (Okuda et al. 2009). This expansion in healthcare information occurred in an educational climate during which medical programs were shortened programs and medical residency hours were reduced, placing additional constraints in learning opportunities (Kneebone 2010). In nursing, faculty shortages, fewer clinical sites and clinical expenses have limited the ability of some schools to provide clinically supervised education (Nehring 2010). Clinical learning was also affected by patients who were no longer as willing to have students "practicing" on them. These factors contributed to "a disconnect between the classroom and clinical environment" (Okuda et al. 2009, 330). Educators in both medical and nursing programs faced the problem of developing creative approaches to best meet the educational needs of students in this changing landscape, which lent support to simulation as a technique to enhance clinical learning.

Another major influence that favored the development of simulation has been the renewed emphasis on quality and safety that followed the Institute of Medicine's (IOM) influential report, *To Err is Human: Building a Safer Health System* (Durham and Alden 2008; Kohn, Corrigan, and Donaldson 2000; Nehring 2010; Okuda et al. 2009). The

report acknowledged the importance of learning environments that seek to improve communications, establish a more effective work culture, report errors, and develop feedback mechanisms to ensure that learning results from mistakes (Kohn, Corrigan, and Donaldson 2000, 178). Significantly, the report advocated that academic and health care institutions "use simulations whenever possible," specifying that "health care organizations and teaching institutions should participate in the development and use of simulation for training novice practitioners, problem solving, and crisis management, especially when new and potentially hazardous procedures and equipment are introduced" (Kohn, Corrigan, and Donaldson 2000, 179). This influential report summarized much of what is known about errors in healthcare today, advising that healthcare professionals should "anticipate the unexpected" (174). Moreover, it submits "that ALL technology introduces new errors, even when its sole purpose is to prevent errors" (175). As a result of these findings, simulation has emerged as a valuable method to train healthcare professionals on new technologies, to improve skills related to tasks and crisis management, and consequently, to reduce error and harm (175). Later IOM reports also called for reform in the preparation of healthcare professionals to address the continuing safety issues in health care today (Nehring 2010). An Agency for Healthcare Research and Quality report on patient safety (Hughes 2008) also endorsed simulation in nursing education as a means to reduce medication errors. The report considers simulation essential to enhance critical thinking, decision making skills, communication skills and team work, thereby improving safety in health care and patient outcomes (Durham and Alden 2008). Durham and Alden explain that simulation exercises can teach the psychomotor skills needed for safe medication administration, but more so,

simulation can provide practice opportunities for handling the clinical decision making that surrounds medication administration.

A deepened understanding and appreciation of the ethical issues in health care has had a considerable influence on the development and utilization of simulation in healthcare education. Ziv et al. establish that there is an "ethical imperative" to provide carefully designed simulated learning experiences whenever feasible (2006, 252). The authors address the enduring conflict that positions the need for students to work directly with patients against the goal of providing optimal, safe care for patients. Simulation is presented as useful for "mitigating these ethical tensions and practical dilemmas" (Ziv et al. 2006, 252). The authors also suggest that simulation exercises can be designed to enhance humanistic training. While Ziv et al. review this ethical obligation from the position of medicine, their position is equally relevant to nursing and other health-related professions. While agreeing that it is essential that students interact and perform their clinical activities with actual patients, the authors argue for the integration of simulation into the curriculum. Four themes are examined in their analysis: best standards (to include patient care, education, and skills evaluation), error management and patient safety, patient autonomy, and social justice and resource allocation (252), which are examined below.

First, the perspective of best standards claims that the patient's wellbeing is the primary consideration, despite the need for student training. Harm that occurs from inexperienced care providers or that arises from a learning experience is justified "only after maximizing approaches that do not put patients at risk" (Ziv et al. 2006, 253). Students should have adequate skills-training sessions in simulated experiences, thereby

improving their competencies when working with actual patients. Fundamental to this concept is the message that "patients are to be protected whenever possible and they are not commodities to be used as conveniences of training" (253). In addition, best standards practices in education maintain that teachers have a moral obligation to provide optimal learning experiences for students. Simulation has a valuable role in providing essential learning experiences when there are inconsistent or absent opportunities in actual clinical settings. The authors suggest that with simulation-based learning "the process and structure of medical education then becomes a series of progressive choices by educators rather than a response to ad hoc clinical availabilities" (253). As a result, educators move closer to attaining the best standards for education. Furthermore, Ziv et al. acknowledge the best standards practice of using simulation-based testing to assess clinical competency. The Objective Standardized Clinical Examinations (OSCE), for example, uses standardized patients in simulated clinical experiences to assess a medical student's competency level in clinical skills.

Second, Ziv et al. (2006) illustrate the value of simulation in error management and improving patient safety in high-risk professions, such as medicine. Despite the efforts of educators and students, errors occur in clinical settings. In situations with actual patients, errors are either avoided by the educator's oversight or stopped in progress (Ziv et al. 2006). Simulation exercises, however, permit errors to arise in safe settings, allowing students to see the results of their actions. Errors are reviewed in simulation settings as part of the debriefing period, prompting students to learn from mistakes. In addition to individual simulation exercises, simulations can also be developed that illustrate patient safety and error management at a systems level (254).

Third, the ethical imperative considers the issue of patient autonomy. The authors observe that despite the patient's right to refuse treatment and the right to refuse to be cared for by a student, in customary practice, the process for informing the patient is automatic and not individualized, so much so that patients may not fully understand their rights. Moreover, patients may not be aware a student is treating them. In such circumstances, the pressure to provide learning experiences may compromise the patient's right to a truly informed consent. Ziv et al. offer that simulation can protect the right to autonomy by providing alternative learning opportunities, thereby reducing the need to practice procedures on patients. Simulation exercises can also assist in teaching how best to obtain informed consent to help ensure the quality of the informed consent process and to respect patient autonomy (Ziv et al. 2006).

Lastly, the argument for an ethical imperative speaks to the issues of social justice and resource allocation. Ziv et al. observe that patients in academic hospitals have a disproportionate burden in terms of the risks associated with "novice training" (2006, 254). Furthermore, the patient population in academic hospitals is comprised of a disproportionate number of poor and disadvantaged individuals. While not mentioned by the authors, it is also likely that poor and disadvantages patients may be less aware of their rights to refuse treatment or may feel obliged to agree to procedures as part of charity care. This issue of distributive justice calls for more simulation-based training to prevent the objectification of patient as commodities for training (Ziv et al. 2006). The authors also point to the benefits of simulation in reducing the need to use live animals for training, noting the availability of sophisticated simulator models.

While the above analysis of influencing factors is not exhaustive, it is evident that these important factors have had either a direct or tangential effect on one another, and together they have propelled simulation to the forefront of medical and nursing education. Other "driving forces" have been identified to integrate simulation more fully into health care that reflect the goals of professional schools, professional societies, healthcare institutions, simulation societies, liability insurers, and the public, among others (Gaba 2004). These include improving patient care and patient safety, improving performance and efficiency, maintaining competency, remaining competitive, and reducing errors and costs (Gaba 2004, i8). Although a major goal of simulation is to improve the student's proficiency in order to reduce the likelihood of error and harm and to improve patient care, it would be unrealistic to expect that students are able to perform optimally in clinical settings following a simulated experience. Simulation as a complement to actual clinical performance, however, alters the "learning curve" for students (Issenberg et al. 2005, 22) and helps to ensure that the first time a student performs a skill with a real patient, the risk for error has been reduced (Durham and Alden 2008; NCSBN 2005).

## D. Types of Simulation Activities Employed in Healthcare Education

As noted in section A above, there are many typologies that are used to identify and describe the various forms of simulation. Simulations can be identified by the use of technology, the level of complexity, the level of fidelity, and the use of computerassistance. Additionally, there are forms of simulation that combine or integrate various simulation components, referred to as hybrids, such as using a task trainer or a standardized patient with virtual reality. Most texts on simulation provide descriptions of

the various types, but there is often overlapping among the categories. Role play is included in some categorizations of simulation (see Alinier et al. 2006; Nehring 2010) although depending upon how it is understood and defined, it may not provide students the opportunity to assume a functional role and instead be merely "play acting." Adding further uncertainty, some reviewers place role play and standardized patients in the same category, although the use of standardized patients provides a high-fidelity simulation experience.

Simulation that uses a device (simulator) is often described by the type of simulator (high-fidelity human patient simulation, for example). The names by which certain types of simulation are known have changed over the years, and it is likely that new forms of simulation will emerge in the future. It is important, however, for educators to have an understanding of the levels of available fidelity and the complexity among the various forms to assist them in planning simulation-based learning exercises. Durham and Alden (2008), Maran and Glavin (2003) and Nehring (2010) provide useful descriptions, which provide much of the support for this section. As Maran and Glavin point out, Penn State University has developed an extensive resource at its simulation website on types of simulators, which is also consulted. The ordering of the types of simulation most commonly used will follow a nomenclature that reflects the general complexity of the technology.

i. Verbal simulation – This is simulation that relies on the participants only. It is typically conducted with no technology. Role play and "what if" discussions comprise this category (Gaba 2004).

ii. Simulated patient and standardized patient – These are individuals, whether paid actors or volunteers, that are "carefully trained to take on the characteristics of a real patient in order to provide an opportunity for a student to learn or be evaluated on skills firsthand," creating "veritable" learning experiences (Wallace 1997, 6). Barrows introduced this type of simulation and originally referred to these individuals as "programmed patients" (1964, 803). The term "simulated patient" tends to be used as the generic term. The term "standardized patient" was suggested by Geoffrey Norman, a psychometrician, to "capture one of the technique's strongest features, the fact that the patient challenge to each student remains the same" (Wallace 1997, 6). The concept of standardization in the patient role allows for consistency in the clinical experience, attempting to ensure that each student has a similar experience. Other names that have been seen in the literature include patient instructor and patient educator, denoting the role of the simulated or standardized patient to provide feedback on the assessment and if desired, to guide the student's performance. Additionally, the substitute patient may be called professional patient, surrogate patient, and teaching associate (Wallace 1997, 5). The standardized patient can be a healthy person or selected because of his or her underlying medical conditions. Originally designed to evaluate the medical student's clinical skills and patient management, standardized patient simulations are used in teaching and assessing physical assessment skills, communication skills and professionalism (including breaking bad news), ethics, interviewing skills, diagnosis, patient management, patient counseling and patient education (Decker et al. 2008; May, Park, and Lee 2009; Wallace 1997). They are also used for learning "sensitive" tasks, such as pelvic examination and male genito-rectal examination, and domestic violence

training (Issenberg and Scalese 2008; May, Park, and Lee 2009). While more extensively used in medical programs, the use of standardized patients is growing in nursing programs, more so in graduate programs (Decker et al. 2008).

iii. Task trainer - This simulator is also called a part-task trainer or skill trainer and reproduces part of a task, skill, or an environment rather than a complete, integrated experience (Durham and Alden 2008; Gaba 2004; Maran and Glavin 2003). The fidelity of the task-trainer can vary from low to high and complexity can range from simple to complex. In general, they represent body parts or are full-body models or training versions of equipment, such as endoscopes. Task trainers usually teach one particular psychomotor skill or an aspect of a technique and are "static," in that they do not elicit a response from the learner. They are popular because they are usually portable and affordable. Well-known examples of a task-trainer are Harvey, the cardiovascular assessment simulator, and Resusci® Anne, used for training in cardiopulmonary resuscitation. Other examples include anatomical models of body parts, such as a pelvis to perform urinary catheterization or a pelvic examination and an arm to perform intravenous cannulation. Task trainers can be combined with simulated patients to enhance the realism and provide opportunities to engage interpersonal and communication skills (Kneebone 2002).

iv. Patient Simulator or human patient simulator (HPS) – Also known as "integrated systems," this advanced method integrates the computer with a mannequin, computer screen, or virtual reality system to represent a patient (Maran and Glavin 2003; Penn State, What's Available). The systems range from simple to complex and from low to high fidelity. The patient simulators can be used for individual or team-based teaching.

Mannequin-based patient simulators are used with more frequency in medical and nursing programs. The patient simulators are either model-driven, in which the mannequin responds automatically to the learner's interventions (or lack of action) or instructor-driven, in which the simulator response as directed by the facilitator. The complexity and fidelity of the system is reflected in the mannequin cost. Mid-range or intermediate systems have speech availability, chest movement, heart, lung and bowel sounds, pulses, blood pressure, and have computer screens to display vital signs. Higher level systems add more physiologic function, movement and realism. The highest systems are quite complex and are capable of providing physiologic and pharmacologic responses to interventions and administered medications through a computer identification system. With the addition of environmental fidelity, these systems create a realistic clinical situation (Maran and Glavin 2003).

v. Computer-based patient simulators – The interactive software or web-based programs allow students to work independently, thereby accommodating their own pace of learning and learning needs (Decker et al. 2008). Multiple students at a time can work independently through the simulation or as groups. Computer-based simulations are often used to teach concepts, critical thinking, and decision making skills, and less often to teach psychomotor skills (Penn State, What's Available). They provide immediate feedback to the student and also compile statistics for the instructor (Decker et al. 2008; Maran and Glavin 2003). Examples of computer-based simulation include SimHosp and Second Life (Nehring 2010). Many examples of computer-based simulations are available online (Penn State, What's Available) and include programs on cardiac

auscultation, cardiac arrest, and obstetric emergencies. The cost for this type of simulation is generally reasonable.

vi. Virtual reality-based patient simulators – This form of simulation employs a computer-generated patient and environment that creates a high-fidelity experience, using sensory stimuli and at times three-dimensional imaging. Although the contemporary versions of virtual reality are sophisticated and are designed to immerse the student in the experience, it had its beginnings as early as the 1950s (Rosen 2008). Virtual reality systems may also integrate task trainers to enhance the authenticity of the experience (Maran and Glavin 2003). The use of haptic feedback introduces the sense of touch, allowing the learner to experience the "feel" of the task or skill (Penn State, What's Available). Virtual reality is available in various forms, which affect the way in which the student experiences the simulation. Immersive virtual reality "integrates the user into the world of the computer," while pseudo virtual reality permits the student to observe the environment but limits how much he or she can affect it (Vozenilek et al. 2004, 1151). Desktop or screen-based virtual reality confines the experience to the computer itself and is limiting (Gaba 2004). Augmented reality superimposes virtual images onto real images by using head-mounted displays or screens (Vozenilek et al. 2004). Some systems allow the facilitator to track the learner's eye gaze as he or she interacts with the virtual patient. While many systems exist that teach technical skills, virtual reality is also used to teach history-taking, assessment, clinical reasoning, communication and interviewing skills (to include motivational interviewing), ethics, and professionalism (Cook, Erwin, and Triola 2010; Deladisma et al. 2007; Fleetwood et al. 2000; McEvoy, Butler, and MacCarrick 2012). Well known virtual reality-based simulations include Pulse!! The Virtual Clinical

Learning Lab, which pulls together students from different healthcare professions, Second Life, SimHosp, and SimClinic (Campbell 2007; Nehring 2010). Virtual reality is often used for procedural instruction that trains the learner in intravenous catheter insertion, gastrointestinal endoscopy or minimally invasive surgery (Passiment, Sacks and Huang 2011). Virtual reality has been used more often in medicine than in undergraduate nursing, likely due to the cost (Nehring and Lashley 2009). There are some programs available for undergraduate nurses, however, such as an intravenous catheter insertion program that includes haptic feedback, which was developed as early as 1996 (Merril and Barker 1996; Nehring and Lashley 2009).

vii. Hybrid forms – As simulation has matured and educators have become more comfortable with the technique, different variations have emerged that integrate two or more types of simulation. This increases the authenticity of the experience, but more so, it broadens its usefulness in reaching the educational goals, engaging the learner's cognitive, psychomotor and behavioral skills. For instance, computer-based models are often combined with task trainers to increase the authenticity of the simulated experience (Maran and Glavin 2003). Standardized patients may be used with mannequin–based human patient simulators to support the development of communication skills and professional attributes in addition to technical skills. In these scenarios, the standardized patient may have the role of a family member, while the human patient simulator functions as the patient (Bowyer et al. 2010). Standardized patients are also used in conjunction with task-trainers to allow students to practice communication skills while performing a skill (Kneebone 2010). Doing so adds context and complexity to the situation. Kneebone cautions that selecting the type of simulation in terms of its

complexity is significant, particularly when students are learning specific technical skills. Students typically focus on mastering the technical aspects of a task prior to managing the same task in the context of a patient-centered scenario where interpersonal and communication skills are required. Simulations need to best meet the level of the learner; those that are deemed too simple may be undervalued and those that are considered too complex may cause unnecessary frustration and a lack of confidence (Kneebone 2010). Hybrid forms of simulation, however, offer a rich learning environment that develops context and an ease of transfer of learning to the actual clinical setting.

## E. Theories That Support the Use of Simulation in Healthcare Education

Many theories have been put forward to support the use of simulation in medicine and nursing. In general, adult learning theories are applied in the professions to describe and understand the relationships between individuals, situated experiences, social interactions, and knowledge (Fenwick 2001). Many of the educational approaches emerge from the experiential learning theories of Dewey (1933), Schön (1983) and Kolb (1984). Experiential learning theories are grounded in "constructivism," which is understood as reflecting on experience "to construct new knowledge" (Fenwick 2001, Constructivism, 9). Dewey, Schön and Kolb emphasize the role of reflection in learning. The model of Dreyfus and Dreyfus (1980, 1986, 2009), which has been adapted by Benner (2001), also supports simulation as an educational approach based on its theory of competency and the process of skill acquisition. In addition, the concept of a signature pedagogy proposed by Shulman (2005a, 2005b) as a model of professional practice is relevant to simulation. Other theories are also evident in the simulation literature. For instance, Ericsson (2004) has been cited by many in medicine and nursing for his work in

deliberate practice (Kneebone 2005; McGaghie et al. 2009; NCSBN 2005). The literature also reveals the use of the transformative learning theory of Mezirow (1991), the adult learning theory or "andragogy" of Knowles (1970) and situated cognition theories (Fenwick 2001; Paige and Daley 2009). This dissertation will demonstrate the relevance of the theories and models of Dewey, Schön, Dreyfus and Dreyfus, Benner and Shulman, each of which is particularly applicable to simulation.

As part of its structure, simulation creates the opportunity for students to reflect on their actions, attitudes and feelings that arise as a result of a simulated clinical experience. By its design, simulation includes a period for reflection that follows the implementation phase, during which the student or a group of students review their approach to solving a problem or their practice performance. This particular design characteristic sets about to accomplish that which Dewey (1933) stipulated as essential to education – a reflective practice. For Dewey, reflective thought is "the active, persistent and careful consideration of any belief or form of knowledge in light of the grounds that support it and the conclusions to which it tends" (9). As understood by Dewey, reflective thinking orders thoughts in such a manner that they link together so that previous outcomes influence the next in a "sustained movement to a common end" (5).

Reflective thinking is purposeful and consists of two phases: "a state of doubt, hesitation, perplexity, mental difficulty, in which thinking originates, and …an act of searching, hunting, inquiring, to find material that will resolve the doubt, settle and dispose of the perplexity" (Dewey 1933, 12). This description aptly fits the complexities and problem solving that surface as students work through a simulated clinical experience and also reflects the mental processes that arise as students are engaged in the teaching

and learning experience. Dewey suggests that experiencing a problem moves the learner to search past knowledge and experiences for possible solutions. In the absence of relevant experience, the student will grasp at a quick answer and "jump" to a conclusion, but Dewey notes, "when one is willing to endure suspense and to undergo the trouble of searching" reflective thought arises (16). Thus, as Dewey observes, reflective thought frees the individual from routine action, directs deliberate thought, preparation and planning, and "enriches things with meaning" (19). The goals of reflective thought, therefore, aid the medical and nursing student in critical reasoning, problem solving and forming well-considered outcomes.

Dewey also felt strongly about the type of projects that educators plan and what features make certain activities valuable or "educative" (217). He contended that educational activities needed to meet four conditions. They must elicit and hold the interest of the individual, have intrinsic value "beyond immediate pleasure," arouse a new curiosity and desire for new information, and involve sufficient time for achievement of its aim (218-19). Dewey's description of worthwhile learning activities guides the educator in planning simulated learning experiences that will assist students in developing and applying new knowledge.

Working from the tradition of Dewey and other experiential learning theorists, Schön (1987b) provides a hypothesis that focuses on the learner's experience as well as on the notion of a reflective practicum. Schön submits that the epistemology of traditional education centers on the vestiges of Technical Rationality, which is distinguished by its reliance on science and research (1983). Schön argues that Technical Rationality minimizes the place of "artistry" in practice. He explains artistry as learning something

new "when you cannot in principle know what it is you're supposed to be learning and yet you must learn it" (1987b, 10). Professional practice is challenged by real-world problems, the "indeterminate zones of practice- uncertainty, uniqueness and value conflicts" that "escape technical rationality" (Schön 1987a, 6). A reflective practicum is designed to provide learning experiences that enhance the artistry required to learn in this "messiness" (1987b, 6). For Schön, the art of learning also includes applying knowledge to concrete situations. The basis for this form of reflective practicum is learning by "doing"; therefore, educators are encouraged to create learning environments in which students learn through experience. The educator's role also involves assisting students to merge these skills with applied science. As Schön observes, applied science "has a special zone of relevance which depends on our ability to do these other things, on the one hand to set problems in ways that the categories of applied science can fix and fit and, on the other hand, to fill with art the gap between theory and technique and concrete action" (1987b, 7). Experiential learning can fill this gap by creating opportunities for a reflective practicum that includes "virtual" or simulated environments in which students learn by "doing" in a safe environment that is supported by an educator in the role of a "coach" (Schön 1987b, 7). Students, in effect, "try to educate themselves before they know what it is they're trying to learn" (Schön 1987b, 10). This process of discovery helps students to bridge the gap between theory and practice. The teacher in the role of a coach interprets the student's actions and shares these observations.

Central to the reflective practicum as defined by Schön are the concepts of "reflection-in-action" and "reflection-on-action." Schön explains reflection-in-action as "tacit and spontaneous" understanding in the midst of doing (1987b, 3). Reflection-inaction often is generated as one's performance produces unexpected results (either positive or negative), leading to a natural reflection on the event as it transpires. This involves "thinking on one's feet" to create experiments and testing them out to solve problems on the spot. Episodic reflection-in-action aids in professional development and expertise and is central to professional practice (Schön 1983). Reflection-on-action, on the other hand, emerges after an event and contributes to understanding and knowledgebuilding. It occurs as the person examines his or her actions, feelings, and the possibilities of alternative courses (Fenwick 2001). Simulation serves as a reflective practicum that provides opportunities for students to reflect-in-action as they work through a situation and, in addition, provides time for reflection-on-action following the event.

Simulation is often selected as a teaching method because of its role in developing critical reasoning and technical skills through practice opportunities. The educational model of Dreyfus and Dreyfus provides an understanding of skill acquisition and the relationship between theory and practice that is central to simulation. The Dreyfuses' model is a five-stage model of skill acquisition that was developed as a result of their work with airline pilots, chess players and automobile drivers (Dreyfus and Dreyfus 2009, 8). The authors maintain that although formal rules assist in skill formation, experience is necessary to progress to skill proficiency (1980, 5; 1986, 19-20). The original formulation of this model in a manuscript on skill learning with pilots reveals the learner's passage through the stages of novice, competency, proficient, expert and mastery (Dreyfus and Dreyfus 1980). The original stages have been modified slightly; in a 1981 manuscript by Stuart Dreyfus, the five stages are identified as novice, advanced

beginner, competence, proficient and expert. These revised stages appear to have been consistently used in the later writings of Dreyfus and Dreyfus and are used in Benner's adaptation of this model and used in this description as well.

Dreyfus and Dreyfus observe that progression through these stages is marked by four mental functions (1980, 1981). First is the learner's ability to relate elements of a situation to previous experience (situational learning) as the learner moves from the novice stage of unsituated learning. Next is the capacity for salience recognition, that is, knowing what elements of a situation are important, which improves with experience (1981, 25). Novice learners struggle to identify the important elements or information in a situation and instead give equal attention to all elements. A learner's progress is also distinguished by the ability to analyze holistically rather than analytically. Finally, progress is also marked by a move from rational to intuitive decision making (1981, 25).

In 1984, Benner applied the Dreyfuses' model in a field study to examine the experiential learning of nurses and engaged narratives to examine nursing knowledge (2001, 293). Benner illustrates the way in which nursing students and nurses progress in moving toward expertise in practice through their defined stages of novice, advanced beginner, competent, proficient and expert. Benner explains that the model reflects the movement from a performer's early use of abstract knowledge to using "past concrete experience" (13). As performers progress through the stages, they move from a preoccupation with the many pieces of a clinical situation to an ability to view the situation holistically. Benner observes that the passage from novice to expert is characterized by a movement away from "*detached* observer to *involved* performer" (13).

Another noted difference between the novice and the expert practitioner is the way in which rules are followed and principles are applied.

From the time of Benner's influential study, the works of Benner and Dreyfus and Dreyfus have been closely associated. The researchers publish together, and the Dreyfuses have applied their theory to nursing practice as well. Dreyfus and Dreyfus contend, "Nursing ... is at the same time a paradigm case of applied theory and an outstanding example of a practice that is in principle beyond the reach of theory and analytical reason" (2009, 19), thus acknowledging the importance of both theoretical knowledge and practice to achieve expertise. They argue that complex practice professions such as nursing and medicine require experiential learning in addition to theory. In light of the similarities in the proposed theories of these researchers, they are examined together to establish their relevance to simulation in medicine and nursing.

The skill acquisition theory observes that the novice learner is dependent upon both rules and principles and focuses on the task only, having no prior situated knowledge to assist in learning new skills. The educator must, therefore, consider the level of the learner when designing a simulated learning exercise so as to not overwhelm the novice learner with extraneous details and to allow the learner to focus on the task at hand. In order to assist the novice, therefore, the educator should reduce distraction and place learning in noncontextual situations (Dreyfus and Dreyfus 2009, 10). Applying this theory, a student who is learning to insert a urinary catheter would first learn the principles involved (sterile technique, for instance) and practice the technical skills isolated from context and distraction. Once there is a level of skillfulness with the basic task, the educator can place the learning in context and introduce patient variables. With

the addition of clinical practice and opportunities to work on real problems, the student will eventually progress to the advanced beginner stage (Dreyfus and Dreyfus 2009). As the learner moves from non-situational learning to recognizing similarities in situations, the learner develops the ability to discriminate meaningful from trivial data (Dreyfus and Dreyfus 1980).

This model underscores that the simulated learning experience should be aligned with the level of the learner to avoid overwhelming the novice or depriving the advanced learner of challenges. Importantly, however, Benner cautions, "What one cannot do is be beyond experience, or be responsible for what has not yet been encountered in practice" (2001, x). With experience, students progress from the novice stage through the remaining stages, with the possibility of reaching the expert stage as practitioners. Expertise is marked by intuitive performance that does not rely upon rules or principles, but a know-how and discretionary judgment that comes only from experience (Benner 2001, xxiii). Dreyfus and Dreyfus and Benner argue strongly for experiential learning. While simulation is not a substitute for actual clinical experiences with patients, simulation can provide learning experiences for students to practice clinical reasoning and technical skills, thereby learning how to recognize what is important in a situation, set priorities, apply theoretical knowledge, and learn to view the whole picture.

In his work with the Carnegie Foundation for the Advancement of Teaching, Shulman guided the comparative studies of the professions tasked with identifying the characteristics that distinguish what he refers to as the "signature pedagogies" of profession education (2005a, 52). The Carnegie Foundation recently published the results of the investigations of nursing in 2009 and medicine in 2010

(http://www.carnegiefoundation.org/publications\_archive). While not a learning theory per say but rather an exposition on professional education, the thesis of a signature pedagogy holds relevance to simulation as a teaching method in medicine and nursing. Signature pedagogies, Shulman explains, are the unique features that characterize teaching and learning in a profession - the fundamental components that teach professionals "to *think*, to *perform* and to *act with integrity*" (2005a, 52). Shulman describes professional education as having intellectual, technical and moral elements that are reflected in the three apprenticeships of the professions:

a cognitive apprenticeship wherein one learns to think like a professional, a practical apprenticeship where one learns to perform like a professional, and a moral apprenticeship where one learns to think and act in a responsible and ethical manner that integrates across all three domains (Shulman, 2005b, n.p.)

As examples of the characteristic forms of signature pedagogies, he offers clinical rounds in medicine, case dialogue in law and design studios in engineering (2005b). The significance of the signature pedagogies, Shulman argues, is their influence "in shaping the character of future practice and in symbolizing the values and hopes of the professions" (2005a, 53). These fundamental characteristics form the "habits of the mind, habits of the heart, and habits of the hand" (2005a, 59). The signature pedagogies, therefore, direct the formation and socialization of professionals as well as their instruction (Shulman 2005b; Benner 2009).

Findings of the Carnegie studies demonstrate that the professions give unequal attention to the intellectual, technical and moral dimensions of education, which may indicate the priorities of the professions (Shulman 2005a, 52). The signature pedagogy of

medicine, for example, reveals more attention is given to clinical performance with less attention to forming professional attributes (2005a, 52). In nursing, the signature pedagogy reveals more emphasis on professional identity and ethical comportment and less attention to the instruction in the sciences that are essential to nursing (Benner et al. 2009). Shulman submits that the professions have a responsibility to give equal attention to all three dimensions that encompass not only the instruction of students, but their socialization into the profession as well (2005a).

The signature pedagogies are categorized as pedagogies of uncertainty, engagement and formation (Shulman 2005b). Uncertainty arises from the nature of the instruction, the interactions of the students and educators in classes and in clinical settings, the interactions of peers, and from the curriculum itself (57). The uncertainty brings with it an opportunity for students to experience unstructured professional practice – the "messiness," value conflict and uniqueness of problems inherent to professional practice to which Schön (1983) referred – and learn to manage this ambiguity which is necessary for professional growth. Shulman observes that in these times of changing "curriculum materials," educators have to creatively adapt the signature pedagogy to the lack of consistent opportunities with patients (2005b, n.p.) Patients are not as readily available to students in inpatient settings and those that are admitted are often in facilities for shorter periods, directly affecting learning opportunities. The signature pedagogy also works to bridge the perceived "gap" between theory and practice that many educators observe; it has the potential to "connect thought and action" (2005b, n.p.).

The signature pedagogy of a profession has far-reaching effects, affecting how students and educators define expertise, the structure of authority and even the design of

the classroom (Shulman 2005a); therefore, it should be carefully considered and implemented. It is distinguished both by what it is and "by what it is *not*," the approaches that are emphasized and what is minimally included. He observes that "a choice necessarily highlights and supports certain outcomes while, usually unintentionally, failing to address other important characteristics of professional performance" (2005a, 55). Presently, the pedagogical approaches in medicine include case studies, web-based instruction, small group instruction, standardized patients and simulation, clinical rounds and patient presentations (Shulman 2005b; Swanwick 2010). In nursing these include simulation, case studies, web-based instruction, pre- and post-clinical conferences, role modeling, and coaching (Benner et al. 2010). Simulation, which is used in both medicine and nursing, is a choice that can create an engaging, interactive and formative learning environment. Simulation is a well-suited methodology to enable the development of the cognitive, practical and moral apprenticeship of nursing and medical students, providing opportunities for educators to place emphasis on the knowledge, skills and attitudes necessary for the development of the artistry of professional practice.

## F. The Structure and Components of Simulation in Healthcare Education

Simulation has well-recognized components that provide its structure and contribute to its success as a teaching method. Hertel and Millis observe that the structure of simulation conforms to a model plan for higher education, "the application of knowledge, interaction with ideas and people, experience, feedback, and reflection" (2002, 11). While different models for simulation have been suggested to guide the educator (see Jeffries 2007; Nehring and Lashley 2010), there are basic elements that are

essential to any simulation framework. The composition of a simulation typically reveals four phases: preparation, implementation of the case study or scenario, debriefing and reflection, and evaluation. There are many specific details involved in simulation, particularly when initially employed by an education program; however, a review of all aspects of the teaching methodology is beyond the reach of this dissertation. For more specific details and directions in designing and implementing the various types of simulation, an educator would benefit from consulting the work of simulation consultants, professional organizational websites, and texts that provide templates for scenario development, guidelines for establishing simulation centers, and also address typical budgetary matters (see Campbell 2008; Hertel and Millis 2002; Jeffries 2007, Jones 1995; Ker and Bradley 2010; Nehring and Lashley 2010; NLN SIRC). The following elements are essential to simulation and are culled from both the simulation literature and personal experiences in designing and using simulations.

## i. Preparation

Effective simulation requires careful preparation, and consistent with this philosophy, the preparation phase of simulation requires particular attention to planning and details. When considering simulation as a teaching pedagogy, the educator must determine, first of all, if simulation is an appropriate teaching method for achieving particular learning objectives and clearly establish its purpose. The following questions may guide the educator in selecting simulation and focusing the purpose of the simulation: Is an interactive, learner-centered experiential method preferred because of the benefits of the immersive experience in stimulating the cognitive, behavioral or affective domains? Is simulation appropriate because clinical opportunities to work with

actual patients are scarce or have the potential for harm? Is simulation selected because observing students in the clinical environment is not dependable or convenient and does not provide opportunities to compare student performances? (or because of difficulties finding appropriate clinical experiences?) Is the purpose to provide opportunities for deliberate practice or practice until reaching competency? Is the purpose to allow directed teaching during a student's performance in a "time in-time out" style as Barrows introduced, or is it selected to allow students to find a safe place to make errors and learn from their mistakes? Will the simulation be utilized primarily for learning and assessment or is the primary purpose for formative or summative evaluation? Will individualized or group feedback and the opportunity for reflection on the performance help to move the educational objectives forward?

Importantly, the selection of simulation, as with all teaching methods, is driven by the learning objectives. Simulation can be understood as an evolving case study that introduces a problem to be solved (Gredler 2004). The preparation of the simulation includes the development of the case study, known as a scenario, which must be appropriate to the learner's level and meet the objectives of the course. Importantly, it is the case study, and not the equipment, environment or technology employed, which serves as the simulation's foundation. The case study may be designed by the teacher, adapted from an available case, or purchased. Templates for scenario development are available to guide the educator (Childs, Sepples, and Chambers 2007; Issenberg and Scalese 2008). Different from case studies that students read and analyze by "looking at an event from the outside," learning with a simulation pulls students into the case study, allowing them to work with the "power and authority of professionals who are trying to

cope with a developing situation" (Jones 1995, 12-13). The learning objectives should be formulated to take advantage of this immersive experience, which may be designed to meet cognitive, affective and psychomotor skills. As with all other teaching methods, the educator needs to consider the learners' abilities with regard to knowledge and skill level, their familiarity with the teaching mode (whether standardized patients, mannequin-based patient simulators or virtual reality), and the type of preparation the students and faculty will need to more successfully achieve the learning outcomes.

The scenario serves as the teaching-learning template. It provides the case study or problem, the participant roles, the role of the facilitator, the design of the simulated patient, the set up of the simulated environment, and the equipment and materials needed for the scenario (Dubose, Sellinger-Karmel, and Scoloveno 2010). It lays out the progression of the simulation, identifies the cues that students receive to help them progress through the case, determines the immediate feedback to the students' actions (or inaction) and identifies those features that determine the resolution of the problemsolving exercise. During the design of the scenario, the educator can manipulate the events of a case study to best meet the learners' needs. Prompts or cues can be embedded in the case documents (the medical record, assessment data and shift report, for instance) and within the scenario at an appropriate level that conforms to the students' learning ability (Larew et al. 2006). The level of difficulty of the scenario can be adapted for the students' ability by using a series of "escalating prompts" that range from subtle baseline cues to more specific cues (Larew et al. 2006, 17). This strategy attempts to "provide positive learning experiences that challenge students at their current level of functioning," avoiding undo frustration and anxiety (17). More experienced students may progress

through a scenario with fewer and more subtle cues, while inexperienced students may require more prompting, consistent with Benner's theory (2001) in which learners progress from novice to expert. The educator can also compress or accelerate the time in which normal events should transpire, modify the learning experience to make it simpler or more complex, and conveniently " 'design out' negative learning experiences that often occur in actual practice" (Hertel and Millis 2002, 5). The preparation of the scenario is a highly creative process, but the educator must be careful to ensure the accuracy of the data, concepts, and events using reference texts as needed (Dubose, Sellinger-Karmel, and Scoloveno 2010). It is important as well to make certain that the relationships between participants and the relationships between processes and concepts are valid to maintain the credibility of the simulation for the students (Gredler 2004).

The fidelity and complexity of the simulation must also be determined. This decision is affected by the learning objectives and also by the resources available to the educator. While many educators aim to replicate reality as much as possible to assist students in suspending disbelief, this is not always necessary (Hertel and Millis 2002). Hertel and Millis assert that environments that are too complex or too real may interfere with a student's ability to achieve the objectives. Simulation, rather, should aim for "reality of function," a concept advanced by Jones (1985, in Hertel and Millis, 25-26) that ensures an environment which enables students to perform their roles well. Kneebone offers a metaphor for simulation that is useful to planning as he compares simulation to the artist's canvas, suggesting, "simulation is less like a photograph of clinical care than a painting of it – a process that requires selection and interpretation" (2010, i47). He advises that simulation should recreate the critical aspects of an

experience but does not need to replicate all facets of it. Most educators attend to the need to provide a close representation of physical and environmental fidelity, attempting to replicate the physical appearance of the patient, body part or environment. Educators also address functional fidelity, attempting to imitate the task, the patient's actions or the response to the student's interventions as well as psychological fidelity, which addresses the student's perception of the simulation (Allen, Buffardi, and Hays 1991; Gredler 2004). Gredler adds that it is important to provide for fidelity of the relationships among the participant roles and the causal factors, striving for logical, credible relationships among people, things and ideas. The simulator and environment are prepared in advance of running the scenario and reset between groups of students.

High-fidelity simulation may be desired for most simulations, but it is not essential, and its use depends on the complexity of the skills to be learned. A fine motor task may require a high-fidelity task trainer that produces haptic feedback (touch) to provide more accuracy (Maran and Glavin 2003). The fidelity of this experience can be enhanced by the use of a standardized patient, alongside the task trainer, as Kneebone et al. (2002) suggest, allowing for the development of interpersonal and communication skills. The beginning learner, however, may become frustrated by a high-fidelity experience while learning a task and might benefit from a lower-fidelity trainer to initially learn a technical skill. As the student develops confidence with the procedural skills, the simulation can progress to a higher-fidelity experience with more complexity, perhaps with the addition of a contextually-based situation, which would allow the student to develop the necessary professional attributes and communication skills

essential in clinical interactions without experiencing undo frustration (Dubose, Sellinger-Karmel, and Scoloveno 2010; Kneebone et al. 2002).

Effective preparation also considers the roles of students and facilitators, which must be delineated and explained to the participants prior to beginning the simulation. In addition, simulations that use a simulated or standardized patient require preparation or coaching of the actor, who must be carefully trained to portray the characteristics of a patient or family member. The use of high-fidelity human patient simulators often requires the preparation of the technical staff to assist in managing the scenario.

Importantly, students should understand the purpose of the simulation and how it has been designed to achieve particular educational outcomes and recognize the expectations of the learner in terms of preparation (Dubose, Sellinger-Karmel, and Scoloveno 2010, 197; Wilt and King 2012). They must understand their responsibilities in assuming their assigned roles and, in addition, recognize their responsibilities as learners to genuinely engage in the activity. Students should recognize that a primary goal of simulation is to nurture, develop and socialize the student into a professional role. The assigned role will often require them to perform behaviors for which as students they are not legally permitted; therefore, students may need specific preparation in these areas (Dubose, Sellinger-Karmel, and Scoloveno 2010). In a simulation for undergraduate nursing students, for example, a simulation might require the student to contact a physician for orders; in a simulation for medical students, a medical student might lead a cardiac arrest team, giving orders, "calling" the end of the code, and conveying the news to the family.

To benefit from simulation, the students should be adequately prepared to interact with the simulated patient and environment. They should have practice time in an orientation session to experiment with the mannequin and any equipment that is used to increase their familiarity and confidence and to plan for the simulation (Wilt and King 2012). Some schools produce a video orientation to their particular simulation equipment and environment and make this available for students to view independently in preparation (Wilt and King 2012). Students also need to understand the level of fidelity and what information is and is not available to them. In preparation for simulation-based education, students are typically required to complete pre-simulation assignments that fill any conceptual gaps and review applicable coursework that supports the scenario (communication guidelines, such as "SBAR" – Situation-Background-Assessment-Recommendation). In addition, students should have adequate time to plan care for the patient and establish priorities.

The preparation details may be moderate or quite extensive depending on the complexity of the scenario and the students' experience with simulation, yet all simulations require consideration of these factors. Some educators estimate that it takes approximately the same amount of time to develop a scenario as it does to prepare a complex lecture; therefore, educators should allow adequate time for development (Rauen 2001). Experienced educators in simulation (Issenberg and Scalese 2008; Jeffries 2007; Nehring and Lashley 2010) address the full spectrum of elements that must be considered when preparing for simulation-based learning, including the equipment and props, available technology, space and scheduling considerations, available time to schedule the simulation and debriefing sessions, faculty development, and expenses.

## ii. Implementation

The primary interactive element of simulation is the engagement of students in the scenario as they work to solve a problem. If appropriate to the simulation, the scenario may begin with an audiotaped or verbal patient report. Students are encouraged to review the patient's chart if this is part of the case study and prioritize their approach to the patient. During the scenario, the educator adopts the facilitator role. Dubose, Sellinger-Karmel and Scoloveno (2010) describe the facilitator's involvement in a simulation on a continuum from completely instructor-led to completely student-driven. The facilitator can offer prompts or cues to guide the students in the right direction, an approach that is used more often with students who have had little exposure to simulation. The authors report that the students respond positively to this assistance. As students gain more experience, they are capable of following through the case study without guidance, thereby taking responsibility for their learning.

During the implementation phase, the activity becomes student-centered and student-directed. Peer-to-peer teaching occurs and students work to problem-solve together, mirroring real world decision making (Hertel and Millis 2002). Hertel and Millis, therefore, recommend allowing the simulation to progress without the facilitator's assistance, urging the facilitator to refrain from intervening; however, the decision on facilitator involvement will likely be determined by the students' experience and the complexity of the scenario. Some questions by the students may need immediate clarification, such as concerns about the equipment that would affect the progress of the scenario. Hertel and Millis suggest that students be gently guided to consider their role, clarify their concerns, or seek out additional information (53).

The facilitator's role is also influenced by whether the simulation is designed for teaching and learning or if it is meant for evaluation (Jeffries 2005). If students are being evaluated on their performance, the facilitator will generally let the scenario unfold without offering cues or direction. In simulations that are used for teaching and learning, the facilitator may opt to provide additional prompts, allow for a "time-out," or let the activity progress unaided.

During this phase of the simulation, the participants receive feedback from their engagement in the activity. The feedback may be positive or negative in response to their actions or failure to act. A mechanism for feedback may be built into the simulation itself. Pre-programmed integrated simulators, for instance, are capable of responding immediately to the learner's action, while others require a facilitator or operator to manually make changes in the patient (Childs, Sepples, and Chambers 2007). For example, in a simulation using a high-fidelity patient simulator, a student may observe that the patient is having respiratory difficulty and immediately raise the head of the bed, apply a pulse oximeter to measure oxygen saturation, and replace the patient's nasal cannula to correct the patient's hypoxemia. As a result of the correct actions, the patient's breathing pattern and oxygen saturation level will return to normal, providing positive feedback to the student. In the same scenario, the student who fails to note the patient's respiratory distress or observes the condition but does not take action will experience a deterioration of the patient's condition unless the facilitator provides additional cues to alert or direct the student's actions. In scenarios using standardized patients, the script indicates the type of feedback in the form of patient response that students receive from their interactions.

Although reflection is an essential part of the activities that follow the implementation phase, students are also engaged in reflection as they actively participate in the scenario. Simulations are designed to create problems and introduce uncertainty. As both Dewey and Schön posit, reflection arises spontaneously from doubt and perplexity and assists students to creatively search for solutions. Students reflect upon and review their knowledge and past experiences, looking for similarities or patterns that can be adapted to the present setting and move from routine behavior to deliberate thought. As noted in the discussion of experiential learning theories, Schön refers to this process as "reflection-in-action." Reflecting on their knowledge, experiences, and the feedback that results from their actions while in the midst of the simulation assists students in identifying and managing the problems that arise in the case study. This process of discovery builds new knowledge and experiences.

## iii. Debriefing, Reflection, and Assessment

Simulation is a reflective practicum, and the reflective and assessment processes that students engage in following the simulated experience help to connect classroom theory to clinical practice. This period of reflection and feedback is known as "debriefing" and is shaped as a "guided discussion" (Hertel and Millis 2002, 62). Many educators and students believe that the debriefing period is an essential component of a simulation, perhaps its most important feature (Jeffries and Rizzolo 2006; Morse 2012; Onori, Pampaloni, and Multak 2012; Shinnick et al. 2011). Morse (2012) observes that empirical evidence examining the debriefing component of simulation tends to show that students value the critical self-reflection and constructive feedback that result from this guided discussion. Lasater observed in a small focus group study that while students

appreciate affirmative feedback, they identified a need for "definitive and straightforward" feedback, one student in particular noting, "I would have benefited from knowing the shortcomings of my choices" (2007, 274).

The debriefing serves several purposes, including time for "settling" of students' emotions, reflection, and assessment of learning (Hertel and Millis 2002, 60). In addition to revealing the learning that occurs during the simulated experience, students are guided in transferring this new knowledge to actual clinical experiences (Kerr and Bradley 2012). The debriefing period reinforces these connections, helping to bridge the theory-practice gap and serving as a place for students to reflect on their decisions, choices and actions. During this time, students engage in discovery about themselves and "the soundness of their own thinking" (Lederman, in Hertel and Millis 2002, 60). These activities capture Schön's notion of reflection-on-action.

In general, the structure of a debriefing session follows a format of facilitated discussion and reflection. Hertel and Millis (2002) provide a useful format for the guided discussion that focuses on three facets of the students – "as people, as the roles they simulated, and as students" (2002, 66). Focusing on the participants as individuals, the facilitator provides time for emotional release, recognizes their feelings and elicits the changes in their feelings that occur during the simulation. Identifying and sharing emotions can reduce stress and allow the discussion to continue. Thus, one of the first questions often asked in the debriefing session is "How do you feel?" Moving forward, the discussion focuses on the students' roles and the motivations, critical thinking and decision making processes that directed the plan of action (Hertel and Millis 2002, 68-69). The discussion reveals the students' thought processes, perceptions of data and cues,

and also weaknesses in the scenario design. Finally, the debriefing focuses on the students as "students." The facilitator's role in directing this phase is to clarify the learning objectives, relate the activity to prior learning and review the way in which this new experience and knowledge can be applied in authentic clinical experiences (Hertel and Millis 2002, 70). During this process, the facilitator guides the students in reviewing their performance to identify strengths as well as gaps in knowledge or skills and reviews key learning points (Durham and Alden 2008). Debriefing is critical in clarifying misconceptions or misinformation to prevent negative transfer of incorrect information to the clinical setting (Hertel and Millis 2002; Morse 2012). Simulations often generate perplexing problems (such as value conflicts) that reflect real world issues, which are not easily resolved. The debriefing session, therefore, serves an important function in allowing students to engage in dialogue about these issues in a supportive setting. Moreover, the guided discussion provides the facilitator with a window into the students' thought processes, identifies patterns of clinical reasoning, and the strengths and weaknesses that need to be addressed.

The simulation literature reveals several guides for structured debriefing to assist the facilitator (Durham and Alden 2008; Hertel and Millis 2002). It is important to note that while students expect and value feedback during the debriefing, feedback should be constructive and warmly delivered (Durham and Alden 2008; Onori, Pampaloni, and Multak 2012). Some variables in debriefing include the use of video-recordings to facilitate the review process. Students can watch the recordings independently and critique their performances prior to gathering for a group debriefing session. Some highfidelity patient simulators also maintain a log of the events and students' actions, which

may be reviewed during the debriefing (Jeffries 2005). In addition, facilitators may chose to include the use of written reflection in the form of journaling, for instance, to promote insight in critical thinking, transference of learning to the clinical setting and further socialization into the professional role.

At some point in the simulation, often following the debriefing, students can be offered additional time to work through the scenario again with or without the facilitator's presence. Some students will appreciate the opportunity for additional practice to gain more confidence and skill competency as they apply their new learning and insights. Viewing the problems presented in the case study with a fresh point of view allows students a second chance to experiment with new approaches (Hertel and Millis 2002, 6). The additional opportunity may develop the students' confidence and enhance their ability to connect this knowledge to the clinical practice setting. To accommodate this process in an environment with limited simulation technology, this educator sets up a modified version of the simulation for students to replay the scenario independently while the facilitator uses the main simulation environment to conduct simulations with other groups of students. Students have expressed their appreciation for the additional practice opportunities.

Sandars (2009) raises important ethical considerations concerning a reflective practicum. He cautions that some types of reflection, whether group discussions or reflective journals or diaries, expose a student's vulnerability; therefore, the educator must be careful to protect confidentiality. Sharing about events with adverse or negative outcomes or poor performance must be handled sensitively. Reflective practices require that the facilitator create a safe environment for students.

iv. Evaluation

Apart from the formative feedback that students receive in the debriefing session, summative evaluation of the students' performances may be conducted. Simulation that is conducted for formative evaluation places the educator in the role of evaluator rather than facilitator (Jeffries 2005). Students often experience some anxiety with simulation (Lasater 2007). Not surprisingly, therefore, students are more anxious when their performance is formally evaluated (Durham and Alden 2008). In nursing, most simulations are conducted for teaching and learning, and less often, for performance evaluation (Jeffries 2007). In medicine, however, simulations have often been used to evaluate competencies as noted earlier. Feedback that is provided to a student following a summative evaluation for competency testing, however, is provided privately to maintain confidentially (Dreifuerst 2009).

Simulations that include standardized patients are another rich source of personalized feedback for students. Standardized patients offer direct feedback on their perceptions of the students' performance. This formative evaluation is in addition to the feedback that is provided by the facilitator during the debriefing session. The responses from the standardized patients may be verbal and/or written as part of a checklist (Saewart and Rockstraw 2012). The student also completes a self-evaluation of the performance.

In addition, students should be given the opportunity to evaluate all components of the simulation, which acknowledges the importance of their contributions in the student-centered learning activity. This information provides significant feedback for the

facilitator on the students' perceptions of the activity and its ability to meet the learning objectives. In addition to students' perceptions, the evaluation can be structured to assess the specific learning that resulted from the simulation.

# G. The Empirical Studies That Illustrate the Use of Simulation in Healthcare

## Disciplines

The scholarship examining simulation in healthcare education is extensive. The areas of study include its effectiveness as a teaching and learning method, its validity in evaluating competencies, the most beneficial means to apply simulation, the use of theory to support simulation research, comparisons of different types of simulation in achieving learning outcomes and more. For the purposes of this dissertation, the primary focus of the review of the scholarship is on the efficacy of simulation as a teaching method in achieving learning outcomes. Systematic reviews of simulation in medicine and nursing are examined; however, specific research studies that enrich the perspective on simulation in medicine and nursing are also examined.

The simulation scholarship differs somewhat between the disciplines of medical and nursing education. Medical education has a longer history than nursing education of utilizing standardized patients, incorporating high-fidelity simulation, and employing simulation for summative assessment of clinical competencies in the licensure exam process (Sanford 2010). The scholarship in these areas is more extensive, therefore, in medicine than in nursing. Yet, there are similarities in the nature of the research, which has moved from simple assessments of student perceptions of simulation to qualitative and quantitative studies of its effectiveness in achieving learning outcomes (Shinnick,

Woo, and Mentes 2011) Less frequently, simulation research is conducted that measures its effects on patient outcomes.

Systematic reviews may apply the Kirkpatrick model or a variation of this model to assess the outcome measures of a simulation intervention (Issenberg et al. 2005; May, Park, and Lee 2009). Models are used to classify the level of effect produced by the study intervention (written case study, high-fidelity simulation, standardized patient, virtual reality, for example). The Kirkpatrick model, for example, classifies the range of effect as follows, moving from participation to changes that produce patient outcomes: Level 1, participation in educational experiences; Level 2a, change in attitudes; Level 2b, change of knowledge and/or skills; Level 3, behavioral change; Level 4a, change in professional practice; and Level 4b, benefits to patients (Issenberg et al. 2005, 14). Level 1 and 2 changes are much more likely to result from a simulation intervention, whereas Level 4 changes are more difficult to demonstrate. An example of a Level 4 change is illustrated in the study by Andreatta et al. (2011). The researchers demonstrated positive outcomes for patients in an educational intervention that incorporated frequent, randomly scheduled mock code simulations for pediatric medical residents. Although the mock codes involved a multidisciplinary team, the researchers focused on the development of the pediatric medical residents' skills. A variety of scenarios were designed to address areas of difficulty or deficits. The simulations were video recorded and followed by debriefing sessions that were conducted for formative assessment (34). Initially, the educational initiative demonstrated a 30% increase in survival rates associated with cardiopulmonary resuscitation. With further development of the program and an increase in the number of

annual mock code simulations, the pediatric survival rate following cardiopulmonary events rose to 50% (35).

Systematic reviews of simulation often center on a particular type of simulation, rather than examining simulation in general. Reviews of simulation may focus on high-fidelity patient simulators, standardized patients, virtual reality or hybrid approaches. Additionally, the simulation scholarship may examine the components and features of simulation separately, studying debriefing, reflection, evaluation, fidelity and more. For ease of discussion, the studies are presented by discipline. An examination of research studies on the specific use of simulation in *ethics* education in medicine and nursing, however, is included as part of Chapter Six.

#### i. Medical research

In medicine, systematic reviews, critical reviews and a meta-analysis are available that examine the use of high-fidelity simulation, virtual reality and standardized patients. In addition, best evidence in medical education (BEME) reports on simulation and on the use of the Objective Structured Clinical Examination (OSCE) have been conducted. The BEME report on the OSCE is included, in particular, because of its emphasis on the use of standardized patients. An examination of the reviews and the BEME reports follows.

In a BEME review that examined the literature from 1969 to 2003 on high-fidelity medical simulations, Issenberg et al. (2005) reported on the aspects and applications of simulation that lead to effective learning. Although this review shared little of the evidence on simulation's effectiveness in achieving learning outcomes, it contributes valuable insight into the best practice features of simulation. The systematic review

included 109 articles that were empirical and predominantly quantitative and comparative in design, which used high-fidelity simulation as an educational intervention (17). This excluded review articles and those that used simulation for assessment only. The majority of the articles appeared in surgical and biomedical engineering journals, followed by anesthesiology, internal medicine and medical education (19). Consistent with this, almost 80% of the articles concerned "practical procedures," followed by management skills, clinical skills and knowledge of basic sciences (19). To be included in the systematic review, the articles also had to produce results in knowledge, skills and attitudes, which were classified using the Kirkpatrick criteria described earlier. The study did not share the results of this classification.

The review process identified ten essential characteristics or features of simulation that educators should consider when planning simulation-based instruction: feedback, repetitive practice, curriculum integration, practicing with increasing levels of difficulty, multiple learning strategies, capturing clinical variation, controlled environment, individualized learning, clearly defined outcomes, and simulator validity (fidelity). Some of these features have already been discussed in this chapter or will be reviewed in the analysis of the advantages of simulation, but a few of the features deserve attention here. Almost half of the study articles (47%) reported that feedback is simulation's most important feature, allowing students time for self-appraisal (Issenberg et al. 2005, 21). The review showed that "the source of the feedback is less important than its presence" (23). The review demonstrated that 39% of the studies stress the value of repetitive (and deliberate) practice, which has been associated with significant improvement in clinical performance (cardiac assessment skills and cardiac life support,

for example) and improves both skill acquisition and maintenance (21). An important finding in 25% of the review articles is the need to integrate simulation-based education into the curriculum. It should not be considered "*extra*-ordinary" but a typical and required part of learning (23). Adapting the range of difficulty level to the student's readiness or stage as a learner is a significant feature and consistent with the Dreyfuses' model of skill acquisition. Issenberg et al. (2005) reinforce the need for educators to set objective criteria against which students are able to "demonstrate performance mastery" (23). Moreover, they add that by increasing the variety of clinical situations to which students are exposed, there is greater likelihood of achieving standardization of curricula, a particular concern in schools with restricted access to certain patient populations (23).

McGaghie et al. (2010) conducted a review of research and scholarship on simulation-based education research published from 2003 to 2009. The review authors compared the results of this systematic review with previous reviews including the BEME published in 2005 (see Issenberg et al. 2005, above) and confirmed that research has improved by comparison (50). With this review, the authors assembled a list of twelve best practices and features of simulation, the first five of which are identical to earlier reviews: feedback, deliberate practice, curriculum integration, outcome measurement, simulation fidelity, skill acquisition and maintenance, mastery learning, transfer to practice, team training, high-stakes testing, instructor training, and education and professional context (52). The characteristic of feedback is well-described in this report, making use of a four-step model by Rudolph et al. that the review maintains is well-supported in evidence: "(i) note salient performance gaps related to predetermined objectives; (ii) provide feedback describing the gap; (iii) investigate the basis for the gap by exploring the frames and emotions contributing to the current performance level, and (iv) help close the performance gap through discussion or targeted instruction about principles and skills relevant to performance" (Rudolph et al. 2008 in McGaghie et al. 2010, 54).

The review by McGaghie et al. is very useful to educators and researchers, as it identifies the current level of research in the area, yet it also presents the "gaps in understanding," those aspects of simulation-based education that are still unclear or unanswered and warrant investigation (2010, 52). For instance, while the review provides evidence for debriefing and feedback, it also finds gaps in current research about how much feedback is required, calling for additional research in this area. Deliberate practice is shown to be beneficial in developing and maintaining skills, knowledge and affective behaviors (55). Questions remain, however, about the period of time over which practice should occur for the best results (55). Curriculum integration is accepted as necessary as evidenced by the systematic review, yet concerns exist regarding the best means to accomplish this goal. The review acknowledges the difficulty in establishing reliable outcome measurements, calling this "one of the greatest challenges now facing the field" (56). McGaghie et al. affirm, "Reliable data are vital for, firstly, providing accurate feedback to learners about educational progress and, secondly, making arguments for valid research results" (56).

The review also revealed that most simulation was conducted for teaching and learning of technical skills, with much less attention to professional attributes and skills such as communication or cultural sensitivity, for example. Concerns over skill maintenance or "decay" emerged from the review, leading to questions about conditions

that contribute to loss of competency. McGaghie et al. (2010) note that despite the acceptance of mastery learning and recognition that learners vary in the rate in which they achieve mastery, few studies demonstrate this approach to learning. Questions remain on the variables (cognitive aptitude, motor skill, experience) that affect the time it takes a learner to master a competency (53, 57). In their assessment of transfer to practice, the authors recognize the difficulty in designing studies that achieve a positive effect on patient care, and more so, in measuring this effect. The review, however, provides several examples of improvements in patient care and outcomes. Simulation that was conducted for central venous catheter insertion was associated with lower procedural errors and reduced catheter-associated infection rates; mock code sessions led to better compliance with treatment guidelines; and virtual reality training improved surgical skills (58). The best practices for team training were identified and elaborated on in the review, with gaps in understanding concerning maintaining team skills and the interchangeability of team members.

Moreover, McGaghie et al. (2010) shared simulation's usefulness in high-stakes testing, a form of summative evaluation. In addition to simulations with standardized patients (OSCEs) that are used for clinical competency training for medical residents in the United States, simulation is currently used for certification testing in anesthesiology in Israel and in internal medicine in Canada (59). As outcome measurements improve, simulation may be used more frequently for high-stakes testing. Instructor training was evaluated in the review, identifying a gap in understanding of whether educators should be trained specifically for simulation or if general teaching skills are preferred. Lastly, the review examined the educational and professional context, concluding that context

contributes significantly to learning. Educators should strive for authenticity – that is, learning should occur in a context similar to their practice setting (60). This mirrors Barrows' view that students need opportunities "to learn in the same manner as the student is going to practice" (Wallace 1997, 10). McGaghie et al. acknowledge that the compilation of the list of best practices and identification of gaps may be biased by their selection of included research and scholarship and also from their experience as educators. Nevertheless, the extensive and detailed review paints a clear understanding of the state of simulation in medical education and the path to best practices.

In a review of simulation in medical education, Okuda et al. (2009) examined 113 articles on the use of simulation in medical education. The review was limited to simulations that utilized partial task trainers, high-fidelity simulators and computer screen-based simulations and excluded standardized patients. The authors found evidence to support the effectiveness of simulation in increasing knowledge, confidence and comfort levels with procedures. Simulation was also effective in improving performance of skills, teamwork and communication (Okuda et al. 2009). The authors noted that a small number of the studies on simulation showed improvements in clinical outcomes. The authors advocate for more studies aimed at demonstrating a relationship between simulation and improved patient outcomes.

In a compilation of several BEME reviews, an Agency for Healthcare Research and Quality (AHRQ) report and other reviews, McGaghie et al. (2009) assessed the available evidence on simulation and medical education and affirmed the above review findings. They added that success with simulation may be "dose-response" related, such that more exposure to simulation produces better outcomes (65S). In addition, based on

the work of Issenberg (2006), the reviewers pointed out that successful outcomes are a "product of simulation technology (*eg*, devices, standardized patients), teachers prepared to use the technology to maximum educational advantage and curriculum integration" (65S).

A meta-analysis of quantitative research demonstrated that simulation used as deliberate practice is superior to the traditional teaching method of "see one, do one, teach one" (McGaghie et al. 2011, 706). Simulation-based deliberate practice is performed for skill improvement. Some of its characteristics include "focused, repetitive practice," reliable measurements, feedback and monitoring (707). The meta-analysis of 14 studies included residents, medical students and a small number of internal medicine fellows. The results indicated that deliberate practice improved laparoscopic surgery skills, improved performance and application of advanced cardiac life support protocols during actual codes, improvement in cardiac auscultation skills, and improved performance of central venous catheter insertion, hemodialysis catheter insertion and thoracentesis. The authors report that the results are powerful and demonstrate a very high confidence level (95%) (708).

Virtual reality has also been examined in a systematic review by Cook, Erwin and Triola (2010), which included literature to February 16, 2009. The review focused on the effectiveness of virtual reality and its design features that contribute to learning outcomes. The review articles comprised research with medical students, nursing students, residents, nurse practitioner students, physicians and other healthcare professionals. Cook, Erwin and Triola evaluated the effectiveness of virtual reality on knowledge, clinical reasoning and skills in three situations – compared to no intervention,

compared to a non-computer intervention (for example, lecture, standardized patient, or handouts) and compared to a computer intervention (usually another virtual reality). Studies that assessed virtual reality without a comparison intervention demonstrated substantial gains in knowledge, clinical reasoning and skills (1595, 1599). When compared to either non-computer or computer-based interventions, the results were negligible. The authors could not elicit quantitative data on the comparisons with computer interventions but found that learning outcomes improved with enhanced feedback, mastery learning, advance organizers and contrasting cases (1599).

May, Park and Lee (2009) conducted a review of the literature on the use of standardized patients in health care education, which included undergraduate and graduate students as well as practitioners and faculty in medicine, nursing, dentistry, pharmacy, physiotherapy, dietetics, allied health and speech pathology. Research on standardized patients has established this educational method as highly effective in medical education and to a more limited degree in nursing (May, Park, and Lee 2009; McGaghie et al. 2011; Okuda et al 2009). May, Park and Lee (2009) noted that while standardized patients are highly valued in medicine and nursing for teaching and learning and also evaluation, there have been no reviews that provide evidence of their contributions in teaching and learning. Most of the review articles on standardized patients conducted up to this point concerned their use in evaluation (487).

The review comprised English language articles from 1999 to 2005 and was based on sixty-nine studies. Seventy-three percent of the articles were from medicine, fifteen percent were from nursing, and the remaining from other healthcare disciplines (May, Park, and Lee 2009, 489). Sixty-six percent of the total review articles focused on the

application of standardized patients with students; the remainder were with residents (17%), practitioners (12%) and teaching faculty (2%). The results indicate that standardized patients are used most often to teach communication skills (55%), clinical skills (32%) and physical examination skills (17%). In addition, they are used to teach medical ethics, teaching skills, cultural competency, and multidisciplinary cooperation (489). The review noted that in addition to portraying the patient, the standardized patient provided feedback in almost half of the studies. In a smaller number of studies (8.7%), the standardized patients instructed the students (489).

The review authors observed that both subjective (self-reports of satisfaction and benefit) and objective measurements (in the form of OSCEs and cognitive tests) were conducted. The study outcomes were assessed by a model adapted from the Kirkpatrick and Freeth levels (May, Park, and Lee 2009, 489). The majority of the studies (59%) noted self-reported levels of satisfaction and the perceived benefits of using standardized patients for teaching (Level 1 changes). Many of the studies (71%) reported a gain in attitudes (increased confidence, comfort with communications and conducting sensitive procedures) and cognitive and psychomotor skills (Level 2 changes). Five studies (7.3%) showed outcomes that resulted in a sustained change of behavior when assessed months after the teaching-learning intervention (Level 3 changes) (489-490).

The review by May, Park and Lee (2009) demonstrated that standardized patients have been used to teach communication and interpersonal skills in situations of breaking bad news, counseling, domestic violence, depression and patient education. Standardized patients have been engaged in scenarios on clinical bioethics, hospice and palliative care, and working with verbally disabled patients. More traditional uses of standardized

patients include disease management and physical assessment. The review authors observe, however, that the articles show a lack of rigorous design and explicit theoretical base.

Lane and Rollnick (2007) examined the literature on the use of standardized patients and/or role play on the communication skills of healthcare students and practitioners. Twenty-three studies were included that compared the use of standardized patients and/or role play to no intervention, to didactic methods and to interactive techniques. For this analysis, only the results from studies that used standardized patients are discussed. The review found that medical students who participated in training using standardized patients showed statistically significant improvements in breaking bad news, domestic violence training, sexual health counseling, and history-taking skills over the control groups who did not have the intervention (15). Dental students who had standardized patient experiences displayed greater examination and record-keeping skills than students who consulted with actual patients, but there were no significant differences between the groups in communication skills (15). The review authors reported that dental students who voluntarily took a motivational interviewing course on smoking cessation that used standardized patients incorporated more of the techniques in their interactions with patients; patients were also "more actively involved" (15). The review noted that medical residents who participated in smoking cessation training that used standardized patients showed significantly better counseling skills in smoking cessation than those who received only a didactic intervention. Additonally, patient-centered communication training for nurses using standardized patients resulted in increased use of open-ended

questioning, psychosocial questions and "affective talk" than nurses without training (16).

When standardized patient interventions were compared to didactic methods, significant improvements were seen in violence screening skills of medical students and HIV risk assessment and counseling in physicians (Lane and Rollnick 2007). When compared to interactive methods (feedback, group discussions, videotaped reviews), the results are more variable. One study in the review (Kleinman et al. 1996) that assessed students' technical and interpersonal skills with patients undergoing pelvic examination incorporated standardized patients in two ways - one group was taught by a standardized patient who functioned as both patient and teacher, and the other group observed a pelvic examination performed by a teacher on a standardized patient. The study results were measured following the students' rotation and indicated that students who were taught by the standardized patient had greater interpersonal skills (Lane and Rollnick 2007, 17). Other studies in the review that compared the use of standardized patients to interactive techniques showed no significant difference between groups. It should be noted that there were few studies that compared the standardized patient intervention to interactive techniques. Lane and Rollnick mentioned the difficulty in summarizing the findings in these particular studies. While some studies were well-designed, many had methodological weaknesses that made it difficult to establish "concrete conclusions" (16). Overall, the review demonstrated that most of the studies that compared the use of standardized patients to no intervention or didactic methods demonstrated a significant improvement in communication skills (16).

Finally, a BEME report on the use of OSCE in medical education reviewed the literature from 1975 to 2001 (Harden and Harden 2003). This review does not reflect the impact of research resulting from the incorporation of OSCE into the clinical competency exams for medical licensure, which occurred later, in 2004. The BEME report indicates trends, however; thus the results are briefly described here. It is interesting to note that 81% of OSCEs used a "live person" of which 70% were standardized patients (Harden and Harden 2003, 13-14). The remaining OSCEs used actual patients. The OSCEs were used most often to assess history taking and physical examination (54%), followed by communication (39%) and patient management (22%) (15). Rarely was the OSCE used to assess personal development (1%) and "role of the doctor" (0.3%) (15). The authors reported that the OSCE was primarily used to assess clinical competence and less often to evaluate teaching, teaching methods or a feature of the curriculum (12%). The report indicated that the purpose of the OSCE is primarily "to evaluate levels of clinical competence (51% of all recorded purposes), to evaluate some aspect of the curriculum or teaching or a teaching technique (12%), [and] to monitor progress, assess learning and provide feedback (11%)" (16). The BEME results showed that the OSCE was used most often as a pilot and less often for summative evaluation (31%) and formative evaluation (20%) (15). The BEME review also indicated predominantly positive responses to OSCE, citing its validity, reliability, objectivity, and positive acceptance by students, staff, and patients. Some negative responses included concerns with reliability (despite being accepted as highly reliable, there are concerns with consistency in marking the participant's behavior) and that it is time- and labor-intensive and expensive (19-23).

ii. Undergraduate Nursing Research

The compilation of literature reviews on simulation in nursing includes studies that examine how simulation is conceptualized as a teaching paradigm or a learning paradigm (Kaakinen and Arwood 2009), the use of theory to support simulation research (Kaakinen 2009; Rourke, Schmidt, and Garga 2010), the results of qualitative and quantitative studies (Cant and Cooper 2009; Lapkin et al. 2010; Sanford 2009; Shinnick, Woo, and Mentes 2011), simulation design characteristics (Jeffries and Rizzolo 2006; Smith and Roehrs 2009), and educational practices (Jeffries and Rizzolo 2006). The systematic reviews are limited in that they consider studies in English only. The scholarship on standardized patients is limited to the review by May, Park and Lee (2009) who examined the application of standardized patients in both nursing and medicine. A three-year project on simulation in nursing (Jeffries and Rizzolo 2006), a significant venture of the NLN and Laerdal Medical, is reviewed here as well.

A review of nursing literature on simulation from 2000-2007 found that simulation has been applied and studied more often as a teaching paradigm than as a learning paradigm (Kaakinen and Arwood 2009). The review authors found that very few articles mention a learning theory; however, almost eighty percent of the articles referred to simulation as a teaching method (2). Kaakinen and Arwood observe that simulation is most often assessed for its effectiveness in teaching a particular skill rather than in promoting cognitive or affective learning. Consistent with this paradigm, the effectiveness of simulation as a teaching method is assessed by measuring the learner's performance and mastery of skills. Kaakinen and Arwood propose that simulation should also be designed for student-centered learning that cultivates the cognitive processes and conceptual learning essential for critical thinking and problem solving (17). With this

shift to a learning paradigm, a simulation would be guided by learning theories and would reflect this in its measurement of assessment. The authors analyzed the following learning theories for their suitability in learning-centered simulation: Bandura's social learning theory, Kolb's experiential learning theory, Knowles' adult learning theory, Lewin's social change theory, Dewey's social construct theory, cognitive styles theory, Schön's reflective practice, and Benner's performance-based model (12-16). Features such as reflection, student-determined learning goals and feedback might hold more importance in the learning paradigm, depending on the theoretical framework.

Rourke, Schmidt and Garga (2010) examined the use of theory-based research in high-fidelity simulation for the period of 1989 to 2009. The review authors evaluated the theoretical frameworks for twenty studies that met their inclusion criteria and found that only ten percent of the studies adequately engaged theory-based research (6). The remaining studies reflected either a minimal use of theory (45%) or no theory at all (45%). Rourke, Schmidt and Garga explain that although educators and nursing students are "fervent about the efficacy" of simulation, research has not been able to support its perceived value (2). The authors argue for developed, consistent application of theory to bring external validity to this area of study.

A three-year, multi-site research study on simulation in nursing by the National League of Nursing and Laerdal Medical was directed toward expanding the scholarship in simulation and developing simulation materials for educators (Jeffries and Rizzolo 2006). The project evaluated the following aspects of simulation: design features, educational practices (active learning, collaboration, diverse ways of learning, high expectations), cognitive gains and self-reported measurements of satisfaction, self-

confidence and judgment. The simulation design features assessed in the study included objectives/information, support, problem solving, feedback, and fidelity. The study compared teaching post-operative care to small groups of nursing students using a paper/pencil case study simulation, static mannequin-based simulation, or high-fidelity mannequin-based simulation with a guided reflection activity. The project was conducted in four phases, generating student outcome data for Phase III and Phase IV. The results of the final phase of the study are shared here.

In the final phase (Phase IV) of the study, each student participated in two of the three forms of simulation and compared the experiences. Overall, those in high-fidelity patient simulation rated their satisfaction with the learning activity and their confidence significantly higher than the paper/pencil case study group and reported active learning, affirming its importance. The high-fidelity simulation group rated fidelity, feedback, support and objectives significantly higher than the paper/pencil case study group. The students who worked with high-fidelity simulation also indicated that fidelity and feedback were the most important features.

Those who participated in the paper/pencil case study rated collaboration higher and also reported higher expectations to perform well compared to those in the highfidelity simulation. The group identified less problem solving opportunities but rated their judgment performance higher than those working in high-fidelity simulation. Jeffries and Rizzolo (2006) suggested that this finding might be related to the format for case study, which has information included within it compared to the discovery that is required in active learning methods, such as high-fidelity simulation.

The study also noted that the student's assigned role in the simulation did not result in a significant difference on knowledge gain, satisfaction or confidence; however, those who assumed the observer role rated collaboration lower than the other students. Jeffries and Rizzolo (2006) advised that educators construct activities to allow those in the observer role to better collaborate in the simulation. Consistent with the views of most educators, the project also indicated that debriefing and feedback are the most important design features. Students rated the active learning feature of simulation as the most important aspect. The authors submitted that the features of simulation, such as active learning, reflection and feedback, develop students' self-confidence and enhance learning. Based on the research findings, educators are advised to carefully address educational practices, such as active learning, collaboration, diverse ways of learning, and high expectations when choosing simulation and to incorporate the features of simulation when developing simulation experiences.

Cant and Cooper (2009) conducted a systematic review of the literature on the effectiveness of medium to high-fidelity simulation in nursing compared to other educational methods. The original unrestricted search of studies on simulation in nursing and health care published between 1999 and January 2009 yielded over 2000 articles, indicating extensive literature on the subject. When exclusion criteria were applied, twelve quantitative studies on medium to high-fidelity mannequin-based simulation were identified for the review. The review evaluated the effectiveness of simulation, form of simulation, validity of the measurement assessment, and timing of the assessment. Cant and Cooper reported that although each study in the systematic review compared mannequin-based simulation to another teaching method, three-fourths of the studies

compared simulation with the 'usual' teaching methods, which were identified as either lecture, student group interaction, case studies, clinical debriefing, tests or self-learning packets. In two cases the methods were not defined (6). More than half of the studies employed a validated assessment measure; however, the remaining studies did not clearly indicate the reliability of the assessment instrument. The review articles assessed simulation's effectiveness on the following: knowledge, clinical skills, critical thinking, satisfaction or self-confidence.

Cant and Cooper shared that all studies in the review report showed statistical improvements in knowledge, skills, critical thinking and confidence, demonstrating simulation's value as a teaching-learning method; however, when compared with other methods, statistically significant gains "over and above" the other methods were variable (2009, 8). Fifty percent of the studies reported additional statistically significant gains in knowledge, critical thinking, satisfaction or confidence when compared to the control groups (11). More than half of the studies compared high-fidelity simulation (HFS) to other interactive learning techniques, such as skill stations, videotaped simulations, group discussions, case study presentations, virtual education, and clinical seminars that share features of HFS, which perhaps weakened the expected effect of this method and likely "reduced the comparative effect" (11). When simulation was compared to traditional lecture, it demonstrated a statistically significant improvement in knowledge (11; see Brannan, White, and Brezanson 2008). The review also reported that student's assigned role does not appear to affect learning outcomes, noting that it is likely the exposure to simulation that contributes to learning (12). Based on the evidence in their review, Cant and Cooper concluded that simulation is an effective teaching method when applied in

the right context and when used with best practice guidelines that are aligned with teaching goals (13).

Lapkin et al. (2010) conducted a systematic review of published and unpublished studies from 1999 to 2009 to investigate the efficacy of high-fidelity human patient simulation mannequins (HPSM) on clinical reasoning skills. The review uncovered 1600 articles that reference simulation and clinical reasoning skills; however, the review criteria of randomized and quasi-randomized control trials with undergraduate nursing students reduced the potential pool to 21 articles. Further inclusion criteria (the use of high-fidelity patient simulation with undergraduate nursing students) reduced the number to 8 articles. No studies were designed exclusively to study clinical reasoning, thus the attributes of clinical reasoning and outcomes that are associated with HPSM (confidence, judgment, satisfaction) were examined. As a result, the outcomes measurements included critical thinking, clinical skill performance, knowledge acquisition, self-reported confidence levels, and student satisfaction with simulation (e209). Three studies examined the effect of HPSM on critical thinking and noted varying effects. Two of the three studies found significant improvement in critical thinking (Howard 2007; Schumacher 2004), and one showed no difference from the control (Ravert 2008). Two studies demonstrated a statistically significant improvement on clinical skill performance (Alinier et al 2004; Radhakrishnan, Roche, and Cunningham 2007). All four studies that evaluated the effect of HPSM on knowledge acquisition reported significant gains (Brannon, White, and Bezanson 2008; Hoffmann, O'Donnell and Kim 2007; Howard 2007; Jeffries and Rizzolo 2006). One of these studies (Jeffries and Rizzolo 2006)

reported knowledge gains in all three groups - HPSM, static patient simulation, and paper/pencil case study (the summary of this study is included above).

The review by Lapkin et al. (2010) identified three studies that assessed selfconfidence levels and found inconsistent results. One study reported higher levels of selfconfidence in students using HPSM (Jeffries and Rizzolo 2006), whereas the other studies reported no statistically significant differences between the HPSM and control groups (Alinier et al. 2004; Brannan, White, and Bezanson 2008). Finally, two studies in the review examined student satisfaction with simulation (Howard 2007; Jeffries and Rizzolo 2006). Both studies reported "overwhelming" student satisfaction when asked to assign value to simulation or rate their satisfaction with the learning activity (Lapkin et al. 2010, e220). Lapkin et al. share that while the results are inconclusive regarding HPSM's effect on improving clinical reasoning, HPSM is associated with a statistically significant improvement in critical thinking, clinical skill performance and knowledge levels.

Shinnick, Woo and Mentes (2011) conducted a literature review on mannequinbased Human Patient Simulation (HPS) to examine its effect on knowledge gains and transferability, skill attainment, critical thinking, and self-efficacy (measured as confidence). Inclusion criteria were a quantitative design and a sample size of at least ten prelicensure nursing students, which yielded eight quantitative studies. The review authors noted a short, ten-year history of HPS research in nursing (66). The majority of the initial studies culled in the search were descriptive or qualitative, which reported subjective data that examined participants' attitudes and perceptions of satisfaction, confidence levels, and transferability of knowledge to the clinical setting. Other articles

included research on the development of evaluation instruments and issues related to the technology and procedural aspects of simulation (66). The authors' assessment of the simulation literature in nursing revealed that debriefing is the most valued component of simulation, that students experience increased self-efficacy with simulation, and that simulation is perceived by students as valuable, beneficial, enjoyable, and realistic (66-70). While one study found no statistical difference in students' perceptions of the transferability of knowledge to the clinical setting, several studies revealed that students found simulation to be of value for future clinical experiences. The authors also noted the variability in HPS's effect on critical thinking and knowledge gain in the small number of quantitative studies. The following effects of simulation were found: Simulation improved clinical skills and competence (Alinier et al. 2006) and improved cognitive skills (Brannan, White and Bezanson 2008), whereas no significant difference was observed in knowledge gains among different learning activities (Griggs 2002; Jeffries and Rizzolo 2006), and no significant difference was demonstrated in critical thinking (Ravert 2004). Brown and Chronister (2009) demonstrated improvement in critical thinking in one group only (second semester seniors).

Sanford (2010) examined the literature on high-fidelity simulation in nursing education for qualitative and quantitative research findings. Although Sanford does not indicate the inclusive dates for the literature review, the study lends support to trends in this area. Sanford reports that most studies are opinion-based, with few rigorous research or evidence-based studies to indicate positive learning outcomes on simulation in nursing education. One study by Smith and Roehrs (2009) noted the significance of a simulation's design characteristics to its effectiveness in achieving outcomes. The study

outcomes were self-reported and included self-confidence and student satisfaction. The results indicated that the problem solving features in a simulation had a strong association with self-confidence, while problem solving features and objectives considered together significantly affected both self-confidence and satisfaction (Smith and Roehrs 2009, 77). The authors indicate the importance of carefully developing a simulation's features.

#### iii. Summary of research

The review of the research demonstrates that simulation in medical and nursing education is an effective approach for teaching and learning. There is evidence that simulation is associated with improvement in specific competencies and is valued by students. Simulation has been successfully employed to teach not only psychomotor or technical or procedural skills, but to develop cognitive and affective skills. The literature in medical education is far more extensive and illustrates the benefits of simulation with gains or improvements in the following areas: skill training for proficiency, knowledge and cognitive skills, critical thinking, communication skills, interpersonal skills, interviewing, clinical skills (history taking and physical examination), counseling, domestic violence screening, team performance, confidence, satisfaction, and comfort in performing sensitive procedures. The nursing literature has a briefer history, but the research reveals that simulation is beneficial and leads to improvements in knowledge, cognitive skills, critical thinking, clinical skills and competence, self-confidence, and satisfaction.

The literature shows that the features of simulation are relevant to its success; in particular, debriefing and feedback are considered the most important features of

simulation, valued by students and faculty alike. Deliberate practice is effective in developing and maintaining skills. Interestingly, a learner's role in a simulation does not affect the learning outcomes.

There is still a need for continued research and scholarship on simulation in medical and nursing education. The reviews illustrate that simulation has primarily resulted in changes (improvements or gains) for students, yet there is evidence that simulation can produce positive changes in patient care and outcomes. Several reviewers call for simulation design and measurement to capture this effect. The literature also identifies "gaps in understanding" that point to areas for future research (McGaghie et al. 2010, 52).

The ability to directly compare and interpret research findings is hampered by the amount of diversity in experimental design and interventions as well (Cant and Cooper 2009, 11). Many studies in nursing compared high-fidelity simulation (HFS) to other interactive learning techniques, such as skill stations, videotaped simulations and clinical seminars that share features of HFS, which perhaps weakened the expected effect of this highly interactive method and likely "reduced the comparative effect" (Cant and Cooper 2009, 11). Moreover, although research is ongoing in developing reliable outcome measurements, according to McGaghie et al., this is "one of the greatest challenges now facing the field" (2010, 56). As simulation is used more often for summative assessment and high-stakes testing, assurance is needed of the reliability and validity of measurement instruments and procedures.

H. Advantages and Challenges of Simulation in Healthcare Education

The analysis of simulation thus far has alluded to many of the advantages and challenges associated with this unique teaching method. Simulation provides for a holistic education that encompasses a range of learning goals – transmitting knowledge, improving reasoning and skills, while at the same time promoting affective behaviors and attitudes. It also offers a holistic approach to teaching and assessment of learning where a learner's performance can be assessed for a constellation of factors – knowledge, clinical reasoning, professional virtues, and skills. Simulations that place patient problems in context, revealing the complexities of healthcare, offer much more to students than written case studies and group discussions. When compared to other forms of teaching, simulation presents a rich tapestry for student learning and assessment of teaching effectiveness.

Simulation offers benefits to educators and students and ultimately, to patients. Some of the advantages to educators arise from the structure and instructional design characteristics of simulation and its ability to support educators in meeting educational goals. Simulation fulfills the purpose of creating explicit learning opportunities to aid in the transfer of learning to the clinical practice setting. The benefits of simulation have often been described as allowing for predictable, consistent, reliable, standardized and safe experiences (Okuda et al. 2009). According to Okuda et al., from the patient's view, simulation "increases the likelihood of a minimum competency level" (2009, 339). Simulations can be developed to ensure that all students are prepared at a basic level to care for frequently occurring clinical situations (history taking, physical examination and technical skills, for example). Educators can also use simulation to expose every student to critical patient situations that are difficult to find in the clinical area, thus expanding

the breadth of their education (Issenberg and Scalese, 2009; Lasater 2007). Simulation has the advantage of enabling educators to eliminate extraneous details that confound learning with real patients and compress time to facilitate directed attention on specific teaching and learning objectives.

The advantages of repetitive, deliberate practice are well-recognized and unique to simulation. Practice opportunities have been demonstrated to enhance skill performance and goal attainment. Simulation provides the unique opportunity for educators to observe students in action and witness clinical skill performance, professional attributes, and identify strengths and weaknesses. In this way, simulation offers advantages in assessing the effectiveness of teaching to meet educational goals (Barrows 1964). Debriefing and reflection sessions that follow allow educators to hone in on critical thinking and decision making processes, reinforce learning and clarify misconceptions.

Simulation has advantages for students, as well. Its primary advantage to learners is the opportunity to improve their performance in a safe environment and to make mistakes and learn from them (Hovancsek 2007). In many situations, it surpasses learning by traditional methods with gains in knowledge, clinical skills, communication skills, affective learning, critical thinking, collaboration, team performance, and confidence. Many types of simulation offer the advantage of being individualized to the learner's needs and providing personalized feedback (Hertel and Millis 2002). Students experience firsthand the outcome of their responses, attitudes, decisions and actions, often immediately, allowing them to experiment with different approaches. Students submit that the debriefing and feedback following a simulation is essential to learning. The

guided discussion provides opportunities for reflection and self-critique, which is essential for building connections from theory to practice in the clinical setting and for assessment of thought processes. Simulation also makes clear for the student any gaps in learning (Durham and Alden 2008). As a student-centered teaching method, simulation is highly interactive and has the advantage of placing the responsibility of learning with students, giving students control and aiding motivation to learn (Hertel and Millis 2002). Moreover, students value simulation, and for many, it is an enjoyable way to learn. Although some students report anxiety with simulation (Durham and Alden 2008; Rauen 2001), simulation may ultimately reduce anxiety by increasing confidence (Campbell 2008; Durham and Alden 2008). Many of the present generation of learners are comfortable with technology (Durham and Alden 2008) and expect that available simulation technologies be integrated into their education.

Furthermore, simulation is advantageous for patients. The ultimate goal of educating students is to improve the quality of patient care and outcomes. Providing simulated experiences prior to the clinical setting can contribute to patient safety by reducing risk and harm. In their argument on the ethical imperative to use simulation, Ziv et al. share that simulation offers "best standards of care and training, error management and patient safety, patient autonomy, and social justice and resource allocation" (2006, 253). The process of educating medical and nursing students creates ethical dilemmas for educators, which simulation can help to minimize or avoid (Ziv et al. 2006). Care delivered by inexperienced or incompetent students may cause harm to patients, generally conceptualized as physical harm, but patients are also at risk for psychological, sociocultural, and moral harm. Educators are therefore ethically obliged to provide

learning opportunities for students to reduce these risks and promote optimal care for patients. With simulation, educators can observe firsthand any deficits in competencies that could contribute to poor care. Simulation has been recommended by the Institute of Medicine to reduce medical error, which should translate to lowered healthcare costs (Gaba 2004; Kohn, Corrigan, and Donaldson 2000). Although research that demonstrates direct benefits to patients is limited, there is evidence of improved student performance, knowledge, critical thinking and more. Furthermore, deliberate practice has been shown to significantly improve skill performance and teamwork, leading to improved patient care and outcomes (Andreatta et al. 2005).

While simulation has many benefits as a teaching method, it has distinct challenges and difficulties that may limit the extent of its implementation. The major challenges include the factors of time and expense. Simulation is a time and laborintensive process, requiring substantial effort for preparation and implementation. Implementation generally involves repeating scenarios with several groups of students; therefore, scheduling multiple groups requires a substantial investment in time. If simulations involve patient actors (whether professional standardized patients or volunteers), additional time is required to prepare the actors.

Simulation can be associated with considerable expenses, which vary depending on the type of simulation that is selected, the level of sophistication, and the available resources of the institution. Standardized patient simulations have expenses associated with hiring and training actors, the use of video recording equipment, support staff, and space. The use of volunteers, staff or student actors as patients can reduce the costs. The expenses associated with simulators, haptic systems and virtual reality vary significantly,

from affordable (task trainers and low to medium-fidelity patient simulators) to very expensive (immersive high-fidelity systems in fully equipped rooms). The costs are higher when simulation is first integrated into a program, as expenses are generated for the purchase of simulator equipment, associated technology, and space to store and use the equipment (Issenberg and Scalese 2009). Continuing costs occur for maintenance and upkeep of the simulators and technological equipment. Issenberg and Scalese (2009) point also to the personnel expenses associated with simulations – faculty to develop and conduct simulations, training of additional faculty to assist in facilitation, and personnel to maintain the equipment and facility.

Some limitations of simulation are related to the equipment itself. Although task trainers and low to medium-fidelity simulators are portable, more sophisticated simulators are fragile and cumbersome to transport, making their use outside of dedicated simulation space difficult in some cases. (Issenberg and Scalese 2009). With care, however, many high-fidelity mannequins can be moved to different places. Finding portable tables to move pieces from one space to another is helpful. Issenberg and Scalese (2009) also note that technical equipment often has set uses and may not be suited for a wide range of learning.

Simulation equipment and technology also have intrinsic limitations that affect student learning. In mannequin-based simulation, students note the limitations presented by a simulator's inability to convey nonverbal communication (Lasater 2007). In addition, although the simulator may use pre-programmed responses or may be voiced by the facilitator, students often are uncomfortable communicating with the mannequin. Many patient simulators are also limited in the ability to display physical changes in

color, edema, and neurovascular reflexes (Lasater 2007). The appearance of a simulator may contribute to some confusion in a scenario. Simulators may be masculine, feminine or gender-neutral, but they are often bulky with large features. Despite efforts to feminize the mannequin with breasts and wigs for simulation involving a female patient, some students have difficulty recalling the patient's gender during the scenario.

Furthermore, standardized patient simulation in medical education has faced criticism for its inability to authentically represent the power dynamics of the patientphysician relationship (Hanna and Fins 2006; Wear and Varley 2008). While the student and faculty may perceive the standardized patient scenario as realistic, Hanna and Fins (argue that simulated experiences alter the "usual power dyamics" in the patientphysician encounter (2006, 265). The standardized patient, they observe, holds the balance of power in the form of knowledge of the problem and the expected competencies of the student and ultimately control's the outcome of the encounter. In many simulations, the standardized patient also evaluates the student's performance. They suggest that a standardized patient may respond differently to the student than would an actual patient, affecting the authenticity of the encounter. Hanna and Fin claim that students focus on impressing their evaluators, disingenuously interacting with standardized patients. The result of simulation-based education, they argue, is learning "to be good physicians from the outside" rather than better physicians "from inside the heart" (2006, 267). Furthermore, they suggest that a standardized patient may respond differently to the student than would an actual patient, which affects the authenticity of the encounter.

Exercises that offer practice opportunities for students outside of the clinical setting represent but do not substitute for reality, and therefore, they are limited by their lack of complete authenticity. While the arguments of Hanna and Fins (2006) raise important concerns regarding the validity of simulation for the evaluation of clinical competency, similar claims can be made toward most nonclinical-based learning. Students who are engaged in classroom dialogues, small group discussions or written case analyses, for example, may also give more weight to how they are perceived and assessed and adjust their performance. Yet, experiential learning that occurs outside of the clinical arena provides opportunities for habituation of professional behaviors. Students have opportunities to "try out" suggested techniques and modify their approaches based on immediate feedback. The shift in power in the patient-practitioner dynamic, however, and its influence in achieving learning outcomes should be identified and addressed to assist students in transferring skills to the clinical setting.

Finally, as the review of empirical research reveals, one of the challenges facing simulation at present is the lack of a substantial body of evidence-based research, presently more so in nursing, but evident in both medicine and nursing. Many research studies have been limited to small convenient samples with limited comparisons. In light of the high costs associated with some forms of simulation, Issenberg and Scalese call for additional research demonstrating "evidence that the investment will yield valuable results" (2009, 34).

## **Chapter Six – Simulation in Healthcare Ethics Education**

A. The Relevance and Appropriateness of Simulation to Healthcare Ethics Education

The last chapter has shown that simulation is an integral component of medical and nursing curricula. Moreover, it is accepted as a sound educational strategy that is relevant to the pedagogies of engagement and formation and is recognized as an effective and valid method for teaching cognitive, technical and behavioral skills. Importantly, simulation as a teaching method is relevant in the development of healthcare professionals. It provides interactive learning opportunities where participants may experience gains in knowledge and confidence and transfer this learning to future clinical situations. Simulation use is also associated with improvements in procedural skill development. Furthermore, and particularly relevant to this thesis, the simulation literature supports the practice of affective and behavior skills, leading to gains in confidence, clinical reasoning, empathy, communication skills, interpersonal skills and professional attributes. Students also express satisfaction when learning with simulation.

The integration of simulation in medical and nursing instruction has been supported by adult and experiential learning theories, which acknowledge the value of its components, such as the opportunities for practice, reflection on learning, and feedback. It has likewise also been shown that simulation holds many advantages over traditional methods of instruction in developing professional competencies. Importantly, the simulation experience promotes the assimilation of knowledge and the synthesis of new learning that can be applied in the clinical setting. Although the simulation literature in medicine and nursing focuses more on its application in technical skill development, this

dissertation thus far has demonstrated its appropriateness in other aspects of professional competence as well. The structural components of simulation extend its appropriateness, for example, to developing the reflective practice, clinical reasoning skills and problem solving in the context of undefined or ambiguous situations. As this chapter will reveal, not only is simulation an excellent method of instruction in the traditional areas of clinical practice, its efficacy can also be appreciated in ethics education.

While the advantages of simulation have been expounded in Chapter Five, a few of its themes and motifs deserve to be highlighted again to illustrate the specific relevance of using simulation to teach ethics to medical and nursing students. Gredler's definition of simulation holds particular significance to ethics education and serves to guide this analysis of simulation (2004, 571). Central to her understanding of simulation is the unfolding case study that reflects the ambiguity and complexity of actual patientpractitioner situations. Gredler's definition is fitting for ethics simulations in light of its well-developed features and the recognition of simulation's role in assisting students to work through open-ended problems for which a solution is not readily known. Ethics in medicine and nursing unfolds in situations that are by their nature uncertain and openended. As Schön observed, the real world presents itself with many "indeterminate" areas of professional practice, characterized by "uncertainty, uniqueness and value conflict" (1987a, 6). Caldicott and Danis also remind the educator that teaching and learning experiences should mimic reality, arguing for more "nuanced teaching" rather than presenting ethical problems as "narrow sets of choices between competing values, virtues or principles or laws" (2009, 284). Simulation can be designed to reflect the choices that confront the patient and practitioner, as well as the factors that influence the dynamic

interaction between them and present these in context that stress the need for a nuanced approach to ethics education and practice.

Ethical practice in health care can be complex and challenging for students, particularly as they work to competently apply ethical principles, skills or strategies in the clinical setting. Patients and the ethical decisions that they must contend with need to be considered in the contexts of their particular illness and their specific relationships with others. As Cassell remarked, "Ethics is about relationships" (2007, 21). Ethics educators can use simulation to structure a learning environment that contains those elements of clinical situations that call upon "lived ethics," especially elements which arise within the relationship of the patient and the practitioner (Drane 1998, 15). Relationships are an essential consideration in ethics and include the patient's relationships with various family members, healthcare practitioners and even friends (Cassell 2007, 22). Therefore, it is important that practitioners demonstrate respect for the patient's relationships with others. Cassell observed the integral nature of "context, illness and other people" in the patient's exercise of autonomy, in particular, noting that self-determined individuals are naturally influenced by others (2007, 21). Simulation emerges as an appropriate method for teaching ethics because its structure accents the relational aspects of ethics by placing the novice practitioner in a relationship with the simulated patient. The patientpractitioner interaction is central to simulation; therefore, scenarios can be designed to help students learn the skills to develop trusting relationships with patients, develop empathy and to recognize and respect the contribution of relationships to patients' actions. Scenarios can be created that demonstrate the importance of family in the patient's decision making process, emphasizing for students the need that practitioners

have to understand not just the patient, but the important role that relationships have in the decisions a patient makes (Cassell 2007, 21). Simulation provides practice opportunities for students to learn the skills and the best dispositions to adopt in their future roles as nurses and physicians, and thereby successfully engage with patients.

As Drane remarked, most circumstances that practitioners encounter in their clinical practices require neither rules nor strategies, but a "way of being that fits or is appropriate" (1988, xi). Simulation allows the educator to manipulate the elements being simulated in the teaching for the purpose of reinforcing particular virtues and professional attributes so as to express these ways of "being." For instance, simulation can be designed to promote ethical awareness and sensitivity, to nurture respect of patient dignity and to foster empathy and a caring attitude in a difficult clinical situation. The virtuous practitioner "sees what is morally salient in most situations, and... is predisposed by virtue to act appropriately (Prior 2007, 59). While it may be more difficult to teach virtues than to teach ethical decision making frameworks, some virtues, such as compassion, empathy, respect, trust, honesty, advocacy and courage, align well with this teaching and learning method. Simulation can be used, therefore, to develop and sustain professional virtues and provide situations in which the learner can practice exercising these virtues. Simulations can also be designed to provide practice with prudential reasoning as well as problem-solving skills in a realistic context. It is through this simulated clinical experience that students can practice recognizing and clearly articulating the problem in their minds (Dennis Novack 2011, pers. comm). The learner can experience the process of selecting and applying ethical principles and virtues or

balancing benefits and harms, thus building knowledge, cognitive skills and behavioral skills.

Simulation benefits students by providing safe situations in which they develop self-awareness of their biases, communication strengths and weaknesses, and learn how it is that they make ethical decisions (Novack 2011, pers. comm.). Through simulated patient experiences students learn first-hand about the barriers that are often present in ethical dilemmas involving patients and families. Differences in values and opinions, biases, and frustration or anger can create barriers to effective decision making. The structure of simulation enables learners to reflect-on-action and develop an understanding of their behavior and decision making skills and the justification for their action, i.e., whether they acted on their underlying core values and beliefs, if they acted from a sense of duty and so forth (Novack 2011, pers. comm.).

An advantage of simulation over experiences with actual patients is the opportunity for many students to prepare for a specific learning experience, discuss the unfolding case study and receive immediate feedback. Additionally, scenarios can be adapted to meet the needs of the learner, and problem issues can be repeated and reinforced. Educators can identify frequently occurring problems or issues that are especially troubling to students and develop simulations to assist students in managing these. As Caldicott and Faber-Langendoen found, twelve percent of the ethical conflicts reported by students in one institution involved issues of moral courage (2005, 869). Students reported hesitating to speak up due to their position in the practitioner hierarchy and fear of retaliation. Simulation can be used to assist students to develop confidence, moral courage and effective communication skills to better manage future similar – but

real – situations. Students may feel awkward experimenting within the professional role in initial learning situations, but with repeated exposure, feedback and reflection on ethical competencies, students may become more comfortable adopting and internalizing their role, including the exercise of moral virtues (courage, advocacy, empathy) and prudential reasoning. They experience the results of their attitudes, decisions and actions and, consequently, are able to form habits that are fundamental to their future role as moral agents in clinical encounters.

Furthermore, as a teaching method, simulation offers a holistic educational approach that benefits the student and educator. While this interactive method is known to be effective for teaching a specific skill, it also has the advantage of allowing the educator to teach *across* the different educational domains. Supporting Shulman's (2005b) conceptualization of professional education, simulation can be adapted to meet the educational goals of the cognitive, practical, and moral apprenticeships. For instance, the educator can develop a simulation to reinforce knowledge of ethics principles and theory and improve reasoning skills, while at the same time promote explicit virtues and attitudes, such as moral sensitivity and empathy. Simulation also provides a holistic means to assess learning and evaluate the effectiveness of ethics teaching. The educator can use simulation, for example, to assess not only the level of knowledge acquired and the skills achieved in applying ethical principles but also assess the learner's exercise of professional virtues of character, such as compassion, openness and respect and the use of prudential reasoning skills that are essential in achieving positive patient outcomes.

## B. Ethical Reasons for the Application of Simulation to Ethics Education

Although simulation has been shown to be an appropriate teaching method for ethics education based upon its relevance to the development of the healthcare professional and the benefits to teaching and learning that are associated with its pedagogical features, its employment in nursing and medical curricula can be also be ethically justified. The introduction and use of simulation as an educational method can be justified by appealing to certain professional virtues and moral principles, as will be explained in this section. In addition, its incorporation in ethics education is justified by a consideration of the ethical responsibilities of those teaching ethics in nursing and medicine, which will also be examined below. As will be explained in this section, ethics educators in nursing and medicine are also able to fulfill their own professional obligations to provide useful educational and learning experiences to their students by successfully implementing simulations in their ethics teaching.

The primary ethical rationale for the use of simulation is to *nurture and sustain* the moral and intellectual virtues in medical and nursing students to better achieve quality patient care. Consistent with the ultimate goal of ethics education - to improve the quality of patient care - the goal of simulation-based ethics education is to assist students in recognizing and developing the necessary skills and dispositions to lessen and prevent errors and harm, to reduce risks and burdens to patients (reflecting the principles of beneficence and nonmaleficence), to prevent injustices, respect patient autonomy, and effectively engage in ethical decision making.

A proposal by Ziv et al. (2006) calling for an ethical imperative to integrate simulation in medical education is relevant here. As noted in Chapter Five, Ziv et al. based their position on four factors, each of which holds true here as well. The reasons that they provided included best standards for patient care, medical education and skills evaluation; error management and patient safety; patient autonomy; and social justice and resource allocation (252). The ethical imperative arose primarily from the perspective of using simulation in its conventional sense to teach and develop technical and behavioral skills. The authors' analysis considered simulation-based training primarily in regard to communication, management, cooperation, interviewing, history-taking, physical examination and procedural skills (253). The authors also promoted the use of simulation to teach informed consent, citing the goal of nurturing the learner's "ethically sensitive approach" to the process of obtaining informed consent (254). The foundation laid out by Ziv et al. can be extrapolated and taken as a premise in an argument supporting the use of simulation in ethics education and in demonstrating that there is also a moral imperative to use simulation in the ethics education of medical and nursing students. Some of the themes proposed by Ziv et al. are highlighted in this section, which describes the ethical imperative to use simulation to teach virtues, prudential reasoning (or ethical analysis) and ethical decision making.

It is well appreciated that there is an ethical obligation to reduce risk, minimize error, and remove the risk of burden to patients. The principle of nonmaleficence holds that physicians and nurses should "first do no harm," and as Ziv et al. report, this principle underscores patient safety movements (2006, 252). Simulation experiences are developed to provide consistent practice opportunities and skill assessment for healthcare

professionals that ideally lessen or remove the burden or risk of harm to patients and to students. Ziv et al. recognized this ethical obligation and the benefit of simulation to minimize avoidable harm to patients. The authors did not specify the possible types of harm, but they reported on harm that results from medical errors, which is often measured as physical injury. As they noted, patients need to be protected and also deserve to be treated by students who "hone" their skills to an acceptable level of competency prior to working with actual patients (252). Harm to patients, however, can be appreciated in a much broader sense that includes *moral* harm – harm as unfairness, disrespect and to interests – as well as socioeconomic (stigmatization, for example), psychological, and more commonly, physical harm (UNESCO 2008, 24). From this expanded definition it is clear to see the practitioner's responsibility to recognize a more comprehensive range of risks to patients.

The need to avoid moral harm is of particular significance to this discussion of ethics education. Insensitive, unapproachable, or indifferent nurses or physicians can potentially create moral harm. Lack of integrity and compassion and dishonesty also results in harm to patients. While there is limited discussion in the literature of the risks of harm or burdens to patients resulting from ethically insensitive nurses and physicians, it is evident that interactions with such practitioners may result in disrespect, injustice and social stigmatization being directed toward and suffered by a patient. Moreover, insensitive physicians and nurses are at risk for overlooking problems, ethical or otherwise, and providing substandard patient care. Simulation enables educators to place students in explicit learning experiences that require moral sensitivity and moral engagement to develop and nurture the appropriate dispositions and attributes. At the

same time, simulation allows educators to assess the moral comportment of students prior to working with patients. Simulation may also be used to help students achieve the advocacy skills and moral courage needed to manage situations where uncaring or indifferent care providers place patients at risk for harm or substandard care.

Ziv et al. (2006) also observed that simulation in medical education could potentially reduce injustices associated with novice training. They report that a disproportionate number of poor individuals receive care in academic health systems where they are more likely to be treated by novices in training rather than experienced practitioners, which results in an increased burden of risk for this patient population (254). Ziv et al. focus on risks associated with procedural or technical skills. This argument, however, also holds sway in ethics training. Ethics educators should be aware of the risks associated with insufficient clinical ethics skills, such as protecting autonomy, the process for informed consent, and withholding and withdrawing treatment, and their obligation to protect patients and prevent problems of distributive justice within a clinical setting. Simulated experiences in ethics curricula may help to reduce the burden of risk associated with novice training in academic medical institutions and minimize the injustices that are disproportionately experienced by already disadvantaged patients in this population.

Simulation also has the potential to enhance prudential reasoning and ethical problem-solving. As noted in Chapter Four, prudence is a fundamental intellectual virtue, which nurses and physicians must exercise to properly apply the virtues of character, ethical principles and theories. Nurses and physicians who lack prudence are at risk for making hasty or careless determinations in applying virtues and principles and may create

negative outcomes for patients. For example, the virtue of honesty, while critical to the patient-nurse or patient-physician relationship, must be applied with prudence in individual situations. Information shared at the wrong time or in the wrong way can create harm (Pellegrino and Thomasma 1993, 87). Likewise, physicians and nurses who lack prudence may carelessly apply the standards and guidelines of care and fail to consider the individual needs and circumstances of the patient, negatively impacting the quality of care. Learning to skillfully apply virtues, principles and standards to individual patients requires experience. As Benner asserted, "What one cannot do is be beyond experience, or be responsible for what has not yet been encountered in practice" (2001, x). Simulation can be designed to specifically strengthen prudential reasoning skills. It enables students to recognize and clearly articulate ethical issues. Creating practice opportunities to develop moral sensitivity and prudential reasoning skills enhances the practitioner's competency in responding to the individual needs of patients.

The use of simulation to teach ethics is promoted by the educator's responsibility to embed the ethical principles of beneficence, nonmaleficence, justice and autonomy in students' practice. Educators are morally obligated to ensure that students practice according to the ethical standards of the professions in order to provide quality care for patients. In their covenant with society, educators and academic institutions also have an ethical obligation to sustain the virtues and values of the profession (Begley 2006; Pellegrino and Thomasma 1993). Educators have a responsibility to produce virtuous practitioners, ensuring, for example, that their graduates practice with honesty and integrity. This responsibility is based on society's expectation that academic institutions "will provide some assurance of the characters of their graduates" and with this

expectation comes the obligation to "detect at minimum, the most egregious deficits, and attempt to correct them" (Pellegrino and Thomasma1993, 178). This assessment of a student's disposition and character should occur early in a student's education. Simulation provides opportunities for faculty to detect problems and mitigate them early and is particularly helpful in curricula that offer less supervision and oversight of students in the clinical settings.

Simulation has a valuable place in the ethics education of medical and nursing students in that it can be used to monitor and counter any tendencies for ethical erosion and moral malaise. Scenarios can be constructed, for example, that reproduce conflicts of interest or clinical situations that require empathy in challenging circumstances. Simulation is also useful in clarifying the moral boundaries for practitioners and reinforcing the need for courage in the face of morally ambiguous practices. Pellegrino and Thomasma describe some professional practices at the moral margins, such as refusing to treat certain patient groups or cooperating with policies that require inappropriate early discharge of patients (1993, 145). Simulation can be designed that raises moral awareness of these practices and through reflection demonstrates and reinforces the virtues, such as courage, needed to resist these. Employing simulation allows the educator to not only provide practice situations to develop and sustain virtues, but it also facilitates the assessment of students' performances and alerts the educator to moral failings, such as ethical cynicism, erosion of professional virtues, or exercising self-interest over altruism.

Decker (2007) raises several additional ethical considerations related to using simulation that are relevant to this argument. While her comments are directed to the use

of high-fidelity simulation in nursing education and do not specifically address ethics education, they raise interesting perspectives on the educator's responsibilities when using simulation. Decker questions the exercise of virtues, such as trust and compassion, as they relate to educating nursing students. She considers the issue of trust in the nursepatient and faculty-student relationship and questions whether failing to offer simulation in nursing education demonstrates a breach of trust and a form of neglect of patients as well as neglect of students' educational needs. Decker also questions whether educators are acting consistent with the virtue of compassion when they permit students to perform procedures on patients without the benefit of simulated experiences. These concerns are also quite relevant in ethics education. Educators should be mindful of their obligation to maintain a culture of trust and compassion in the student-patient relationship. Simulation can be used to prepare students for clinical ethics skills, such as end-of-life care issues or the process of informed consent, thus preventing neglect of patients and students and sustaining trust and compassion in student-patient and student-educator relationships.

C. The Empirical Studies That Illustrate the Use of Simulation in Healthcare Ethics Education.

While simulation has a growing presence in medical and baccalaureate nursing curricula overall, the literature review shows that it is employed to a lesser degree in ethics education. Consistent with the availability of simulation research in medical and nursing education overall, there has been more research on using simulation as a method in ethics curricula in medicine than in nursing. The literature review shows, however, that there has been minimal empirical research using simulation to teach ethics in either discipline. As a result, the articles described in this section are predominately descriptive

in nature and include student and faculty perceptions of the effectiveness of ethics simulation. The research on simulation in ethics education will be presented according to profession. In light of the limited numbers of articles that specifically use simulation to teach ethics to medical students, some articles using simulation with residents are included in this section.

## i. Research Using Simulation in Ethics Education in Medicine

A small number of articles have been written that expressly review the use of simulation in the ethics education of medical students. Of these, most examine simulation as a method for assessment or evaluation of students' ethical knowledge and skills, while fewer articles review its use as a teaching method. There is a wider body of literature, however, on using simulation to teach different aspects of professionalism to medical students, which overlaps to some extent with ethics education. The Accreditation Council for Graduate Medical Education defines professionalism as "a commitment to carrying out professional responsibilities and an adherence to ethics principles" (2011, IV.A.5.e). This definition further lists the attributes of professionalism as compassion, integrity, respect for others, respect for privacy and autonomy, altruism, accountability, and sensitivity to diverse population (2011, IV.A.5.e). Traditionally, however, professionalism has been seen as a broader construct that is difficult to define (van de Camp et al. 2004). The characteristics of professionalism mentioned most often in the literature include altruism, accountability, respect and integrity (van de Camp et al. 2004, 697). Professionalism also includes ethical practice, medical-legal compliance and clinical competence among its distinguishing features (McKenna and Rosen 2010). McKenna and Rosen observe the need to include values and the ability to resolve value

conflicts among the professional attributes of physicians. While the professionalism literature often mentions professional virtues among professional attributes, it does not use the same terminology; in fact, it is rare to find "virtue" directly mentioned in articles on professionalism.

Additionally, some of the literature on professionalism focuses on professional conduct and misconduct, while other literature focuses on interpersonal and communication skills. Standardized patient simulations have been used to teach professionalism concepts, such as being sensitive to the patient, relationship-building skills (Hochberg et al. 2010) and breaking bad news (Edinger et al. 1999; see Baer et al. 2008 for simulation experiences that use cancer survivors to teach students skills in breaking bad news). These interactive learning experiences are relevant here in that professionalism and competence in communication skills are essential for ethics competency. It is important to consider, however, that while the literature on professionalism may implicitly reflect ethics and hold relevance to the development of professional attributes, it does not explicitly teach ethics. Furthermore, simulations that are used to primarily teach professionalism may not include reflection as part of the learning experience, an important component of simulation that enriches the students' learning related to transferring and applying professional virtues, knowledge and skills to the clinical setting. A review of relevant articles in simulation and ethics education in medicine follows. Several articles on professionalism that are particularly relevant to ethics education are included in this section.

In an early article on the ethics education of internal medicine residents, Arnold et al. developed an ethics curriculum to raise awareness of the ethical dimensions of clinical

practice, support residents in applying ethical principles in clinical practice, and assist the residents to "communicate sensitively" about ethical concerns (1988, 92). The curriculum incorporated education in medical ethics (such as informed consent, truth telling, confidentiality, resource allocation, end-of-life care) and communication skills. The program used videotaped role play of standardized patient interactions to provide the residents with opportunities to practice the behavioral skills necessary for ethical practice, especially when discussing difficult ethical issues (94). Arnold et al. explained that the residents reviewed their videotapes, although the authors did not describe the reflection process that was implemented. If sufficient opportunities were provided for reflection and feedback, this teaching method, described as "role play," would serve as a fitting example of simulation-based ethics education. The curriculum program used a variety of techniques in addition to the taped sessions with standardized patients; however, the authors did not include the residents' evaluation of each technique. The program was positively viewed by the residents for its contribution to their clinical practice. Moreover, the residents suggested that the learning activities be a required part of the residency program (95).

Gordon and Tolle (1991) used simulation with volunteer simulated patients to train medical residents in advance directives. At the time of this study, the authors reported that the group of medical residents felt unprepared to discuss advance directives with patients or their families. Despite the age of the study and the likelihood that current medical residents might feel better prepared, the results of the educational program are relevant to this dissertation. The simulation's purpose was teaching and improving skills and sharing individual feedback; the authors note that it was not conducted for evaluation

purposes. The educational strategy that Gordon and Tolle implemented contained the important elements of simulation, including practice scenarios, debriefing and evaluation. The authors provided introductory and preparatory work for the residents and reviewed the students' personal goals, their plans for conducting interviews and the areas in which they wanted feedback. Gordon and Tolle described a holistic educational approach to improve knowledge, skills and attitudes about advance directives. The medical residents overwhelmingly rated the experience as positive and found it to be "realistic, relevant, and useful" (567). The residents' self-rated assessments showed improvements in all areas. Of note, one of the participants commented, "Although role-playing is always difficult, it is very useful to actually practice saying things. Thinking about what you would say is different than actually saying it" (568).

In 1991, Edinger et al. (1999) developed an ethics education program for medical students that integrated standardized patient interactions for the express purpose of teaching and learning ethics rather than for evaluation. The authors observed that at this time in medical education many standardized patient sessions were conducted for teaching and evaluation of basic clinical skills, while few programs were using standardized patients for the teaching of ethics. The ethics program included medical students in the pre-clinical and clinical years. Students participated in a series of standardized patient simulations that represented ethical dilemmas in clinical practice. The topics included advance directives, breaking bad news, informed consent, physician-assisted suicide, obtaining "do not resuscitate" status, and patient autonomy in the context of psychiatry. The program also included standardized patient sessions that afforded students the opportunity to practice communication skills with patients in difficult

clinical contexts, such as domestic violence and obtaining a sexual history. Depending on the topic, students either interviewed the standardized patient or observed a classmate. In some situations, the students observed a physician as a role model in the interaction. Discussion sessions followed the standardized patient interactions with an emphasis on reflection on feelings, emotions, behavior, communication skills, relationship-building and the effectiveness of the students' responses. The authors did not report any measurement data, but they observed that "the interactive experience… provides an opportunity for trial and error, understanding and appreciation, and personal and professional growth in the area of ethics, without the reliance on real patients to bear the brunt of a student's ethical education" (4).

Fleetwood et al. (2000) compared the efficacy of computer-based learning with small group discussions in an ethics education course with second-year medical students. The "small group discussion" group received eight bioethics lectures and eight small group discussions. In the "computer-based learning" group, two computer-based scenarios on confidentiality and assisted suicide that featured simulation-based learning were used in place of two small group discussions on the same topics. The computerbased learning program required the medical student to adopt the role of the physician and interact with an "onscreen" patient who responded with text, audio and video (97). The student interacted with the simulated onscreen patient by text and received individualized feedback from a variety of professionals on his or her ethical responses and communication skills. All students were evaluated with a final exam. The authors found that the students who participated in the computer-based learning scored better on test items on confidentiality, while no difference was found between the groups on test

items related to assisted suicide (98-99). All students were also assessed based on their interactions in a standardized patient simulation. Standardized patients evaluated the students' knowledge of confidentiality, communication skills, and also completed a patient satisfaction survey. Students who completed the two computer-based learning sessions scored better on the bioethics content; however, no difference was seen in the communications skills between the groups. The authors report that the small group discussion group scored better in patient satisfaction. Fleetwood et al. found no difference between the two groups in self-reported preparation in managing issues of confidentiality with "actual" patients, including communication skills, but the computer-based learning group reported feeling better prepared to manage the legal aspects of confidentiality (101).

Another initiative to incorporate interactive experiential learning of ethics was designed by Fleetwood, Novack and Templeton (2002), which employed standardized patient simulations to provide practice opportunities for third-year medical students. The authors' goal was to "bridge the gap between classroom instruction and bedside encounters with patients" (1100). In this program students participated in eight OSCEs, four in medical ethics and four in psychiatry. They worked through the scenarios either individually, in pairs, or in groups of three. When in groups of three, two students were observers only. The medical ethics topics included "confidentiality and HIV, informed consent, medical futility, advance care planning, physician-assisted suicide, 'do not resuscitate' orders, disclosure and cultural expectations, and medical students' training dilemmas" (1100). The four psychiatry scenarios included depression, alcoholism, domestic violence, and dementia. Faculty members observed the students through

monitors or 2-way mirrors and later provided individualized feedback (1101). A debriefing session followed the students' interactions and included ethical and legal concepts, communication skills, as well as "personal attitudes and biases that enhance or inhibit effective communication" (1101). Students positively rated the experience (4.3/5) (1101). It should be noted that although the authors referred to the learning experience as an "OSCE", which typically implies a focus on evaluation, the sessions included a 90-minute debriefing session, demonstrating an emphasis on teaching and formative assessment. The OSCE is defined in this article as an objective structured clinical exercise, not examination. Significantly, the faculty developed a guide with case materials and made this available to other medical programs at cost.

The standardized patient simulation experience in ethics reported by Fleetwood, Novack and Templeton (2002) and described above is currently conducted as part of the psychiatry clerkship in Drexel University's medical program and was observed by this writer in November 2011. Although the majority of students participate at Drexel University's main campus in Philadelphia, students at its Pittsburgh campus participate via teleconference with the standardized patients. The simulations are presently facilitated by Dennis Novack, who confirmed that the focus is teaching, learning and formative evaluation (pers. comm.). Dr. Novack notes that faculty watch full cases or parts of the simulations and observe the values, behaviors and communication skills of students. They also are able to observe generic mistakes that challenge most students. Typical communication barriers include managing confrontation, differences in values and opinions and presenting bad news. The sessions lead to discussions on the ethical concepts that arise in clinical situations, as well as norms and decision making. The faculty members are able to use the simulation sessions to learn whether students act based on their underlying core values and beliefs, are duty-bound and how they justify their actions (Novack, pers. comm.). At the end of the debriefing session, the students evaluate the educational exercise, which continues to be positively received.

As part of a comprehensive assessment of professional competence in secondyear medical students, Epstein et al. (2003) developed a 2-week intensive program that consisted of activities that integrated many characteristics of professional competence. Among the activities in this program were eight standardized patient interactions and a team-based simulation. The standardized patient interactions and post-encounter activities were designed to assess the students' competence in applying science knowledge, clinical decision making, ethics and more. The team-based simulation focused on collaboration, clinical reasoning skills, and problem solving. Standardized patient encounters and team-based simulations were videotaped for later review. Students received feedback from the standardized patients on their ability to facilitate patientcentered communications and also reviewed the videotaped interactions individually with faculty. In addition, students were assessed by their peers on interpersonal sensitivity, including demonstrations of respect, compassion, empathy, sensitivity and appropriateness of behavior. The authors note that the program included an "intensive period of reflection and feedback" intended to produce a "strong formative effect" (188). The program produces an individualized learning plan for each medical student that reflects the student's strengths, weaknesses and priorities (189). The program integrated ethics, interpersonal attributes and professionalism with science knowledge and skills over a variety of exercises, thus it is not possible to examine the specific effect of the

standardized patient simulations on the teaching and learning of ethics. It is interesting, however, to note that the students rated the standardized patient experiences as most valuable (4.25/5) of all of the program activities, followed by the individual review of the videotaped session (4.11/5), underscoring the value of simulation exercises and feedback for students (193).

The Ethics Objective Structured Clinical Examination (OSCE) deserves particular mention here as well. As introduced in Chapter Two, the Ethics OSCE has been used to evaluate the clinical ethics skills of medical students, residents and residency candidates (Singer et al. 1993; Singer et al. 1996). Based on the original OSCE model of Harden and Gleeson, the Ethics OSCE uses standardized patients in simulated case studies to assess the participant's ability to respond to clinical ethics problems (Singer et al. 1993). Early studies by Singer et al. (1996) have demonstrated the Ethics OSCE's face and content validity, inter-rater reliability and construct validity. Singer et al. (1996) also conducted a study to further test the reliability of the model (testing inter-station reliability) and at the same time compare the performance of students from different medical programs. Interestingly, the findings revealed differences among the students from the various programs on the mean scores on particular scenarios, allowing the researchers to identify possible weaknesses in the different curricula. The authors share, "Because clinical ethics requires cognitive, behavioral, communication, and professional skills, as well as virtue of character, it is perhaps the most difficult domain of medical practice to evaluate. Our research using the OSCE proves that the behavioral aspects of clinical ethics can be evaluated" (Singer et al. 1996, 498). However, the researchers were unable to demonstrate inter-station reliability and as a result, they could not endorse the Ethics

OSCE as a single method of evaluation (1996, 498). The authors suggest, however, that in light of the available feedback from the standardized patients, OSCEs can be used for teaching ethics to students (1993, 26).

As mentioned earlier, some of the research from the professionalism literature is included in this section because of its pertinence to professional virtues or quality of care. While the focus of the following article is on simulation for evaluation purposes, the results of this study reveal the potential contribution of professional attributes to quality of care. Van Zanten et al. (2005) assessed the reliability of the Objective Structured Clinical Examination (OSCE) to measure the professional attributes of international medical school graduates who applied to graduate medical programs in the United States. The professional attributes of truthfulness, rapport, having a personal manner, and achieving patient satisfaction were assessed in interactions with standardized patient actors. The study found that doctors who had recently graduated from medical programs or were younger scored higher in interviewing, rapport, patient satisfaction and counseling. Female candidates scored higher than males in all interpersonal skills, especially on rapport, which was defined as "ability to be on his or her way to establishing a caring relationship with a patient" (23). Of interest to this dissertation, the study revealed that the professional attributes assessed in the clinical examinations were moderately correlated with the candidate's performance of clinical skills, specifically demonstrating that these professional attributes resulted in enhanced history taking, physical examination and written communication skills (25). Although van Zanten et al. focused on the efficacy of OSCEs to assess professionalism, this finding suggests that professional attributes, such as respect, a caring stance, and truthfulness, can contribute to

improved practitioner performance in clinical skill performance. One can hypothesize, however, that improved clinical performance, which was moderately correlated with professional attributes, can contribute to improved patient care.

Gisondi et al. (2004) conducted a simulation with first, second and third-year emergency medicine residents to evaluate the residents' response to ethical dilemmas and their professionalism in critical care situations. The simulations used a hi-fidelity patient simulator. The simulations were used for evaluation and not as a teaching-learning strategy. This study is relevant here, however, because it demonstrates the way in which ethical issues can be embedded into previously developed high-fidelity simulations. The authors developed ethical cases on informed consent, patient confidentiality, withdrawing care, orders to "do-not-attempt-resuscitation," and performing procedures for practice on the newly deceased which were then incorporated into emergency medicine crisis resource management scenarios (932). When the medical case problem in the scenario was resolved, the facilitators introduced the ethical problem. The authors suggest that the "inconspicuous" addition of the ethical issue allowed for "unrehearsed and genuine" responses from the participants (935). The researchers found that the scenarios highlighted strengths and deficits in the residents' professional responses to the ethical dilemmas. They observed, for example, a high number of residents who conveyed confidential information in the scenario, a finding that was not available through prior professionalism evaluations (936). Not surprisingly, second and third-year residents displayed the "critical action" for the ethical problems more frequently than first-year residents (934). Interestingly, the researchers found that third-year residents displayed professional behaviors in response to certain ethical issues more "reflexively," proposing

that perhaps residency training provided the experiences that they needed. Gisondi et al. raised an interesting and important observation related to the residents' reflexive responses to the ethical issues of informed consent and end-of-life care. The authors questioned whether ethics instruction could be delivered such that learners could learn to react to clinical ethics problems much as they would to other clinical issues. They asked, "If the clinical response to ventricular fibrillation can be taught to occur reflexively, could a similar professional response be taught for commonly encountered ethical dilemmas?" (936). Their question suggests that with structured practice through simulation, residents (and other healthcare practitioners, by extension) could develop improved clinical ethics skills and professionalism to the same level of competence as is expected with medical problems (cardiac dysrhythmias, for instance). Clinical practice through simulation could provide the opportunities to sensitize medical students and residents to routinely and reflexively recognize and respond to these concerns.

Analyzing the experiences in the scholarly literature of simulation in ethics education reveals some trends and successes. The majority of the articles examined in this review identified self-reported improvements in learning. The results also suggested that simulation was relevant, beneficial and provided a valuable means for both students and educators to assess the effectiveness of learning. Not surprisingly, these studies revealed an emphasis on ethical decision making and clinical ethics skills. Consistent with Eckles et al.'s (2005) review which identified a virtue/skill dichotomy in medical ethics education, most ethics curricula are orientated toward teaching ethical decision making skills and less inclined toward teaching professional virtues. The application of simulation in medical ethics education is reminiscent of this finding. The majority of the

studies (Arnold et al 2004; Edinger et al. 1999; Fleetwood et al. 2000; Fleetwood, Novack, and Templeton 2002; Gisondi et al. 2004; Gordon and Tolle 1991) employed simulation to enhance knowledge and improve the cognitive and behavioral skills needed to competently manage ethical problems that the medical students or residents would more often encounter (informed consent, advance directives, withdrawing or withholding treatment), as well as some that occur less frequently (physician-assisted suicide). Although the simulations were conducted for the purpose of skill-building related to ethical problem solving, some authors reported the inclusion of discussion sessions that provided opportunities for the participants to discuss their feelings and emotions, values and value conflicts (Edinger et al. 1999; Fleetwood, Novack, and Templeton 2002). Some authors provided extensive debriefing sessions that emphasized how students make decisions. For example, Fleetwood, Novack and Templeton (2002) included an in-depth debriefing and reflection session that discussed professional values and value conflicts, bias, opinions and underlying belief system and the influence of these on the patientphysician relationship.

Although the simulation scholarship illustrates a trend toward skill development, several authors conducted simulation to teach and assess professional attributes and values. The study articles included assessment of teaching and learning that emphasized professional virtues and attributes, such as respect, compassion, empathy, interpersonal sensitivity (Epstein et al. 2003). Van Zanten et al. (2005) stressed truthfulness, building a caring relationship, and interpersonal skills in a simulation that was used primarily for evaluation.

Ethics educators may incorporate a discussion of professional virtues in ethics simulations, but this is not readily evident in the literature. Whether simulation is conducted with an emphasis on virtues or skills, it is evident from the ethics simulation scholarship that students value and appreciate learning through simulation. Many study authors have reported student satisfaction in their research, which is a positive and encouraging finding.

## ii. Research Using Simulation in Ethics Education in Nursing

There are a limited number of research articles that explicitly review the use of simulation in the ethics education of nursing students. Other experiential strategies, such as role play, have been used in nursing to teach related concepts, such as communication skills as discussed in Chapter Four. This review identified one qualitative study by Smith et al. (2012). In light of the small numbers of articles, additional literature employing simulation with nurses (Gropelli 2009) is included in this section.

Perlman (2008) designed simulations to teach skills used in ethics consultation to second-year, preclinical baccalaureate nursing students enrolled in an ethics course. The simulations were integrated into the course to develop the students' critical thinking, anticipate ethical problems and respond appropriately, illustrate the integration of ethical and clinical decision making, help students appreciate the complexity of the health care environment, demonstrate collaboration in problem solving and spur their interest in joining ethics committees (12). Nine case scenarios were written for the simulate ethics consultations that reflected prevalent ethics issues such as confidentiality and privacy, informed consent, capacity determination, and end-of-life care issues, among others. The

nursing students assumed the role of ethics consultants, clinicians, patients, surrogates and family members. The simulations required that participating students prepare up to 2 weeks ahead of time for their roles, an approach that would allow them to better adopt the functional role, consistent with Gredler's definition of simulation. The students serving as ethics consultants reviewed the details of the clinical case and prepared their decision making strategy. The course utilized a "clinical pragmatism framework" to guide the ethical decision making process (10). Certain students who serve as evaluators provide immediate feedback, which is followed by a discussion in the next class session that focuses on the ethical support for the students' decisions (13). The large class size limited the opportunity for all students to assume the ethics consultant role. Perlman suggested a few changes for the simulations, in particular, that standardized patients be used for the patient role. He also suggested that the simulations be video-recorded for improved feedback. Perlman shares that the course evaluations are "uniformly positive" (14).

In an immersive clinical laboratory experience, Vanlaere, Coucke and Gastmans (2010) provided simulated experiences for the purpose of developing or enhancing ethical reflection and empathy in care providers. The simulation included nursing students, nursing assistant students, experienced nurses, and allied health professionals. The learning experience is directed toward the care of older adults, who report that healthcare providers "knowingly or unknowingly contribute to their vulnerability" (2010, 325). Vanlaere, Coucke and Gastmans have described the aspects of "good care' as responding to this vulnerability, being empathic, attentive, and understanding the patient's "situation, perspective, and vulnerability" (325). They maintain that empathy arises from ethical reflection and is an essential part of virtue ethics. To address this

aspect of "good care," Vanlaere, Coucke and Gastmans developed a "care-ethics lab" simulation for care providers for the purpose of encouraging ethical reflection on care practices, developing a caring stance, and promoting empathy (326). The care-ethics lab experience consists of a voluntary two-day simulation activity referred to as an "empathy session," which focuses on empathy and the ethics of caring for others. Participants engage in the experiential learning exercise that includes an overnight stay in the simulation lab, which is designed to model a nursing home or long-term care facility with bedroom facilities and group living spaces for meals and entertainment. Experienced care providers (nurses and allied health workers) assume the role of nursing home residents, and each is provided with a specific client profile, which they may further tailor to direct the care activities they will receive (feeding, transferring assistance, bathing or restraints, for example). The nursing students and nursing assistant students take on the role of the care providers who must plan and administer all care. The authors explain that the emphasis for the students as simulated care providers is on "practicing' an attitude of involvement and adjusting behavior to this attitude within a 'safe', pedagogical context" (330). This statement reflects the priority of the care-ethics perspective on ethical reflection and empathy over the procedural provision of care and guides the students' learning.

Following the simulation, facilitators conduct a reflection activity with the simulated care providers and care receivers separately and then jointly. The reflection sessions examine the participants' perceptions and the nature of care practices with a particular emphasis on relationships. The authors report that the experienced care providers (nurses and allied health professionals) reflect on their values and feelings and

"gain greater insight into the content of their own moral awareness" (Vanlaere, Coucke, and Gastmans 2010, 333). It is not unusual for nursing programs to place students temporarily in the patient role as a clinical laboratory exercise so that they can experience nursing care from the patient's perspective, such being fed or transferred. The care-ethics lab simulation, however, is a unique immersive experience that places care ethics *central* in the learning exercise, emphasizing empathy and ethical reflection and promoting an increased understanding and awareness of the caring dimension within care practices. Those in the role of nursing home residents record their reflections at different points in the experience, and include their motivation for participating and their expectations (330). The student participants engage in additional follow up reflective activities. A few months after the simulation, the simulated care receivers participate in a group discussion in which they review their current work practice and the elements that either encourage or limit ethical reflection and caring practices. As a result of the simulation, some care providers have formed groups that have implemented measures to improve the quality of care for the nursing home residents (332). Although this article did not include any qualitative or quantitative measurements of the simulation's effects on the participants or on quality of care, this finding suggests that there were some changes in professional behavior as a result of the simulation (Kirkpatrick level 4a) that, consequently, may produce positive outcomes for patients (see Issenberg et al. 2005 regarding the Kirkpatrick criteria for effectiveness of intervention that was discussed in Chapter Five). The authors also propose that the care-ethics lab simulation assists in ethical reflection that aids the care provider in examining the context in which she or he provides care and understanding "how this context can become a habit in establishing good care" (334).

Gropelli (2009) developed a simulation as part of an elective ethics course for graduate nursing students that engaged the students in ethical decision making in the setting of simulated ethics committees. The ethics committee simulations included topics such as "do not resuscitate" orders, advance directives, futile medical care, withholding and withdrawing treatment, palliative care, and the rights of prisoners as patients (6). The nurses acquired the roles of the ethics committee members (physician, nurses, clergy, social worker, attorney and member of the community) and also the roles of the patient, family members, attending physicians. This learning activity was conducted with nurses with at least two years of experience and many with over 15 years of practice; the author noted that the diversity of the collective experiences of the participants enhanced the activity. The student actors' roles as family members and patient were unscripted and unrehearsed, requiring the simulated ethics committee members to adjust and adapt their responses as in real life (6). Students in the role of ethics committee members prepared by reviewing the central issues in the cases and employing an ethical decision making model. Gropelli notes that students were actively engaged and accountable; they were responsible for making all of the arrangements for the mock ethics committee (contacting members, setting up the room, contacting security). The author shared that the students accurately demonstrated their roles, made recommendations by consensus, communicated the recommendations and documented the results (7).

A debriefing session was held following the scenario. This simulation-based activity included preparation, an enacted scenario, feedback and reflection, all recommended components of simulation. Gropelli reported that the participants' discussions in the debriefing session centered on their decision making and

communication skills. During the debriefing session, the students critiqued the simulation and offered feedback to peers. The evaluations of this simulation were positive, and the students reported that the activity supported their use of ethical decision making frameworks and assisted them in developing therapeutic communication skills (2009, 8). Moreover, Gropelli noted that some students shared informally that following this experience they made referrals to ethics committees (8). This outcome reflects an increased sensitivity and awareness of ethical problems and the exercise of professional virtues and attributes to act on the behalf of the patient. Increasing nurses' referrals to ethics committees has the potential to increase the quality of patient care and patient outcomes (Kirkpatrick level 4b, benefits to patients), a significant outcome related to ethics education. Some participants reported that they volunteered to attend ethics committee meetings to observe the proceedings, while another student reported joining an ethics committee. The above anecdotal evidence represents a change in the behavior of some participants (Kirkpatrick level 4a, change in professional practice). Increasing nurses' involvement with ethics committees might also enhance the quality of patient care. The effect of simulated learning activities in ethics on raising interest and involvement in ethics should be studied further in the future.

The results of Gropelli (2009) are similar to the work of Vanlaere, Coucke and Gastmans (2010) in that they demonstrate the potential of simulation to exert a higher level of educational effect as assessed by the Kirkpatrick criteria; thus, they are particularly relevant to this thesis. They support the claim that using simulation to teach ethics can be an effective means to provide interactive practice opportunities that may lead to positive changes in the behavior of the learner and ultimately to improvements in

patient care.

In a qualitative study comparing simulation to in-person case discussion and online computer-based learning, Smith et al. (2012) designed a high-fidelity mannequinbased simulation (HFS) to teach legal and ethical concepts to third-year baccalaureate nursing students. The case study for the simulation focused on advance directives, and the authors' aim was to increase recognition of the ethical and legal issues that surround this concept and increase cultural competence and respect. Nursing students were divided into three groups (HFS group, in-person group, online group), and their responses were compared on three points: knowledge of the ethical and legal issues that emerged in the case study, the ability to integrate and apply other nursing content, and reflection on the learning experience (positive and negative aspects, for example). The study results revealed that all three educational approaches increased knowledge and integrated prior learning. The authors observed that in response to an open-ended reflection question, that students in each group experienced different types of learning, each with its own benefit. The in-person group appreciated the ability to discuss the case and found the format helpful in applying content. The online group found the approach convenient and also valued the opportunity for discussions. The high-fidelity simulation group valued its "interactive, real-life nature" (2012, 394). Although the in-person online groups reported negative feedback, the simulation experience received none, and the authors reported that this experience received the most positive responses. On a 5-point Likert scale, the mean scores of students' perceptions of the learning experiences demonstrated that the highfidelity simulation was rated most favorably (4.5), followed by the in-person discussion group (4.2) and the online computer-based learning group (3.6). The authors found that

the high-fidelity experience showed a statistically significant higher rating than the other methods (p < 0.05) (395). The faculty working with the simulation group concluded that it was a "personal" experience, which was not the case in the other groups. They observed "students embrace, identify with, and assume their assigned roles" (396). They also related that the students in the simulation discussed the whole case in the debriefing and were earnest in their reflections, whereas the other groups engaged in discussion as an ongoing process. This led to a greater release of emotion during the debriefing in the simulation group. The decision making of those in the simulation differed greatly from the other groups. Those in the simulation had to make decisions in a time-sensitive manner, unlike those in the other groups, and experienced decision making in a more realistic context.

Finally, a different perspective of the effects of simulation on the ethics education of nursing students is offered in a study by Haddad (2010) that involved both pharmacy and nursing students. In this simulation, which is part of an ethics course for pharmacy students, nursing students assumed the patient role. Haddad notes that the pharmacy students were unaware that the young female patients with whom they were interacting were in fact nursing students, and they instead operated under the impression that standardized patients were performing the patient role. As Haddad noted, the pharmacy students, therefore, interacted with the standardized patients as laypersons, rather than as fellow health science students (482), an important learning consideration. The ethics simulation centered on a female patient who is taking a teratogenic drug, isotretinoin (Accutane), and suspects that she might be pregnant. The patient has sought out a pharmacist (pharmacy student) for counseling about the drug's side effects. Although

Haddad designed the simulations primarily for the ethics education of the pharmacy students, in this study, she examined the standardized patient role on nursing students, investigating its effect on their learning and self-reflection. Each nursing student repeated the patient role approximately seven times. At the completion of the simulations, the nursing students recorded their reflections on their learning, the emotions experienced in the patient role, their assessment of the pharmacy students' interactions, what they learned about themselves and the effect of this experience on future patient encounters (484). The nursing students observed that their identification with the role grew with each experience and questioned how they would actually feel in this particular situation responses that indicated an empathic response to the patient's situation. They also reported sensing the pharmacy students' frustration and anger during the simulations. A week later, the nursing students participated in a focus group discussion where they related their feelings and conveyed their learning. The nursing students recounted that from their experience as simulated patients they were sensitive to their interactions with patients and the appropriateness of their communication and teaching, and importantly, the way in which as healthcare professionals their personal values and attitudes influence patient encounters. One nursing student summarized this observation, noting that despite the effort to "put your own beliefs and values aside, it still filters through with just the kind of suggestions you make" (485).

The nursing scholarship on ethics simulation is more recent and small in comparison to that of medicine. While ethics education is integral to nursing, the use of simulation to teach ethics is in the early stages of development. Unlike the ethics simulation scholarship in medicine, there were no articles identified at the time of this

review that engaged standardized patients in the ethics education of nursing students. With the recent introduction of standardized patient simulations in other areas of nursing education, it is likely that this may emerge as a new type of ethics simulation. Nevertheless, the small sample provides interesting findings that suggest implications for ethics educators. Of the simulation exercises included in the review articles, two were conducted to promote ethical decision making and communication skills (Gropelli 2009; Perlman 2008), one was conducted to enhance ethical sensitivity (Vanlaere, Coucke, and Gastmans 2010), another was conducted to assess the effectiveness of simulation to increase knowledge (Smith et al. 2012), and one was used to assess the effect of the student's role as a standardized patient on self-reflection (Haddad 2010). Despite the range of goals for the ethics simulations, four of the simulations implied that the activities enhanced ethical awareness, sensitivity or ethical reflection (Gropelli 2009; Haddad 2010; Perlman 2008; Vanlaere, Coucke, and Gastmans 2010). The language used in the research by Vanlaere, Coucke and Gastmans was more aligned with virtue ethics than most of the ethics simulation articles found in the medical literature. The authors used simulation primarily to develop empathy, attentiveness, ethical reflection, moral awareness, rather than to resolve ethical dilemmas or developing specific clinical ethics skills. Even in a study that was developed to teach clinical ethics skills (to pharmacy students), the outcomes suggested the development of empathy. Haddad's (2010) research with nursing students as standardized patients suggested that as a result of the experience, nursing students developed empathic responses and increased awareness of how values influence clinical encounters with patients.

All of the research articles with regard to ethics simulation in nursing revealed

that the participants positively responded to simulation and provided positive feedback on the simulation activities. In the single empirical study, the students who participated in a ethics simulation rated it higher than those in the other learning methods (Smith et al. 2012). It is reasonable to propose that even in this early phase of ethics simulation, nursing students appreciate and value simulation as a teaching method. Ethics educators should be encouraged by the positive results of their colleagues in using simulation to teach ethics. While simulations involving standardized patients in nursing ethics education have not been documented in the literature thus far, educators can look to the successes in medical ethics education for guidance. It is important, however, for nursing ethics to promote a balance between educational approaches that emphasize professional virtues and those that promote clinical ethics skills. The following section provides suggestions for using simulation to teach ethics to medical students and nursing students. The suggested approaches strive for a balance in the virtue/skill dichotomy and work to reduce the tendency to exclusively teach decision making skills and management of ethical dilemmas with less consideration of the ethical comportment of the healthcare practitioner.

D. Suggestions for the Implementation of Simulation in Healthcare Ethics Education

The dissertation has demonstrated that simulation has value and utility in the ethics education of medical students and baccalaureate nursing students. It has shown that there are strong pedagogical and ethical reasons for its incorporation into ethics curricula. It has also been argued that the ethical imperative proposed by Ziv et al. (2006) is equally applicable here. While there have been few studies that examined its effectiveness in

changing professional behavior, simulation has been reported to enhance the acquisition of professional virtues and attributes and to develop the ethical decision making skills of health care professionals (Edinger et al. 1999; Hochberg et al. 2010). Simulation has also been described as effective in improving the learner's therapeutic communication skills (Gordon and Tolle 1991), relationship-building skills and enhancing empathy (Vanlaere, Coucke and Gastmans 2010). There is evidence of simulation's effectiveness in enhancing ethical sensitivity, ethical awareness and ethical reflection (Vanlaere, Coucke and Gastmans 2010). Importantly, as suggested by some anecdotal reports that were identified in the review of the literature on simulation and ethics education, simulation has in some instances created changes in the participant's professional practice (for example, responding with increased ethical awareness, initiating ethics committee referrals, joining ethics committees and creating quality care initiatives). Furthermore, the anecdotal evidence has also suggested that simulation has the potential to benefit patients directly by increasing the quality of patient care (through an ethics committee referral, for instance).

The reflective component of simulation demonstrates its power in transforming theory into clinical practice. As one student commented in the study by Haddad (2010) included above, simulation provides an opportunity to recognize the effect that bias and personal values have on the information that practitioners provide to patients. This awareness illustrates just how powerful the experiential format of simulation can be. It is possible that this student may have not come to realize just how difficult it is to convey respect and an open, empathic stance when immersed in a situation that challenges one's personal values without having had this firsthand, albeit, simulated experience and the

opportunity to reflect on action. Similarly, despite their considerable professional experience, the nurses and allied health professionals in Vanlaere, Coucke and Gastmans' care ethics lab simulation reported benefiting from the opportunity for genuine reflection on their caring practices and consequently implemented practice level changes as a result.

It is interesting to note that the simulation studies which reported effects other than changes in knowledge and attitude occurred in simulations that were conducted with nurses and other health care professionals rather than with students (Gropelli 2009 and Vanlaere, Coucke, and Gastmans 2010, for example). It might be the case that changes associated with simulation at levels that positively affect professional practice and produce patient benefits are easier to observe or follow in experienced professionals. It would understandably be more difficult to demonstrate the same level of effect in simulations that are conducted with medical and medical students because of the difficulties associated with longitudinal studies and the confounding effects of other influencing factors.

While there are limited studies on the effectiveness of using simulation to teach ethics, educators in both medicine and nursing should be encouraged by the descriptive results and anecdotal reports of faculty and students. The studies share the benefits of simulation to awaken learners' ethical sensitivity and to provide practice experience in both the virtues and skills essential to achieve ethics competency. As with any educational approach, the simulation must be carefully designed to best achieve the learning objectives. There are several ways, however, in which educators can integrate simulation into ethics education.

Simulation can be designed primarily to teach ethics, or ethics can be integrated into scenarios that also address a medical or nursing problem. The first type of simulation typically focuses on the ethical dilemmas that surface in particular clinical situations, and attention is primarily directed to the development of clinical ethics skills. Topics may include informed consent, confidentiality, truth telling, end-of-life care issues (withholding or withdrawing treatment, advance directives, "do-not-resuscitate" orders), and physician-assisted suicide, for example. Fleetwood, Novack and Templeton (2002) provide an example of this type of ethics simulation. Another type of simulation that primarily involves ethics addresses the "everyday" ethical dimensions of patient care. This type of simulation can be used to illustrate how ethics arises from the patientpractitioner interaction and comprises most clinical care decisions or situations. An example of an everyday ethics case is presented by Fry, Veach and Taylor (2011), which brings forth the ethics concerns embedded in the routine decision to turn and reposition a patient who is experiencing pain. The authors present this as a written case study, but a case such as this is easily formatted into a simulation, which provides students the opportunity to develop ethical awareness and sensitivity as they are placed in the role of the nurse.

A second type of simulation incorporates ethics as part of a scenario that initially presents a medically-oriented problem to solve. As the students resolve the medical issue, the simulation continues to evolve as an ethical issue is introduced. Gisondi et al. (2004) used this approach successfully with emergency medicine residents. The authors reported that the natural way in which the ethical issue was incorporated resulted in "unrehearsed" responses from the participants. This method would also be appropriate for nursing

students. Simulations that center on providing nursing care to resolve a problem can evolve with the addition of an ethical issue. The addition of ethical dimensions to medical and nursing scenarios may help to sensitize the learners to ethics and develop students and practitioners who routinely recognize ethical concerns and respond almost automatically (Gisondi et al. 2004). The unexpected introduction of an ethical problem reinforces a holistic care model and "de-compartmentalizes" ethical decision making in students.

There is a wide-range of ethics topics that are suitable for simulation. Some of these have been included in the literature review. As this dissertation has emphasized, it is important, however, that students have opportunities to practice exercising professional virtues and prudential reasoning. Therefore, the topics that are suggested in this section are broad and are applicable to both medical students and nursing students. Rather than a list of typical scenario topics, the following compilation provides a direction for the focus of simulation and includes reflection questions for educators to use to facilitate discussion. This approach not only stresses knowledge and behavioral skills that are important in students' education, but it also develops the moral comportment of students and assists in developing moral insight and moral agency. The suggestions for the focus of ethics simulations include:

Developing moral awareness and sensitivity. To accomplish this, students should be
placed in simulations that incite moral sensitivity, moral awareness and moral
engagement. Students must be challenged to recognize moral issues in clinical
situations that do not obviously present an ethical dimension. Simulation scenarios
can be designed to nurture the appropriate dispositions and attributes in students.

One example is a scenario that centers on applying fall precautions in a patient who is weak and lightheaded when getting out of bed, but who continues to try to move from bed independently (Fry, Veach, and Taylor 2011). The obvious issues in the case center on the patient's physical safety, yet there are also important considerations on the benefits and harms of the interventions that constrain an individual's independence. An emphasis on teaching and assessing for moral sensitivity should become an integral part of every simulation, without regard to its primary focus. Reflection questions can include: "Does this case have a moral dimension?" "What moral issues are present in this situation?" "Does this situation present any moral harms for the patient?" "What issues were easy for you to recognize?" "What ethical elements surprised you in this case?"

2. Prudence or practical reasoning. Because prudence is a primary virtue that directs the application of the moral virtues in the patient-practitioner relationship, it is critical to the deliberative processes particular to solving ethical problems. It is important, therefore, that all simulations emphasize its value. As Deveterre observes, prudence "clarifies the overall good we are aiming at for our lives, and it manages our feelings, behaviors, and habits in each situation" (2010, 32). It is a goal-oriented virtue that emphasizes *eudaimonia*, understood as well-being, flourishing or "good" and is essential in health care. Properly applied, prudence creates the sense of accomplishing the good for all who have a stake in the decisions. As noted in Chapter Four, prudence or practical wisdom is different from theoretical knowledge and does not follow a prescriptive or deductive approach that relies on principles,

rights and norms (Deveterre 2010). Simulation, therefore, with its complexity and open-ended methodology, becomes an excellent educational approach in which to practice this virtue. Simulations can promote prudential reasoning and the use of an inclusive, expansive framework that is goal-oriented toward achieving good or wellbeing and emphasize the virtues required to attain this goal. The debriefing and reflection component of simulation reinforces prudential reasoning and helps the learner to recognize which virtues to apply in a particular situation and review the appropriate action to take. Reflection questions can include: "What virtues are essential for the professional in this particular situation?" "What professional attributes are needed to successfully gather information?" "Did you work to convey respect, compassion, openness?" "What makes this particular situation unique and in what way does this affect the decision making process?" "Does the situation require balancing of harms and risks? "Are the standards and guidelines appropriate in this situation?" ""What conflicting values arise here "What mistakes did you make?" "What biases or preconceived ideas influenced your decision making process?" "How did the differences in values that you and the patient hold affect this process?" "What norms guided your decision making?" "What underlying core values influenced your perspective of the situation?" "How would you explain or justify your action?"

3. Respect of person. Students may be challenged to convey a respectful attitude in a simulation scenario in which the patient is a known drug abuser who the student suspects is seeking narcotics. Reflection questions can include: "What

preconceptions did you bring to the situation?" "What bias did you hold, and how did it affect your ability to convey respect for the person?" "How do you see other health care professionals behaving in similar situations?" "Can moral harm result from stigmatization or bias?"

- 4. Respect for autonomy. Students can be exposed to this fundamental principle through simulations that center on informed consent or decision making regarding care determinations. Reflection questions can include: "Why does patient autonomy play a central role in the delivery of health care?" "What professional virtues are important to express in protecting patient autonomy?" "How can your openness and sensitivity alert you to the other influences that affect the patient's decisions?"
- 5. Establishing and maintaining trust. This virtue can be taught using any case scenario, whether it focuses primarily on a patient's medical problem or on an ethics issue. Simulations that intend to help students develop trust are especially important with beginning students. Reflection questions can include: "What did you do in this simulation to establish trust?" "What are some signals that trust between the care provider and the patient has been achieved?" "How did your communication help or hinder this?" "Why is trust so vital in the patient-practitioner relationship?"
- 6. Caring and compassion. Simulations should be practiced with the goal to learn "good care" as described by Vanlaere, Coucke and Gastmans (2010, 325). Reflection questions can include: "How natural is it to care?" "How do you convey a caring

stance to a patient?" "What virtues and attitudes did you exercise to convey "good care?"

- 7. Empathy. Many simulated clinical situations require empathy on the part of the practitioner, such as end-of-life care situations or patients in physical or emotional pain. Reflection questions can include: "What professional virtues are important in this situation?" "How do you know if you are successfully relating an empathic stance?" "Whose perspective matters here?" "How can you verify that you are empathically conveying concern?"
- 8. Honesty and truth telling. Simulations can be designed in which the patient's request for information may produce harm. A case scenario concerning a young adult who presents to the emergency department with chest pain can serve as an example. Because cocaine use is a potential cause of chest pain in young patients without a history of cardiac disease, a urine toxicology screening test is routinely ordered. The healthcare professional is aware, however, that if the patient knows the reason for the urine specimen, the patient may refuse care or refuse to provide the specimen, which affects the quality of care. Other options include simulated situations in which the learner is conflicted in truth telling, torn between a sense of duty to the patient and duty to the hierarchy, whether professional or organizational. Reflection questions can include: "How do you know when to disclose information?" "Where is your obligation in a particular situation?" "Would there ever be a situation in which

it would be permissible to fabricate a 'white lie' to tell a patient in order to avoid some harm coming to that patient?'

- 9. Moral courage. Simulations can be designed that introduce difficult clinical situations in which students can practice effective communication skills and also gain the confidence needed to do the right thing. Specific simulations can be used, for example, where a subordinate witnesses some mistake or medical error on the part of a superior. Reflection questions include: "What support do you need to act with moral courage?" "How much evidence is necessary to blow the whistle on a superior when they might engage in questionable clinical behavior?" "Do you feel better having done the right thing?" "Why is it difficult to do the right thing?" "What barriers did you perceive?"
- 10. Advocacy. Simulations in which the learner must act on the patient's behalf to protect patient rights, provide benevolent care or prevent harm are appropriate. A simulation might include advocating for particular care options that involves approaching a senior staff member for a referral or plan of care that has been unreasonably dismissed. Reflection questions can include: "What are your obligations in this situation?" "Why are you obligated?" "What principles are at stake?" "What support do you need to be effective in this situation?"
- 11. Integrity. Students can learn to better understand this virtue through a simulation that requires them to follow through on a commitment to a patient. Reflection questions

can include: "What is integrity and what does it mean to have integrity in a medical setting?" "Did you follow through on your promise?" "What barriers affected your ability to act with integrity?"

- 12. Relationship-building practices. As Cassell notes, "Ethics is about relationships"; therefore, it is essential that students attend to their relationships skills in all clinical situations. Simulations should routinely include discussion and reflection on the nature of the patient-provider relationship as well as the influence of the patient's other relationships on his or her decisions. Reflection questions can include: "How can you improve your understanding of the impact that patient relationships have on their care and care determinations?" "Did your communication convey an appreciation for the patient's relationships with significant individuals?" "Who are the stakeholders in a patient's decision making?"
- 13. Everyday ethics. Simulations can be used to illustrate that ethics is an integral part of the patient-practitioner relationship and that all patient situations have an ethical dimension. Ethics teaching, therefore, is not limited to the more visible, dilemmaoriented issues. All simulations, even those designed to teach medical problem solving or procedural skills, have the potential to include ethics teaching and when viewed this way, can enhance the ethical sensitivity of students. The cases by Fry, Veach and Taylor represented earlier demonstrate everyday ethical issues.

- 14. Conflict of interest simulation can be used to present concrete conflicts of interest and also actions that are ambiguous (and legal), yet sit at the moral margins.
  Reflection questions include: "How are issues at the moral margins different from typical conflicts of interests?" "What virtues are necessary to recognize and avoid a conflict of interests?"
  - E. Final Considerations and Recommendations for Further Research

This chapter has demonstrated how simulation can be engaged in ethics education. Suggestions such as the above expand the possibilities for the ethics educator, moving away from the predominant model of dilemma-oriented and skill-oriented ethics education and toward a more holistic educational approach. Simulations that focus on dynamic ethical dilemmas are also important and as the literature review has illustrated, students value the opportunities to practice learning how to manage specific dilemmas that will arise in their practice. Educators are advised, though, to consider the above areas of focus when teaching about ethical dilemmas, moving from teaching that is skillfocused, to teaching that also emphasizes ethical sensitivity and professional virtues. Depending on the educator's objectives, simulation may be conducted to solely emphasize ethics or ethics can be incorporated into existing medical or nursing cases. Ethics educators are advised to use a holistic approach that supports the cognitive, technical (ethics skills) and moral apprenticeships. Simulation also lends itself to a holistic assessment of the effectiveness of teaching, particularly formative assessment. This dissertation has sought to examine and defend the value and efficacy of using simulation in the ethics education of medical and nursing education. The results of the analysis have shown that simulation is a relevant and effective means to achieve the goals of ethics education. While simulation presents some difficulties that are not encountered in traditional case analysis, it presents opportunities and benefits that oral or written analysis cannot offer. The benefits of experiential learning through simulation may not be obvious or easy to measure; however, anecdotal evidence suggests that the participant may transform learning into changes in professional behavior following simulated experiences.

As the literature review reveals, medical ethics curricula have included more simulated learning experiences than in nursing, primarily employing the use of standardized patient simulations that center on ethical dilemmas and professionalism. The nursing literature is in the early stages of ethics simulations. It is recommended, therefore, that nursing educators develop ethics curricula that engage ethics simulations in baccalaureate nursing curricula. This will require support for faculty development as well as financial support, not unlike what is currently required for simulation in general. One way to initially begin this process is to identify the ethical dimensions of existing nursing simulations, thus saving time on creating new ethics scenarios at the outset. Through debriefing and reflection activities, facilitators can engage students through the reflection questions to develop an awareness of the ethical dimensions present in many clinical situations. Another suggestion is for nurse educators to integrate an ethical issue into an existing simulation, much like the approach that Gisondi et al. (2004)

used, in which an ethical problem surfaces after the resolution of the dominant clinical problem. These suggestions allow ethics to be practiced in simulation without the need to create new scenarios. With more time and experience, educators can also create scenarios that focus on particular clinical ethics issues, such as confidentiality or advance directives, which provide practice in applying professional virtues and ethics skills to address specific issues. Resources are available through schools of medicine that have developed specific scenarios (see Fleetwood, Novack, and Templeton 2002). Ethics texts are another source for case scenarios (Fry, Veach, and Taylor 2011, for example). The case studies included in ethics texts can be adapted for simulations, which contribute a considerable curricular expense. Another option is to involve drama students. Drama students could be included as part of their curriculum work or as a service learning activity that contributes to the professional development of healthcare professionals.

The area of ethics simulation offers rich opportunities for further study, particularly related to ethics simulations that use an approach that emphasizes both professional virtues and skills. As the review of literature has shown, there is limited empirical research that has been conducted with ethics simulation. Recommendations for future study on ethics simulation include assessing its ability to enhance the student's awareness of the moral dimensions in clinical situations and its effectiveness in promoting the display of professional virtues in future simulated or actual clinical situations. Research can also be conducted on the effectiveness of ethics simulation to assist students in role transformation and to withstand the negative effects of the hidden curriculum, for example. As educators employ simulation more frequently to teach ethics,

it is hoped that students will more naturally recognize and respond appropriately to the ethical dimensions of patient care and competently address and resolve ethical problems. The work in this dissertation is a contribution to the attainment of this goal.

## **Reference List**

- Accreditation Council for Graduate Medical Education. 2011. Common Program Requirements. Accessed December 12, 2011. http://www.acgme.org/acwebsite/ home/common\_program\_requirements\_07012011.pdf
- Agich, George J. and Heidi Forster. 2000. "Conflicts of Interest and Management in Managed Care." *Cambridge Quarterly of Healthcare Ethics* 9 (02): 189.
- Agosta, Lou. 2011. "Empathy and Sympathy in Ethics." In *Internet Encyclopedia of Ethics*. Accessed Mar. 27, 2011. http://www.iep.utm.edu/emp-symp/#SH3b.
- Alinier, Guillaume, Barry Hunt, Ray Gordon, and Colin Harwood. 2006. "Effectiveness of Intermediate-Fidelity Simulation Training Technology in Undergraduate Nursing Education." *Journal of Advanced Nursing* 54 (3): 359-369. doi:10.1111/j.1365-2648.2006.03810.x.
- Allen, John, Louis Buffardi, and Robert Hays. 1991. "The Relationship of Simulator Fidelity to Task and Performance Variables." Report No. ARI-91-58. Alexandria VA: United States Army Research Institute for the Behavioral and Social Sciences.
- American Association of Colleges of Nursing. 2002. "Hallmarks of the Professional Nursing Practice Environment." Accessed February 26, 2012. http://www.aacn.nche.edu/publications/white-papers/ hallmarks-practice-environment.
- American Association of Colleges of Nursing. 2008. *The Essentials of Baccalaureate Education for Professional Nursing Practice*. American Association of Colleges of Nursing. Accessed April 2, 2011. http://www.aacn.nche.edu/education/pdf/ baccessentials08.pdf
- American Association of Critical-Care Nurses. 2005. "The 4A's to Rise Above Moral Distress." Accessed March 5, 2012. http://www.aacn.org/WD/Practice/Docs/ 4As\_to\_Rise\_Above\_Moral\_Distress.pdf.
- American College of Obstetricians and Gynecologists. 2011. "Empathy in Women's Health Care." *Obstetrics and Gynecology* 117: 756-61.
- American Medical Association. "Current Topics in Advocacy." American Medical Association. Accessed April 10, 2012. http://www.ama-assn.org/ama/pub/ advocacy/current-topics-advocacy.page
- American Medical Association. 2001. *Code of Medical Ethics*. American Medical Association. Accessed March 5, 2011. http://www.ama-assn.org/ama/pub/physician-resources/medical-ethics/code-medical-ethics.page?.

- American Nurses Association. "ANA Advises Federal Agencies." American Nurses Association, Inc. Accessed April 10, 2012. http://www.nursingworld.org/comments.
- American Nurses Association. 2001. Code of Ethics for Nursing with Interpretive Statements. Silver Spring, MD: Nursebooks.org.
- American Society for Bioethics and Humanities. 2009. ASBH Task Force on Ethics and Humanities Education in Undergraduate Medical Programs. Glenview, IL: American Society for Bioethics and Humanities.
  - —. 2006. Core Competencies for Health Care Ethics Consultation. Glenview, IL: American Society for Bioethics and Humanities.
- Anderson, David, Robert Cavalier, and Preston Covey. 1996. A Right to Die? The Dax Cowart Case. Cdr edition. Routledge. http://www.routledge.com/books/details/9780415917537/.
- Andreatta, Pamela, Ernest Saxton, Maureen Thompson, and Gail Annich. 2011.
   "Simulation-Based Mock Codes Significantly Correlate with Improved Pediatric Patient Cardiopulmonary Arrest Survival Rates." *Pediatric Critical Care Medicine: A Journal of the Society of Critical Care Medicine and the World Federation of Pediatric Intensive and Critical Care Societies* 12 (1): 33-38. doi:10.1097/PCC.0b013e3181e89270.
- Aristotle. 1941. "Nicomachean Ethics." In *The Basic Works of Aristotle*, edited by Richard McKeon, 935-1112. New York: Random House.
- Arnold, Robert M., Lachlan Forrow, Steven A. Wartman, and Joan Teno. 1988.
   "Teaching Clinical Medical Ethics: A Model Programme for Primary Care Residency." *Journal of Medical Ethics* 14 (2): 91-96.
- Ashcroft, Richard E. 2000. "Teaching for Patient-Centred Ethics." *Medicine, Health Care, and Philosophy* 3 (3): 287-295.
- Athanassoulis, Nafsika. 2004. "Virtue Ethics." *Internet Encyclopedia of Ethics*. Accessed April 20, 2012. http://www.iep.utm.edu/virtue/.
- Baer, Alan N., Jack P. Freer, David A. Milling, William R. Potter, Hillary Ruchlin, and Karen H. Zinnerstrom. 2008. "Breaking Bad News: Use of Cancer Survivors in Role-Playing Exercises." *Journal of Palliative Medicine* 11 (6): 885-892. doi:10.1089/jpm.2007.0253.
- Barnes, P. W., N. M. Arpante, V. C. Lewis, and A. Rosenfield. 1995. "Standards of Care in Reproductive Health Services." *The Western Journal of Medicine* 163 (3 Suppl): 28-32.

- Barrows, Howard S. and Stephen Abrahamson. 1964. "The Programmed Patient: A Technique for Appraising Student Performance in Clinical Neurology." *Journal* of Medical Education 39 (8): 802-803. Accessed June 30, 2012. http://journals.lww.com/academicmedicine/Citation/1964/08000/The\_Programme d\_Patient\_A\_Technique\_for\_Appraising.15.aspx.
- Baxter, Pamela E. and Sheryl L. Boblin. 2007. "The Moral Development of Baccalaureate Nursing Students: Understanding Unethical Behavior in Classroom and Clinical Settings." *The Journal of Nursing Education* 46 (1): 20-27.
- Beck, Rainer S., Rebecca Daughtridge, and Philip D. Sloane. 2002. "Physician-Patient Communication in the Primary Care Office: A Systematic Review." *The Journal of the American Board of Family Medicine* 15 (1): 25-38.
- Begley, A. M. 2006. "Facilitating the Development of Moral Insight in Practice: Teaching Ethics and Teaching Virtue." Nursing Philosophy: An International Journal for Healthcare Professionals 7 (4): 257-265. doi:10.1111/j.1466-769X.2006.00284.x.
- Benner, Patricia. 1998. "When Health Care Becomes a Commodity: The Need for Compassionate Strangers " In *The Changing Face of Health Care: A Christian Appraisal of Managed Care, Resource Allocation, and Patient-Caregiver Relationships*, edited by John F. Kilner, Robert D. Orr and Judith Allen Shelly, 119-135. Grand Rapids, Michigan: William B. Eerdmans Publishing Co.
- From Novice to Expert. Benner, Patricia. 2001. From Novice to Expert: Excellence and Power in Clinical Nursing Practice. Comm. ed. Menlo Park, CA: Addison-Wesley.

 2012. Pedagogical Uses of Lecture. Accessed May 30, 2012. http://www.educatingnurses.com/articles/pedagogically-sound-uses-of-lecture/.

- Benner, Patricia, Molly Sutphen, Victoria Leonard-Kahn, and Lisa Day. 2008.
   "Formation and Everyday Ethical Comportment." *American Journal of Critical Care: An Official Publication, American Association of Critical-Care Nurses* 17 (5): 473-476.
- Benner, Patricia, Molly Sutphen, Victoria Leonard, and Lisa Day. 2010. *Educating Nurses: A Call for Radical Transformation*. San Francisco: Jossey-Bass.
- Bickel, Janet. 1986. Integrating Human Values Teaching Programs into Medical Students' Clinical Education. Washington D.C.: American Association of Medical Colleges.

- Bowyer, Mark W., Janice L. Hanson, Elisabeth A. Pimentel, Amy K. Flanagan, Lisa M. Rawn, Anne G. Rizzo, E. Matthew Ritter, and Joseph O. Lopreiato. 2010.
  "Teaching Breaking Bad News using Mixed Reality Simulation." *The Journal of Surgical Research* 159 (1): 462-467. doi:10.1016/j.jss.2009.04.032.
- Brannan, Jane D., Anne White, and Judy Bezanson. 2008. "Simulator Effects on Cognitive Skills and Confidence Levels." *Journal of Nursing Education* 47 (11): 495-500.
- Brody, Jane K. 1988. "Virtue Ethics, Caring, and Nursing." *Scholarly Inquiry for Nursing Practice* 2 (2): 87-101.
- Brook, Robert H., Elizabeth A. McGlynn, and Paul G. Shekelle. 2000. "Defining and Measuring Quality of Care: A Perspective from US Researchers." *Journal for Quality in Health Care* 12 (4): 281-295.
- Brooks, David. 2010. "The Cognitive Revolution and Civic Life Today." Hastings Center, New York. http://www.youtube.com/watch?v=71P1EDxctao.
- Brown, Diane and Connie Chronister. 2009. "The Effect of Simulation Learning on Critical Thinking and Self-confidence when Incorporated into an Electrocardiogram Nursing Course." *Clinical Simulation in Nursing* 5 (1): e45e52. doi:10.1016/j.ecns.2008.11.001
- Brown, Kate H. and Diane Gillespie. 1997. ""We Become Brave by Doing Brave Acts": Teaching Moral Courage through the Theater of the Oppressed." *Literature and Medicine* 16 (1): 108-120.
- Brunero, Scott, Scott Lamont, and Melissa Coates. 2010. "A Review of Empathy Education in Nursing." *Nursing Inquiry* 17 (1): 65-74. doi:10.1111/j.1440-1800.2009.00482.x.
- Buck, George H. 1991. "Development of Simulators in Medical Education." *Gesnerus* 48 Pt 1: 7-28.
- Buckley, Sharon, Jamie Coleman, Ian Davison, Khalid S. Khan, Javier Zamora, Sadia Malick, David Morley, et al. 2009. "The Educational Effects of Portfolios on Undergraduate Student Learning: A Best Evidence Medical Education (BEME) Systematic Review. BEME Guide no. 11." *Medical Teacher* 31 (4): 282-298. doi:10.1080/01421590902889897.
- Burden, Amanda R. 2011. "Simulation in Anesthesiology." Anesthesiology News Special Edition October: 23-27. http://www.anesthesiologynews.com/ download/Simulation\_ANSE2011\_WM.pdf.

- Caldicott, Catherine V. and Marion Danis. 2009. "Medical Ethics Contributes to Clinical Management: Teaching Medical Students to Engage Patients as Moral Agents." *Medical Education* 43 (3): 283-289. doi:10.1111/j.1365-2923.2008.03277.x.
- Caldicott, Catherine V. and Kathy Faber-Langendoen. 2005. "Deception, Discrimination, and Fear of Reprisal: Lessons in Ethics from Third-Year Medical Students." *Academic Medicine* 80 (9): 866-873.
- Callahan, Daniel. 1980. "Goals in the Teaching of Ethics." In *Ethics Teaching in Higher Education*, edited by Daniel Callahan and Sissela Bok. New York: Plenum Press.
- Callister, Lynn Clark, Karlen E. Luthy, Pam Thompson, and Rae Jeanne Memmott. 2009. "Ethical Reasoning in Baccalaureate Nursing Students." *Nursing Ethics* 16 (4): 499-510. doi:10.1177/0969733009104612.
- Cameron, Miriam E., Marjorie Schaffer, and Hyeoun-Ae Park. 2001. "Nursing Students' Experience of Ethical Problems and Use of Ethical Decision-Making Models." *Nursing Ethics* 8 (5): 432-447.
- Campbell, Alastair V., Jacqueline Chin, and Teck-Chuan Voo. 2007. "How Can We Know That Ethics Education Produces Ethical Doctors?" *Medical Teacher* 29 (5): 431-436. doi:10.1080/01421590701504077.
- Campbell, S. M., M. O. Roland, and S. A. Buetow. 2000. "Defining Quality of Care." *Social Science & Medicine* 51 (11): 1611-1625. doi:10.1016/ S0277-9536(00)00057-5.
- Campbell, Suzanne Hetzel. 2007. "Clinical Simulation." In *Clinical Teaching Strategies in Nursing*, edited by Kathleen B. Gaberson and Marilyn H. Oermann, 123-140. New York, NY: Springer Publishing Co., LLC.
- Cant, Robyn P. and Simon J. Cooper. 2010. "Simulation-Based Learning in Nurse Education: Systematic Review." *Journal of Advanced Nursing* 66 (1): 3-15. doi:10.1111/j.1365-2648.2009.05240.x.
- Carpenito-Moyet, Lynda Juall. 2009. Nursing Care Plans and Documentation: Nursing Diagnosis and Collaborative Problems. 5th ed. Philadelphia: Wolters Kluwer Health/ Lippincott Williams and Wilkins.
- Cassell, Eric J. 2007. "Unanswered Questions: Bioethics and Human Relationships." *Hastings Center Report* 37 (5): 20-23.
- Chan, C. S., Y. T. Wun, A. Cheung, J. A. Dickinson, K. W. Chan, H. C. Lee, and Y. M. Yung. 2003. "Communication Skill of General Practitioners: Any Room for Improvement? How Much Can It Be Improved?" *Medical Education* 37 (6): 514-526.

- Charon, Rita. 2001. "Narrative Medicine: A Model for Empathy, Reflection, Profession, and Trust." *Journal of the American Medical Association* 286 (15): 1897-1902.
- Charon, Rita and Renee C. Fox. 1995. "Critiques and Remedies: Medical Students Call for Change in Ethics Teaching." *JAMA: The Journal of the American Medical Association* 274 (9): 767, 771.
- Chauhan, Getta and Ann Long. 2000. "Communication is the Essence of Nursing Care. 1: Breaking Bad News." *British Journal of Nursing (Mark Allen Publishing)* 9 (14): 931-938.
- Chen, Daniel, Robert Lew, Warren Hershman, and Jay Orlander. 2007. "A Cross-Sectional Measurement of Medical Student Empathy." *Journal of General Internal Medicine* 22 (10): 1434-1438. doi:10.1007/s11606-007-0298-x.
- Childs, Janis C., Susan B. Sepples, and Kristy Chambers. 2007. "Designing Simulations for Nursing Education." In *Simulation in Nursing Education*, edited by Pamela R. Jeffries, 35-58. New York, NY: National League for Nursing.
- Cohen, Mitchell J. M., Abigail Kay, James M. Youakim, and John M. Balacius. 2009. "Identity Transformation in Medical Students." *American Journal of Psychoanalysis* 69 (1): 43.
- Coles, Robert. 1995. "Medical Ethics and Living a Life." In *On Doctoring: Stories, Poems, Essays*, edited by Richard Reynolds and John Stone. 2nd ed., 280-286. New York: Simon and Schuster.
- Cook, David A., Patricia J. Erwin, and Marc M. Triola. 2010. "Computerized Virtual Patients in Health Professions Education: A Systematic Review and Meta-Analysis." Academic Medicine: Journal of the Association of American Medical Colleges 85 (10): 1589-1602. doi:10.1097/ACM.0b013e3181edfe13.
- Cooper, J. B. and V. R. Taqueti. 2004. "A Brief History of the Development of Mannequin Simulators for Clinical Education and Training." *Quality & Safety in Health Care* 13 (Suppl 1): i11-8. doi:10.1136/qhc.13.suppl\_1.i11.
- Cordingley, Lis, Catherine Hyde, Sarah Peters, Bryan Vernon, and Christine Bundy. 2007. "Undergraduate Medical Students' Exposure to Clinical Ethics: A Challenge to the Development of Professional Behaviours?" *Medical Education* 41: 1202-1209.
- Coulehan, Jack and Peter C. Williams. 2001. "Vanquishing Virtue: The Impact of Medical Education." *Academic Medicine* 76 (6): 598-605.
- Coutts, Mary Carrington. 1991. "Teaching Ethics in the Health Care Setting: Part 1: Survey of the Literature." *Kennedy Institute of Ethics Journal* 1 (2): 171-185.

- Crandall, Sonia J. and Gail S. Marion. 2009. "Commentary: Identifying Attitudes Towards Empathy: An Essential Feature of Professionalism." *Academic Medicine* 84 (9): 1174-1176. doi:10.1097/ACM.0b013e3181b17b11.
- Davidson, R. and M. E. Mills. 2005. "Cancer Patients' Satisfaction with Communication, Information and Quality of Care in a UK Region." *European Journal of Cancer* 14: 83-90.
- Decker, Sharon. 2007. "Simulation: Education and Ethics." In Simulation in Nursing Education: From Conceptualization to Evaluation, edited by Pamela R. Jeffries, 11-19. New York, N.Y.: National League of Nursing.
- Decker, Sharon, Susan Sportsman, Linda Puetz, and Lynda Billings. 2008. "The Evolution of Simulation and its Contribution to Competency." *The Journal of Continuing Education in Nursing* 39 (2): 74-80.
- Deladisma, Adeline M., Marc Cohen, Amy Stevens, Peggy Wagner, Benjamin Lok, Thomas Bernard, Christopher Oxendine, et al. 2007. "Do Medical Students Respond Empathetically to a Virtual Patient?" *The American Journal of Surgery* 193 (6): 756-760. doi:10.1016/j.amjsurg.2007.01.021.
- Devettere, Raymond J. 2010. *Practical Decision Making in Health Care Ethics*. 3rd ed. Washington, D.C.: Georgetown University Press.
- Devita, Michael A. 2009. "Society for Simulation in Healthcare Presidential Address, January 2009." *Simulation in Healthcare: Journal of the Society for Simulation in Healthcare* 4 (1): 43-48. doi:10.1097/SIH.0b013e318197d315.
- Dewey, John. 1933. *How We Think: A Restatement of the Relation of Reflective Thinking to the Educative Process.* New York: D. C. Heath and Co.
- Doane, Gweneth, Bernadette Pauly, Helen Brown and Gladys McPherson. 2004. "Exploring the Heart of Ethical Nursing Practice: Implications for Ethics Education." *Nursing Ethics* 11 (3): 240-253.
- Doron, Israel. 2007. "Court of Ethics: Teaching Ethics and Ageing by Means of Role-Playing." *Educational Gerontology* 33 (9): 737-758. doi:10.1080/03601270701364479.
- Doukas, David J., Laurence B. McCullough, and Stephen Wear. 2010. "Reforming Medical Education in Ethics and Humanities by Finding Common Ground with Abraham Flexner." *Academic Medicine* 85 (2): 318-323. doi:10.1097/ACM.0b013e3181c85932.
- Drane, James F. 1988. *Becoming a Good Doctor: The Place of Virtue and Character in Medical Ethics*. Kansas City, MO: Sheed and Ward.

- Dreifuerst, Kristina Thomas. 2009. "The Essentials of Debriefing in Simulation Learning: A Concept Analysis." *Nursing Education Perspectives* 30 (2): 109-114.
- Dreyfus, Hubert L. and Stuart E. Dreyfus. 1986. *Mind Over Machine: The Power of Human Intuition and Expertise in the Era of the Computer*. New York, NY: The Free Press.
- ———. 2009. "The Relationship of Theory and Practice in the Acquisition of Skill." In *Expertise in Nursing Practice: Caring, Clinical Judgment, and Ethics*, edited by Patricia Benner, Christine A. Tanner and Catherine A. Chesla. 2nd ed., 1-23. New York: Springer Publishing Co.
- Dreyfus, Stuart E. 1981. "Formal Models vs. Human Situational Understanding: Inherent Limitations on the Modeling of Business Expertise." Unpublished report supported by United States Air Force, Office of Scientific Research, Bolling AFB, Washington, DC.
- Dreyfus, Stuart E. and Hubert L. Dreyfus. 1980. "A Five-Stage Model of Mental Activities Involved in Directed Skill Acquisition." Unpublished report supported by United States Air Force, Office of Scientific Research, Bolling AFB, Washington, DC.
- Drought, Theresa. 2006. "The Application of Principle-based Ethics to Nursing Practice and Management: Implications for the Education of Nurses." In *Essentials of Teaching and Learning in Nursing Ethics: Perspectives and Methods*, edited by Anne J. Davis, Verena Tschudin and Louise De Raeve, 81-96. UK: Churchill Livingstone.
- DuBois, James M. and Jill Burkemper. 2002. "Ethics Education in U.S. Medical Schools: A Study of Syllabi." *Academic Medicine* 77 (5): 432-437.
- Dubose, Deatrah, Laurie D. Sellinger-Karmel, and Robert L. Scoloveno. 2010. "Baccalaureate Nursing." In *High-Fidelity Patient Simulation in Nursing*, edited by Wendy M. Nehring and Felissa R. Lashley, 189-209. Sudbury, MA: Jones and Bartlett Pub.
- Durham, Carol Fowler and Kathryn R. Alden, eds. 2008. "Enhancing Patient Safety in Nursing Education through Patient Simulation." In *Patient Safety and Quality: An Evidence-Based Handbook for Nurses*, edited by Rhonda G. Hughes. Rockville, MD: Agency for Healthcare Research and Quality. http://www.ncbi.nlm.nih.gov/books/NBK2628/#ch51.rl1.
- Eckles, Rachel E., Eric M. Meslin, Margaret Gaffney, and Paul R. Helft. 2005. "Medical Ethics Education: Where Are We? Where Should We Be Going? A Review." *Academic Medicine* 80 (12): 1143-1152.

- Edinger, Walter, JoDee Roberston, Joy Skeel, and Judith Schoonmaker. 1999. "Using Standardized Patients to Teach Clinical Ethics." *Medical Education Online* 4 (4): 1-5. Accessed Sept. 9, 2012. http://med-ed-online.net/index.php/meo /article/view/4306/4497.
- Epstein, Ronald M., Elaine F. Dannefer, Anne C. Nofziger, John T. Hansen, Stephen H. Schultz, Nicholas Jospe, Laura W. Connard, Sean C. Meldrum, and Lindsey C. Henson. 2004. "Comprehensive Assessment of Professional Competence: The Rochester Experiment." *Teaching and Learning in Medicine* 16 (2): 186-196. doi:10.1207/s15328015tlm1602 12.
- Ericsson, K. Anders. 2004. "Deliberate Practice and the Acquisition and Maintenance of Expert Performance in Medicine and Related Domains." *Academic Medicine: Journal of the Association of American Medical Colleges* 79 (10 Suppl): S70-81.
- Fenwick, T. (2001). "Experiential learning: A critical review of theory and practice." ERIC Clearinghouse on Adult, Career, and Vocational Education, Information Series No. 385. Columbus Ohio: ERIC. Accessed July 18, 2012. http://www.calpro-online.org/eric/textonly/mp\_fenwick\_01.asp.
- Fisher, Mark J., Erin A. Taylor, and Patricia L. High. 2012. "Parent-Nursing Student Communication Practice: Role-Play and Learning Outcomes." *The Journal of Nursing Education* 51 (2): 115-119. doi:10.3928/01484834-20111216-04; 10.3928/01484834-20111216-04.
- Fleetwood, Janet, Dennis Novack, and Bryce Templeton. 2002. "Bringing Medical Ethics to Life: An Educational Programme Using Standardised Patients." *Medical Education* 36 (11): 1100-1101.
- Fleetwood, Janet, Wayne Vaught, Debra Feldman, Edward Gracely, Zach Kassutto, and Dennis Novack. 2000. "MedEthEx Online: A Computer-Based Learning Program in Medical Ethics and Communication Skills." *Teaching & Learning in Medicine* 12 (2): 96-104. doi:10.1207/S15328015TLM1202 7.
- Fowler, Marsha D. and Verena Tschudin. 2006. "Ethics in Nursing: An Historical Perspective." In *Essentials of Teaching and Learning in Nursing Ethics: Perspectives and Methods*, edited by Anne J. Davis, Verena Tschudin and Louise De Raeve, 13-25. UK: Churchill Livingstone.
- Fowler, Marsha D. M., ed. 2008. *Guide to the Code of Ethics for Nurses: Interpretation and Application*. Silver Spring, MD: The American Nursing Association, Inc.
- Fox, Ellen, Robert M. Arnold, and Baruch Brody. 1995. "Medical Ethics Education: Past, Present, and Future." *Academic Medicine* 70 (9): 761-768.

- Fox, Ellen, Melissa M. Bottrell, Kenneth A. Berkowitz, Barbara L. Chanko, Mary Beth Foglia, and Robert A. Pearlman. 2010. "IntegratedEthics: An Innovative Program to Improve Ethics Quality in Health Care." *The Innovation Journal: The Public Sector Innovation Journal* 15 (2): article 8. http://www.innovation.cc/ scholarly-style/fox integrated8ethics 8 final.pdf.
- Fox, Ellen, Melissa Bottrell, Mary Beth Foglia, and Rebecca Stoeckle. "Preventive Ethics: Addressing Ethics Quality Gaps on a Systems Level." National Center for Ethics in Health Care, US Department of Veterans Affairs. Accessed August 11, 2012. http://www.ethics.va.gov/docs/integratedethics/
  Preventive\_Ethics\_Addressing\_Ethics\_Quality\_Gaps\_on\_a\_Systems\_Level\_2007 0808.pdf.
- Fry, Sara T. 1989. "Teaching Ethics in Nursing Curricula. Traditional and Contemporary Models." *The Nursing Clinics of North America* 24 (2): 485-497.
- 2004. "Nursing Ethics." In *Encyclopedia of Bioethics*, edited by Stephen G. Post.
   3rd ed. Vol. 4, 1898-1903. New York: Macmillan Reference USA. Gale
   Reference Library. Web. Accessed 20. August 2011.
- Fry, Sara T., Robert M. Veach, and Carol Taylor. 2011. Case Studies in Nursing Ethics. 4th ed. Sudbury, MA: Jones and Bartlett Learning.
- Gaba, David M. 2004. "The Future Vision of Simulation in Health Care." *Quality & Safety in Health Care* 13 (Suppl 1): i2-10. doi:10.1136/qhc.13.suppl\_1.i2.
- Gallagher, Ann. 2006. "The Teaching of Nursing Ethics: Content and Method. Promoting Ethical Competence." In *Essentials of Teaching and Learning in Nursing Ethics: Perspectives and Methods,* edited by Anne J. Davis, Verena Tschudin and Louise De Raeve, 223-240. UK: Churchill Livingstone.
- Gardner, Roxane and Daniel B. Raemer. 2008. "Simulation in Obstetrics and Gynecology." Obstetrics and Gynecology Clinics of North America 35 (1): 97-127. doi:10.1016/j.ogc.2007.12.008.
- Garrett, Joan. 2010. "Students Learn Medical Ethics through Forum Theatre Class." *Community College Week* 22 (17): 10-20.
- Gastmans, Chris, Bernadette Dierckx de Casterle, and Paul Schotsmans. 1998. "Nursing Considered as Moral Practice: A Philosophical-Ethical Interpretation of Nursing." *Kennedy Institute of Ethics Journal* 8 (1): 43-69. doi:10.1191/0969733002ne539oa.
- Gastmans, Chris. 2002. "A Fundamental Ethical Approach to Nursing: Some Proposals for Ethics Education." *Nursing Ethics* 9 (5): 494-507. doi:10.1191/0969733002ne539oa.

- Gelhaus, Petra. 2012. "The Desired Moral Attitude of the Physician: (I) Empathy." *Medicine, Health Care and Philosophy* 15: 103-113.
- Gillam, Lynn. 2009. "Teaching Ethics in the Health Professions " In *A Companion to Bioethics*, edited by Helga Kuhse and Peter Singer. 2nd ed. Malden, MA: Wiley-Blackwell.
- Gisondi, Michael A., Rebecca Smith-Coggins, Phillip M. Harter, Robert C. Soltysik, and Paul R. Yarnold. 2004. "Assessment of Resident Professionalism using High-Fidelity Simulation of Ethical Dilemmas." *Academic Emergency Medicine* 11 (9): 931-937.
- Gilligan, Carol. 1982. In a Different Voice: Psychological Theory and Women's Development. Cambridge, MA: Harvard University Press.
- Glick, Shimon M. 1994. "The Teaching of Medical Ethics to Medical Student." *The Journal of Medical Ethics* 20: 239-243.
- Göçmen, Doğan. 2007. The Adam Smith Problem. New York: Tauris Academic Studies.
- Goldie, John, Lisa Schwartz, Alex McConnachie, and Jillian Morrison. 2001. "Impact of a New Course on Students' Potential Behaviour on Encountering Ethical Dilemmas." *Medical Education* (35): 295-302.
- Goldie, John. 2000. "Review of Ethics Curricula in Undergraduate Medical Education." *Medical Education* 34: 108-119.
- Gooch, Paul W. 1987. "Irony and Insight in Plato's Meno." *Laval Théologique Et Philosophique* 43 (2): 189-204. Accessed June 1, 2012. http://www.erudit.org/revue/ltp/1987/v43/n2/400301ar.pdf.
- Gordon, Goeffrey H. and Susan W. Tolle. 1991. "Discussing Life-Sustaining Treatment. A Teaching Program for Residents." Archives of Internal Medicine 151 (3): 567-570.
- Gredler, Margaret E. 2004. "Games and Simulations and their Relationships to Learning." In *Handbook of Research on Educational Communications and Technology*, edited by David H. Jonassen. 2nd ed., 571-581. Mahwah, N. J.: Lawrence Erlbaum Assoc., Inc.
- Griggs, Roseanne Ruth. "The Effects of the use of a Human Patient Simulator on the Acquisition of Nursing Knowledge in Undergraduate Nursing Students at a University in Illinois." PhD diss., Southern Illinois University, 2003. ProQuest (3100754).

- Grol, Richard. 2001. "Improving the Quality of Medical Care: Building Bridges among Professional Pride, Payor Profit, and Patient Satisfaction." *JAMA: The Journal of the American Medical Association* 286 (20): 2578-2585.
- Grol, Richard and Michel Wensing. 1995. "Implementation of Quality Assurance and Medical Audit: General Practitioners' Perceived Obstacles and Requirements." *The British Journal of General Practice: The Journal of the Royal College of General Practitioners* 45 (399): 548-552.
- Gropelli, Theresa. "Enhanced Student Learning of Nursing Ethics Using Active Simulation." Paper presented at the 34<sup>th</sup> Annual International Conference, "Navigating Innovations in Teaching and Learning," Simon Fraser University, Vancouver, British Columbia, July 14-19, 2009.
- Guyatt, Gordon, Elie A. Akl, Jack Hirsh, Clive Kearon, Mark Crowther, David Gutterman, Sandra Zelman Lewis, Ian Nathanson, Roman Jaeschke, and Holger Schünemann. 2010. "The Vexing Problem of Guidelines and Conflict of Interest: A Potential Solution." *Annals of Internal Medicine* 152: 738-741. doi:10.1059/0003-4819-152-11-201006010-00254.
- Guyatt, Gordon H., Susan L. Norris, Sam Schulman, Jack Hirsh, Mark H. Eckman, Elie A. Akl, Mark Crowther, et al. 2012. "Methodology for the Development of Antithrombotic Therapy and Prevention of Thrombosis Guidelines: Antithrombotic Therapy and Prevention of Thrombosis, 9th Ed: American College of Chest Physicians Evidence-Based Clinical Practice Guidelines." *Chest* 141 (2 Suppl): 53S-70S. doi:10.1378/chest.11-2288.
- Haddad, Amy. 2010. "What Health Science Students Learn from Playing a Standardized Patient in an Ethics Course." *Cambridge Quarterly of Healthcare Ethics* 19 (4): 481-487. doi:10.1017/S096318011000037X.
- Hafferty, Frederic W. and Ronald Franks. 1994. "The Hidden Curriculum, Ethics Teaching, and the Structure of Medical Education." *Academic Medicine* 69 (11): 861-871.
- Hall, Judith A. and Michael C. Dornan. 1988. "What Patients Like about their Medical Care and How Often They are Asked: A Meta-Analysis of the Satisfaction Literature." Social Science & Medicine 27 (9): 935-939. doi:10.1016/0277-9536(88)90284-5.
- Hall, Judith A., Debra L. Roter, and Nancy R. Katz. 1988. "Meta-Analysis of Correlates of Provider Behavior in Medical Encounters." *Medical Care* 26 (7): 657-675.
- Hamric, Ann B. and Leslie Blackhall J. 2007. "Nurse-Physician Perspectives on the Care of Dying Patients in Intensive Care Units: Collaboration, Moral Distress, and Ethical Climate." *Critical Care Medicine* 35 (2): 422-429.

- Hanna, Michael and Joseph J. Fins. 2006. "Viewpoint: Power and Communication: Why Simulation Training Ought to be Complemented by Experiential and Humanist Learning." Academic Medicine: Journal of the Association of American Medical Colleges 81 (3): 265-270.
- Harden, V and R. M. Harden. 2003. "OSCE Annotated Bibliography with Contents Analysis: BEME Guide No 17." Accessed May 20, 2012. http://www2.warwick.ac.uk/ fac/med/beme/reviews/published/harden/ beme guide no 17 beme guide to the osce 2003.pdf..
- Hassenstab, Jason, Isabel Dziobek, Kimberley Rogers, Oliver T. Wolf, and Antonio Convit. 2007. "Knowing What Others Know, Feeling what Others Feel: A Controlled Study of Empathy in Psychotherapists." *The Journal of Nervous and Mental Disease* 195 (4): 277-281.
- Herrmann, Eleanor Krohn. 1981. "Mrs. Chase: A Noble and Enduring Figure." *The American Journal of Nursing* 81 (10): 1836. http://www.jstor.org/stable/3462728.
- ———. 2008. "Remembering Mrs. Chase. Before there were Smart Hospitals and Sim-Men, there was "Mrs. Chase"." NSNA IMPRINT: 52-55. Accessed June 26, 2012. http://www.nsna.org/Portals/0/Skins/NSNA/pdf/Imprint\_FebMar08\_Feat\_ MrsChase.pdf.
- Hertel, John P. and Barbara J. Millis. 2002. Using Simulations to Promote Learning in Higher Education. Sterling, VA: Stylus Pub.
- Hochberg, Mark S., Adina Kalet, Sondra Zabar, Elizabeth Kachur, Colleen Gillespie, and Russell S. Berman. 2010. "Can Professionalism be Taught? Encouraging Evidence." *The American Journal of Surgery* 199 (1): 86-93. doi:10.1016/j.amjsurg.2009.10.002.
- Hoffmann, Rosemary L., John M. O'Donnell, and Yookyung Kim. 2007. "The Effects of Human Patient Simulators on Basic Knowledge in Critical Care Nursing with Undergraduate Senior Baccalaureate Nursing Students." *Simulation in Healthcare: Journal of the Society for Simulation in Healthcare* 2 (2): 110-114. doi:10.1097/SIH.0b013e318033abb5.
- Hojat, Mohammadreza. 2009. "Ten Approaches for Enhancing Empathy in Health and Human Services Cultures." *Journal of Health and Human Services Administration* 31 (4): 412-450.
- Hojat, Mohammadreza, Daniel Z. Louis, Fred W. Markham, Richard Wender, Carol Rabinowitz, and Joseph S. Gonnella. 2011. "Physicians' Empathy and Clinical Outcomes for Diabetic Patients." *Academic Medicine: Journal of the Association* of American Medical Colleges 86 (3): 359-364. doi:10.1097/ACM.0b013e3182086fe1.

- Hojat, Mohammadreza, Michael J. Vergare, Kaye Maxwell, George Brainard, Steven K. Herrine, Gerald A. Isenberg, Jon Veloski, and Joseph S. Gonnella. 2009. "The Devil is in the Third Year: A Longitudinal Study of Erosion of Empathy in Medical School." *Academic Medicine* 84 (9): 1182-1191. doi:10.1097/ACM.0b013e3181b17e55.
- Hovancsek, Marcella T. 2007. "Using Simulations in Nursing Education." In Simulation in Nursing Education: From Conceptualization to Evaluation, edited by Pamela R. Jeffries, 2-7. New York, NY: National League for Nursing.
- Howard, Valerie Michele. "A Comparison of Educational Strategies for the Acquistion of Medical-Surgical Nursing Knowledge and Critical Thinking Skills: Human Patient Simulator vs. the Interactive Case Study." PhD diss., University of Pittsburgh, 2007. ProQuest (3270096).
- Hughes, Rhonda G., ed. 2008 (April). *Patient Safety and Quality: An Evidence-Based Handbook for Nurses*. Rockville, MD: Agency for Healthcare Research and Quality. http://www.ncbi.nlm.nih.gov/books/NBK2651/.
- Hunter, Kathryn M., Rita Charon, and John L. Coulehan. 1995. "The Study of Literature in Medical Education." Academic Medicine : Journal of the Association of American Medical Colleges 70 (9): 787-794.
- Illingworth, Susan. 2004. Approaches to Ethics in Higher Education. University of Leeds: Philosophical and Religious Studies Subject Centre, Learning and Teaching Support Network. Accessed May 28, 2012. http://prs.heacademy.ac.uk/projects/ethics/index.html.
- International Council of Nurses. 2006. *The International Code of Ethics for Nurses*. Geneva, Switzerland: International Council for Nurses. http://www.icn.ch/ about-icn/code-of-ethics-for-nurses/.
- Issenberg, S. Barry. 2006. "The Scope of Simulation-Based Healthcare Education." Simulation in Healthcare: Journal of the Society for Simulation in Healthcare 1 (4): 203-208. doi:10.1097/01.SIH.0000246607.36504.5a.
- Issenberg, S. Barry, William C. McGaghie, Emil R. Petrusa, David Lee Gordon, and Ross J. Scalese. 2005. "Features and Uses of High-Fidelity Medical Simulations that Lead to Effective Learning: A BEME Systematic Review." *Medical Teacher* 27 (1): 10-28. doi:10.1080/01421590500046924.
- Issenberg, S. Barry and Ross J. Scalese. 2008. "Simulation in Health Care Education." *Perspectives in Biology and Medicine* 51 (1): 31-46.
- Jaeger, Suzanne M. 2001. "Teaching Health Care Ethics: The Importance of Moral Sensitivity for Moral Reasoning." *Nursing Philosophy* 2 (2): 131-142.

- Jameton, Andrew. 1984. Nursing Practice: The Ethical Issues. Prentice-Hall: Englewood Cliffs, N.J.
- Jeffries, Pamela R. 2005. "A Framework for Designing, Implementing, and Evaluating Simulations Used as Teaching Strategies in Nursing." *Nursing Education Perspectives* 26 (2): 96-103.

———. 2007. Simulation in Nursing Education: From Conceptualization to Evaluation. New York: National League for Nursing.

- Jeffries, Pamela R. and Mary Anne Rizzolo. 2006. Designing and Implementing Models for the Innovative use of Simulation to Teach Nursing Care of Ill Adults and Children: A National, Multi-Site, Multi-Method Study. New York, N.Y.: National League for Nursing.
- Johnstone, Megan-Jane. 2009. *Bioethics: A Nursing Perspective*. 5th ed. New York: Churchill Livingstone.
- Jones, Ken. 1995. *Simulations: A Handbook for Teachers and Trainers*. 3rd ed. New Jersey: Nichols Publishing Company.
- Jonsen, Albert R. 2000. *A Short History of Medical Ethics*. New York: Oxford University Press.
- ———. 2007. "A History of Bioethics as Discipline and Discourse." In *Bioethics: An Introduction to the History, Methods, and Practice*, edited by Nancy S. Jecker, Albert R. Jonsen and Robert A. Pearlman, 3-16. 2nd ed. Sudbury, MA: Jones and Bartlett.
- Kaakinen, Joanna and Ellyn Arwood. 2009. "Systematic Review of Nursing Simulation Literature for use of Learning Theory." *International Journal of Nursing Education Scholarship* 6 (1): Article 16. doi:10.2202/1548-923X.1688.
- Kanter, Steven L., Paul F. Wimmers, and Arthur S. Levine. 2007. "In-Depth Learning: One School's Initiatives to Foster Integration of Ethics, Values, and the Human Dimensions of Medicine." *Academic Medicine* 82 (4): 405.
- Kelley, John M., Anthony J. Lembo, J. Stuart Ablon, Joel J. Villanueva, Lisa A. Conboy, Ray Levy, Carl D. Marci, et al. September 2009. "Patient and Practitioner Influences on the Placebo Effect in Irritable Bowel Syndrome." *Psychosomatic Medicine* 71 (7): 789-797. doi:10.1097/PSY.0b013e3181acee12.
- Ker, Jean and Paul Bradley. 2010. "Simulation in Medical Education." In Understanding Medical Education: Evidence, Theory and Practice, edited by Tim Swanwick, 164-180. Hoboken NJ: Wiley-Blackwell.

- Kim, Sung Soo, Stan Kaplowitz, and Mark V. Johnston. 2004. "The Effects of Physician Empathy on Patient Satisfaction and Compliance." *Evaluation and the Health Professions* 27: 237-251.
- Kleinman, Dawn E., Marvin L. Hage, Axalla J. Hoole, and Vicki Kowlowitz. 1996.
   "Pelvic Examination Instruction and Experience: A Comparison of Laywoman-Trained and Physician-Trained Students." *Academic Medicine: Journal of the Association of American Medical Colleges* 71 (11): 1239-1243.
- Kneebone, Roger. 2005. "Evaluating Clinical Simulations for Learning Procedural Skills: A Theory-Based Approach." Academic Medicine: Journal of the Association of American Medical Colleges 80 (6): 549-553.
- ———. 2010. "Simulation, Safety and Surgery." *Quality & Safety in Health Care* 19 (Suppl 3): i47-52. doi:10.1136/qshc.2010.042424.
- Kneebone, Roger, Jane Kidd, Debra Nestel, Suzanne Asvall, Paraskevas Paraskeva, and Ara Darzi. 2002. "An Innovative Model for Teaching and Learning Clinical Procedures." *Medical Education* 36 (7): 628-634.
- Knowles, Malcolm S. 1970. *The Modern Practice of Adult Education: Andragogy vs. Pedagogy*. New York: Cambridge Books.
- Kohn, Linda T., Janet M. Corrigan, and Molla S. Donaldson, eds. 2000. To Err is Human: Building a Safer Health System. Committee on Quality Care in America, Institute of Medicine. Washington, DC: National Academy Press. http://books.nap.edu/openbook.php?isbn=0309068371
- Kolb, David A. 1984. Experiential Learning. Englewood Cliffs, NJ: Prentice-Hall.
- Kopelman, Loretta M. 1999. "Values and Virtues: How Should They be Taught?" Academic Medicine: Journal of the Association of American Medical Colleges 74 (12): 1307-1310.
- Kyle, Gaye. 2008. "Using Anonymized Reflection to Teach Ethics: A Pilot Study." *Nursing Ethics* 15 (1): 6-16.
- Lachman, Vicki D. 2008. "Weighing Obligations & Virtues." Nursing 38 (10): 43.
- Laerdal Medical. "History in Brief." Laerdal Medical. Accessed June 26, 2012. http://www.laerdal.com/about/default.htm.
- Lakhan, Shaheen E., Elissa Hamlat, Turi McNamee, and Cyndi Laird. 2009. "Time for a Unified Approach to Medical Ethics." *Philosophy, Ethics, and Humanities in Medicine: PEHM* 4: 13. doi:10.1186/1747-5341-4-13.

- Lane, Claire and Stephen Rollnick. 2007. "The use of Simulated Patients and Role-Play in Communication Skills Training: A Review of the Literature to August 2005." *Patient Education and Counseling* 67 (1–2): 13-20. doi:10.1016/j.pec.2007.02.011.
- Lapkin, Samuel, Tracy Levett-Jones, Helen Bellchambers, and Ritin Fernandez. 2010.
   "Effectiveness of Patient Simulation Manikins in Teaching Clinical Reasoning Skills to Undergraduate Nursing Students: A Systematic Review." *Clinical Simulation in Nursing* 6 (6): e207-222. doi:10.1016/j.ecns.2010.05.005.
- Larew, Charles, Sherrie Lessans, Debra Spunt, Dawn Foster, and Barbara G. Covington. 2006. "Innovations in Clinical Simulation: Application of Benner's Theory in an Interactive Patient Care Simulation." *Nursing Education Perspectives* 27 (1): 16-21.
- Lasater, Kathie. 2007. "High-Fidelity Simulation and the Development of Clinical Judgment: Students' Experiences." *The Journal of Nursing Education* 46 (6): 269-276.
- Lehmann, Soleymani, Lisa, Willard S. Kasoff, Phoebe Koch, and Daniel D. Federman. 2004. "A Survey of Medical Ethics Education at U.S. and Canadian Medical Schools." *Academic Medicine* 79 (7): 682-689.
- Lempp, Heidi and Clive Seale. 2004. "The Hidden Curriculum in Undergraduate Medical Education: Qualitative Study of Medical Students' Perceptions of Teaching." *BMJ* (*Clinical Research Ed.*) 329 (7469): 770-773. doi:10.1136/bmj.329.7469.770.
- Liaison Committee on Medical Education. 2011. Functions and Structure of a Medical School: Standards for Accreditation of Medical Education Programs Leading to the M.D. Degree. Accessed July 29, 2011. http://www.lcme.org/ functions2011may.pdf.
- Lin, Chiou-Fen, Meei-Show Lu, Chun-Chih Chung, and Che-Ming Yang. 2010. "A Comparison of Problem-Based Learning and Conventional Teaching in Nursing Ethics Education." *Nursing Ethics* 17 (3): 373-382. doi:10.1177/0969733009355380.
- Litzelman, Debra K. and Ann H. Cottingham. 2007. "The New Formal Competency-Based Curriculum and Informal Curriculum at Indiana University School of Medicine: Overview and Five-Year Analysis." *Academic Medicine* 82 (4): 410-421.
- Lomis, Kimberly D., Robert O. Carpenter, and Bonnie M. Miller. 2009. "Moral Distress in the Third Year of Medical School; a Descriptive Review of Student Case Reflections." *The American Journal of Surgery* 197 (1): 107-112. doi:10.1016/j.amjsurg.2008.07.048.

MacIntyre, Alasdair. 2007. After Virtue. 3rd ed. Indiana: University of Notre Dame Press.

- Mahood, Sally C. 2011. "Medical Education: Beware the Hidden Curriculum." *Canadian Family Physician* 57: 983-985.
- Mann, Karen, Jill Gordon, and Anna MacLeod. 2009. "Reflection and Reflective Practice in Health Professions Education: A Systematic Review." *Advances in Health Sciences Education: Theory and Practice* 14 (4): 595-621. doi:10.1007/s10459-007-9090-2.
- Manson, Helen. 2008. "The Need for Medical Ethics Education in Family Medicine Training." *Family Medicine* 40 (9): 658-664.
- Maran, Nicola J. and R. J. Glavin. 2003. "Low- to High-Fidelity Simulation a Continuum of Medical Education?" *Medical Education* 37 (Suppl 1): 22-28.
- Mattick, Karen and J. Bligh. 2006. "Teaching and Assessing Medical Ethics: Where Are We Now?" *Journal of Medical Ethics* 32 (3): 181-185.
- May, Win, Joo Hyun Park, and Justin P. Lee. 2009. "A Ten-Year Review of the Literature on the use of Standardized Patients in Teaching and Learning: 1996-2005." *Medical Teacher* 31 (6): 487-492. doi:10.1080/01421590802530898.
- McElhinney, Thomas K. and Edmund D. Pellegrino. 2001. "The Institute on Human Values in Medicine: Its Role and Influence in the Conception and Evolution of Bioethics." *Theoretical Medicine* 22: 291-317.
- McEvoy, Michelle, Bryan Butler, and Geraldine MacCarrick. 2012. "Teaching Professionalism through Virtual Means." *The Clinical Teacher* 9 (1): 32-36. doi:10.1111/j.1743-498X.2011.00487.x.
- McGaghie, William C., S. Barry Issenberg, E. R. Cohen, J. H. Barsuk, and D. B. Wayne. 2011. "Does Simulation-Based Medical Education with Deliberate Practice Yield Better Results than Traditional Clinical Education? A Meta-Analytic Comparative Review of the Evidence." *Academic Medicine: Journal of the Association of American Medical Colleges* 86 (6): 706-711. doi:10.1097/ACM.0b013e318217e119.
- McGaghie, William C., S. Barry Issenberg, Emil R. Petrusa, and Ross J. Scalese. 2010.
   "A Critical Review of Simulation-Based Medical Education Research: 2003-2009." *Medical Education* 44 (1): 50-63. doi:10.1111/j.1365-2923.2009.03547.x.

- McGaghie, William C., Viva J. Siddall, Paul E. Mazmanian, and Janet Myers. 2009. "Lessons for Continuing Medical Education from Simulation Research in Undergraduate and Graduate Medical Education: Effectiveness of Continuing Medical Education: American College of Chest Physicians Evidence-Based Educational Guidelines." *Chest* 135 (3 Suppl): 62S-68S. doi:10.1378/ chest.08-2521.
- McGlynn, Elizabeth A. 1997. "Six Challenges in Measuring the Quality of Health Care." *Health Affairs* 16 (3): 7-21.
- McGuffey, William Holmes. *McGuffey's First Eclectic Reader, Revised Edition*. Accessed May 20, 2012. http://www.gutenberg.org/cache/ epub/14640/pg14640.html
- McKenna, Jocelyne and H. David Rosen. 2012. "Competency-Based Professionalism in Anesthesiology: Continuing Professional Development." *Canadian Journal of Anaesthesia = Journal Canadien d'Anesthesie* 59 (9): 889-908. doi:10.1007/s12630-012-9747-z.
- McKie, Andrew, Fiona Baguley, Caitrian Guthrie, Carol Jackson, Pamela Kirkpatrick, Adele Laing, Stephen O'Brien, Ruth Taylor, and Peter Wimpenny. 2012.
  "Exploring Clinical Wisdom in Nursing Education." *Nursing Ethics* 19 (2): 252-267. doi:10.1177/0969733011416841.
- Meissner, Karin, Ulrike Bingel, Luana Colloca, Tor D. Wager, Alison Watson, and Magne Arve Flaten. 2011. "The Placebo Effect: Advances from Different Methodological Approaches." *The Journal of Neuroscience* 31 (45): 16117-16124.
- Mercer, Stewart W. and William J. Reynolds. 2002. "Empathy and Quality of Care." *The British Journal of General Practice: The Journal of the Royal College of General Practitioners* 52 (Suppl): S9-12.
- Merril, Gregory L. and Virginia L. Barker. 1996. "Virtual Reality Debuts in the Teaching Laboratory in Nursing." *Journal of Intravenous Nursing: The Official Publication* of the Intravenous Nurses Society 19 (4): 182-187.
- Metersky, Mark L. and Ian Nathanson. 2012. "Introducing the Future of ACCP Clinical Practice Guidelines." *Chest* 141 (2): 285-286.
- Mezirow, Jack. 1991. *Transformative Dimensions of Adult Learning*. San Francisco: Jossey-Bass.
- Miaoulis, George, Jr., Jonathan Gutman, and Margaret M. Snow. 2009. "Closing the Gap: The Patient-Physician Disconnect." *Health Marketing Quarterly* 26 (1): 56-68.

- Miles, Steven, H., Laura Weiss Lane, Janet Bickel, Robert Walker, and Christine Cassel. 1989. "Medical Ethics Education: Coming of Age." *Academic Medicine* 64 (12): 705-714.
- Morse, Catherine Jean. 2012. "Debriefing After Simulated Patient Experiences." In Human Simulation for Nursing and Health Professions, edited by Linda Wilson and Leland Rockstraw, 58-68. New York: Springer Publishing Co.
- National Center for Ethics in Health Care. Sept. 4, 2012. "Integrated Ethics: Improving Ethics Quality in Health Care." US Department of Veterans Affairs. http://www.ethics.va.gov/integratedethics/.
  - ——. Sept. 4, 2012. "Preventive Ethics: Addressing Ethics Quality Gaps on a Systems Level." US Department of Veterans Affairs. http://www.ethics.va.gov/integratedethics/pec.asp.
  - ——. 2010. The ISSUES Approach Pocket Card. US Department of Veterans Affairs. http://www.ethics.va.gov.authenticate.library.duq.edu/docs/integratedethics/Prevent ive\_Ethics\_ISSUES\_Pocket\_Card--20070228.pdf.
- National Council of State Boards of Nursing. 2005. "Clinical Instruction in Prelicensure Nursing Programs." Accessed June 21, 2012. https://www.ncsbn.org/Final Clinical Instr Pre Nsg programs.pdf.
- National League for Nursing. *Simulation Innovation Resource Center*. Accessed May 20, 2012. http://sirc.nln.org/mod/glossary/view.php?id=183& mode=letter&hook=S&sortkey=&sortorder=.
- Naylor, C. David. 1995. "Grey Zones of Clinical Practice: Some Limits to Evidence-Based Medicine." *The Lancet* 345 (8953): 840-842. doi:10.1016/S0140-6736(95)92969-X.
- Nehring, Wendy M. 2010. "History of Simulation in Nursing." In *High-Fidelity Patient Simulation*, edited by Wendy M. Nehring and Felissa R. Lashley, 3-26. Boston: Jones and Bartlett Publishers.
- Nehring, Wendy and Felissa R. Lashley. 2009. "Nursing Simulation: A Review of the Past 40 Years." *Simulation and Gaming* 40 (4): 528-552.
- Noddings, Nel. 1984. Caring: A Feminine Approach to Ethics and Moral Education. Los Angeles: University of California Press.
  - —. 2002. *Educating Moral People: A Caring Alternative to Character Education*. New York, NY: Teachers College Press.

- Oberle, Kathleen and Dorothy Hughes. 2001. "Doctors' and Nurses' Perceptions of Ethical Problems in End-of-Life Decisions." *Journal of Advanced Nursing* 33 (6): 707.
- Okuda, Yasuharu, Ethan O. Bryson, Samuel DeMaria Jr, Lisa Jacobson, Joshua Quinones, Bing Shen, and Adam I. Levine. 2009. "The Utility of Simulation in Medical Education: What is the Evidence?" *The Mount Sinai Journal of Medicine, New York* 76 (4): 330-343. doi:10.1002/msj.20127.
- Onori, Michael J., Fabien Pampaloni, and Nina Multak. 2012. "What is a Standardized Patient?" In *Human Simulation for Nursing and Health Professions*, edited by Linda Wilson and Leland Rockstraw, 24-27. New York, N.Y.: Springer Publishing Co.
- Oxley, Julinna C. 2011. The Moral Dimensions of Empathy: Limits and Applications in Ethical Theory and Practice. New York: Palgrave Macmillan.
- Page, Ray L. "Brief History of Flight Simulation." 2000. In SimTecT 2000 Proceedings, SimtechT 2000 Organizing and Technical Committee, Simulation Conference and Exhibition, Simulation Australia, Sydney. http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.132.5428&rep =rep1&type=pdf.
- Papadakis, M. A., C. S. Hodgson, A. Teherani, and N. D. Kohatsu. 2004.
  "Unprofessional Behavior in Medical School is Associated with Subsequent Disciplinary Action by a State Medical Board." *Academic Medicine* 79 (3): 244-249.
- Parker, Malcolm. 1995. "Autonomy, Problem-Based Learning, and the Teaching of Medical Ethics." *Journal of Medical Ethics* 21 (5): 305-310.
- Passiment, Morgan, Heather Sachs, and Grace Huang. 2011. *Medical Simulation in Medical Education: Results of an AAMC Survey*. Washington, DC: Association of American Medical Colleges. https://www.aamc.org/download/259760/data/ medicalsimulationinmedicaleducationanaamcsurvey.pdf.
- PBS Frontline. "The Suicide Tourist." WGBH Educational Foundation. Accessed January 8, 2011. http://www.pbs.org/wgbh/pages/frontline/suicidetourist/.
- Pellegrino, Edmund D. 1995. "Toward a Virtue-Based Normative Ethics for the Health Professions." *Kennedy Institute of Ethics Journal* 5 (3): 253-277.
  - —. 1998. "The Good Samaritan in the Marketplace: Managed Care's Challenge to Christian Charity " In *The Changing Face of Health Care: A Christian Appraisal* of Managed Care, Resource Allocation, and Patient-Caregiver Relationships, edited by John F. Kilner, Robert D. Dorr and Judith Allen Shelly, 103-118. Grand Rapids, Michigan: Wm. B. Eerdmans Publishing Co.

——. 1999. "The Origins and Evolution of Bioethics: Some Personal Reflections." Kennedy Institute of Ethics Journal 9 (1): 73-88.

- —. 2008a. "Character, Virtue, and Self-Interest in the Ethics of the Professions." In *The Philosophy of Medicine Reborn: A Pellegrino Reader*, edited by H. Tristram Engelhardt Jr. and Fabrice Jotterand, 231-254. Notre Dame, Indiana: University of Notre Dame Press.
- 2008b. "From Medical Ethics to a Moral Philosophy of the Professions." In *The Philosophy of Medicine Reborn: A Pellegrino Reader*, edited by H. Tristram Engelhardt Jr. and Fabrice Jotterand, 147-160. Notre Dame, Indiana: University of Notre Dame Press.
- Pellegrino, Edmund D., Richard J. Hart, Sharon R. Henderson, Stephen E. Loeb, and Gary Edwards. 1985. "Relevance and Utility of Courses in Medical Ethics." *JAMA: The Journal of the American Medical Association* 253: 49-53.
- Pellegrino, Edmund D., Mark Siegler, and Peter A. Singer. 1990. "Teaching Clinical Ethics." *Journal of Clinical Ethics* 1 (3): 175-180.
- Pellegrino, Edmund D. and David C. Thomasma. 1993. *The Virtues in Medical Practice*. New York: Oxford University Press.
- Pendry, Patricia S. 2007. "Moral Distress: Recognizing it to Retain Nurses." *Nursing Economic*\$ 25 (4): 217-221.
- Penn State Milton S. Hershey Medical Center. *Penn State Hershey Simulation Center*. Accessed May 20, 2012. http://www.pennstatehershey.org/web/ simulation/home.
- Perlman, David. 2008. "Experiential Ethics Education: One Successful Model of Ethics Education for Undergraduate Nursing Students in the United States." Monash Bioethics Review 27 (1-2): 9-32.
- Plato. 1949. *Meno*. Translated by Benjamin Jowett. New York: The Library of Liberal Arts.
- Platt, Frederic W. and Constance M. Platt. 1998. "Empathy: A Miracle or Nothing at All." *Journal of Clinical Outcomes Management* 5 (2): 30-33.
- Prior, William J. 2007. "Moral Philosophy and Moral Cultivation." In *Moral Cultivation: Essays on the Development of Character and Virtue*, edited by Brad K. Wilburn, 49-67. Lanham, MD: Lexington Books.

Project Gutenberg eBook. Accessed March 20, 2012. http://www.gutenberg.org/.

- Radhakrishnan, Kavita, Joan P. Roche, and Helene Cunningham. 2007. "Measuring Clinical Practice Parameters with Human Patient Simulation: A Pilot Study." *International Journal of Nursing Education Scholarship* 4 (8): 1-11. doi:10.2202/1548-923X.1307.
- Rauen, Carol A. 2001. "Using Critical Thinking to Teach Critical Thinking Skills: You Just Can't Throw the Book at Them." *Critical Care Nursing Clinics of North America* 13 (1): 93-103.
- Ravert, Patricia. 2008. "Patient Simulator Sessions and Critical Thinking." *Journal of Nursing Education* 47 (12): 557-562.
- Rehmann, Albert J. 1995. "A Handbook of Flight Simulation Fidelity Requirements for Human Factors Research." DOT/FAA/CT-TN95/46. Springfield, VA: National Technical Information Service. Accessed July 9, 2012. http://ntl.bts.gov/lib/000/800/858/tn95\_46.pdf
- Reynolds, William J. and B. Scott. 1999. "Empathy: A Crucial Component of the Helping Relationship." *Journal of Psychiatric & Mental Health Nursing* 6 (5): 363-370. http://search.ebscohost.com/login.aspx? direct=true&db=afh&AN=5583038&site=ehost-live.
- Rosoff, Philip M. 2011. "Is Medical Ethics Education Effective?" *Le Journal Medical Libanais. the Lebanese Medical Journal* 59 (1): 12-16.
- Roter, Debra L., Judith A. Hall, David E. Kern, L. Randol Barker, Karan A. Cole, and Robert P. Roca. 1995. "Improving Physicians' Interviewing Skills and Reducing Patients' Emotional Distress. A Randomized Clinical Trial." *Archives of Internal Medicine* 155 (17): 1877-1884.
- Rourke, Liam, Megan Schmidt, and Neera Garga. 2010. "Theory-Based Research of High Fidelity Simulation Use in Nursing Education: A Review of the Literature." *International Journal of Nursing Education Scholarship* 7 (1): Article11. doi:10.2202/1548-923X.1965.
- Saewart, Karen J. and Leland J. Rockstraw. 2012. "Development of Evaluation Measures for Human Simulation: The Checklist." In *Human Simulation for Nursing and Health Professions*, edited by Linda Wilson and Leland J. Rockstraw, 28-36. New York: Springer Publishing.
- Sandars, John. 2009. "The Use of Reflection in Medical Education: AMEE Guide No. 44." *Medical Teacher* 31: 685-695.
- Sanford, Pamela G. 2010. "Simulation in Nursing Education: A Review of the Research." *The Qualitative Report* 15 (4): 1006-1011.

- Satava, Richard M. 2009. "The Revolution in Medical Education—The Role of Simulation." *Journal of Graduate Medical Education* 1 (2): 172-175.
- Savitt, Todd L., ed. 2002. *Medial Readers' Theater: A Guide and Scripts*. 2nd ed. Iowa City: Univ. of Iowa Press.
  - ——. 2010. "Medical Readers' Theater as a Teaching Tool." *Cambridge Quarterly of Healthcare Ethics* 19: 465-470.
- Schoenly, Lorry. 1994. "Teaching in the Affective Domain." *Journal of Continuing Education in Nursing* 25 (5): 209-212.
- Schön, Donald A. 1983. *The Reflective Practitioner: How Professionals Think in Action*. New York: Basic Books, Inc.
- ——. 1987a. Educating the Reflective Practitioner. New York: Basic Books.
- ———. 1987b. "Educating the Reflective Practitioner." Presentation to the 1987 Meeting of the American Educational Research Association, Washington, DC. Transcribed by Jan Carrick, January 1998. Accessed July 22, 2012. http://post.queensu.ca/ ~russellt/howteach/schon87.htm
- Schumacher, Lori Beth. "The Impact of Utilizing High-Fidelity Computer Simulation on Critical Thinking Abilities and Learning Outcomes in Undergraduate Nursing Students." PhD diss., Duquesne University, 2004. ProQuest (3151333).
- Seiler, Stephanie N., Bradley J. Brummel, Kerri L. Anderson, Kyoung Jin Kim, Serena Wee, C. K. Gunsalus, and Michael C. Loui. 2011. "Outcomes Assessment of Role-Play Scenarios for Teaching Responsible Conduct of Research." *Accountability in Research* 18 (4): 217-246. doi:10.1080/08989621.2011.584760.
- Self, Donnie J., Fredric D. Wolinsky, and DeWitt C. Baldwin Jr. 1989. "The Effect of Teaching Medical Ethics on Medical Students' Moral Reasoning." Academic Medicine 64 (12): 755-759.
- Sellman, Derek. 2007. "On Being of Good Character: Nurse Education and the Assessment of Good Character." *Nurse Education Today* 27: 762-767. doi:10.1016/j.nedt.2006.10.009
- Serratt, Teresa, Charlene Harrington, Joanne Spetz, and Mary Blegen. 2011. "Staffing Changes Before and After Mandated Nurse-to-Patient Ratios in California's Hospitals." *Policy, Politics, & Nursing Practice* 12 (3): 133-140. doi:10.1177/1527154411417881.
- Shaneyfelt, Terrence, M. 2001. "Building Bridges to Quality." JAMA: The Journal of the American Medical Association 286 (20): 2600-2601.

- Sheldon, Lisa Kennedy. 2011. "An Evidence-Based Communication Skills Training Programme for Oncology Nurses Improves Patient-Centred Communication, Enhancing Empathy, Reassurance and Discussion of Psychosocial Needs." *Evidence Based Nursing* 14 (3): 87-88. doi:10.1136/ebn1156.
- Shinnick, Mary Ann, Mary A. Woo, and Janet C. Mentes. 2011. "Human Patient Simulation: State of the Science in Prelicensure Nursing Education." *The Journal* of Nursing Education 50 (2): 65-72. doi:10.3928/01484834-20101230-01.
- Shulman, Lee S. 2005a. "Signature Pedagogies in the Professions." *Daedalus* 134 (3): 52-59.
- .2005b. "The Signature Pedagogies of the Professions of Law, Medicine, Engineering, and the Clergy: Potential Lessons for the Education of Teachers." Presentation, Irvine, CA, February 6-8, 2005.
- Shumway, James M. and Ronald M. Harden. 2003. "AMEE Guide no. 25: The Assessment of Learning Outcomes for the Competent and Reflective Physician." *Medical Teacher* 25 (6): 569-584.
- Siegler, Mark. 2001. "Lessons from 30 Years of Teaching Clinical Ethics." Virtual Mentor 3 (10). Accessed July 27, 2011. http://virtualmentor.ama-assn.org/ 2001/10/medu1-0110.html.
- Singer, Peter A., Robert Cohen, Anja Robb, and A. Rothman. 1993. "The Ethics Objective Structured Clinical Examination." *Journal of General Internal Medicine* 8 (1): 23-28
- Singer, Peter A., Anja Robb, Robert Cohen, G. Norman, and Jeffrey Turnbull. 1996. "Performance-Based Assessment of Clinical Ethics using an Objective Structured Clinical Examination." *Academic Medicine* 71 (5): 495-498.
- Sixma, Herman J., Jan J. Kerssens, Crétien van Campen, and Loe Peters. 1998. "Quality of Care from the Patients' Perspective: From Theoretical Concept to a New Measuring Instrument." *Health Expectations* 1 (2): 82-95.
- Smith, Adam. 1790. *The Theory of Moral Sentiments*. 6th ed. London: A. Millar. http://www.econlib.org/library/Smith/smMSCover.html.
- Smith, Katharine V., Jacki Witt, JoAnn Klaassen, Christine Zimmerman, and An-Lin Cheng. 2012. "High-Fidelity Simulation and Legal/Ethical Concepts: A Transformational Learning Experience." *Nursing Ethics* 19 (3): 390-398. doi:10.1177/0969733011423559.
- Smith, Patrick. 2010. "Medical Ethics as Theatre." Chattanooga Times Free Press. Accessed May 28, 2012. http://www.timesfreepress.com/news/2010/ mar/07/video-medical-ethics-theatre/.

- Smith, Sherilyn, Kelly Fryer-Edwards, Douglas S. Diekema, and Clarence H. Braddock, 3rd. 2004. "Finding Effective Strategies for Teaching Ethics: A Comparison Trial of Two Interventions." *Academic Medicine* 79 (3): 265-271.
- Smith, Sherrill J. and Carol J. Roehrs. 2009. "High-Fidelity Simulation: Factors Correlated with Nursing Student Satisfaction and Self-Confidence." *Nursing Education Perspectives* 30 (2): 74-78.
- Solomon, Patricia. 2011. "Problem-Based Learning." In *Innovative Teaching Strategies in Nursing and Related Health Professions*, edited by Martha J. Bradshaw and Arlene J. Lowenstein. 5th ed., 137-145. Boston: Jones and Bartlett.
- Southgate, L. J., S. R. Heard, P. D. Toon, and M. R. Salkind. 1987. "A Student-Led Approach to Teaching." *Journal of Medical Ethics* 13 (3): 139-143.
- Squier, Roger W. 1990. "A Model of Empathic Understanding and Adherence to Treatment Regimens in Practitioner-Patient Relationships." *Social Science & Medicine* 30 (3): 325-339.
- Stepien, Kathy A. and Amy Baernstein. 2006. "Educating for Empathy. A Review." Journal of General Internal Medicine 21 (5): 524-530.
- Stewart, Moira, Judith Belle Brown, Allan Donner, Ian R. McWhinney, Julian Oates, W. Wayne Weston, and John Jordan. 2000. "The Impact of Patient-Centered Care on Outcomes." *The Journal of Family Practice* 49 (9): 796-804.
- Stoddard, Hugh, A. and Toby Schonfeld. 2011. "A Comparison of Student Performance between Two Instructional Delivery Methods for a Healthcare Ethics Course." *Cambridge Quarterly of Healthcare Ethics* 20 (3): 493-501.
- Storch, Janet L. 2009. "Ethics in Nursing Practice." In A Companion to Bioethics, edited by Helga Kuhse and Peter Singer. 2nd ed., 551-562. Malden, MA: Wiley-Blackwell.
- Storch, Janet L. and Nuala Kenny. 2007. "Shared Moral Work of Nurses and Physicians." *Nursing Ethics* 14 (4): 478-491.
- Street, Richard L., Jr., Gregory Makoul, Neeraj K. Arora, and Ronald M. Epstein. 2009.
  "How does Communication Heal? Pathways Linking Clinical-Patient Communication to Health Outcomes." *Patient Education & Counseling* 74: 295-301.
- Sulmasy, Daniel P. and Eric S. Marx. 1997. "Ethics Education for Medical House Officers: Long-Term Improvements in Knowledge and Confidence." *Journal of Medical Ethics* 23 (2): 88-92.
- Swanwick, Tim. 2010. Understanding Medical Education: Evidence, Theory and Practice. London: Wiley-Blackwell.

- ten Have, Henk. 2008. "Unesco's Ethics Education Programme." *Journal of Medical Ethics* 34: 57-59.
- The Joint Commission. 2010. Advancing Effective Communication, Cultural Competence, and Patient- and Family-Centered Care: A Roadmap for Hospitals. Oakbrook Terrace, IL: The Joint Commission. http://www.jointcommission.org/ assets/1/6/aroadmapforhospitalsfinalversion727.pdf.
- Tsai, Tsuen-Chiuan and Peter H. Harasym. 2010. "A Medical Ethical Reasoning Model and its Contributions to Medical Education." *Medical Education* 44 (9): 864-873.
- Turner, Judith A., Richard A. Deyo, John D. Loeser, Michael Von Korff, and Wilbert E. Fordyce. 1994. "The Importance of the Placebo Effects in Pain Treatment and Research." *Journal of the American Medical Association* 271 (20): 1609-1614.
- Ulrich, Connie M., Carol Taylor, Karen Soeken, Patricia O'Donnell, Adrienne Farrar, Marion Danis, and Christine Grady. 2010. "Everyday Ethics: Ethical Issues and Stress in Nursing Practice." *Journal of Advanced Nursing* 66 (11): 2510-2519. doi:10.1111/j.1365-2648.2010.05425.x.
- UNESCO. 2008. *Bioethics Core Curriculum: Section 1: Syllabus, Ethics Education Programme.* UNESCO. http://unesdoc.unesco.org/images/0016/001636/ 163613e.pdf.
- Van De Camp, Kalinka, Myrra J. F. J. Vernooij-Dassen, Richard P. T. M. Grol, and Ben J. A. M. Bottema. 2004. "How to Conceptualize Professionalism: A Qualitative Study." *Medical Teacher* 26 (8): 696-702. doi:10.1080/01421590400019518.
- van Mook, Walther N. K. A., Willem S. de Grave, Scheltus J. van Luijk, Helen O'Sullivan, Valerie Wass, Lambert W. Schuwirth, and Cees P. M. van der Vleuten. 2009. "Training and Learning Professionalism in the Medical School Curriculum: Current Considerations." *European Journal of Internal Medicine* 20 (4): e96-e100. doi:10.1016/j.ejim.2008.12.006.
- van Zanten, Marta, John R. Boulet, John J. Norcini, and Danette McKinley. 2005. "Using a Standardised Patient Assessment to Measure Professional Attributes." *Medical Education* 39 (1): 20-29. doi:10.1111/j.1365-2929.2004.02029.x.
- Vanlaere, Linus and Chris Gastmans. 2007. "Ethics in Nursing Education: Learning to Reflect on Care Practices." *Nursing Ethics* 14 (6): 758-766. doi:10.1177/0969733007082116.
- Vanlaere, Linus, Trees Coucke, and Chris Gastmans. 2010. "Experiential Learning of Empathy in a Care-Ethics Lab." *Nursing Ethics* 17 (3): 325-336. doi:10.1177/0969733010361440.

- Vaughan, Rose Hélène. 1935. *The Actual Incidence of Moral Problems in Nursing: A Preliminary Study in Empirical Ethics*. Washington, D.C.: Catholic University Press.
- Veach, Robert M. and Sharmon Sollitto. 1976. "Medical Ethics Teaching: Report of a National Medical School Survey." JAMA: The Journal of the American Medical Association 235 (10): 1030-1033.
- Vozenilek, John, J. Stephen Huff, Martin Reznek, and James A. Gordon. 2004. "See One, do One, Teach One: Advanced Technology in Medical Education." Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine 11 (11): 1149-1154. doi:10.1197/j.aem.2004.08.003.
- Wallace, Peggy. 1997. "Following the Threads of an Innovation: The History of Standardized Patients in Medical Education." *Caduceus* 13(2): 5-28.
- Ward, Julia, Julianne Cody, Mary Schaal, and Mohammadreza Hojat. 2012. "The Empathy Enigma: An Empirical Study of Decline in Empathy among Undergraduate Nursing Students." *Journal of Professional Nursing* 28 (1): 34-40. doi:10.1016/j.profnurs.2011.10.007.
- Weir, William. 2012. "At 101, 'Mrs. Chase' is a Medical Marvel." *The Harford Courant*. Accessed June 26, 2012. http://articles.courant.com/2012-03-29/health/hc-mrschase-hartford-hospital-0328-20120327 1 doll-nursing-student-mannequins.
- Wells-Federman, Carol L., Eileen M. Stuart, John P. Deckro, Carol Lynn Mandle, Margaret Baim, and Cynthia Medich. 1995. "The Mind-Body Connection: The Psychophysiology of Many Traditional Nursing Interventions." *Clinical Nurse* Specialist 9 (1): 59-66.
- White, Barbara J. 1998. "A Nurse's Experience." In *The Changing Face of Health Care:* A Christian Appraisal of Managed Care, Resource Allocation, and Patient-Caregiver Relationships, edited by John F. Kilner, Robert D. Orr and Judith Allen Shelly, 17-24. Grand Rapids, Michigan: William B. Eerdsmans Publishing.
- Wilkes, Michael, Etan Milgrom, and Jerome R. Hoffman. 2002. "Towards More Empathic Medical Students: A Medical Student Hospitalization Experience." *Medical Education* 36 (6): 528-533.
- Wilkinson, Judith M. 1987-1988. "Moral Distress in Nursing Practice: Experience and Effect." *Nursing Forum* 23 (1): 16-29.
- Williams, Carol L. 1979. "Empathic Communication and its Effect on Client Outcomes." *Issues in Mental Health Nursing* 2 (1): 15-26.

- Wilt, Kathryn E. and Michalene King. 2012. "Time Well Spent: Integrating Simulation into an Accelerated 1-Year BSN Program." *Clinical Simulation in Nursing* 8 (3): e103-107. doi:10.1016/j.ecns.2010.10.002.
- Woodring, Barbara C. and Richard C. Woodring. 2011. "Lecture: Reclaiming a Place in Pedagogy." In *Innovative Teaching Strategies in Nursing and Related Health Professions*, edited by Martha J. Bradshaw and Arlene J. Lowenstein. 5th ed., 113-135. Boston: Jones and Bartlett Publishers.
- Woods, Martin. 2005. "Nursing Ethics Education: Are We Really Delivering the Good(s)?" *Nursing Ethics* 12 (1): 5-18. doi:10.1191/0969733005ne754oa.
- Zachariae, R., C. G. Pedersen, A. B. Jensen, E. Ehrnrooth, P. B. Rossen, and H. von der Maase. 2003. "Association of Perceived Physician Communication Style with Patient Satisfaction, Distress, Cancer-Related Self-Efficacy, and Perceived Control Over the Disease." *British Journal of Cancer* 88 (5): 658-665. doi:10.1038/sj.bjc.6600798.
- Ziv, Amitai, Paul R. Wolpe, Stephen D. Small, and Shimon Glick. 2006. "Simulation-Based Medical Education: An Ethical Imperative." Simulation in Healthcare: Journal of the Society for Simulation in Healthcare 1 (4): 252-256. doi:10.1097/01.SIH.0000242724.08501.63.